Case No. 84739

IN THE SUPREME COURT OF THE STATE OF NEVENTIAL SUPREME COURT OF THE STATE OF NEVENTIAL STATE STATE OF NEVENTIAL STATE STATE OF NEVENTIAL STATE OF NEVENTIAL STATE STATE OF NEVENTIAL STATE STATE STATE OF NEVENTIAL STATE STATE OF NEVENTIAL STATE ST

STATE ENGINEER, et al.

Appellants,

vs.

LINCOLN COUNTY WATER DISTRICT, et al.

JOINT APPENDIX

VOLUME 10 OF 49

Docket 84739 Document 2022-35237

Water-Resources Assessment and Hydrogeologic Report for Cave, Dry Lake, and Delamar Valleys

PRESENTATION TO THE OFFICE OF THE NEVADA STATE ENGINEER

Prepared by



November 2007

SE ROA 11689

JA_4451

Water-Resources Assessment and Hydrogeologic Report for Cave, Dry Lake, and Delamar Valleys

Submitted to: Tracy Taylor, P.E. State Engineer State of Nevada Department of Conservation & Natural Resources **Division of Water Resources** 901 S. Stewart Street, Suite 2002 Carson City, Nevada 89701

Pertaining to: Groundwater Applications 53987 through 53992 in Cave, Dry Lake, and Delamar Valleys

November 2007

Prepared by: Southern Nevada Water Authority Water Resources Division P.O. Box 99956 Las Vegas, Nevada 89193-9956

when G. Bun

Andrew G. Burns, Water Resources Division Manager

James M. Watrus, Sr. Hydrologist

Gary L. Wufour Gary L. Dixon, Principal Geologist

11- B.07

11-13-07 Date

Nov. 12, 2007 Date

ES.1.0 INTRODUCTION

In 1989, the Las Vegas Valley Water District (LVVWD) filed six applications (53987 through 53992, inclusive) for the appropriation of groundwater resources in Cave, Dry Lake, and Delamar valleys. By agreement with LVVWD, Southern Nevada Water Authority (SNWA) has assumed full interest in these applications. Since filing the applications, LVVWD and SNWA have been actively supporting the applications by establishing hydrologic monitoring networks to acquire groundwater data through the completion of hydrologic and geologic investigations within the three basins. The major work elements have included (1) construction of nine monitor wells for the acquisition of additional geologic, hydrologic and water-chemistry data; (2) geologic mapping and extensive geophysical surveys to define the hydrogeologic framework; (3) vegetation mapping (Cave Valley), and (4) completion of hydrologic site inventories and monitoring, through cooperative agreements with the U.S. Geological Survey (USGS) and Nevada Division of Water Resources (NDWR). Data and information from these programs and previous studies serve as the basis for the analyses and conclusions presented in this report.

This report presents the technical basis and justifications supporting the aforementioned groundwater applications, and is comprised of three parts.

- Part A presents a regional-scale assessment of the natural recharge, groundwater discharge, and interbasin flow of the White River Flow System. This assessment was conducted to estimate the volumes of precipitation recharge for Cave, Dry Lake, and Delamar Valleys using new data and information collected during the more than forty years since the USGS Reconnaissance Series Reports were published for these basins.
- Part B describes the geologic frameworks of Cave, Dry Lake, and Delamar valleys; and the
 occurrence and movement of groundwater within these three basins.
- Part C presents an analysis of the potential water-related effects of groundwater production
 associated with SNWA's groundwater applications on senior water-rights and selected points
 of interest. Also included in this part of the report, is a plan to monitor the effects of pumping
 and resource management measures to mitigate adverse impacts, should they occur.

The remainder of this executive summary follows the organization of the document. A summary of each part of the document is provided by report section.

Executive Summary

SE ROA 11691

ES-1



ES.2.0 PART A - REGIONAL ASSESSMENT OF THE GROUNDWATER RESOURCES OF THE WRFS

ES.2.1 Introduction

This section presents the purpose, scope and objectives of the regional assessment of the groundwater resources of the White River Flow System (WRFS). The purpose of the assessment was to derive updated groundwater budget estimates for the basins of the WRFS, particularly Cave, Dry Lake, and Delamar valleys. The updated groundwater budget estimates incorporate new information produced during the more than 40 years since the USGS Groundwater - Reconnaissance Series Reports for Cave, Dry Lake, and Delamar valleys were first published (Eakin, 1962; 1963a). This information includes meteorologic, hydrologic, and geologic data, and many studies resulting in improved estimates of precipitation distribution and groundwater discharge. This new information warrants a reevaluation of the groundwater resources for these basins; however, to do so requires adopting a regional approach (i.e., flow system) with respect to precipitation and natural recharge.

ES.2.2 Previous Investigations

This section provides summary descriptions of numerous studies related to Cave, Dry Lake, and Delamar valleys and the WRFS. These studies have included water-resource investigations, geologic and hydrogeologic investigations, recharge and discharge estimations, and other hydrologic studies relevant to the groundwater-resources assessment presented in this report. The most pertinent of these are as follows:

- Nevada precipitation maps developed by Hardman (1936, 1962, and 1965)
- Reconnaissance Reports for Cave, Dry Lake, and Delamar Valleys (Eakin, 1962; 1963a)
- A study of the WRFS by Eakin (1966)
- The U.S Air Force MX-missile siting program
- The Great Basin Regional Aquifer System Analysis study (RASA)
- The LVVWD Cooperative Water Project (CWP)
- · An evaluation of the White River and Meadow Valley Wash Flow Systems (LVVWD, 2001)
- Isotopic evaluation of the groundwater budgets derived by LVVWD (2001) (Thomas, 2001)
- Basin and Range Carbonate Aquifer System Study (BARCASS) (Welch and Bright, 2007)

ES.2.3 Technical Approach

This section describes the technical approach and methods used to complete the groundwater-resources assessment. The groundwater-balance method was selected to derive groundwater budgets for the WRFS and its basins. As applied in this analysis, pre-development conditions were assumed in which the total groundwater inflow to the system (i.e., groundwater recharge and boundary inflow) equals the total groundwater outflow from the system (i.e., groundwater evapotranspiration [ET] and boundary outflow). Given that the volume of boundary outflow constitutes a relatively small component of the WRFS groundwater budget, the scope of the analysis included the entire WRFS in the application of this method to ensure that the

typically high uncertainty associated with this budget component was minimized. An optimization approach was implemented using the Excel solver to derive a relationship between recharge efficiencies and precipitation that yielded a balanced groundwater budget for the flow system. This was completed using a spatial distribution of precipitation, estimates of groundwater ET, and external boundary flow. The solution includes recharge efficiencies expressed as a power function of precipitation. The constant and exponent that define this function were determined using the Excel Solver. Independent estimates of boundary flow (internal and external) were used to constrain the solution. The resultant recharge efficiencies were applied to the precipitation distribution to derive a spatial distribution of recharge from which basin recharge estimates could be computed and used to develop basin groundwater budgets. As a final step, uncertainty analyses were conducted to evaluate the effect of the uncertainty associated with the most uncertain variables on the calculated recharge efficiencies and final groundwater budgets.

ES.2.4 Data Compilation and Analysis

This section describes the meteorologic, hydrologic, and geologic data and information that were compiled and analyzed to derive estimates required by the groundwater-balance method applied in this assessment. These include independent estimates of the spatial distribution of precipitation, basin groundwater ET, and boundary flow volumes.

The Precipitation-elevation Regression on Independent Slopes Model (PRISM) was selected as the precipitation distribution for the WRFS. This precipitation source is preferred because precipitation distributions developed using the PRISM method utilize modern tools and incorporate more recent data and information not reflected in previous mapping efforts. More specifically, the most recent normal grid (800-m 1971 to 2000 - Version 2; May 3, 2007) was used as it is considered to be the best quality product to date. It is available online at the following internet address: http://www.prism.oregonstate.edu/products/matrix.phtml. Precipitation-station data were compiled and used to assess the validity of the PRISM-based distribution. Precipitation values extracted from the PRISM precipitation grid were compared to the period of record mean annual precipitation for precipitation stations located within the WRFS and vicinity. The PRISM precipitation values were within 10 percent of the period of record mean annual value for most of these stations, indicating a relatively good model fit with the observed data.

Average annual estimates of pre-development groundwater ET for basin discharge areas were derived as follows:

- Regional groundwater discharge areas were delineated and ET classes were defined using previous studies, Landsat imagery, aerial photography, other published information, and field work conducted by SNWA.
- 2. Available ET rate data obtained from the following sources were used in the derivation of basin ET estimates:
 - USGS WRIR 01-4239 (Reiner et al., 2002)
 - USGS SIR 2007-5078 (Moreo et al., 2007)
 - USGS OFR 2007-1156 Draft Report (Welch and Bright, 2007)

Executive Summary

SE ROA 11693

ES-3



- 3. ET rates were scaled to reflect basin conditions using potential evapotranspiration (PET) rates calculated using an equation relating PET to latitude and altitude using data from DRI Publication No. 41198 (McCurdy and Albright, 2004).
- 4. Basin ET volumes were calculated as the sum of the products of ET area multiplied by ET rate for the corresponding ET classes in that basin.
- Average annual precipitation for the ET areas was derived from the PRISM precipitation distribution and subtracted from the total ET volume to calculate estimates of groundwater ET.

The five defined ET classes and their respective range of ET rates (total ET) are as follows: (1) open water (6.63 to 9.06 ft/yr), (2) wetland/meadow (2.02 to 2.34 ft/yr), (3) dense meadow/riparian (2.82 to 3.89 ft/yr), (4) medium-density phreatophytes (1.03 to 1.18 ft/yr), and (5) bare soil/low-density phreatophytes (1.00 ft/yr). Pre-development maps delineating groundwater discharge areas and the ET classes within, were developed using satellite imagery, aerial photography, and field observations. Irrigated lands within these areas were removed from the maps and replaced with an ET class consistent with the natural vegetation encompassing the irrigated areas. Predevelopment range of 100,600 (Eakin, 1966; Rush, 1968) and 176,305 (LVVWD, 2001). Except for two basins, White River Valley and the Muddy River Springs Area, the estimated groundwater ET volumes compare reasonably well with the estimates reported in the Reconnaissance Series Reports. Estimates of groundwater ET for the three project basins are 1,285 afy for Cave Valley, and zero for Dry Lake and Delamar valleys.

Estimates of spring flow based on spring discharge records were derived for groundwater discharge areas within basins of the WRFS that contain a significant number of gaged springs. These estimates were assumed to constitute lower bounds for the total groundwater discharge volumes (i.e. groundwater ET + outflow). Included in the evaluation were the regional groundwater discharge areas of White River and Pahranagat valleys, and the Muddy River Springs Area. The total spring discharge for the largest springs in White River Valley was estimated to be 34,462 afy, a value that is lower than the estimated groundwater ET, 67,342 afy. For Pahranagat Valley, the total spring discharge estimate of about 25,000 afy (Eakin, 1963b, p. 19-20) is lower than the estimated volume of groundwater ET, 28,516 afy. For the Muddy River Springs Area, the majority of the spring discharge exits the valley in the form of ET and surface water flow in the Muddy River. The estimated average annual flow in the Muddy River near Moapa gage is about 34,000 afy (Wells, 1954, p. 566) and the estimated groundwater ET is 6,000 afy; thus, the total groundwater discharge is estimated at 40,000 afy. This value also constitutes the lower bound for the total groundwater inflow to this basin for predevelopment conditions.

Locations where groundwater flow occurs across basin boundaries were identified based on the prevailing hydrogeology, including rock types and geologic structures. Flow volumes were estimated for selected locations where sufficient information was available. These estimates were applied either in the solver as constraints and their initial conditions, or as additional information for calculating individual basin groundwater budgets. They are as follows:

ES-4

Executive Summary

- Outflow from Cave Valley to White River Valley (through Shingle Pass) at 4,000 afy
- Outflow from White River Valley to Pahroc Valley ranging between 6,300 (Maxey and Eakin, 1949, p. 45) and 40,000 afy (Eakin, 1966, p. 265)
- Inflow to Dry Lake Valley from Pahroc Valley at 2,000 afy
- Minimum total inflow to the Muddy River Springs Area at 40,000 afy
- Maximum outflow from Coyote Spring Valley to Hidden Valley at 15,000 afy.
- Inflow from Lower Meadow Valley Wash at 9,200 afy (4,000 afy to the Muddy River Springs Area, and 5,200 afy to California Wash)
- Minimum outflow to the Colorado River at 25,000 afy

ES.2.5 Data Analysis and Results

This section describes how the estimates of precipitation distribution, groundwater ET, and selected volumes of interbasin flow described in the previous section were used to setup and initialize the Excel solver and derive a solution. Summary descriptions of the analysis and results follow.

For this analysis, the data requirements for the Excel solver included precipitation data (area and rate) for specified intervals, the target value for groundwater ET, initial estimates for the parameters, and values for the constraints. Precipitation data were generated for areas of potential recharge using the PRISM grid. The potential recharge area for a given basin was defined by an area that excluded the valley floor, groundwater ET areas, and areas where precipitation is less than 8 inches. Within this area, 1-in, precipitation bands were defined and used as input to the solver.

The target value for the solver is the estimated value of groundwater ET for the WRFS. The target cell contains a formula expressing groundwater ET as the total recharge volume, plus the total inflow, minus the total outflow. The target value is 139,424 afy. The inflow is 9,200 afy and the outflow is a constrained parameter that is derived as part of the solution.

Parameters represent the unknown variables. For this analysis, the primary parameters are the coefficients of the power function expressing the recharge efficiencies as a function of precipitation. Secondary parameters are outflow from Coyote Spring Valley to Hidden Valley, and the total outflow to the Colorado River. Initial estimates for the power function coefficients were derived from the Maxey-Eakin precipitation-recharge relationship, and are 8.0×10^{-5} for the constant and 3.62 for the exponent. To initialize the solver, the parameters representing inter-basin flow were assigned their constraints as initial estimates. Constraints imposed on the solution are as follows:

- 1. the power function coefficients are positive,
- 2. the outflow from White River Valley is between 6,300 and 40,000 afy,
- 3. the outflow from Coyote Spring Valley to Hidden Valley is less than or equal to 15,000 afy,
- 4. the total inflow to the Muddy River Springs Area is greater than or equal to 40,000 afy,

Executive Summary

ES-5



- 5. the total outflow to the Colorado River is greater than or equal to 25,000 afy, and
- 6. the maximum recharge efficiency is less than or equal to 0.63.

Additional fixed constraints were placed on the volumes of underflow at specified locations where flow volumes were estimated. These flow volumes and locations are as follows: (1) 5,200 afy of inflow from Lower Meadow Valley Wash to California Wash, and (2) 4,000 afy of inflow from Lower Meadow Valley Wash to Muddy River Springs Area. This results in a total inflow to the WRFS of 9,200 afy. The solution is derived through an optimization process in which the coefficients of the power function and external boundary flows are adjusted within the constraints, until the formula in the target cell is solved. The solution is defined by the resultant recharge efficiencies and distribution, and a balanced WRFS groundwater budget. The recharge distribution and additional flow routing were then used to derive the basin groundwater budgets external from the solver.

The resulting power function compares relatively well with the work of Faybishenko (2007), Davisson and Rose (2000), and Lichty and McKinley (1995). The recharge efficiencies were computed using the power function derived by the solver, and are summarized here for the following precipitation zones: 45 percent for >20 in. zone, 14 percent for 15 to 20 in. zone, 4 percent for 12 to 15 in. zone, and 0.4 percent for 8 to 12 in. zone. The efficiencies are different from the standard Maxey-Eakin efficiencies because the precipitation distributions and groundwater ET estimates are different. However, the estimated recharge volumes compare well with those reported in the literature. The total recharge calculated for the WRFS is 155,224 afy, which falls within the reported range of values (104,650 [Eakin, 1966; and Rush, 1968] to 201,500 afy [LVVWD, 2001]). Except for two basins, the estimated recharge volumes for all basins with recharge volumes greater than 1,000 afy, fall within the reported ranges. The two exceptions are Garden and Dry Lake valleys. The derived outflow volumes are as follows: (1) 15,000 afy from Coyote Spring Valley to Hidden Valley and, (2) 25,000 afy of total outflow to the Colorado River. A groundwater budget for the WRFS and its individual basins was constructed using the recharge, groundwater ET and boundary flow estimates derived by this analysis. All of the WRFS budget components fall within the ranges of values reported in the literature (Table ES.2-1). The derived groundwater budget components for all basins of the WRFS, including the project basin are presented in Figure ES.2-1.

Budget Component	Estimated Value (afy)	Reported Range (afy)
Recharge	155,224	104,650 to 201,500
Inflow	9,200	7,000 to 32,000
Groundwater ET	139,424	100,600 to 176,305
Outflow	25,000	11,100 to 49,000

Table ES.2-1 Estimated Groundwater Budget for the WRFS

Water-Resources Assessment and Hydrogeologic Report for Cave, Dry Lake, and Delamar Valleys



Figure ES.2-1 WRFS Groundwater Budget

Executive Summary

ES-7



ES.2.6 Uncertainty Analyses

This section describes uncertainty analyses that were performed to evaluate the estimated budget components of the WRFS and project basins summarized in the previous section. The approach and results are summarized.

The purpose of the uncertainty analyses was to evaluate the effect of the uncertainties associated with selected variables on the calculated recharge efficiencies and final groundwater budgets. The most uncertain variables were selected for these analyses based on the range of their estimated values, and included the following:

- 1. maximum recharge efficiency,
- 2. groundwater ET estimate for White River Valley,
- 3. groundwater ET estimate for the Muddy River Springs Area,
- 4. the total inflow to WRFS, and
- 5. the total outflow from WRFS.

Ten alternate scenarios were implemented for each low and high value of the five tested variables. The derived solutions were then compared to the base-case solution. An additional scenario designed to evaluate the BARCASS interpretation (Welch and Bright, 2007) for the northern part of the WRFS was implemented. For this scenario, the BARCASS groundwater ET estimates for Long, Jakes, White River, and Cave valleys were substituted, and the full volume of outflow from Cave Valley was re-directed to the White River Valley.

The results of the analysis may be summarized in terms of the resulting ranges of estimated recharge for the WRFS and the three project basins. The largest uncertainty is related to the estimates of groundwater ET in White River Valley, in which the total recharge for the WRFS ranges from 124,882 to 184,897 afy, as compared to the base-case value of 155,224 afy. For Cave, Dry Lake, and Delamar valleys, the uncertainty for recharge ranges from 28,721 to 44,584 afy, as compared to the base-case value of 36,727 afy. The range of uncertainty is about 20 percent. The scenario designed to evaluate the BARCASS interpretation yields slightly higher values of recharge compared to the base-case, but the results are essentially the same with differences in the total recharge and the recharge in the project basins only 8,367 and 2,214 afy, respectively.

ES.2.7 Conclusions

ES-8

The major findings of this study related to the precipitation distribution, the groundwater ET and the recharge distribution of the WRFS and its basins are summarized below.

- The groundwater-balance method used to complete the water-resources assessment described in this report is essentially the same approach used by Maxey and Eakin (1949) almost 60 years ago. While the approach is the same, this analysis incorporates more accurate data of better resolution, and newer technology that allows for more detailed calculations.
- The PRISM precipitation distribution (800-m 1971 to 2000 normals Version 2; May 3, 2007) best represents the precipitation distribution for the WRFS for the purposes of this assessment.

Executive Summary

Water-Resources Assessment and Hydrogeologic Report for Cave, Dry Lake, and Delamar Valleys

- The PRISM grid provides a good representation of the long-term mean annual precipitation distribution for the WRFS. This grid represents, within ten percent, the period of record mean annual value for most precipitation stations within the WRFS and vicinity.
- The PRISM precipitation grid provides a much better spatial distribution of precipitation than the Hardman precipitation maps (1936, 1965), particularly at the higher precipitation intervals.
- The predevelopment estimates of groundwater ET developed for the WRFS and its basins compare very well with previous estimates. An exception is the estimate for White River Valley which is 30,000 afy higher than Eakin (1966) due to the differences in the mapped acreage. For this study and other more recent studies (e.g. BARCASS), the mapped area for White River Valley is more than four times the acreage relied upon by Eakin (1966).
 - Predevelopment groundwater ET for the WRFS is estimated to be 139,424 afy, which falls within the literature range of 100,600 (Eakin, 1966; Rush, 1968) and 176,305 afy (LVVWD, 2001). Estimates of groundwater ET for the three project basins are as follows: 1,285 afy for Cave Valley, and zero for the other two.
- The solution derived using the groundwater-balance method yielded a recharge-precipitation relationship, recharge efficiencies and a recharge distribution that are believed to be the best to date for the WRFS.
 - The solution incorporates new data and technology, and is calibrated to new and better estimates of discharge.
 - The recharge-precipitation relationship is somewhat different from that of Maxey and Eakin (1949), but very comparable to more recent relationships and field-based data developed by others (Faybishenko, 2007; Davisson and Rose, 2000; Lichty and McKinley, 1995).
 - The derived recharge efficiencies are summarized for the following precipitation zones: 45 percent for >20 in. zone, 14 percent for 15 to 20 in. zone, 4 percent for 12 to 15 in. zone, and 0.4 percent for 8 to 12 in. zone.
 - The calculated groundwater recharge for the WRFS under predevelopment steady-state conditions is about 155,000 afy (rounded to the nearest thousand). The total recharge falls within the reported range of values (104,650 to 201,500 afy).
 - The basin recharge volumes compare well with the recharge estimates reported in the literature for most basins. The exceptions are Garden Valley and Dry Lake Valley which are higher based on this analysis.
 - The total recharge volume calculated for the project basins is 36,727 afy, which is larger than that calculated by Eakin (1962 and 1963a) using the topographic map available at the

Executive Summary

ES-9



time. However, it is within about 3,000 afy of the total derived (33,400 afy) by applying the standard Maxey-Eakin recharge efficiencies to the Hardman (1962) precipitation map.

- A groundwater budget for the WRFS and its individual basins was constructed using the recharge, groundwater ET and boundary flow estimates derived by this analysis. All of these components fall within the literature range of values.
- The largest uncertainty is associated with the White River Valley groundwater ET based on the range of estimated values (37,000 to 79,560 afy). This uncertainty results in ranges of recharge of 124,882 to 184,897 afy for the WRFS; and 28,721 to 44,584 afy for the three project basins, or a range of uncertainty of plus or minus about 20 percent.

ES.3.0 Part B - Hydrogeologic Framework and Groundwater Occurrence and Movement for Cave, Dry Lake, and Delamar Valleys

ES.3.1 Introduction

Part B of this report describes the geologic framework and the occurrence and movement of groundwater within Cave, Dry Lake, and Delamar valleys. The geologic framework served as the basis from which an understanding of the hydrogeology was developed, particularly the geologic features controlling local and regional groundwater flow in these basins.

The geologic framework is presented as a surface geologic map and a series of geologic cross sections (Plate 1), depth-to-basement maps, and profiles interpreting seismic and audiomagnetotelluric data. Hydraulic data from wells and springs are used in conjunction with the geologic framework to describe the occurrence and movement of groundwater within the basins and vicinity. The hydrogeologic description is relied upon in Part C of this report, which evaluates the potential water-related effects of developing the SNWA applications in these basins.

ES.3.2 Geologic Framework

This section provides an overview of the geologic information available for Cave, Dry Lake, and Delamar valleys based on a report prepared by SNWA entitled "Geology of White Pine and Lincoln Counties and Adjacent Areas, Nevada and Utah: The Geologic Framework of Regional Groundwater Flow Systems" (Dixon et al., 2007).

Cave Valley is bounded in the east by the southern extension of the Schell Creek Range which is comprised of an east-dipping and heavily faulted sequence of Precambrian to Tertiary rocks. The dominant fault is on the western flank of the range, and in the northeast part of the valley, the range is cored by Precambrian to Cambrian quartzite. In the west, Cave Valley is bounded by the Egan Range, which is a complexly faulted, east-dipping horst comprised of Cambrian to Permian rocks overlain by Tertiary volcanic rocks. The basin has been effectively partitioned into two sub-basins by the

ES-10

Executive Summary

northeast-striking oblique-slip Shingle Pass Fault, which has displaced a part of the Egan Range, forming an east-dipping fault block that extends northeast across and underneath the valley where it terminates against the west range-front fault of the Schell Creek Range. This fault block contains the Mississippian Chainman Shale formation which, in this area, is considered a confining unit because of its thickness and hydraulic properties. The northern sub-basin is relatively shallow as the depth to basement is typically no greater than 3,300 ft-bgs. The basin fill is comprised of mostly sands and gravels, and occasional cemented strata near the surface. The southern sub-basin is significantly deeper, with the depth to basement extending to almost 20,000 ft-bgs. The basin fill is comprised of mostly sands and gravels, but with greater clay content than the northern sub-basin.

Dry Lake and Delamar valleys are bounded in the west by the North Pahroc Range and the South Pahroc Range, respectively. The North Pahroc Range is a west-dipping horst consisting of upper Paleozoic rocks overlain by Tertiary volcanic rocks. The South Pahroc Range consists of a series of west-dipping volcanic rocks, which extend southward to southern Delamar Valley where they are terminated by the east-northeast trending Pahranagat Shear Zone. To the east, Dry Lake and Delamar valleys are bounded from north to south by the Fairview, Bristol, Highland, and Chief Ranges, and the Delamar Mountains to the south. To the north, the Fairview Range consists of Devonian to Pennsylvanian carbonate rocks, which are interrupted by the Indian Peak Caldera Complex. To the south, the Bristol, Highland and Chief Ranges consist of Cambrian carbonate rocks underlain by Precambrian to Cambrian quartzites. Further south, the Delamar Mountains consist of east-dipping Late Proterozoic to Cambrian rocks and Tertiary volcanic rocks associated with the Caliente and Kane Springs Wash caldera complexes.

The geometry of Dry Lake and Delamar valleys was defined, in part, by gravity surveys and depth to basement estimates which indicate the presence of low saddles to the north adjoining northern Dry Lake Valley (i.e. Muleshoe Valley) and to the south adjoining Delamar Valley. Two grabens are present in Dry Lake Valley, one along the margins of the valley and another along the central axis of the valley. The depth to basement in these areas are estimated to range from about 3,300 to almost 10,000 ft-bgs. In southwest Delamar Valley, a bowl-shaped basin is inferred, and the maximum depth to basement is estimated to be almost 7,000 ft-bgs, with much of the area only about 5,000 ft-bgs. The basin fill underlying these areas consists primarily of sands and gravels, with increasing clay content and thicker sequences of volcanic tuffs to the south. Based on the geologic framework, Dry Lake and Delamar valleys are one contiguous groundwater basin.

ES.3.3 Groundwater Occurrence and Movement

This section describes a conceptualization of the groundwater systems underlying Cave, Dry Lake, and Delamar valleys developed through an inventory of groundwater sites and an assessment of the current and recent water-level conditions for both the basin-fill and carbonate-rock aquifers.

In the northern sub-basin of Cave Valley, depth to water in the basin fill ranges from 2 to 221 ft-bgs, with water-level elevations ranging from 6,896 to 6,119 ft-amsl. Depth to water in the single carbonate well is 1,051 ft-bgs, with a water-level elevation of 5,406 ft-amsl. This carbonate water-level is most likely reflective of the hydraulic head within the Shingle Pass Fault. In the southern sub-basin, depth to water in the basin fill ranges from 8 to 327 ft-bgs, with water-level elevations ranging from 6,221 to 5,704 ft-amsl. Depth to water in the two carbonate wells located on

Executive Summary

SE ROA 11701

ES-11



the east and west side of the southern sub-basin ranges from 140 to 220 ft-bgs, respectively, with water level elevations ranging from about 5,850 to 5,790 ft-amsl. There is a north to south hydraulic gradient and direction of groundwater flow in the basin fill. The hydraulic heads in the carbonate rocks appear to be structurally controlled by the fault block associated with the Shingle Pass Fault, as reflected by the water-level elevations north and south of the block. Flow through this fault block is unlikely because of the presence of the Mississippian Chainman Shale formation.

Water-level data indicate a slight increasing trend of about 10 ft since the early 1980s. Because there is virtually no groundwater development within the basin, this trend is most likely reflective of the natural variability of the groundwater system.

Depth to water in Dry Lake and Delamar valleys ranges from 3 to 1,316 ft-bgs, and deepens to the south. Water-level elevations range from 6,630 to 3,486 ft-amsl and indicate a north to south gradient and direction of groundwater flow in both the basin fill and carbonate aquifer. This conceptualization of flow is supported by the data and is consistent with previous investigations by Eakin (1966), Scott et al. (1971), and Harrill et al. (1988).

ES.3.4 Conclusions

This section presents the summary and conclusions of Part B. Based on the geologic framework and groundwater data compiled as part of this effort the following is concluded:

Cave Valley is effectively partitioned into two sub-basins by the northeast-striking oblique-slip Shingle Pass Fault, which has displaced a part of the Egan Range, forming an east-dipping fault block that extends northeast across and underneath the valley where it terminates against the west range-front fault of the Schell Creek Range.

- Based on water-level data, this fault block does not appear to significantly influence flow in the basin fill, but does control water-levels in the carbonate-rock aquifer as evidenced by the gradient defined by two carbonate wells on the north and south side of the block.
- Groundwater flow across this fault block is unlikely due to the presence of the Mississippian Chainman Shale formation which, in this area, is considered a confining unit because of its thickness and hydraulic properties.
- While groundwater flow is unlikely across the fault block, it is likely from north to south along the western range-front fault of the Schell Creek Range based on the elevation of Cave Spring and carbonate wells in the south.
- Interbasin outflow to the southern third of White River likely occurs in small amounts through the Shingle Pass Fault zone, estimated here to be no more than 4,000 afy.
- The remaining interbasin outflow occurs in the southern part of the valley to northern Pahroc Valley. This flow is estimated to be at least 9,400 afy.

ES-12

Executive Summary

• Inflow from southern Steptoe Valley is unlikely due to the lithology and structure of the geologic framework underlying northern Cave Valley and an inferred groundwater divide created by groundwater recharge in this area. The inferred groundwater divide cannot be verified with the presently available water-level data.

Dry Lake and Delamar valleys are one contiguous basin based on the geologic framework as defined by the range-front faults and depth to basement maps, and groundwater water data that indicate north-south gradients in the basin fill and carbonate-rock aquifers.

- All groundwater within the basins is derived from local recharge except for a small amount of
 interbasin inflow from northern Pahroc Valley at a location coincident with a series of
 northwest-trending right-lateral faults that form the boundary between southern Cave Valley,
 northern Pahroc Valley, and northern Dry Lake Valley. This inflow is estimated to be about
 2,000 afy or less.
- Groundwater flows from Dry Lake Valley to Delamar Valley where flow is controlled by the Caliente and Kane Springs Wash caldera complexes to the east, and the South Pahroc Range and associated range-front faults to the west. The geologic framework in this area precludes flow to Pahranagat Valley to the west and Lake and Patterson Valleys to the east.
- Interbasin outflow from Delamar occurs along and across the Pahranagat Shear Zone and into northern Coyote Spring Valley and possibly the very southern part of Pahranagat Valley. This outflow is estimated to be about 24,000 afy, and is comprised of inflow from northern Pahroc Valley and recharge derived locally from within the hydrographic area boundaries.

ES.4.0 PART C - ANALYSIS OF POTENTIAL PUMPING EFFECTS AND MONITORING AND MANAGEMENT

ES.4.1 Introduction

Part C of this report presents an analysis of the potential water-related effects associated with the SNWA applications in Cave, Dry Lake, and Delamar valleys, including a description of existing water rights, an analysis of the potential effects, and a description of a program that would be implemented to monitor and manage effects as they occur. The focus of these analyses is primarily limited to Cave, Dry Lake, and Delamar valleys, although some of the data that were used in the analyses are from locations in adjacent valleys.

ES.4.2 Summary of Water Rights

This section provides descriptions of the existing water rights within Cave, Dry Lake, and Delamar valleys, with emphasis on currently committed underground water rights. Based on abstracts obtained from the NDWR, the majority of the permitted water rights in Cave, Dry Lake, and Delamar valleys are for springs located within the mountain blocks on either side of the valleys. The abstracts also indicate that the manner of use for most of these permits is for stock watering (NDWR, 2007a).

Executive Summary

SE ROA 11703

ES-13



Committed duties for underground rights in these valleys include 46.58 afy in Cave Valley, 56.56 afy in Dry Lake Valley, and 7.24 afy in Delamar Valley (NDWR, 2007b).

ES.4.3 Effects Analysis

This section includes a simplified effects analysis of the groundwater production requested under the SNWA applications in Cave, Dry Lake, and Delamar valleys. The simplified analysis used the Theis (1935) equation to evaluate the effects of continuously pumping the application volume from the points of diversion in each of the three basins for a period of 75 years. The locations of interest for the analysis included existing permits for valley floor springs, underground permits, and areas of environmental concern as identified in SNWA (2007). Although the drawdowns were quantitatively simulated using the Theis (1935) equation, the estimates are still considered qualitative because of the many assumptions that are inherent to this equation.

The conceptual model applied to the Theis analysis involved two separate analyses one for Cave Valley and a combined analysis for Dry Lake and Delamar valleys. This is consistent with the interpretation for Dry Lake and Delamar valleys that these two hydrographic areas are connected geologically and hydrologically, making them one contiguous basin. Simulated flow barriers were placed along the eastern and western margins of southern Cave, Dry Lake, and Delamar valleys to simulate the lack of interbasin flows in those directions. This interpretation is consistent with the flow interpretations presented in Sections 2.0 and 3.0 of Part B of this report. The flow barriers were simulated in the analysis using image wells, which are imaginary discharging wells placed on the opposite side of the flow boundaries from the application points of diversion, at the same distance from, and perpendicular to, the simulated boundary. The resulting drawdown at any point within the boundaries is the algebraic sum of the drawdowns produced at that point by the application point of diversion and its image. In this case, the use of image wells results in greater drawdowns within the valley than would be observed without the simulated flow barriers.

While the alluvial and carbonate-rock aquifers were evaluated separately in this analysis, the drawdowns simulated in each were summed for each location of interest. This is a conservative assumption because it is unlikely that pumping an alluvial point of diversion would have much effect on a location of interest sourced in the carbonate-rock aquifer, and visa versa.

The data required for the Theis analysis included aquifer properties derived from two site specific tests conducted on carbonate rock and alluvial aquifers in Dry Lake Valley, as well as storativity values obtained from a literature review of the properties for the carbonate rock aquifer. The corresponding values for transmissivity and storage coefficient are 170,000 ft²/d and 0.013 for the alluvial aquifer, and 8,690 ft²/d and 0.03 for the carbonate-rock aquifer. Additional assumptions included a 1,000 ft open interval for each well with a radii of 0.83 ft, and the continuous pumping of these wells for a period of 75 years.

The findings of the simplified effects analyses are summarized as follows:

 Simulated drawdowns at the end of 75 years of continuous pumping in Cave Valley ranged from 41 to 50 ft for points of diversion located 3.53 and 5.97 mi away respectively. To further evaluate the effects of the proposed SNWA pumping on underground water rights, the well

ES-14

Executive Summary

construction and water-levels must be considered. However, this information is not available for these water-right points of diversion.

- Simulated drawdowns at the end of 75 years of continuous pumping in Dry Lake Valley ranged from 8 ft at water-right number 5936 located in the northern portion of Dry Lake Valley to 48 ft at water-right number 8698 which is located 6.44 mi from the nearest proposed SNWA application point of diversion. Well completion information was available for three of the five underground water right locations and based on an analysis of this information it is unlikely that any adverse effects will occur at these locations. Water-right numbers 8698 and 35771 both have sources identified as springs. Inspection of these points on topographic maps suggests that they may be reservoirs fed by springs located within the mountain block. If they are reservoirs, then pumping the proposed production wells, would not effect these locations. The springs in Dry Lake Valley were unlikely to experience adverse effects due to their nature, geographic location, and the geologic complexity between the springs and the proposed SNWA applications points of diversion.
- Simulated drawdowns at the end of 75 years of continuous pumping in Delamar Valley ranged from 53 to 65 ft. Grassy Spring is the one area of environmental concern in Delamar Valley (SNWA, 2007). The drawdown simulated at Grassy Spring was 61 ft. The spring is located over 450 ft higher in elevation than the nearest proposed pumping well. The discharge measurements and temperatures for Grassy Spring indicate it is derived from local precipitation. These factors suggest a different water source than the source underlying the application point of diversion. Furthermore, the depth to water at the application point of diversion is likely to be on the order of 1,000 ft. Therefore, it is highly unlikely that any effects will occur as a result of pumping the proposed SNWA applications. The remaining water rights in Delamar Valley are associated with springs. Limited water level data in Delamar Valley, as described in Section 2.0 of Part B, indicates a deep water table. This would suggest that the springs are likely derived from a perched system that would be unaffected by pumping from the proposed SNWA application points of diversion.

This scenario is conservative with respect to the evaluation of water-related effects in that the annual production rate is the maximum permitted duty, and the scenario does not incorporate water-resource management strategies that could be invoked to minimize or reduce effects. The simulated drawdowns are believed to be at the upper bounds of what is expected due to the single aquifer assumption in this analysis where drawdowns in the carbonate rock aquifer and the alluvial aquifer were combined. There are several major limitations to the use of the Theis equation to calculate drawdowns over a hydrographic area. Drawdowns simulated using the Theis equation are only approximations of the potential drawdowns. In reality, the magnitude of the drawdowns caused by pumping at a given observation point depends on several factors:

- · The distance between the pumping well and the observation point
- · The presence of major heterogeneities including barriers to groundwater flow
- The actual aquifer properties in the region of pumping
- · The magnitude, length, and continuity of pumping

Executive Summary

SE ROA 11705

ES-15



ES.4.4 Monitoring and Management

This section describes SNWA's commitment to monitoring and management of the development of these water rights. SNWA will abide by all permit conditions attached to the water-right permits and work within the water-right administrative process set forth by the Nevada State Engineer to (1) manage its use of the water resources, and (2) avoid unreasonable lowering of the water table such that adverse impacts to senior water-right holders do not occur. Additionally, SNWA will seek to manage its use of the permitted water resources in a manner that does not adversely affect environmental resources. SNWA will coordinate the development of a program to monitor long-term fluctuations in water levels and spring discharge at selected monitoring sites in Cave, Dry Lake, and Delamar valleys and vicinity. SNWA installed two monitor wells in each of Cave, Dry Lake, and Delamar valleys in 2005 as well as a monitor well in northern Pahranagat Valley. Additionally, SNWA cooperatively funds USGS monitoring of wells and springs as part of a Joint Funding Agreement between SNWA, USGS, and NDWR. The monitoring sites are regularly evaluated to ensure data adequacy and appropriate spatial coverage. The monitoring of these sites may be included in the permit conditions prescribed by the Nevada State Engineer. Figure ES.4-1 depicts the current monitoring sites within Cave, Dry Lake, and Delamar valleys and vicinity.

ES.4.5 Conclusions

The major finding of the effects analysis is that it is highly unlikely that any adverse effects will occur to existing permitted underground and spring water-right points of diversion or environmental areas of concern. The monitoring program established as per the permit conditions set forth by the Nevada State Engineer will allow for the management of the water resources to avoid unreasonable effects.

ES-16

Executive Summary





Figure ES.4-1 Current Monitoring Sites

Executive Summary

ES-17



ES.5.0 UNAPPROPRIATED GROUNDWATER RESOURCES IN CAVE, DRY LAKE, AND DELAMAR VALLEYS

The unappropriated groundwater resources for Cave, Dry Lake, and Delamar valleys is defined here as the precipitation recharge derived locally within each hydrographic area boundary, minus the existing committed underground rights. An estimate of the unappropriated groundwater resources for each valley based on the analyses presented in this report is presented in Table ES.5-1.

Table ES.5-1 Estimate of Unappropriated Groundwater Resources in Cave, Dry Lake, and Delamar Valleys

HA Name	Precipitation Recharge (afy)	Committed Underground Water Rights (afy)	Unappropriated Groundwater (afy)
Cave Valley	14,659	46.58	14,612
Dry Lake Valley	15,667	56.56	15,610
Delamar Valley	6,401	7.24	6,394
Total	36,727	110.38	36,616

Note: Values were not rounded, to allow calculations to be tracked during analysis and reporting.

The total unappropriated groundwater resources estimated for Cave, Dry Lake, and Delamar valleys is sufficient to satisfy the SNWA applications filed in these basins.

ES-18

Executive Summary



Part A

Regional-Scale Evaluation of Natural Recharge, Discharge, and Interbasin Flow for the White River Flow System

Prepared by: Andrew G. Burns

November 2007

SOUTHERN NEVADA WATER AUTHORITY Groundwater Resources Department Water Resources Division • snwa.com

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

CONTENTS

List of	Figure	s		
List of	Tables			
List of	Acron	ms and Abbreviations ix		
1.0	Introduction			
	1.1 1.2 1.3	Background1-1Purpose and Scope1-3Document Organization1-3		
2.0	Previo	us Investigations		
	2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9 2.10	USGS/DCNR Reconnaissance Investigations2-1Nevada Water Resources Report No. 32-2Desert Institute Publication No. 41054.2-2U.S. Air Force MX Missile Siting Program2-2Great Basin Regional Aquifer System Analysis2-3LVVWD Cooperative Water Project2-3USGS Open-File Report 96-469.2-4LVVWD Coyote Spring Valley Report2-4Desert Research Institute Publication No. 411692-5BARCAS Study2-5		
3.0	Technical Approach			
	3.1 3.2	Objective 3-1 Review and Selection of Recharge Method 3-1 3.2.1 Recharge Methods 3-1 3.2.2 Groundwater Balance Methods 3-2 3.2.2.1 Maxey-Eakin Method and Derivatives 3-2 3.2.2.2 Other Water Balance-Based Methods 3-3 3.2.3 Soil Water-Balance Methods 3-4		
		3.2.3.1 INFIL Code 3-5 3.2.3.2 Basin Characterization Model 3-5 3.2.4 Chloride Mass-Balance Method 3-5 3.2.5 Selection of Recharge Method 3-5		
	3.3	Technical Approach using the Groundwater-Balance Method.3-63.3.1Method Application3-73.3.2Method Implementation3-8		
4.0	Data C	ompilation and Analysis 4-1		
	4.1	Precipitation Data Analysis4-14.1.1Objective4-14.1.2Review and Selection of Precipitation Method4-14.1.2.1Methods4-14.1.2.2Selection of Precipitation Method4-2		



ii

CONTENTS (CONTINUED)

		4.1.3	Precipitation Data 4-3		
	4.2	Region	al Groundwater Discharge 4-3		
		4.2.1	Groundwater ET 4-3		
			4.2.1.1 Objective		
			4.2.1.2 Technical Approach 4-6		
			4.2.1.3 Regional Groundwater Discharge Areas and		
			ET Classes 4-6		
			4.2.1.4 Compilation of ET and PET Rate Data 4-7		
		1.4	4.2.1.5 Derivation of Groundwater ET Estimates 4-10		
		4.2.2	Spring Discharge 4-13		
			4.2.2.1 Objective		
	10		4.2.2.2 Compilation of Spring Discharge Records 4-13		
	4.3	Interba	sin Flow		
5.0	Data	Data Analysis and Results 5-1			
	5.1	Data A	nalysis		
		5.1.1	Delineation of Areas of Potential Recharge 5-1		
		5.1.2	Precipitation Data		
		5.1.3	Solver Target ET _{gw}		
		5.1.4	Solver Parameters 5-5		
		5.1.5	Constraints		
	5.2	Solution	n Process		
	5.5	Kesults	and Discussion		
		532	Recharge Efficiencies 5-11		
		533	Recharge Distribution 5-14		
		5.5.5	5 3 3 1 WRFS Basin Recharge Estimates 5-14		
			5.3.3.2 Recharge in Cave, Dry Lake, and Delamar Valleys		
		5.3.4	Groundwater Budgets		
6.0	Unce	rtainty Ar	nalvses		
	C 1	01:01	<u> </u>		
	6.1	Objecti	ve		
	6.2	Approa	Cn		
	0.5	6 3 1	Groundwater ET Estimates 62		
		632	Total Inflow from the Lower Meadow Valley Wash Flow System 6-3		
		633	Total Outflow to the Colorado River 6-5		
		634	Maximum Recharge Efficiency 6-6		
		6.3.5	SNWA/BARCASS Scenario		
	6.4	Summa	ry		
7.0	Summ	nary and (Conclusions		
8.0	Refer	ences			

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

CONTENTS (CONTINUED)

Appen	lix A - Precipitation Station Data				
A.1.0	Precipitation Station Data				
Appen	lix B - Groundwater Discharge Areas and Evapotranspiration Rates				
B.1.0	IntroductionB-1				
	B.1.1 Groundwater Discharge Areas B-1 B.1.2 ET Rates B-1				
B.2.0	References				
Appen	lix C - Spring Data				
C.1.0	Spring Discharge				
C.2.0	References				
Appen	lix D - Interbasin Groundwater Flow through Selected Boundaries				
D.1.0	IntroductionD-1				
D.2.0	Interbasin Flow within the WRFSD-1				
	D.2.1 Outflow From Coyote Spring Valley to Hidden Valley D-1 D.2.2 Interbasin Flow among Cave, Dry Lake, and Delamar Valleys D-7 D.2.2.1 Cave Valley D-7 D.2.2.1.1 Outflow to White River Valley through Shingle Pass D-9 D.2.2.1.2 Outflow to Northern Pahroc Valley D-11				
	D.2.3 Dry Lake and Delamar Valleys D-11 D.2.3.1 Inflow from Pahroc Valley D-13 D.2.3.2 Outflow from Dry Lake Valley to Delamar Valley D-13 D.2.3.3 Outflow from Delamar Valley D-13				
D.3.0	WRFS Boundary Flow				
	D.3.1 Inflow from Lower Meadow Valley Wash				
	D.3.2 Outflow to Colorado River D-19 D.3.2.1 Black Mountains Area/Rogers and Blue Point Springs D-19 D.3.2.2 Outflow from Lower Moapa Valley D-21				
D.4.0	ReferencesD-24				

iii



CONTENTS (CONTINUED)

Appen	dix E -	Description of Excel Analysis Files (CD-ROM Contents)			
E.1.0	Introduction				
E.2.0	Solution File				
	E.2.1	File ContentsE-1E.2.1.1Worksheet 1 - "1-Explanation".E-2E.2.1.2Worksheet 2 - "2-Groundwater-ET-Estimates"E-2E.2.1.3Worksheet 3 - "3-Precipitation-Recharge-Details"E-2E.2.1.3.1Table ContentsE-2E.2.1.3.2Generation of Precipitation DataE-3E.2.1.4Worksheet 4 - "4-Solution"E-5E.2.1.4.1Area 1 - Main Solver AreaE-5E.2.1.4.1Area 1 DescriptionE-5E.2.1.4.2Initial Estimates for Coefficients of Power FunctionE-6E.2.1.4.3Area 2 - Recharge Efficiency CalculationsE-9C.1.4.1Area 3 - Groundwater Budget CalculationsE-9			
	E.2.2	Solution ProcessE-9E.2.2.1Executing the SolverE-9E.2.2.2Deriving Basin Groundwater BudgetsE-10			
E.3.0	Uncer	ainty Analysis Excel Files E-11			
E.4.0	Refere	nces			

iv

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

FIGURES NUMBER TITLE PAGE 1-1 Location of the White River Flow System, Cave, Dry Lake, and Delamar Valleys . . 1-2 3-1 Location of Precipitation Stations and Mean Annual Precipitation 4-1 within the WRFS and Vicinity 4-4 PRISM Precipitation Distribution within the WRFS and Vicinity, and 4-2 Percent Difference between PRISM and Precipitation-Station Data 4-5 ET Measurement Sites within the WRFS and Vicinity 4-9 4-3 4-4 Meteorological Station Locations and Distribution of PET within the WRFS 4-11 4-5 Potential Locations of Interbasin Flow Based on the Hydrogeologic Framework . . 4-15 5-1 Areas of Potential Recharge within the WRFS 5-3 Precipitation Distribution within Potential Recharge Areas within the WRFS 5-4 5-2 5-3 Interbasin Flow Constraints Applied in Solver 5-8 5-4 5-5 5-6 Digitized Hardman (1962) Precipitation Map for Cave, Dry Lake, and Delamar Valleys 5-16 Estimated Groundwater Budgets for Basins within the WRFS 5-20 5-7 B.1-1 Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Long ValleyB-3 Distribution of Predevelopment Groundwater Evapotranspiration B.1-2 and Location of Selected Springs in Jakes Valley B-4 Distribution of Predevelopment Groundwater Evapotranspiration B.1-3 and Location of Selected Springs in White River Valley......B-5 White River Valley Discharge Area - Inset 1......B-6 B.1-4 White River Valley Discharge Area - Inset 2. B-7 B.1-5

v



FIGURES (CONTINUED)

NUMBE	R TITLE	PAGE
B.1-6	White River Valley Discharge Area - Inset 3	B-8
B.1-7	White River Valley Discharge Area - Inset 4	B-9
B.1-8	Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Cave Valley	B-10
B.1-9	Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Garden Valley	B-11
B.1-10	Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Pahranagat Valley	B-12
B.1-11	Distribution of Predevelopment Groundwater ET and Location of Selected Springs in Muddy River Springs Area, California Wash, Lower Moapa Valley, and Black Mountains Area	., B-13
D.2-1	Interbasin Outflow and Water-Level Elevations of Selected Wells in Southern Coyote Spring Valley and Vicinity	D-2
D.2-2	Interbasin Flows and Water-Level Elevations of Selected Wells and Springs in Cave Valley and Vicinity	D-8
D.2-3	Watershed for Selected Springs in White River Valley	D-10
D.2-4	Interbasin Flows and Water-Level Elevations of Selected Wells and Regional Springs for Dry Lake and Delamar Valleys and Vicinity	D-12
D.2-5	Inflow from Pahroc Valley to Northern Dry Lake Valley	D-14
D.2-6	Delamar and Pahranagat Valleys and the Pahranagat Shear Zone	D-15
D.3-1	Two Sets of Synoptic Discharge Measurements along Muddy River	D-18
D.3-2	Measurement Sites along the Muddy River in Lower Moapa Valley	D-20
E.2-1	Generation Process of One-Inch Precipitation Bands	E-4
E.2-2	Power Function Fit to Maxey-Eakin Precipitation-Recharge Points (Table E.2-1)	E-7
E.2-3	Excel Solver Window	E-10

TABLES TITLE NUMBER PAGE 4-1 4-2 Groundwater Discharge Areas in Acres, by ET Class for the WRFS 4-8 ET and PET Rates Used in the Derivation of ET Estimates for the WRFS 4-8 4-3 4-4 Average PET Rates for Groundwater Discharge Areas and Scaled ET Rates 4-12 4-5 Groundwater ET Estimates for the WRFS 4-12 5-1 5-2 Comparison of Estimates of Precipitation Recharge Volumes, in Acre-Feet per Year, for Basins in the WRFS 5-12 Comparison of Recharge Volumes, in Acre-Feet per Year, for Basins 5-3 Comparison of Precipitation and Recharge Estimates for Cave, Dry Lake, 5-4 and Delamar Valleys..... 5-17 Estimated Groundwater Budget for the WRFS 5-19 5-5 6-1 6-2 6-3 6-4 7-1 Groundwater Budget Components for Cave, Dry Lake, and Delamar Valleys 7-4 C.1-1 Mean Annual Estimates of Spring Discharge for Three Basins of the WRFS..... C-2 C.1-2 Miscellaneous Discharge Measurements of Selected Springs in the WRFS C-4

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

vii



TABLES (CONTINUED)

NUMB	ER TITLE PAGE
C.1-3	Annual Discharge of Spring Gaging Stations in the WRFS
D.2-1	Water Level Data for Selected Wells and Springs
D.2-2	Discharge Data for Selected Springs in White River ValleyD-9
D.3-1	Selected Wells Located in Lower Moapa Valley and Black Mountains Area near the Muddy River
D.3-2	Hydraulic Gradient Calculations in Lower Moapa Valley and Black Mountains Area
E.2-1	Data for Maxey-Eakin Power Function E-6
E.2-2	Maximum Recharge Efficiency Reported E-8
E.3-1	Range of Uncertainty for Selected Budget Components E-12

viii

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

ACRONYMS

BARCAS	Basin and Range Carbonate Aquifer System
BARCASS	Basin and Range Carbonate Aquifer System Study
BCM	Basin Characterization Model
BLM	Bureau of Land Management
DCNR	Department of Conservation and Natural Resources
DEM	digital elevation model
DRI	Desert Research Institute
ET	evapotranspiration
GAP	Gap Analyses Project
GIS	geographic information system
HA	hydrographic area
LVVWD	Las Vegas Valley Water District
MVWD	Moapa Valley Water District
NA	not applicable, not available
NAD83	North American Datum of 1983
NAVD88	North American Vertical Datum of 1988
NCDC	National Climatic Data Center
NDWR	Nevada Division of Water Resources
NLCD	National Land Cover Data
NRCS	Natural Resources Conservation Service
PET	potential evapotranspiration
PRISM	Precipitation-elevation Regression on Independent Slopes Model
RASA	Regional Aquifer-System Analysis
SNOTEL	SNOwpack TELemetry
SNWA	Southern Nevada Water Authority
USAF	U.S. Air Force
USFS	U.S. Forest Service
USGS	U.S. Geological Survey
UTM	Universal Transverse Mercator
WRCC	Western Regional Climate Center
WRFS	White River Flow System
WY	Water Year

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

ABBREVIATIONS

afy	acre-feet per year
amsl	above mean sea level
bgs	below ground surface
cfs	cubic feet per second
ft	foot
gpd	gallons per day
gpm	gallons per minute
km	kilometer
in.	inch
m	meter
mi	mi
yr	year



1.0 INTRODUCTION

The purpose of this report is to present a regional assessment of the precipitation, natural recharge, groundwater discharge, and interbasin flow of the White River Flow System (WRFS) of eastern Nevada (Figure 1-1). This regional assessment was completed in support of water-right hearings related to Southern Nevada Water Authority (SNWA), applications 53987 through 53992, inclusive, in Cave (hydrographic area [HA] 180), Dry Lake (HA 181) and Delamar (HA 182) valleys. Cave, Dry Lake, and Delamar valleys are also referred to as the "project basins" in this report. In Nevada, a hydrographic area is delineated based on topography, has a HA number, and is named after the valley it represents. Even though the groundwater basins underlying the HAs may have different boundaries than the HAs, these are generally unknown. Thus, for the purpose of deriving groundwater budgets, the groundwater basins are assumed to coincide with the HAs. Furthermore, in this report, the terms "HA," "basin," and "valley" are used interchangeably to refer to the individual groundwater basins.

1.1 Background

A great amount of new information has been produced during the more than 40 years since the U.S. Geologic Survey (USGS) Groundwater - Reconnaissance Series Reports for Cave, Dry Lake, and Delamar valleys were first published (Eakin, 1962; 1963a). This information includes new meteorologic, hydrologic, and geologic data and many studies resulting in improved estimates of precipitation distribution and groundwater discharge. This new information warrants a re-evaluation of the groundwater resources for these basins; however, to do so requires adopting a regional approach (i.e., flow system) with respect to precipitation and natural recharge.

The Reconnaissance Series Reports (Eakin, 1962 and 1963a) contain estimates of the precipitation and natural recharge for Cave, Dry Lake, and Delamar valleys, based on a version of the precipitation map of Nevada prepared by Hardman (1936) and the standard recharge efficiencies developed by Maxey and Eakin (1949). To form a new assessment of the natural recharge of these basins requires incorporating an improved precipitation distribution. To do so requires, in turn, that new recharge efficiencies be developed so that when applied to the precipitation distribution, the resultant recharge estimate is balanced by estimates of groundwater discharge. For the three basins addressed in this report, the principal component of groundwater discharge is subsurface outflow, which is one of the more uncertain components of any groundwater budget. To minimize the effects of this uncertainty, a regional approach was used to derive the recharge distribution. This ensures that the most uncertain component (i.e., subsurface outflow) of the groundwater discharge represents the smallest portion of the total groundwater discharge. This is the case for the WRFS, as most of the groundwater discharge occurs as groundwater evapotranspiration (ET) rather than subsurface underflow.









Figure 1-1

Location of the White River Flow System, Cave, Dry Lake, and Delamar Valleys

Section 1.0

1.2 Purpose and Scope

Estimates of precipitation and groundwater discharge for the WRFS were evaluated as part of a regional water-resource assessment to derive natural recharge estimates and groundwater budgets for basins comprising the WRFS. New recharge efficiencies were derived as part of this regional assessment based on the improved estimates of groundwater ET and precipitation distribution. These efficiencies were applied to the precipitation distribution for the purpose of deriving a distribution of natural recharge from which basin recharge estimates could be computed and used in developing basin groundwater budgets.

The WRFS was originally defined by Eakin (1966) as a flow system starting from Long Valley in the north, and ending with the Muddy River Springs Area in the south (Figure 1-1). Eakin's (1966) flow system consisted of thirteen hydrographic areas including Long Valley, Jakes Valley, Cave Valley, White River Valley, Garden Valley, Coal Valley, Pahroc Valley, Pahranagat Valley, Dry Lake Valley, Delamar Valley, Kane Springs Valley, Coyote Spring Valley, and the Muddy River Springs area (also called Upper Moapa Valley). In Eakin's version of the WRFS, the Muddy River Springs Area was assumed to be the terminal discharge area for the flow system. However, more recent studies (Harrill et al., 1988, Figures 3 and 6, Sheet 1; LVVWD, 2001, p. 1-2, Figure 6-1, Errata sheets; Buqo, 2004, p. 57) interpret the Colorado River (now Lake Mead) to be the terminal discharge location of the WRFS. This means that, in addition to the spring water leaving the Muddy River Springs Area along the Muddy River, some groundwater flows past and/or around the Muddy River Springs to basins located downgradient, and discharges to the Colorado River. These basins, which are the California Wash, Lower Moapa Valley, Hidden Valley, Garnet Valley, and Black Mountains Area, were added to Eakin's (1966) interpretation of the WRFS (Figure 1-1). This augmented version of the WRFS was used in this study.

1.3 Document Organization

This report contains eight sections and five appendices. A brief description of each section and appendix follows:

- Section 1.0 is this introduction describing the background, purpose, and scope of the water-resource assessment presented in this report.
- Section 2.0 provides summary descriptions of the previous hydrologic investigations related to the three project basins and to the WRFS.
- Section 3.0 describes the technical approach applied in the data compilation and analysis activities used to complete this water-resources assessment.
- Section 4.0 describes the compilation, processing, and analysis of the available meteorologic, hydrologic, and geologic data used to estimate basin precipitation, groundwater ET, and groundwater outflow.

Contion	4	0	
Section	1	.0	

SE ROA 11722

1-3



- Section 5.0 describes the process and results of the analysis used to derive an average annual recharge distribution for the WRFS, and the resultant groundwater budgets for the basins within the WRFS.
- Section 6.0 presents the results of an uncertainty analysis of the recharge distribution and groundwater budgets derived in Section 5.0, based on the uncertainties associated with estimates of selected variables used in the analysis.
- Section 7.0 provides a summary and the conclusions of this water-resources assessment.
- Section 8.0 provides a list of references cited in this report.
- Appendix A provides information about precipitation stations used in this study.
- Appendix B provides descriptions of the groundwater discharge areas of the WRFS and a compilation of ET rates for the study area and vicinity from the literature.
- Appendix C provides spring discharge measurements for selected springs in White River and Pahranagat valleys, and the Muddy River Springs Area.
- Appendix D provides estimates of interbasin flow across selected basin boundaries used as constraints and/or initial conditions in the recharge calculations.
- Appendix E presents details about the calculations performed to derive the recharge efficiencies, the recharge distribution and the basin budgets, including descriptions of the Excel files.

1-4

Section 1.0
2.0 PREVIOUS INVESTIGATIONS

Numerous studies related to Cave, Dry Lake, and Delamar valleys and adjacent basins have been conducted since the late 1930s. These studies have included water-resource investigations, geologic and hydrogeologic investigations, recharge and discharge estimations, and other hydrologic studies. The following sections describe, in general terms, the purpose, objectives, data sources, and conclusions for the most pertinent of these studies.

2.1 USGS/DCNR Reconnaissance Investigations

During the late 1940s to the early 1980s, the USGS, in cooperation with the State of Nevada Department of Conservation and Natural Resources (DCNR), completed reconnaissance-level hydrologic evaluations or reevaluations of nearly every valley in Nevada. The purpose of the studies was to provide a general appraisal of the groundwater resources as quickly as possible (Eakin, 1963a, p. 2). The results of these studies are presented in two report series, the USGS Water Resources Bulletin Series and the USGS Ground-Water Resources - Reconnaissance Series.

The USGS Ground-Water Resources - Reconnaissance Report Series describes estimates of groundwater recharge, groundwater discharge, and perennial yields for each valley or area in Nevada. The reconnaissance-level hydrologic evaluations of Cave Valley were reported by Eakin (1962), and those for Dry Lake and Delamar valleys were reported by Eakin (1963a). The recharge estimates presented in these reports were based on a method developed by Maxey and Eakin (1949), using the water balance method and an empirical relationship between precipitation and groundwater recharge. A more detailed description of the Maxey and Eakin (1949) is provided in Section 3.2.2.1.

Average annual groundwater recharge estimates on the order of 14,000, 5,000, and 1,200 afy were reported for Cave, Dry Lake, and Delamar valleys, respectively (Eakin, 1962, p. 12; 1963a, p. 17). Eakin (1962, p. 12; 1963a, p. 18) also states that groundwater ET in these valleys probably does not exceed a few hundred acre-feet per year and that discharge by subsurface underflow may closely approach the estimated average annual groundwater recharge. In the perennial yield section of these reports, it is stated that the perennial yield for these basins would be limited based, in part, on presumed constraints related to the economic feasibility of pumping sufficient volumes of groundwater to lower water levels so that groundwater discharge is captured.

Eakin (1966) evaluated the potential for a regional interbasin groundwater system based on the results of the reconnaissance groundwater appraisals. The Paleozoic carbonate rocks are reported as the principal means of transmitting groundwater in the interbasin regional system. The probable means of discharge was reported as underflow from Cave Valley to White River Valley (14,000 afy), Dry Lake Valley to Delamar Valley (5,000 afy), and Delamar Valley to southern Pahranagat Valley (6,000 afy) (Eakin, 1966, pp. 261 and 265).

Section 2.0

SE ROA 11724



2.2 Nevada Water Resources Report No. 3

In this report, Scott et al. (1971) provided a hydrologic summary for the 232 HAs in Nevada. The report was one in a series of reports prepared for the development of a Nevada State Water Plan. The report, titled "Nevada's Water Resources," included precipitation, surface water runoff, and groundwater recharge data in addition to perennial and system yield data for each of the HAs. The sources of most of the data reported by Scott et al. (1971) are the Bulletins and Reconnaissance Series reports. Eakin (1962, 1963a, and 1966) are cited for Cave, Dry Lake, and Delamar valleys; the results of these studies are summarized, consistent with those reported previously, in a series of tables. Perennial yields are reported as 2,000 afy (Cave Valley), 2,500 afy (Dry Lake Valley), and 3,000 afy (Delamar Valley) (Scott et al., 1971, p. 23).

2.3 Desert Institute Publication No. 41054

Under contract with the Las Vegas Valley Water District (LVVWD), the Desert Research Institute (DRI) produced DRI Publication No. 41054 (Hess and Mifflin, 1978) describing a feasibility study about groundwater development from the carbonate aquifers of Nevada. The report includes a compilation of information on the carbonate rocks in eastern and southern Nevada and a plan for further studies. Information compiled includes data on large carbonate springs, carbonate caves, wildcat oil and gas wells, and mines found in the study area. Based on this information, Hess and Mifflin (1978) concluded that thick sequences of carbonate rocks occur beneath the alluvial basins and much of the volcanic rocks of eastern Nevada. Information gathered during the drilling of deep petroleum holes indicate that intervals of cavernous carbonate rock exist to depths greater than 10,000 ft (Hess and Mifflin, 1978, p. 19).

2.4 U.S. Air Force MX Missile Siting Program

In the late 1970s and early 1980s, hydrogeologic evaluations were conducted in support of the U.S. Air Force (USAF) MX Missile Siting Program. The purpose of these evaluations was to assess the potential for water-supply development in 36 HAs in the Great Basin region that were proposed for the deployment of the MX missile system. This program involved literature reviews, exploratory drilling, aquifer testing, groundwater sampling for water-quality analysis, and the development of groundwater flow models to assist in predicting potential impacts of pumping in some valleys. Development of groundwater from the basin-fill aquifers was the preferred water-supply source; most of the valleys had adequate unappropriated groundwater supplies in the basin fill to meet estimated MX water requirements (Ertec Western, Inc., 1981b, p. 32). These studies are documented in several reports by Ertec Western, Inc. (1981a through e) and summarized by Bunch and Harrill (1984).

Aquifer testing of the basin-fill aquifer was performed in Cave, Delamar, and Dry Lake valleys and of the carbonate-rock aquifer in Dry Lake Valley (Ertec Western, Inc., 1981c, pp. 37, 52, and 60). The potential for the development of the carbonate-rock aquifer was rated as high for Cave and Dry Lake Valleys and as moderate for Delamar Valley (Ertec Western, Inc., 1981b, p. 58). A moderate potential for development of the carbonate-rock aquifer in Delamar Valley was attributed to the lack of thick sequences of carbonate rocks in the valley. Ertec Western, Inc. (1981c, pp. 52–53) stated that Delamar Valley is in a known groundwater flow regime and that areas of high-density faulting were observed.

Section 2.0

2.5 Great Basin Regional Aquifer System Analysis

The Great Basin Regional Aquifer System Analysis (RASA) study took place in the 1980s and was undertaken as part of the USGS national RASA program. The main purpose of this study was to develop a geologic, hydrologic, and geochemical framework for regional aquifer systems nationwide (Harrill et al., 1988) to support effective future groundwater management (Harrill and Prudic, 1998). The results of the RASA study are described in nearly 60 reports including the USGS Professional Paper 1409 series (1409A through H). The first report of this series, Harrill and Prudic (1998), provides a summary of the RASA study.

Previous RASA reports for the Great Basin region include the USGS Hydrologic Atlas HA-694 series which consists of the following documents:

- USGS Hydrologic Atlas HA-694-A (Plume and Carlton, 1988): This atlas includes one map with a detailed description. The major hydrogeologic units present in the Great Basin region are delineated and described. Hydrogeologic units that constitute regional aquifers or act as barriers to the movement of groundwater are also identified and discussed.
- USGS Hydrologic Atlas HA-694-B (Thomas et al., 1986): This atlas includes two maps with detailed descriptions. The first map (Sheet 1) shows the general distribution of water levels in the basin-fill deposits of the Great Basin Region. The second map (Sheet 2) is a depiction of the potentiometric surface in consolidated rocks of the carbonate rock province.
- USGS Hydrologic Atlas HA-694-C (Harrill et al., 1988): This atlas contains one map with a
 detailed explanation. The maps depicts interpretations of general groundwater flow directions
 in basin-fill deposits, and flow directions and magnitudes of interbasin flow in consolidated
 rocks. Flow through permeable consolidated rock from Cave, Dry Lake, and Delamar valleys
 to White River, Delamar, and Pahranagat valleys, respectively, are reported. The recharge
 rates and interbasin flow rates reported by Harrill et al. (1988) are consistent with those
 reported by Eakin (1962, 1963a, and 1966).

Another RASA report for the Great Basin region is that of Prudic et al. (1995). This report presents a conceptual evaluation of regional groundwater flow based on a numerical groundwater flow model. The two-layer model was used to simulate the concept of numerous shallow flow regions superimposed upon fewer deep-flow regions (Prudic, 1995, p. 95). Cave, Dry Lake, and Delamar valleys were modeled as part of the White River flow subregion of the Colorado River deep-flow region (Prudic, 1995, p. 66). Groundwater flow in the White River subregion was generally north to south in both model layers, and more groundwater flow was simulated in the lower layer than in any other in the study area (Prudic, 1995, p. 71). The Reconnaissance Reports provided the basic estimates of recharge and discharge (Prudic, 1995, p. 6).

2.6 LVVWD Cooperative Water Project

LVVWD published a series of 19 reports in support of groundwater applications filed with the Nevada State Engineer's Office in 1989 and as part of its Cooperative Water Project (CWP). These hydrologic-assessment reports were prepared for valleys in which groundwater applications were

Section 2.0

SE ROA 11726



filed at the time. The objectives of these reports were (1) to compile, review, and interpret available geologic, hydrologic, and land use data to determine hydrologic characteristics and (2) to develop computer models that simulate steady-state groundwater flow. Brothers et al. (1993) describes the hydrologic assessment conducted for Cave Valley, and Brothers et al. (1996) describes the hydrologic assessment of Dry Lake and Delamar valleys. Two-layer models were developed to simulate groundwater flow in the alluvium and the consolidated rocks. The models were developed to replicate as closely as possible the hydrologic basin budget as defined by Eakin (1962) for Cave Valley and Eakin (1963a) for Dry Lake and Delamar valleys, while attempting to match the existing groundwater levels (Brothers et al., 1993 and 1996).

2.7 USGS Open-File Report 96-469

In cooperation with LVVWD, USGS collected and compiled groundwater data from the 254 wells drilled as part of the MX Missile-Siting Program. Tumbusch and Schaefer (1996) present these data in tables and the well locations on a map. The data include water level, well depth, and site status measured between 1980 and 1996 for four large-diameter production and observation wells in Cave Valley, nine in Dry Lake Valley, and three in Delamar Valley. Groundwater data for two small-diameter monitor wells in Cave Valley were also reported. In addition, digital recording devices were installed in select wells, one in Cave Valley and three in Dry Lake Valley, to monitor detailed changes in groundwater levels. Tumbusch and Schaefer (1996) also report the location, site information, and status of the recording devices.

2.8 LVVWD Coyote Spring Valley Report

In support of groundwater applications in Coyote Spring Valley, LVVWD performed a study to estimate a water-resource budget for the WRFS and Meadow Valley Flow System (LVVWD, 2001). The goal of this study was to define the regional hydrology and geology of the flow systems, estimate their groundwater and surface water budgets, and simulate potential pumping impacts to the regional groundwater and surface water resources related to the LVVWD applications. Additional precipitation data, geologic investigations, geochemistry, and interpretive techniques not previously available were included in the evaluation (LVVWD, 2001, p. 1-1).

The water-resource budget for the area shows a groundwater recharge estimate of 324,000 afy and groundwater ET estimate of about 275,000 afy, leaving about 49,000 afy of discharge from the two flow systems into Lake Mead (LVVWD, 2001, p. 6-2). Estimates of groundwater recharge and ET volumes were greater than those of previous investigators. A modified Maxey-Eakin approach was used to estimate recharge volumes of 20,000, 13,000, and 5,000 afy for Cave, Dry Lake, and Delamar valleys, respectively (LVVWD, 2001, p. 4-25). Volumes of predevelopment ET and groundwater outflow were reported as 5,000 and 15,000 afy (Cave Valley); 1,000 and 12,000 afy (Dry Lake Valley); and 1,000 and 16,000 afy (Delamar Valley), respectively (LVVWD, 2001, pp. 4-37, 4-38, 4-40).

2.9 Desert Research Institute Publication No. 41169

Thomas et al. (2001) describe an approach to evaluate the recharge and ET rates estimated by LVVWD (2001) using a deuterium mass-balance model for water budgets of the regional flow systems in southern Nevada. For this model, groundwater flow paths from Cave Valley to Pahroc Valley, Dry Lake Valley to Delamar Valley, and Delamar Valley to Coyote Valley were evaluated (Thomas et al., 2001, p. 10). A deuterium mass-balance model was produced that was consistent with deuterium values measured in samples collected from the carbonate rock aquifers of the WRFS (Thomas et al., 2001, p. 33). Thomas et al. (2001) pointed out that the model was non-unique and that the modeling results only show that the deuterium data are consistent with the proposed rates and groundwater sources, but does not necessarily confirm that the estimates are correct.

2.10 BARCAS Study

The Basin and Range Carbonate Aquifer System (BARCAS) study is a project created as a result of federal legislation enacted in December 2004 (Section 131 of the Lincoln County Conservation, Recreation, and Development Act of 2004, Public Law 108-424). The purpose of the BARCAS study is to investigate the groundwater flow system underlying White Pine County, Nevada, or Lincoln County, Nevada, and adjacent areas in Utah. The Lincoln County Land Act (U.S. Congress, 2004) states that:

The study shall-(a) focus on a review of existing data and may include new data; (b) determine the approximate volume of water stored in the aquifers in those areas; (c) determine the hydrogeologic and other controls that govern the discharge and recharge of each aquifer system; and (e) develop maps at consistent scale depicting aquifer systems and the recharge and discharge areas of such systems.

The BARCAS study was conducted by USGS, DRI, and the Utah State Engineer's Office, in collaboration with Bureau of Land Management (BLM), and included several tasks including data compilation and collection, identification of aquifers present and their hydrologic properties, delineation of groundwater recharge and discharge areas and rates, synthesis of all information into a conceptual model of the regional flow systems including the water budget, and reporting of the study results.

The draft phase of the BARCAS study has been completed, and the reports are currently available in draft form. The main reports are as follows:

- Geophysical framework investigations (Watt and Ponce, 2007)
- Recharge distribution (Flint and Flint, 2007)
- Mapping of ET units (Smith et al., 2007)
- ET rate measurements (Moreo et al., 2007)
- · Water-level surface maps (Wilson, 2007)
- Delineated irrigated acreage (Welborn and Moreo, 2007)
- · Summary report (Welch and Bright, 2007).

Section 2.0

SE ROA 11728



The BARCAS study area includes four of the WRFS basins. They are Long, Jakes, White River and Cave valleys.

Section 2.0

3.0 TECHNICAL APPROACH

This section includes a review of the various methods used to estimate precipitation recharge for the WRFS, and the rationale for the selection of the groundwater balance method used in this analysis. Also included is a description of the specific technical approach applied in the data compilation and analysis activities used to derive estimates of natural recharge, groundwater discharge, and interbasin flow to assess the water resources of the WRFS.

3.1 Objective

The objective of the water-resource assessment was to estimate the components of the groundwater budget of the WRFS and each of its basins. The estimated budget components represent predevelopment, long-term mean annual conditions, and include precipitation recharge, groundwater discharge by ET, and interbasin inflow and outflow. For such conditions, groundwater budgets may be derived by equating the total recharge to the system (precipitation recharge + inflow) to the total discharge from the system (groundwater ET + outflow). Given that precipitation recharge cannot be measured directly, it is estimated as a function of the other components of the budget. The objective of the approach is to derive a relationship between recharge efficiencies and precipitation that yields a balanced water budget for the WRFS. The solution is constrained by a predefined relationship and conditions on interbasin flow where known.

3.2 Review and Selection of Recharge Method

Relevant methods of estimation of groundwater recharge are described, followed by the rationale for the method selected.

3.2.1 Recharge Methods

Recharge methods that have been used to estimate basin-scale groundwater recharge in the WRFS and vicinity are based on the law of conservation of mass and may be categorized as follows:

- Groundwater balance methods
- · Soil water-balance methods
- Chloride mass-balance method.

Brief descriptions of the three methods and their implementation by various authors are provided in the following sections.

Section 3.0

SE ROA 11730



3.2.2 Groundwater Balance Methods

The groundwater balance method is applied to a groundwater basin, usually under estimated predevelopement steady-state conditions, to derive an estimate of the basin's recharge volume. This volume is calculated as the difference between the total volume of groundwater discharge (i.e., groundwater ET plus subsurface outflow) and the volume of subsurface inflow. A prominent groundwater balance method developed and applied to basins in Nevada is the Maxey-Eakin method (Maxey and Eakin, 1949). This method and its derivatives are summarized, followed by brief descriptions of selected other groundwater balance methods used elsewhere.

3.2.2.1 Maxey-Eakin Method and Derivatives

The Maxey-Eakin method (Maxey and Eakin, 1949; Eakin et al., 1951) was designed to estimate groundwater recharge from precipitation for hydrographic areas of Nevada. Estimates of groundwater recharge based on the Maxey-Eakin method were published by the Nevada Department of Conservation and Natural Resources in the Ground-Water Resources - Reconnaissance Report Series and the Water Resource Bulletins from the mid-1940s through the mid-1970s. Maxey-Eakin methods include the "standard" method and modified versions.

The standard Maxey-Eakin method (Maxey and Eakin, 1949) is based on a precipitation map developed by Hardman (1936). This map, which is described in more detail in Section 4.1.2.1, delineates six precipitation zones, ranging from 0 to over 20 in. of precipitation per year. Using this map, the five precipitation zones above 8 in. in a given hydrographic area are identified as recharge areas. The acreage for each precipitation zone is then measured and multiplied by its average precipitation rate. The resulting precipitation volume is then multiplied by the recharge efficiency for the zone (i.e., the percentage of precipitation that becomes groundwater recharge). The resulting recharge volumes are then summed to yield an estimate of the total recharge volume from precipitation for that hydrographic area, including recharge by direct infiltration and infiltration of runoff. The standard Maxey-Eakin efficiencies were derived by balancing the recharge volume to estimates of discharge volume for 13 basins in Nevada (Maxey and Eakin, 1949). The standard Maxey-Eakin method is not designed to provide a realistic spatial distribution. It does, however, provide first-order approximations of basin recharge volumes (Avon and Durbin, 1994). Also, during the Kane Spring Valley water-rights hearing, it was recognized that the standard Maxey-Eakin efficiencies should only be used with the Hardman precipitation map (Nevada State Engineer, 2007a pp. 12 and 13; Nevada State Engineer, 2007b, pp. 29 and 30).

One major modification of the standard Maxey-Eakin method involves the use of altitude zones on a topographic map to approximate the precipitation zones and calculate their areas. Examples of this variation of the Maxey-Eakin method (1949) are the ones used for Cave, Dry Lake, and Delamar valleys in the Reconnaissance Reports (Eakin 1962; 1963a). Other investigators used variations of the standard Maxey-Eakin method by modifying the precipitation and the recharge efficiencies. They include D'Agnese et al. (1997, pp. 52, 53, and 55), Hevesi et al. (2002, p. 27), Donovan and Katzer (2000, p. 1142), Berger (2000, p. 18), the LVVWD (2001, p. 4-26), Dixon and Katzer (2002, p. 36-43), and Katzer and Donovan (2003, p. 44). Of particular interest are Hevesi et al. (2002), Donovan and Katzer (2000), and Wilson and Guan (2004), who converted the recharge efficiency

Section 3.0

step function, defined in the standard Maxey-Eakin method, to similar power functions expressing recharge as a continuous function of precipitation.

3.2.2.2 Other Water Balance-Based Methods

Other selected methods of estimating recharge from precipitation were applied in Idaho and India. They include studies by Contor (2004) and Kumar and Seethapathi (2002), who used power functions to describe the relationship between recharge and precipitation.

Contor (2004)

In support of the Eastern Snake Plain Aquifer Model Enhancement Project in Idaho, Contor (2004) adapted a relationship used by Rich (1951) to describe a basin's total yield. Contor (2004) simplified the relationship to represent recharge on non-irrigated lands as a function of precipitation as follows:

Recharge =
$$K \times Precipitation^{N}$$
 (Eq. 3-1)

where,

K = the empirical slope parameter

N = the empirical exponent

Considering that recharge cannot physically be greater than precipitation, the slope of the recharge-precipitation relationship should never be greater than one. At the point at which recharge equals precipitation, the exponential relationship is replaced by a straight line having a slope of 1. Furthermore, for a given relationship, the area between the 1 to 1 straight line extended to zero and the exponential curve represents the portion of precipitation that does not become recharge. This represents the water that is stored in the soil or lost to potential evapotranspiration (PET).

Kumar and Seethapathi (2002)

Kumar and Seethapathi (2002) derived an empirical relationship to estimate groundwater recharge from rainfall for the Upper Ganga Canal command area using a seasonal groundwater balance spanning over several seasons: 1972-73 to 1983-84. They found that recharge increases with rainfall in a non-linear fashion. The recharge efficiencies they calculated for monsoon season ranged between 0.05 to 0.19. Kumar and Seethapathi (2002) then derived an empirical relationship between recharge and rainfall by fitting the estimated values of recharge and the values of rainfall using the non-linear regression method. The corresponding equation is as follows:

$$R = 0.63(P - 15.28)^{0.76}$$
(Eq. 3-2)

where,

R = the groundwater recharge from rainfall in monsoon season (in.)

P = the mean rainfall in monsoon season (in.)

Section 3.0

SE ROA 11732



The term 15.28 in. represents the magnitude of rainfall below which recharge does not occur. This equation is similar to that of Contor (2004), except that it assumes that recharge only occurs above a certain level of precipitation (15.28 in.).

Anderson et al. (1992)

As part of a RASA study for alluvial basins located in southwest Arizona and vicinity, Anderson et al. (1992) developed an equation for estimating mountain-front recharge as a function of precipitation using the water budget method. Their approach consisted of developing a relationship between the mean annual mountain-front recharge volume and the total annual volume of precipitation for several watersheds when the precipitation is greater than 8 in./yr. They initiated the equation starting with the available data points and adjusted its coefficients until both the individual basin budgets and the total budget for all basins balanced. Their data points include recharge values derived from models and a few basin estimates. They derived two forms of the equation, one using the total precipitation volume for the basins, and a second one using only the precipitation volume for precipitation rates larger than 8 in. The 8 in. cutoff was arbitrary but yielded a better fit to the data, and was therefore, used to estimate recharge for the study area. The volume of precipitation below 8 in. was attributed to losses to soil-moisture deficits and ET. The resulting equation is as follows:

$$Log R = -1.40 + 0.98 \times Log P$$
 (Eq. 3-3)

where,

Recharge volume in afy R =Precipitation in afy (where P > 8 in./yr). P

Taking the inverse log of each side of the equation yields a power function similar to that of Contor (2004):

$$R = 0.042P^{0.98}$$
 (Eq. 3-4)

where,

3-4

= Mean direct mountain front recharge in afy R P

Mean annual precipitation in afy

Soil Water-Balance Methods 3.2.3

This soil water-balance method focuses on the processes that control net infiltration through the uppermost layers of surficial materials in a given area. These processes include precipitation, snow melt, snow accumulation, and soil water storage. The soil water balance must be successively applied to relatively short time periods for the method to yield reasonable estimates of recharge over long periods of time. This method is used to calculate the amount of water available at each time step, for potential recharge and/or runoff, or water to be carried to the next time step. The soil water balance

Section 3.0

has been implemented to estimate basin recharge in Nevada using two models: the INFIL code and the Basin Characterization Model (BCM). Brief descriptions of these two models follow.

3.2.3.1 INFIL Code

The INFIL code (Scanlon et al., 2006) calculates potential groundwater recharge including volume and distribution. INFIL uses a 24-hour (daily) time step to allow for an accurate simulation of the snow accumulation and melting processes. INFIL calculates runoff and distributes it to a stream network, and simulates recharge through the streambeds. An INFIL model may be calibrated to the available streamflow data, or measurements of soil-moisture content. INFIL has been extensively used to estimate recharge for the Yucca Mountain Project (Flint et al., 2002; BSC, 2004). The INFIL code is the most detailed and refined of all the methods discussed here. However, this method requires a tremendous amount of data and intensive computational resources.

3.2.3.2 Basin Characterization Model

BCM is a geographic information system-based, (GIS) distributed-parameter, water-balance method of estimating basin recharge using monthly climatic boundary conditions (Flint and Flint, 2007). BCM is in essence a simplification of the INFIL code. BCM differs from the INFIL code in that monthly climate data are used, only one soil layer is used, and surface-water is not an explicit parameter. BCM simulates total potential recharge, which is a combination of in-place recharge and runoff. One major shortcoming is the non-inclusion of streams in the code, which renders calibration very difficult, if not impossible. Despite its more simplified form, BCM also requires large amounts of data and significant computational resources. The BCM code has been used by Flint et al. (2004) to derive recharge estimates for basins in the Desert Southwest, and by (Flint and Flint, 2007) for basins in the BARCAS study area. Neither of these BCM models was calibrated to observed conditions, namely groundwater discharge.

3.2.4 Chloride Mass-Balance Method

The chloride mass-balance method is used to estimate groundwater recharge in arid and semiarid environments. Data requirements include the concentration of chloride in pore water, estimates of precipitation volume, and total chloride input to the system from the atmosphere including dry fallout and precipitation. This method has been used in several studies to derive reconnaissance estimates of natural recharge for desert basins in Nevada including those by Dettinger (1989), Maurer and Berger (1997), Russell and Minor (2002), and Mizell et al. (2007).

3.2.5 Selection of Recharge Method

The groundwater-balance method (Section 3.2.2) was selected as the approach for deriving an estimate of natural recharge for the WRFS because it provides the best means of deriving a calibrated recharge estimate by incorporating measurable budget components, namely groundwater ET. As applied in this analysis, a Precipitation-elevation Regression on Independent Slopes Model (PRISM) precipitation distribution was used and a trial-and-error approach taken to solve for a relationship between recharge efficiencies and precipitation that would produce a total recharge estimate that

SE ROA 11734



balances with the total groundwater discharge estimate for the flow system. This method also provided an initial spatial distribution of recharge based on the spatial distribution of precipitation.

The application of the standard Maxey-Eakin efficiencies to the PRISM precipitation distribution was rejected because this would result in an over estimation of the natural recharge. As the Maxey-Eakin method (Maxey and Eakin, 1949) is an empirically-derived solution calibrated to the Reconnaissance Reports Series estimates of groundwater discharge using the Hardman (1936) precipitation map, it was concluded that the standard Maxey-Eakin recharge efficiencies should only be applied to the Hardman precipitation map (Nevada State Engineer, 2007a, pp. 12 and 13; Nevada State Engineer, 2007b, pp. 29 and 30). If new recharge estimates are to be derived based on updated precipitation maps, the appropriate recharge efficiencies should be obtained using the water-balance method and updated estimates of groundwater discharge.

The BCM method offers the best approach for distributing recharge as it uses spatial distributions for the parameter datasets considered in the model. However, the recharge values derived by this method are unconstrained by observed data. In addition, BCM-based models have never been calibrated to the groundwater budget components for a basin. For this reason, the BCM method was not used in this analysis.

The chloride mass-balance method offers an alternative method of deriving recharge estimates, but the method was rejected as it was concluded that the chloride concentration observations for precipitation and groundwater are too few to represent the spatial variability of these input parameters for the WRFS.

Other investigators have derived empirical relationships between precipitation and recharge. While these were not used explicitly in this analysis, they do offer insight into alternate ways to relate recharge to precipitation. For this analysis, it was concluded that a nonlinear equation would best reflect the relationship between recharge efficiency and precipitation. Many investigators have expressed this relationship as a non-linear function rather than a linear relationship or a step-wise function such as that defined by the standard Maxey-Eakin method. Hevesi et al. (2002) modified the standard Maxey-Eakin method by developing an exponential curve to define recharge as a continuous function of precipitation. This approach was also used by Donovan and Katzer (2000, p. 1142) and Wilson and Guan (2004, p. 8). Contor (2004, p. 3) adapted a relationship used by Rich (1951) to describe a basin's total yield and simplified the relationship to represent recharge on non-irrigated lands as a nonlinear function.

3.3 Technical Approach using the Groundwater-Balance Method

The groundwater-balance method is based on fundamental concepts of hydrology and is a standard approach for estimating unknown groundwater-budget components (i.e., recharge) using estimates of other budget components that can be measured within the bounds of reasonable uncertainty (i.e., precipitation, ET). The water-balance method is more reliable for closed groundwater basins (i.e., basins with no boundary flow). It can, however, yield reasonable results when applied to a basin or flow system where the amount of boundary flow is known to be small relative to the total budget for that basin or flow system. For this assessment, the water-balance method was used to derive a relationship between recharge efficiencies and depth of precipitation that yielded a balanced

3-6

Section 3.0

groundwater budget for the flow system. This was completed using a spatial distribution of precipitation and estimates of groundwater ET and outflow.

3.3.1 Method Application

In applying the groundwater-balance method, predevelopment conditions were assumed in which the total groundwater inflow to the system (i.e., precipitation recharge and boundary inflow) equals the total groundwater outflow from the system (i.e., groundwater ET and boundary outflow). Thus:

$$R_{T} = ET_{ow} + Outflow - Inflow \qquad (Eq. 3-5)$$

where,

 R_T = the total recharge from precipitation (afy) ET_{gw} = the total groundwater ET (afy) Outflow = the total groundwater outflow (afy) Inflow = the total groundwater inflow (afy)

For a given precipitation zone where precipitation is assumed to be effective and contribute to recharge, the volume of recharge may be expressed as a function of precipitation and recharge efficiency as follows:

$$R_{Z} = P_{Z} \times Eff_{Z} \times A_{Z}$$
 (Eq. 3-6)

where,

The recharge efficiencies can vary depending on the depth of precipitation that can be partitioned into any number of zones based on the precipitation distribution. The total recharge for the flow system, R_T , is the sum of all recharge that falls within each zone (Z_i) calculated by Equation 3-6. The total recharge is equal to the total discharge, and can be represented by the following:

$$R_{T} = \sum_{i=1}^{n} (P_{Z_{i}} \times Eff_{Z_{i}} \times A_{Z_{i}}) = ET_{gw} + Outflow - Inflow$$
(Eq. 3-7)

The recharge efficiencies for each zone can be calculated using estimated values for ET_{gw} . Outflow, Inflow, P_z, A_z, and the following general assumptions:

Section 3.0

SE ROA 11736



- Eff_Z is between 0 and 1 fraction of precipitation;
- Eff_z increases with an increase in the depth of precipitation;
- Outflow and Inflow are greater than or equal to zero.

3.3.2 Method Implementation

The process followed to derive a recharge distribution for the WRFS and its basins is outlined in Figure 3-1. The groundwater-balance method was used to derive the recharge efficiencies and was implemented using the Excel Solver.

The Excel Solver is designed to find optimal solutions to numerical problems such as the one defined by Equation 3-7, in which the main variables requiring a solution are the recharge efficiencies for the WRFS. The solver finds an optimal value for a formula in one cell of the worksheet called the target cell. The solver works with a group of cells that are related, either directly or indirectly, to the formula in the target cell. Values in these cells are called parameters, which the solver adjusts to produce the desired result defined by the target cell formula. Constraints can be added to restrict the values of the parameters the solver uses. Worksheet cells containing the constraints can refer to other cells that affect the target cell.

To initialize and run the solver, the target cell, parameters, constraints, and initial parameter estimates must first be defined (Figure 3-1). To do so, the necessary data were compiled and analyzed to estimate values for these inputs. This included compiling the following information and data:

- Precipitation station data and a spatial distribution for the area encompassing the WRFS
- Hydrologic data for the WRFS to assist in defining constraints
- · Geologic information and data to assist in identifying likely areas of interbasin flow
- · Maps delineating groundwater discharge areas and ET classes
- · ET rate data, PET data, and a PET distribution encompassing the groundwater discharge areas
- · Estimates of boundary inflow and outflow
- Digital elevation model (DEM).

3-8

Additional information on the Excel Solver, including examples, can be found in the Excel 2003 version help menu and/or the "Microsoft Excel 2003 Bible" (Walkenbach, 2003; pp. 518 to 530).

Upon completion of the data compilation and analysis, estimates for the target cell, parameters and constraints were defined, and the solver initialized. The estimated groundwater ET volume was set to be the target. Recharge efficiencies were the primary parameters. Selected volumes of interbasin flow were identified as secondary parameters, some of which were constrained. Constraints on interbasin flow volumes were developed using spring and streamflow gage records, and groundwater elevations. The resultant recharge efficiencies (part of solution) were applied to the precipitation distribution to derive a recharge distribution. Using the recharge distribution, recharge estimates for each basin were calculated and used in developing the groundwater budget for the entire flow system and each of its basins. A groundwater budget for a given basin was defined by the estimates of recharge, groundwater inflow, groundwater ET, and groundwater outflow. Groundwater inflow is equal to the sum of the outflows of the contributing basins, while the groundwater outflow is the difference between the sum of the recharge and inflow, and the groundwater ET.

Section 3.0



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System



This Page Intentionally Left Blank

3-10

Section 3.0

SE ROA 11739

JA_4501

4.0 DATA COMPILATION AND ANALYSIS

This section describes the data compilation and analysis effort conducted in support of implementing the technical approach described in Section 3.0. Meteorologic, hydrologic, and geologic data and information were compiled and analyzed to derive estimates and work products required for the water-balance method applied in this water-resource assessment, and are presented in the following sections.

4.1 Precipitation Data Analysis

Precipitation is the main, if not the sole, source of groundwater recharge to the WRFS and is one of the important inputs needed to derive the groundwater recharge estimates for the flow system. This section describes the specific objective of the precipitation data analysis, the review of relevant methods and the method selection, and presents the results of the analysis.

4.1.1 Objective

The objective of the precipitation data compilation and analysis was to derive a spatial distribution of precipitation for the WRFS that is representative of long-term mean annual conditions. This distribution is needed for the purposes of deriving estimates of groundwater ET and a spatial distribution of recharge.

4.1.2 Review and Selection of Precipitation Method

Relevant methods of estimating precipitation are described, followed by the rationale for the method selected.

4.1.2.1 Methods

Methods to derive precipitation distributions in Nevada include maps developed in the past by (Hardman, 1936, 1962, and 1965), the PRISM method (Daly et al., 1994, 1997 and 1998), and other precipitation-altitude regression models that have been developed for local-scale analyses.

Precipitation Mapping

A precipitation map for the entire state of Nevada was first produced by Hardman (1936). The map was hand-drawn and was compiled from U.S. Weather Bureau records, USGS topographic maps, and Nevada Experiment Station forage type maps. Six precipitation zones, ranging from 0 to over 20 in. of precipitation per year, were defined as follows: less than 5 in., 5 to 8 in., 8 to 12 in., 12 to 15 in., 15 to 20 in., and over 20 in. The map, also published in Hardman and Mason (1949, p. 10), was later

Section 4.0

SE ROA 11740



updated by Hardman (1962 and 1965) to provide more detail in the southern Nevada area. This updated map was later revised by the Nevada State Engineer (Scott et al., 1971). The first Hardman map (1936) was used to develop and apply the Maxey-Eakin method (Maxey and Eakin, 1949) for estimating annual groundwater recharge in Nevada.

PRISM

4-2

The PRISM method is a mapping model designed to generate spatial distributions of precipitation, temperature, or other climate variables (Daly et al., 1994, 1997, and 1998). The PRISM method was developed through a partnership between the Natural Resources Conservation Service (NRCS) National Water and Climate Center and the Prism Group at Oregon State University. PRISM is claimed as "*the production of the best and, in many cases, the first high-spatial-resolution climate map products for the United States*" (http://www.wcc.nrcs.usda.gov/climate/prism.html). The basic information used in the model consists of point measurements of precipitation, temperature, or other climate variables made at observation stations. In addition to point data, the PRISM model utilizes a DEM, and incorporates natural phenomena affecting the spatial distributions including rain shadows, coastal effects, and temperature inversions. Products include grids of precipitation, temperature, and other climate variables for a given period of time; and monthly, yearly, and event-based climatic parameters. The PRISM model has been used to estimate precipitation for each state of the United States, including Nevada (Daly et al., 1998).

Precipitation-Altitude Regression Models

Precipitation-altitude regression models have been developed for many areas of Nevada to derive precipitation distributions and to estimate precipitation volumes, including but not limited to those of Quiring (1965), Daly et al. (1994), Maurer and Halford (2004), and SNWA (2006). In most instances, these relationships were developed to verify estimates derived by other methods or to revise historical estimates by incorporating new data, information, and technology. The regression models are defined by equations that express the relationship between precipitation and altitude based on station data compiled from various sources. The equation can be applied to DEMs to derive precipitation distributions. Typically, the regression models are developed for local-scale (e.g., hydrographic area) analyses where the data density is relatively high.

4.1.2.2 Selection of Precipitation Method

The PRISM method (Daly et al., 1994, 1997, and 1998) was selected for the precipitation distribution of the WRFS. The PRISM datasets including those of precipitation represent state-of-the art distributions at the basin and regional scales and are widely-accepted and recognized world-wide as the highest-quality spatial climate datasets currently available. The U.S. Department of Agriculture, for example, adopted PRISM as their official climatological dataset. Precipitation distributions developed using the PRISM method utilize modern tools and incorporate more recent data and information not reflected in previous mapping efforts. More recent data include additional stations and precipitation records. Development of a separate precipitation-regression model was rejected as it would be limited by the data availability and would be duplicative of the PRISM work.

Section 4.0

4.1.3 Precipitation Data

Both precipitation station data and PRISM precipitation grids were compiled for this analysis. A PRISM precipitation grid was used to provide a spatial distribution of precipitation, while the station data were used to assess the validity of the PRISM-based distribution.

Precipitation station data were compiled, and the mean annual value for the period of record was derived for each station. Summary data for the stations located within and near the WRFS are listed in Table A.1-1 and presented in Figure 4-1.

Several PRISM precipitation grids are available on the Internet (such as at http://www.ocs. oregonstate.edu/prism/products/). The grids include precipitation distributions for various periods of time and at different resolutions. All PRISM precipitation grids are based on the 1-degree DEM grid at website http://edcwww.cr.usgs.gov/Webglis/glisbin/guide.pl/glis/hyper/guide/usgs_dem. The station precipitation data used to generate the PRISM maps are not available to the public. The most recent normal grid (800-m 1971 to 2000 normals, Version 2; May 3, 2007) is considered to be the best-quality product to date and was deemed appropriate for the purposes of this regional water-resource assessment. Figure 4-2 depicts the PRISM precipitation distribution within the WRFS and vicinity and a relative comparison of the model fit at the precipitation station locations.

4.2 Regional Groundwater Discharge

Within the WRFS, regional groundwater discharge occurs in the form of ET, spring discharge, and inter-basin outflow. This section describes the data compilation and analysis that was performed to derive estimates of regional groundwater discharge for the WRFS, including the delineation of regional groundwater discharge areas and ET-class zones, compilation of ET and PET rate data, and derivation of a spatial distribution of PET. Additionally, spring discharge data were compiled for the purpose of constraining the lower bound of groundwater discharge within these areas. The underlying assumption is that the spring discharge is fully consumed by ET within the discharge area or becomes part of the underflow or surface water discharge to the next downgradient basin.

4.2.1 Groundwater ET

Groundwater ET within the WRFS is the largest source of groundwater discharge. As such, it plays a prominent role in the water-balance approach used in this water-resource assessment. The objective and technical approach of the data compilation and analysis used to derive groundwater ET estimates are described in the following sections.

4.2.1.1 Objective

The objective of the ET data compilation and analysis was to derive an estimate of the total groundwater ET volume for the WRFS for use as the target value in the water balance method to estimate groundwater recharge from precipitation.

Section 4.0

SE ROA 11742





Note: See Table A.1-1 for names of precipitation stations.

Figure 4-1 Location of Precipitation Stations and Mean Annual Precipitation within the WRFS and Vicinity

4-4

Section 4.0



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

PRISM Precipitation Distribution within the WRFS and Vicinity, and Percent Difference between PRISM and Precipitation-Station Data

Section 4.0

4-5



4.2.1.2 Technical Approach

To estimate the mean annual groundwater ET for the WRFS, it was necessary to estimate mean annual groundwater ET for each of its basins that are representative of predevelopment conditions. Deriving estimates of groundwater ET for the groundwater discharge areas of the WRFS required the compilation and analysis of ET and PET rate data from previous investigations, analysis of satellite imagery, and field investigation. This resulted in interim work products that were used to derive the estimates and included maps delineating the boundaries of the regional groundwater discharge areas and the ET classes within those boundaries, and a spatial distribution of PET. This work included the following steps:

- 1. Delineate regional groundwater discharge areas and define ET classes.
- 2. Compile ET rate data from published literature for defined ET classes.
- 3. Compile PET data and derive spatial distribution of PET.
- 4. Scale ET rates using PET distribution.
- 5. Calculate volume of precipitation falling within ET areas.
- 6. Derive groundwater ET estimates.

4.2.1.3 Regional Groundwater Discharge Areas and ET Classes

Delineation of the regional groundwater discharge areas and ET classes for the WRFS relied upon maps and information from previous investigations including the USGS Reconnaissance Reports Series and Water Resource Bulletins, Nichols (2000), and LVVWD (1994 and 2001). Boundaries were refined through the use of satellite imagery, aerial photography, field investigation, and assumptions used to account for irrigated lands that were nonexistent under predevelopment conditions. The Southwest Regional Gap Analyses Project (GAP) data (USGS, 2004) and the National Land Cover Data (NLCD, 1992) were used in instances where other information was insufficient. The primary use of the GAP and NLCD data was to aid in the identification of playas located in the study area. They were also used to confirm other datasets or flag otherwise undetected potential phreatophytic areas for further investigation. Boundaries of many, but not all of the discharge areas were field-verified during the summer of 2004 and modified as needed using high-resolution global positioning system equipment. Five ET classes were defined and delineated using the Normalized Difference Vegetation Index and Landsat 7 Thematic Mapper 2002 satellite imagery. These classes are presented in Table 4-1.

The groundwater ET areas were then adjusted by determining where agricultural land had likely displaced what would otherwise be phreatophytes under predevelopment conditions. For these agricultural lands, the acreage was reclassified to the ET class reflected in the natural vegetation surrounding the fields. Agricultural lands outside the groundwater discharge areas were not included in the total acreages. Maps depicting the resulting groundwater discharge areas and the spatial distribution of the ET classes were prepared for the HAs that have groundwater discharge areas, including:

- Long Valley (HA 175)
- Jakes Valley (HA 174)
- White River Valley (HA 207)

4-6

Section 4.0

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

ET Class	Description				
Open Water	Bodies of open water fed by groundwater sources (springs, seeps, etc.)				
Wetland/Meadow	Area of shallow groundwater near bodies of open water consisting of wetland vegetation, marshland, or dense meadows.				
Dense Meadow/Riparian	Area dominated by dense meadow and woodland vegetation; includes riparian corridors consisting of salt cedar, desert willows, cottonwood, and mesquite trees with underlying shrubs and grasses; includes areas near open water consisting of dense wetland vegetation.				
Medium-Density Phreatophytes	Area dominated by desert shrubland vegetation, including mixed stands of medium density greasewood, rabbit brush, or other phreatophyte species.				
Bare Soil/Low-Density Phreatophytes	Area dominated by bare soil and low- to moderate-density desert shrubland, including greasewood, rabbit brush, or other phreatophytic species.				

Table 4-1 ET Classification

- Cave Valley (HA 180)
- Garden Valley (HA 172)
- Pahranagat Valley (HA 209)
- Muddy River Springs Area (Upper Moapa Valley) (HA 219)
- California Wash (HA 218)
- Lower Moapa Valley (HA 220)
- · Black Mountains Area (HA 215).

These maps and descriptions of the groundwater discharge areas of each valley are provided in Appendix B. The total acreage delineated for the groundwater discharge areas and each ET class are provided in Table 4-2, listed by HA name and from the northern portion of the flow system to the south.

4.2.1.4 Compilation of ET and PET Rate Data

A literature survey for annual ET rates measured in and near the WRFS was conducted, and the rates were compiled into the dataset presented in Table B.1-2. These ET rates were reviewed to select a single most appropriate rate for each of the five ET-unit classes defined for this analysis. The rate selected for each class was based on the similarity between the vegetation types and climate. The selected ET rates are presented Table 4-3, and the measurement sites are depicted in Figure 4-3.

Except for the "Bare Soil/Low Density Vegetation" class, the rates from the ET measurement sites were scaled to the groundwater discharge areas of the WRFS using values of PET extracted from a spatial distribution of PET. A total ET rate of 1.00 ft/yr was assumed for the "Bare Soil/Low Density Vegetation" class for all such areas in the WRFS based on the average rates for sparse and medium dense desert shrubland classes reported in Welch and Bright (2007). The PET distribution was derived from a linear regression model of PET versus latitude Y and altitude Z expressed by the following equation:

PET = 104.3531 - 0.82922734Y - 0.004186006Z	(Eq. 4-1)
---	-----------

4-7

Section 4.0



HA Name	Open Water	Wetland/ Meadow	Dense Meadow/ Riparian	Medium- Density Phreatophytes	Bare Soil/ Low-Density Phreatophytes	Total ^a	
Long Valley		131		10,723	6,739	17,594	
Jakes Valley	44	346		621	4	971	
Cave Valley		1,084		9,651	5,914	16,648	
White River Valley	313	27,958		87,786	28,636	144,692	
Garden Valley			929		21	950	
Pahranagat Valley	1,296	ىك	7,101	÷	260	8,658	
Muddy River Springs Area		<u>2</u> 2	1,785	+*	166	1,951	
California Wash	4		1,342		37	1,379	
Lower Moapa Valley	137		6,900		496	7,533	
Black Mountains Area	1.0	47	409		54	464	

Table 4-2 Groundwater Discharge Areas in Acres, by ET Class for the WRFS

"Values were not rounded, to allow calculations to be tracked during analysis and reporting.

Table 4-3	
ET and PET Rates Used in the Derivation of ET	Estimates for the WRFS

ET-Unit Class	Selected ET Site	Description of Selected ET Site	Total ET or Evaporation Rate (ft/yr)	Source	PET Rate Extracted from PET Grid (ft/yr)
Open Water	Peterson Reservoir	Open water supplied by spring discharge.	8.60	Reiner et al., 2002, Table 4, p. 29	5.42
Wetland/ Meadow	SPV-3	Wetland/meadow land cover surrounding riparlan corridors throughout the project area.	2.25	Moreo et al., 2007, Table 7, p. 20	4.00
Dense Meadow/ Riparian	Springdale	Area dominated by dense meadow and woodland vegetation, primarily trees, meadow and marsh grasses, or mixed trees, shrubs, and grasses; trees, desert ash and cottonwood, with some desert willow and mesquite; water table typically ranging from above land surface to about 20 ft-bgs; soil wet to moist.	3.30	Reiner et al., 2002, Table 4, p. 29	4,84
Medium Density Phreatophytes	WRV-1, WRV-2, and SPV-2	Medium-density phreatophytes, greasewood and shrubs like sagebrush.	1.03ª	Moreo et al., 2007, Table 7, p. 20	4,13
Bare Soil/Low- Density Phreatophytes	Long-Term Mean Rate/ BARCASS Region	Areas within the phreatophytic boundaries that exhibited ground cover densities of less than 20% were considered to be either bare soil or having sparse vegetation cover.	1.00 ⁶	Welch and Bright, 2007, Figure 27, p. 58	4.11

⁸Average ET from WRV-1, WRV-2 and SPV-2

^bThis value is the average of the area-weighted average annual ET for ET Units 7 and 8 reported in Welch and Bright (2007)

Section 4.0



Figure 4-3 ET Measurement Sites within the WRFS and Vicinity



SE ROA 11748



where,

PET = potential ET (in./yr) Y = latitude (NAD, 1983) Z = altitude in ft-amsl (NAVD, 1988)

The regression model was derived using estimates of PET reported in McCurdy and Albright (2004). McCurdy and Albright (2004) calculated PET for existing meteorological stations within the WRFS and vicinity using the Kimberly-Penman method (Wright, 1982) and the Hargreaves-Saman (1985) equation. Using GIS gridding operations, the spatial distribution of PET was derived by applying Equation 4-1 to a 100-m resolution DEM based on the USGS 30-m DEM (USGS, 2001). The meteorological station locations and their estimated PET values are depicted in Figure 4-4 with the resultant PET distribution that was derived for the WRFS.

Several measurements of ET rates are available for the study area and vicinity (Table 4-3). Prior to assigning these rates to the ET classes in the study area, they were adjusted to the local conditions. The adjustment consisted of scaling them by multiplying their values by the ratio of the average PET for a given groundwater discharge area and the PET at the measurement site. The values used to calculate this ratio were extracted from the PET distribution grid. The resultant scaled ET rates are presented in Table 4-4.

4.2.1.5 Derivation of Groundwater ET Estimates

Estimates of the annual groundwater discharge for regional discharge areas within the WRFS were derived using the acreages delineated for each of the ET classes, scaled ET rates and the PRISM precipitation grid described in Section 4.1. For a given basin, the total ET for each ET class was calculated by multiplying the acreage by the scaled ET rate. The groundwater ET for each class was calculated by subtracting the volume of precipitation from the total ET. The total groundwater discharge by ET was calculated by summing the groundwater discharge volumes for each class. The estimates of groundwater discharge by ET for each basin are presented in Table 4-5, which also includes the estimates derived by other investigators for comparison.

With the exception of two basins, White River Valley and the Muddy River Springs Area, the estimated groundwater ET volumes compare reasonably well with estimates reported in the Reconnaissance Series Reports (Table 4-5). For White River Valley, the estimated groundwater ET volume of 67,342 afy falls within the range of previous estimates: 37,000 afy (Eakin, 1966; p. 261; Scott et al., 1971, p. 50) and 79,560 afy (LVVWD, 2001; p. 4-35). The large range can be attributed to the different acreages delineated for the groundwater discharge area. These are listed for each basin in Table B.1-1 in Appendix B. The estimate reported by Eakin (1966) is comparatively low, in part, because the delineation of the groundwater discharge area did not include the large area of low-density shrubland that is observed today (Maxey and Eakin, 1949 p. 44). By contrast, all of the more recent studies have included this area, or part of this area. For White River Valley, the total acreage delineated for the groundwater discharge area from about 36,000 (Maxey and Eakin, 1949, p. 44) to 178,000 acres (Smith et al., 2007, p. 15, tbl. 4). The acreage delineated by this study is about 145,000 acres which falls within the range of the three previous estimates. Depth to groundwater is an important a hydrologic factor influencing the distribution of phreatophytes, and

Section 4.0



Figure 4-4

Meteorological Station Locations and Distribution of PET within the WRFS

```
Section 4.0
```

SE ROA 11750



Southern Nevada Water Authority - Water Resources Division

HA Name	Average PET Rate for GW Discharge Area	Open Water	Wetland Meadow	Dense Meadow/ Riparian	Medium Dense Phreatophytes	Bare Soil/Low Density Phreatophytes
Long Valley	3.82		2.14		1.18	1.00
Jakes Valley	3.61		2.02		1.12	1.00
Cave Valley	3.87		2.17		1.03	1.00
White River Valley	4.18	6.63	2.34		1.04	1.00
Garden Valley	4.13	τ.		2.82		1.00
Pahranagat Valley	4.84	7.68		3.30		1.00
Muddy River Springs Area	5.52			3.77	- -	1.00
California Wash	5.57			3.80		1.00
Lower Moapa Valley	5.71	9.06		3.89	14	1.00
Black Mountains Area	5.69		1.00	3.88		1.00

Table 4-4	
Average PET Rates for Groundwater Discharg	ge Areas and Scaled ET Rates

Table 4-5 Groundwater ET Estimates for the WRFS

	SN	WA (2007)	Welch and	LVVWD (2001) ^b (afy) 11,000	Eakin (1966)	
HA Name	Acreage	Groundwater ET ^a (afy)	(2007) ^b (afy)		(1968) (afy) 2,200	
Long Valley	17,594	2,952	1,233			
Jakes Valley	971	392	858	600	0	
Cave Valley	16,648	1,285	1,550	4,823	200	
White River Valley 144,69		67,342	76,701	79,560	37,000 ^c	
Garden Valley	950	1,696	4 7	4,608	2,000	
Pahranagat Valley	8,658	28,516		38,369	25,000	
Muddy River Springs Area	1,951	5,989	HH	5,080	2,300	
California Wash	1,379	4,505		5,760	6,700 ^b	
Lower Moapa Valley	7,533	25,311		26,505	24,000 ^b	
Black Mountains Area	464	1,432			1,200	
Total	200,840	139,424		176,305	100,600	

^aValues were not rounded, to allow calculations to be tracked during analysis and reporting. ^bIncludes areas irrigated by groundwater ^cValue based on spring discharge estimates (Eakin, 1966; p. 261)

could be used to refine the extent of the groundwater discharge areas; however, the depth to water is not well defined in this area.

The groundwater ET volume for Muddy River Springs Area ranges between 2,300 afy (Eakin, 1966, p. 261) and 5,989 afy (this study). Although the value derived in this study is the maximum, it is essentially the same estimate as the 5,080 afy derived by LVVWD (2001, p. 4-35, tbl. 4-8).

4.2.2 Spring Discharge

For groundwater discharge areas where gaged springs are present, estimates of groundwater discharge based on the spring discharge records can be used to estimate the components of the groundwater budget. As stated previously, the spring discharge is assumed to ultimately be part of the underflow and/or ET. Therefore, the total groundwater discharge from a given basin cannot be lower than the measured spring discharge.

4.2.2.1 Objective

The objective of the spring data compilation and analysis was to identify springs within groundwater discharge areas of the WRFS that have discharge records reporting mean annual flows or have the necessary records from which mean annual flows can be calculated. This information was not used explicitly in the water-budget calculations.

4.2.2.2 Compilation of Spring Discharge Records

Many regional springs occur within the basins of the WRFS, particularly within the regional groundwater discharge areas of White River and Pahranagat valleys and the Muddy River Springs Area. Spring data for these areas were compiled from the literature. Maps in Appendix B depict the locations of these springs with respect to the groundwater discharge areas in which they occur.

Available spring discharge records were compiled for springs in the following basins: White River Valley, Pahranagat Valley, and the Muddy River Springs Area. These records are presented in Appendix C. A summary of the spring discharges for the three selected basins follows.

Two relatively important springs occur in the Black Mountains Area hydrographic area (Figure B.1-11), and were accounted for in the estimate of underflow to the Colorado River (see Section 4.3).

White River Valley

The groundwater discharge area in White River Valley contains a number of springs. Spring discharge measurements for 18 springs are listed in Table C.1-1. The most significant of these springs are Hot Creek Spring, Arnoldson Spring, Preston Big Spring, Lund Spring, Flag Springs, Butterfield Spring, Nicholas Spring, and Moon River Spring. The total average annual discharge measured at these springs is 47.6 cfs or 34,462 afy. This discharge is assumed to represent the lower bound of groundwater discharge, as there are many springs within the area whose flow is not measured. It is lower than the estimated groundwater ET, 67,342 afy. The fact that the estimated

Section 4.0

SE ROA 11752



volume of groundwater ET is larger than the volume of regional spring flow, is a good indication of the validity of the ET estimate.

Pahranagat Valley

The groundwater discharge area in Pahranagat Valley contains a number of springs. Most notable are the Hiko Springs, Solar Panel Spring, Crystal Springs, Ash Springs, and Brownie Spring. Other smaller springs and seeps occur in the southern portion of the discharge area. Hiko, Crystal, and Ash springs have the most significant discharge and, when combined, produce a total spring discharge of about 35 cfs or about 25,000 afy (Eakin, 1963b, p. 19-20). This value is lower than the estimated volume of groundwater ET, 28,516 afy and is indicative of the validity of the ET estimate.

Muddy River Springs Area

The Muddy River Springs Area contains several large thermal spring groups and seeps. These springs represent the principal source of groundwater discharge in the southern portion of the WRFS and form the headwaters of the Muddy River. The Muddy River near Moapa gage (09416000) measures the combined spring discharge from the Muddy River Springs Area. From 1913 to 1918, the mean daily discharge at this location was 46.8 cfs and the mean annual discharge was approximately 34,000 afy (Wells, 1954, p. 566). From 1914 to 1962, the mean annual discharge was reported (46.5 cfs) as 33,700 afy (Eakin, 1964, p. 14, 15, 16, 18). These measurements account for the flow observed at the gage but not the consumptive uses by the riparian vegetation along the spring channels and river corridor, or by the phreatophytes that likely existed in the area during predevelopment conditions. Eakin (1964) estimated 2,000 to 3,000 afy was being consumed by phreatophytes between the spring area and the gaging station. The analysis conducted for this study estimates that there was approximately 6,000 afy of groundwater ET prior to the extensive groundwater development within the area that exists today. Based on these estimates and the estimated mean annual flow at the Muddy River near Moapa gage, predevelopment groundwater discharge from the Muddy River Springs Area is on the order of 40,000 afy. In this case, the spring flow volume of 34,000 afy is less than the groundwater ET estimate of 6,000 afy, but the total, 40,000 afy, constitutes a lower bound for groundwater inflow to this basin.

4.3 Interbasin Flow

Locations where groundwater flow occurs across basin boundaries were identified based on the prevailing hydrogeology, including lithology and geologic structure. Figure 4-5 depicts their locations using arrows superimposed onto a hydrogeologic map of the WRFS extracted from Dixon et al. (2007). The relative magnitude of the flow volumes is depicted using small or large arrows. Most of the interbasin flow occurs across internal basin boundaries. External interbasin flow occurs at two locations only, as discussed below. Summary descriptions of interbasin flow at the external boundary of the WRFS and through selected internal basin boundaries are also provided below. The detailed descriptions are presented in Appendix D.

4-14

Section 4.0



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Figure 4-5 Potential Locations of Interbasin Flow Based on the Hydrogeologic Framework

Estimates of External Interbasin Flow

As shown in Figure 4-5, significant external boundary flow into and out of the WRFS likely occurs by (1) inflow to the WRFS from Lower Meadow Valley Wash and (2) outflow from the WRFS to the Colorado River (currently Lake Mead).

Groundwater inflow to the WRFS originates from Lower Meadow Valley Wash. This inflow occurs through the underlying valley-fill sediments and carbonate rocks. The magnitude of this flow has been estimated by Rush (1968 p. 26), Kirk and Campana (1990 p. 370, 371, 372); Thomas et al. (1996, p. C36), Prudic et al. (1995 p. D71), and Thomas et al. (1996 p. C36) at 7,000 afy, 5,500 to 9,000 afy, 13,000 afy, and 8,000 afy, respectively. The average volume of about 9,200 afy is used in this study. A portion of this inflow, 4,000 afy, was estimated to enter the Muddy River Springs Area. The remainder, 5,200 afy, likely enters the WRFS at the boundary of California Wash basin.

Reported estimates of boundary outflow from the WRFS to the Colorado River range from 11,100 afy (Rush, 1968, p. 24 and 26) to 49,000 afy (LVVWD, 2001, p. 6-3) and include both subsurface outflow and a groundwater component of the surface water flows reaching the Colorado River. For this analysis, groundwater outflow to the Colorado River is estimated to be at least 25,000 afy based on the following components: (1) groundwater component in Muddy River of 7,000 afy, (2) total groundwater outflow from Black Mountains Area of 2,000 afy, and (3) subsurface outflow of 16,000 afy. The estimate of 25,000 afy used in this study falls within the range of reported estimates.

Estimates of Internal Interbasin Flow Volumes at Selected Locations

Several areas of potential interbasin flow within the boundaries of the WRFS have also been identified and are depicted on Figure 4-5. It is necessary to estimate some of these interbasin flow volumes to constrain the recharge solution and to distribute the resulting volume to the individual basins. Interbasin flow locations were selected based on the availability of sufficient information to derive estimates of the expected volumes or bounding values. The following interbasin flow locations were selected for estimation: (1) the outflow volume from White River Valley to Pahroc Valley, (2) the total groundwater inflow to the Muddy River Springs Area, (3) outflow from Coyote Spring Valley to Hidden Valley, and (4) the volumes of interbasin flow across the boundaries of the three project basins, Cave, Dry Lake, and Delamar valleys. Estimates of flow volumes for items (1) and (2) were obtained from the literature and are provided below. Flow volumes for items (3) and (4) were estimated as part of this study (Appendix D) and are summarized in this section.

Outflow from White River Valley to Pahroc

Groundwater outflow from White River Valley to Pahroc Valley was estimated by Maxey and Eakin (1949, p. 45) to be about 19,000 afy. Because of the uncertainty of the estimate, Maxey and Eakin (1949) also estimated a lower bound for the outflow based on (1) the total spring flow of Hot Creek Spring, at the time 11,000 afy, (2) an estimate of ET between this spring and the southern end of the valley of 4,000 afy, and (3) an estimated streamflow of about 700 afy exiting the valley. The groundwater outflow from the valley was estimated by subtracting from the total spring flow, the sum of the estimated ET between the spring and the southern end of the valley and the estimated streamflow exiting the valley. This calculation yielded a groundwater outflow estimate of 6,300 afy (Maxey and Eakin, 1949, p. 45). Considering that several other unmeasured springs contribute to the

SE ROA 11755

ET in this lower section of the valley, it is likely that total volume is somewhat greater than 6,300 afy; therefore, this value was considered the minimum for this study. The maximum volume of outflow was estimated by Eakin (1966, p. 265) at 40,000 afy.

Total Inflow to the Muddy River Springs Area

Groundwater inflow to the Muddy River Springs Area is interpreted to originate mostly from Coyote Spring Valley, with a relatively smaller volume from Lower Meadow Valley Wash (Figure 4-5). Because most of the groundwater outflow from the Muddy River Springs Area occurs as Muddy River streamflow and groundwater ET, the lower bound of the estimate of total inflow to the area can be made using Muddy River gage records and an estimate of ET above the gage. As described in Section 4.2.2, the mean annual streamflow at the Muddy River near Moapa (09416000) gage is about 34,000 afy (based on estimates of 34,000 afy by Wells [1954, p. 566] and 33,700 afy by Eakin [1964, p. 15, 16]). As part of this study, approximately 6,000 afy of groundwater ET was estimated to occur prior to the extensive development of groundwater within the area. Therefore, the total inflow to the Muddy River Springs Area used in this study is assumed to be larger than or equal to 40,000 afy.

Outflow from Coyote Spring Valley to Hidden Valley

Outflow from Coyote Spring Valley to Hidden Valley is likely to occur within the carbonate-rock aquifer. Although no carbonate wells have been drilled in Hidden Valley, carbonate wells where aquifer tests have been conducted exist in Coyote Spring Valley, Garnet Valley, and California Wash. Using carbonate wells located in these valleys located upgradient and downgradient from Hidden Valley, the outflow from Coyote Spring Valley to Hidden Valley has been estimated to be about 15,000 afy, which may constitute a maximum value. These estimates are comparable to the estimates of 16,000 afy derived by LVVWD (2001, p. 6-3). A detailed estimate is provided in Appendix D.

Interbasin Flow across the Boundaries of Cave, Dry Lake, and Delamar Valleys

The majority of the interbasin flow for Cave, Dry Lake, and Delamar valleys occurs as groundwater outflow from Cave Valley to the southern third of White River Valley and northern Pahroc Valley, and from southern Delamar Valley to northern Coyote Spring Valley and perhaps also to the very southern part of Pahranagat Valley. Also, a small amount of groundwater inflow to northern Dry Lake Valley likely occurs from northern Pahroc Valley. Elsewhere among these basins, interbasin flow is limited by geologic structure and lithology.

Groundwater outflow from Cave Valley is thought to occur on the western side of Cave Valley through Shingle Pass to the southern third of White River Valley, and through the southern portion of the valley to northern Pahroc Valley. Total outflow from Cave Valley is estimated to be about 13,400 afy. The outflow from Cave Valley to White River Valley through Shingle Pass is estimated at 4,000 afy, which is considered to be a near maximum value. The remainder, 9,400 afy, flows from Cave Valley to northern Pahroc Valley. A detailed estimate is provided in Appendix D.

For Dry Lake Valley, interbasin flow occurs in the northern portion of the valley as inflow from northern Pahroc Valley and as outflow to Delamar Valley to the south. Inflow from Pahroc Valley is estimated to be no more than 2,000 afy, while the outflow to Delamar Valley is estimated as the sum of the inflow and natural recharge, or about 17,700 afy. A detailed estimate is provided in Appendix D.

Section 4.0

SE ROA 11756



For Delamar Valley, interbasin flow occurs as inflow from Dry Lake Valley and as outflow to northern Coyote Spring Valley and possibly the very southern part of southern Pahranagat Valley. The outflow is estimated as the sum of the inflow from Dry Lake Valley and the natural recharge derived from within the hydrographic area boundaries of both basins, or about 24,100 afy. Interbasin flow elsewhere among these basins is precluded by geologic structure and lithology, principally the Caliente and Kane Springs Wash caldera complexes to the east and southeast and the North and South Pahroc ranges and associated range-front faults. A detailed estimate is provided in Appendix D.

4-18

Section 4.0

5.0 DATA ANALYSIS AND RESULTS

This section describes the data analysis conducted to derive recharge efficiencies and estimate the groundwater budget for the WRFS and its basins following the approach described in Section 3.0. Descriptions of the data analysis and the results are provided in the following sections. Details about the calculations conducted in support of the analysis including descriptions of the Excel files are provided in Appendix E.

5.1 Data Analysis

Under steady-state conditions, the total recharge for the WRFS is equal to the sum of the estimates of groundwater ET and outflow, less any inflow to the system. To construct groundwater budgets for individual basins within the WRFS, the solver described in Section 3.0 was used in conjunction with the PRISM precipitation grid and boundary flow estimates to calculate recharge efficiencies for 1-in. precipitation intervals and outflow to the Colorado River. The data processing and analyses performed to estimate these efficiencies, including the solver setup, targets, parameters, and constraints, are described in the following sections.

5.1.1 Delineation of Areas of Potential Recharge

For the purpose of this report, areas of potential recharge are defined as areas where most of the in-place recharge occurs and mountain-front runoff is generated. This area of potential recharge is used to estimate the recharge distribution at the basin scale, not at the local scale. For example, the recharge that may result from infiltration of mountain-front runoff is not distributed to the actual areas where it may occur. For a given basin, potential recharge is assumed to occur in all areas of a given basin except (1) the valley floor, (2) groundwater discharge areas, and (3) areas where the depth of precipitation is less than 8 in.

The valley floor of a given basin was delineated for each basin of the WRFS using the USGS DEM (USGS, 2001), and subsequently excluded as an area of potential recharge. Land-surface slopes were calculated using the DEM grid, and the relatively flat areas of each basin were identified and excluded if the slopes were less than 2 percent. Alluvial fans were not included in the delineation of the valley floor areas because they are typically comprised of coarse-grained material that can accommodate recharge if the depth of precipitation is significant (i.e., greater than 8 in.).

It is also assumed that groundwater ET areas are not areas of potential recharge. This is consistent with the calculation of estimated groundwater ET where the precipitation is deducted from the total ET estimate. Removing the precipitation volume necessarily leads to the removal of any recharge that may occur there.

Section 5.0

SE ROA 11758



It is also assumed that a minimum of 8 in. of precipitation is necessary before groundwater recharge may occur. The first 8 in. of precipitation are assumed to satisfy the soil moisture deficit and losses to the atmosphere (ET) (Anderson et al., 1992, p. B33). This assumption is consistent with the Maxey-Eakin method (1949) which assumes that recharge is zero below 8 in, of precipitation. Thus, areas receiving less than 8 in, of precipitation are not considered to be areas of potential recharge.

The resulting potential recharge areas are depicted in Figure 5-1. The union of the valley floors, groundwater discharge areas, and areas receiving less than 8 in. of precipitation are labeled "areas of no recharge," for the purpose of this report. As stated before, direct recharge from precipitation may actually occur in these areas, depending on the local conditions.

5.1.2 Precipitation Data

The PRISM precipitation distribution serves as the basis for the solver calculations and the derivation of the spatial distribution of recharge. The PRISM grid for the WRFS was contoured to generate 1-in. precipitation intervals starting from a minimum depth of 3 in. to the maximum depth occurring within the flow system. Next, the area corresponding to each interval was calculated, and then adjusted as necessary to exclude the "areas of no recharge," as defined in the previous section. For each basin, the adjusted areas were then exported to Excel to create a table containing the precipitation rate and corresponding area for each 1-in. interval within the basin. This table forms the basis of the calculations performed by the solver. A map depicting the resulting potential recharge area and the 1-in. precipitation bands is presented in Figure 5-2.

5.1.3 Solver Target ET_{gw}

In the solver, the target is represented by the estimated value of total groundwater ET for the WRFS. The target cell contains a formula relating groundwater ET to the other components of the budget; this formula is Equation 3-5 rearranged in the following form:

$$ET_{gw} = R_T + Inflow - Outflow$$
 (Eq. 5-1)

where,

ETgw	=	total groundwater ET for the WRFS (afy)
RT	=	total recharge for the WRFS (afy)
Inflow	=	total groundwater inflow to the WRFS from the Meadow Valley Flow System (afy)
Outflow	=	total groundwater outflow from the WRFS to the Colorado River (afy)

Based on the data compilation and analysis presented in Section 4.0, the target value for ET_{gw} and the estimate of inflow are 139,424 and 9,200 afy, respectively.


Figure 5-1 Areas of Potential Recharge within the WRFS

Section 5.0

SE ROA 11760



5-4



Figure 5-2

Precipitation Distribution within Potential Recharge Areas within the WRFS

Section 5.0

SE ROA 11761

5.1.4 Solver Parameters

Parameters represent the flow system variables that require a solution. For this analysis, the primary parameters are the recharge efficiencies and interbasin flow rates for selected basin boundaries of the WRFS. The parameter solutions are determined through an optimization process in which the recharge efficiencies and boundary flows are adjusted within the predefined constraints described in Section 5.1.5, to ensure that the total recharge is equal to the sum of the total groundwater discharge and outflow, less the groundwater inflow (Equation 3-7).

Considering that the solution to the problem depends on many variables but only a few of them are known within reasonable bounds of uncertainty, the solution is non-unique and many possible representations exist. For example, the solver may identify solutions that are mathematically feasible but not reasonable given what is understood about the physical aspects of the flow system. It is, therefore, important to provide reasonable initial estimates for all parameters as was done in this case.

Recharge Efficiencies

Recharge efficiency, by definition, is the ratio of recharge to precipitation. Recharge may be expressed as a function of effective precipitation. For this analysis, recharge is assumed to follow a form of the power function of Kumar and Seethapathi (2002) (see Equation 3-2). The equation is as follows:

$$R = a(P-8)^{b}$$
 (Eq. 5-2)

where,

- R = recharge (in./yr)
- a = power function constant
- b = power function exponent

P = Precipitation (in./yr)

P-8 = Effective precipitation (in./yr)

For this analysis, it is assumed that precipitation contributes to recharge (effective precipitation) starting at 8 in./yr where the effective precipitation and recharge are assumed to be zero. The 8 in./yr threshold value is based on the work of Anderson et al. (1992). The volume of precipitation below 8 in./yr is assumed to account for losses to soil-moisture deficits and ET. The effective precipitation and recharge increases with increasing precipitation. This equation is also similar to that of Contor (2004) and Anderson et al. (1992), except that these authors express recharge as a function of total precipitation rather than effective precipitation.

To calculate the recharge efficiencies for use in Equation 3-7, each side of Equation 5-2 is divided by precipitation, P, to yield the following equation expressing recharge efficiency:

Eff =
$$\frac{[a(P-8)^{b}]}{P}$$
 (Eq. 5-3)

Section 5.0

SE ROA 11762



where,

Eff	=	recharge efficiency or R/P as a fraction
a	-	power function constant
b	=	power function exponent
Р	=	Precipitation (in./yr)
P-8	22	Effective precipitation (in./yr)

Equation 5-3 was used in the solver for direct calculation of the recharge efficiencies. The primary parameters are the coefficients of Equation 5-3 (i.e., the constant a and the exponent b).

Initial estimates for these two primary parameters were derived from a power function derived from the step-function defined by the standard Maxey-Eakin efficiencies. They are as follows:

- Power function constant: $a = 8.0 \times 10^{-5}$
- Power function exponent: b = 3.62

Details on how these initial estimates were derived using the Maxey-Eakin recharge efficiencies are provided in Appendix E.

Interbasin Flow Volumes

Secondary solver parameters are (1) the interbasin flow volume from Coyote Spring Valley (HA 210) to Hidden Valley (HA 217) and (2) the total outflow volume from the WRFS into the Colorado River.

The outflow volumes were assigned their constraints as initial estimates. Initial estimates for these two secondary parameters are as follows:

- Outflow from Coyote Spring Valley to Hidden Valley = 15,000 afy
- Total Outflow from the WRFS = 25,000 afy

5.1.5 Constraints

5-6

Constraints were placed on the coefficients of the power function, the maximum recharge efficiency, and the two locations of underflow used as secondary parameters in the solver. Additional constraints were placed on underflow out of White River Valley and the total inflow to the Muddy River Springs Area. The constraints are as follows:

- Power Function Constant, a is positive
- Power Function Exponent, b is positive
- Outflow from White River Valley is between 6,300 and 40,000 afy
- Outflow from Coyote Spring Valley to Hidden Valley is less than or equal to 15,000 afy
- Total inflow to Muddy River Springs Area is greater than or equal to 40,000 afy
- Total outflow is greater or equal to 25,000 afy
- Maximum recharge efficiency is less than or equal to 0.63.

Section 5.0

SE ROA 11763

The constraint imposed on the recharge efficiency was derived from a literature review of maximum recharge efficiencies estimated for the region of the study area. The maximum reported value of 63 percent or 0.63 was selected as the constraint value (Table E.2-2). The details are located in Appendix E.

Additional constraints imposed on interbasin flow volumes are as follows:

- · Inflow to California Wash from Lower Meadow Valley Wash is equal to 5,200 afy
- Inflow to Muddy River Springs Area from Lower Meadow Valley Wash is equal to 4,000 afy.

The information supporting the constraints on flow volumes is described in Section 4.3. A summary of the interbasin flow constraints applied in the solver is presented in Figure 5-3.

5.2 Solution Process

The solution process includes two major steps: (1) use the Excel Solver to derive a recharge distribution, and (2) use the resulting recharge distribution to derive basin budgets.

The solver was used to inversely solve for the recharge efficiencies of the WRFS and the selected underflow volumes using the target ET estimates, the parameters, and the constraints described in Section 5.1.5. Because there is more than one unknown parameter, the derived solution is not unique. To converge to a solution, the solver uses successive values of all parameters while seeking a solution. Values of the primary parameters and the power function coefficients, a and b, are used to calculate recharge for each 1-in. precipitation band. This recharge value is then divided by precipitation to obtain a recharge efficiency, which in turn, is used to calculate recharge volumes. The calculated recharge volumes and values of the secondary parameters are tested in the budget. Once the calculated recharge volumes yield a total groundwater ET value that matches the target value, a solution is reached. The final values of all parameters are part of the solution and are used to estimate the final recharge distribution.

The process of deriving a groundwater budget for each basin in the WRFS is conducted in Excel, but outside of the solver. The solver actually provides a recharge distribution for each basin in the WRFS. Assuming that each basin is under predevelopment steady-state conditions, a groundwater budget may be derived for each basin. The process starts from the most up-gradient basin of the WRFS (Long Valley), and ends at the southernmost basins (Lower Moapa Valley and Black Mountains Area). For each basin, any inflow to the basin is added to the estimate of basin recharge, and the basin's groundwater ET value is then subtracted. The remainder is the outflow to the next contiguous basin(s) located downgradient. For basins having more than one outflow boundary, independent estimates of selected outflow boundaries were made. Such is the case for two basins: Cave Valley and Pahroc Valley. Outflow from Cave Valley and Dry Lake Valley. The estimated outflows are as follows:

Section 5.0

5-7

SE ROA 11764





Figure 5-3 Interbasin Flow Constraints Applied in Solver

-8	Section 5.0
	SE ROA 11765

- Outflow from Cave Valley to White River Valley of 4,000 afy
- Outflow from Cave Valley to Pahroc Valley of 9,400 afy
- Outflow from Pahroc Valley to Dry Lake Valley of 2,000 afy.

These estimates are discussed in Section 4.0, and the details of the estimates of underflow are presented in Appendix D.

5.3 Results and Discussion

The results of the analysis are presented and discussed in the following order: (1) recharge-precipitation relationship, (2) recharge efficiencies, (3) recharge distribution, and (4) basin groundwater budgets.

5.3.1 Recharge-Precipitation Relationship

The power function computed by the solver is a recharge-precipitation relationship. The corresponding equation is as follows:

$$R = 0.0026(P-8)^{3.0515}$$
 (Eq. 5-4)

A graph of this relationship is presented in Figure 5-4 which also includes similar relationships and point recharge estimates derived by others for comparison. In addition to precipitation, recharge also depends on soil types and thicknesses, rock types, and climate. Thus, the mathematical relationship between recharge rates and precipitation rates is not uniquely defined and may vary depending on location and the physical characteristics of the terrain. However, as can be seen from Figure 5-4, all relationships shown exhibit similar trends: lower recharge rates at lower precipitation rates, and higher recharge rates at higher precipitation rates. The differences among these relationships could be due to differences in actual precipitation and geology, or different interpretations. A discussion of the relationships shown in Figure 5-4 follows.

The relationship derived by SNWA (green line in Figure 5-4) compares relatively well with the equation reported by Faybishenko (dark purple line) and the data points reported by Davisson and Rose (2000) and Lichty and McKinley (1995). Faybishenko (2007) utilized methods that estimated recharge at the local scale using different types of information. Their estimates constitute point recharge data. The relationship derived by Faybishenko (2007) is a power function of recharge and precipitation similar to the one derived in this analysis. The recharge rates were estimated using precipitation data and PET rates calculated with the Priestley and Taylor formula for selected meteorological stations located in several states including Nevada. Davisson and Rose (2000) estimated the recharge efficiency at selected field observation points in the Fenner Basin in California in an effort to calibrate the Maxey-Eakin efficiency-precipitation curve to the local conditions. Their methods of estimation included the use of a hydrologic mass-balance method and groundwater age dating. Lichty and McKinley (1995) used methods based on the hydrologic mass balance to estimate recharge at specific sites in Central Nevada. Their estimates are in close agreement with the power function derived by this analysis, and are represented by the two datapoints (red triangles in

Section 5.0

SE ROA 11766





Figure 5-4 Recharge-Precipitation Relationship for the WRFS

Figure 5-4) located on each side of the green curve representing the power function derived by this analysis (green curve in Figure 5-4).

The power function derived by this analysis is less comparable to the relationships derived by Maxey and Eakin (1949), Wilson and Guan (2004), and Maurer and Berger (1997). The Maxey and Eakin (1949) relationship shown in dark blue in Figure 5-4 was derived from the standard recharge efficiencies estimated by these authors for basins in Nevada. The Maxey-Eakin method is deemed to be an appropriate recharge estimation technique at a reconnaissance level for most HAs in Nevada (Avon and Durbin, 1994). However, the Maxey-Eakin relationship may not reflect the physical nature of groundwater recharge nor its distribution due to the overly simplified precipitation map (Hardman, 1936) on which it is based. More representative maps, such as the one developed later by Hardman (1965) using vegetation distribution and more precipitation data, were not available at the time. Though this map (Hardman, 1965) is better than the previous one (Hardman, 1936), it had few data, if any, controlling the interpretation of precipitation at the higher elevations. The relationship developed by Wilson and Guan (2004) (shown in light blue in Figure 5-4) is a power function fit to the precipitation-recharge points derived from the standard Maxey-Eakin efficiencies. That is why it matches the Maxey-Eakin (1949) relationship in Figure 5-4. Maurer and Berger (1997) derived a power function of recharge versus precipitation for the Carson Basin in Nevada, based on the 4-km PRISM precipitation grid for Nevada. The form of their power function is similar to the power

Section 5.0

SE ROA 11767

function derived by Wilson and Guan (2004). While the Maurer and Berger (1997) function was developed more recently than the Maxey-Eakin method (1949) and, therefore, uses better data, it does not provide better estimates of recharge. This limitation it due to its ability to distribute recharge to the appropriate locations due to the coarseness of the 4-km PRISM precipitation grid.

In summary, the power function representing the recharge-precipitation relationship derived for the WRFS by this study is based on more accurate estimates of precipitation distribution and groundwater ET than the functions previously applied to this area. In addition, this power function agrees better with local recharge estimates than the Maxey-Eakin (1949) and the Maurer and Berger (1997) relationships of recharge with precipitation. Because of this, it is concluded that the relationship derived by this study provides a better assessment of the recharge for individual basins of the WRFS than does the Maxey-Eakin method, or variants thereof, used in the Reconnaissance Report Series.

5.3.2 Recharge Efficiencies

The recharge efficiencies derived by this analysis are computed by the solver and can be expressed by the following equation:

Eff =
$$\frac{0.0026(P-8)^{3.0515}}{P}$$
 (Eq. 5-5)

For comparison, Table 5-1 lists the standard Maxey-Eakin precipitation zones and efficiencies as well as the recharge efficiencies derived by this analysis. The efficiency derived by this analysis is higher by 20 percent for the >20 in./yr zone, but is very similar for the middle to lower zones. This places greater volumes of recharge at the higher altitudes, where the depth of precipitation is greater. It was expected that the efficiencies would differ from the standard Maxey-Eakin efficiencies because the precipitation distributions and groundwater ET estimates are significantly different.

Precipitation Zone (in./yr)	Standard Maxey-Eakin (%)	This Study (%)
>20	25	45
15 to 20	15	15
12 to 15	7	4
8 to 12	3	0.4
<8	0	0

Table 5-1 Comparison of Recharge Efficiencies

Recharge efficiencies derived for each 1-in precipitation zone were used to compute recharge estimates for the individual basins comprising the WRFS. This was accomplished by multiplying the

Section 5.0

SE ROA 11768



recharge efficiency by the volume of precipitation calculated for each interval. The recharge values were then summed for each basin. Table 5-2 lists those estimates as well as estimates from previous studies. To check the recharge computation, Equation 5-4 was used in conjunction with the PRISM grid to derive a spatial distribution of recharge.

	- 0.04-	Flint and	Flint et al	. (2004)	110.000	No. 6 also	Eakin (1966
HA Name	This Study ^a	Flint (2007)	Mean Year	Time Series	(2001) ^b	(2000)	and Rush (1968)
Long Valley	19,928	25,000	16,289	13,536	23,000	48,000	10,000
Jakes Valley	12,288	16,000	10,974	8,310	24,000	38,500	17,000
White River Valley	41,065	35,000	34,925	30,759	62,000	- 44	38,000
Cave Valley	14,659	11,000	10,264	9,380	20,000		14,000
Garden Valley	24,818		17,974	15,559	19,000		10,000
Coal Valley	3,857	<u> </u>	3,839	3,110	7,000		2,000
Pahroc Valley	4,507		4,432	4,832	8,000		2,200
Dry Lake Valley	15,667		10,627	11,298	13,000		5,000
Pahranagat Valley	5,507		7,043	7,186	7,000		1,800
Delamar Valley	6,401		7,764	6,404	5,000		1,000
Kane Springs Valley	4,189	44	5,421	6,328	7,000	CAA.	2,600
Coyote Spring Valley	2,128		5,184	5,951	4,000	22	2,000
Muddy River Springs Area	38		12	207	200		Minor
Hidden Valley	42		188	571	300		400
Garnet Valley	96	Щ. щ.	294	1,000	300	1. 14 A. 1	400
California Wash	0		23	652	300		<100
Lower Moapa Valley	33			147	1,000		<50
Black Mountains Area	0		54	1,470	400		<100
Total	155,224		135,307	126,700	201,500		104,650

Table 5-2	
Comparison of Estimates of Precipitation F	Recharge Volumes,
in Acre-Feet per Year, for Basins in	the WRFS

^aValues were only rounded to the nearest "ones" to allow calculations to be tracked during analysis and reporting. ^bTable 6-1, p. 6-3

The recharge for each basin was calculated using the recharge grid derived by Equation 5-4 and depicted in Figure 5-5. These values were then compared to the estimates computed using the efficiencies and precipitation volumes as previously described. The two sets of results were essentially the same, with the difference being about 700 afy for the entire flow system.



Figure 5-5 Distribution of Precipitation Recharge within the WRFS

Section 5.0

SE ROA 11770



5.3.3 Recharge Distribution

The total recharge calculated for the WRFS was 155,224 afy, which exactly matched the sum of the target groundwater ET and outflow, minus the inflow. A discussion of the recharge volumes estimated for the individual basins of the WRFS and the project basins follows.

5.3.3.1 WRFS Basin Recharge Estimates

The estimated basin recharge volumes were compared to those reported in the literature. The recharge volumes of interest are listed in Table 5-2, and are summarized as ranges of reported recharge values by basin in Table 5-3, for comparison with the estimates derived by this study. The total recharge of 155,224 afy estimated for the WRFS by this study falls within the range of previous estimates, which is 104,650 (Eakin, 1966) to 201,500 afy (LVVWD, 2001).

HA Name	This Study	Range of Previously Reported Volumes
Long Valley	19,928	10,000 to 48,000
Jakes Valley	12,288	8,310 to 38,500
White River Valley	41,065	30,759 to 62,133
Cave Valley	14,659	9,380 to 19,595
Garden Valley	24,818	10,000 to 19,153
Coal Valley	3,857	2,000 to 7,002
Pahroc Valley	4,507	2,200 to 7,545
Dry Lake Valley	15,667	5,000 to 13,254
Pahranagat Valley	5,507	1,800 to 7,407
Delamar Valley	6,401	1,000 to 7,764
Kane Springs Valley	4,189	600 to 6,757
Coyote Spring Valley	2,128	2,000 to 5,951
Muddy River Springs Area	38	12 to 237
Hidden Valley	42	188 to 571
Garnet Valley	96	294 to 1,000
California Wash	0	23 to 652
Lower Moapa Valley	33	0 to 1,354
Black Mountains Area	0	54 to 1,470

Table 5-3 Comparison of Recharge Volumes, in Acre-Feet per Year, for Basins of the WRFS

Except for two basins, the estimated recharge volumes for all basins with recharge volumes greater than 1,000 afy, fall within the reported ranges. The two exceptions are Garden and Dry Lake valleys. The recharge estimate for Garden Valley is 24,818 afy; which falls outside the 10,000 to 19,153 afy

range (Eakin, 1966 and LVVWD, 2001). The recharge estimate for Dry Lake Valley, 15,667 afy, is larger than the maximum reported value of 13,254 afy (LVVWD, 2001). The magnitude of these differences is relatively small and has little bearing on the total groundwater budget for the flow system.

5.3.3.2 Recharge in Cave, Dry Lake, and Delamar Valleys

Eakin (1962; 1963a) estimated recharge in Cave, Dry Lake, and Delamar Valleys by applying the standard Maxey-Eakin recharge efficiencies to estimates of precipitation computed using the areas for precipitation zones defined by land-surface elevation contours (p. 11 and p. 16, respectively). As was done in previous Reconnaissance Series reports, it was assumed that selected elevation contours corresponded to the precipitation zones defined by the Hardman (1936) precipitation map. In the case of Eakin (1962; 1963a), an improved topographical base map (1:250,000 scale topographic map) had been developed and was used to calculate the area for each precipitation zone. Based on this new topographical base and oral communications with Hardman, it was assumed that the 6,000 to 7,000 ft altitude interval corresponded to the 8 to 12 in. precipitation zone, the 7,000 to 8,000 ft interval to the 12 to 15 in. zone, the 8,000 to 9,000 interval to the 15 to 20 in. zone, and for altitudes greater than 9,000 ft, to the >20 in. zone. Below altitudes of 6,000 ft, less than eight inches of precipitation was assumed. Revised versions of the Hardman precipitation map were published in 1962 and 1965, in the same time period the Reconnaissance Series Reports for the project basins were published by Eakin (1962; 1963a). The Hardman (1962) map was used as part of this analysis to evaluate the relationship between precipitation and altitude assumed by Eakin (1962; 1963a), and is described in the following discussion.

To evaluate these assumptions, the Hardman (1962) precipitation map was geo-referenced and digitized for the area encompassing Cave, Dry Lake, and Delamar Valleys (Figure 5-6). Using the digitized map and GIS utilities, the area for each precipitation zone was extracted and multiplied by the average precipitation for the corresponding zone to calculate the volume of precipitation. The standard Maxey-Eakin efficiencies were then applied to compute estimates of recharge. These estimates are compared in Table 5-4 to the estimates reported in Eakin (1962; 1963a) and the estimates derived by this study.

The summary data listed in Table 5-4 highlights an important difference between the estimates reported by Eakin (1962; 1963a) and the estimates derived from the Hardman (1962) precipitation map. The areas calculated for each corresponding precipitation zone are significantly different between the two estimates. This difference is due to the fact that the topographic contours do not correspond to the Hardman (1962) precipitation zones as Eakin (1962; 1963a) assumed they would. In some cases, areas representing the 8 to 12 in. precipitation zone are excluded by the 6,000 ft contour, while other areas representing the 12 to 15 in. zone actually fall within the 6,000 to 7,000 ft contour interval. This is illustrated by Figure 5-6 which overlays the topographic contours derived from the USGS 30-m DEM onto the digitized version of the Hardman (1962) precipitation map. These topographic contours were compared to the 1:250,000 scale topographic maps used by Eakin (1962; 1963a), and only very slight differences between the two were found; therefore, their use in evaluating the altitude-precipitation assumptions of Eakin (1962; 1963a) are appropriate. The inconsistencies between the topographic contours and the Hardman (1962) map are manifested in the estimates of precipitation and recharge. For Cave Valley, the recharge based on the Hardman (1962)

SE ROA 11772





Figure 5-6 Digitized Hardman (1962) Precipitation Map for Cave, Dry Lake, and Delamar Valleys

	elamar Valleys
	Dry Lake, and D
le 5-4	timates for Cave,
Tabl	nd Recharge Est
	f Precipitation ar
	Comparison of

			Area (acres)		Pre	cipitation (afy)		œ	techarge (afy)	
HA Name	Precipitation Zone (in.)	Eakin (1962; 1963a)	Hardman (1962)	This Study	Eakin (1962; 1963a)	Hardman (1962)	This Study	Eakin (1962; 1963a)	Hardman (1962)	This Stud
	> 20	3,500	4,655	3,410	6,125	8,146	6,120	1,500	2,037	2,117
	15 to 20	19,500	35,573	46,099	28,470	51,937	64,012	4,300	7,791	7,672
Cave Valley	12 to 15	69,000	129,990	122,302	77,280	145,589	136,231	5,400	10,191	4,742
	8 to 12	114,000	58,998	12,941	94,620	48,968	12,401	2,800	1,469	128
	Total	206,000	229,216	184,752	206,495	254,640	218,764	14,000	21,488	14,659
	> 20	200	1	1	350	1	1	100	1	1
	15 to 20	3,200	3,380	49,434	4,672	4,935	67,122	700	740	7,108
Dry Lake	12 to 15	16,000	74,011	189,766	17,920	82,892	210,695	1,300	5,802	7,196
valley	8 to 12	114,000	98,686	233,143	94,620	81,909	210,006	2,700	2,457	1,363
	Total	133,400	176,077	472,343	117,562	169,736	487,823	5,000	8,999	15,667
	> 20	1	1	1		1	1	i	4	ŕ
	15 to 20	1	,	22,170	ł	1	29,717	ĩ	ī	2,871
Delamar	12 to 15	4,000	22,292	75,122	4,480	24,967	83,539	300	1,748	2,891
valley	8 to12	35,000	45,648	86,183	29,050	37,888	79,816	006	1,137	639
	Total	39,000	67,940	183,475	33,530	62,855	193,072	1,200	2,885	6,401
CDAN	ID TOTAL	378,400	473,233	840,570	357,587	487,231	839,659	20,200	33,372	36,727

Section 5.0



map is about 7,500 afy greater than the value estimated by Eakin (1962), and about 4,000 afy and 1,700 afy greater for Dry Lake and Delamar Valleys, respectively. In total, for the three basins, the difference is about 13,200 afy.

The recharge estimates derived by this study compare more favorably to the estimates derived using the Hardman (1962) precipitation map, with the total recharge for the three basins about 3,400 afy greater for this study. The biggest difference between the two estimates is observed in the distribution of the recharge. In Cave Valley, the recharge derived using the Hardman (1962) precipitation map is about 21,500 afy while the estimate derived by this study is about 14,700 afy, a difference of about 6,800 afy. In Dry Lake and Delamar Valleys, the estimates derived by this study are about 6,700 afy and 3,500 afy greater than the estimates derived from Hardman (1962). The difference between these estimates are mostly due to the spatial distribution of precipitation interpreted by Hardman (1962) and that derived by PRISM (2007). Recharge efficiencies contribute to some of the difference, but have less effect on the results because the Maxey-Eakin efficiencies and the efficiencies derived by this study are very similar for the mid-range precipitation zones (i.e., 12 to 15 in.; 15 to 20 in.) where most of the precipitation occurs in these basins.

In summary, the recharge estimates of Eakin (1962; 1963a) are less than those derived using the Hardman (1962) precipitation map. This difference is significant, and occurs due to the fact that the elevation contours assumed to represent the precipitation zones defined by Hardman (1962) do not match the precipitation map. The estimates derived by this study are comparable to estimates derived using the Hardman (1962) precipitation map. However, the estimates derived using the Hardman (1962) map are greater for Cave Valley, and less for Dry Lake and Delamar Valleys. These differences are mostly due to differences observed in the precipitation distributions; the PRISM precipitation distribution generates more precipitation at the mid-range altitudes for this area than does the Hardman (1962) precipitation map.

5.3.4 Groundwater Budgets

A groundwater budget for the WRFS and its individual basins was constructed using the recharge, groundwater ET, and boundary flow estimates derived by this analysis as described in Section 5.2.

The groundwater budget for the WRFS is presented in Table 5-5. As listed in Table 5-5, all components of the groundwater budget for the WRFS fall within the ranges of values reported in the literature.

The groundwater budget for the WRFS and individual basins is presented in Figure 5-7. Figure 5-7 provides the individual basin budgets, and presents interbasin flow arrows depicting the general direction of flow. The outflow for a given basin is listed as a total, even for basins with more than one outflow arrow. The groundwater budgets for the three project basins are described in the following section:

For Cave Valley, no inflow is presumed to occur based on the lithology and structure comprising the surrounding mountain ranges and the hydraulic head potential across basin boundaries. Recharge is estimated to be about 14,700 afy, while groundwater ET is estimated to be about 1,300 afy. The

Section 5.0 SE ROA 11775

	Latimated ore	unawator Baage	4 34 mid 100 3
Budget Component	Estimated Value (afy)	Reported Range (afy)	Reference
Recharge	155,224	104,650 to 201,500	Eakin, 1966 (min); LVVWD, 2001 (max)
Inflow	9,200	7,000 to 32,000	Rush, 1968 (min) ^a ; LVVWD, 2001 (max)
Groundwater ET	139,424	100,600 to 176,305	Eakin, 1966 (min); LVVWD, 2001 (max)
Outflow	25,000	11,100 to 49,000	Rush, 1968 (min); LVVWD, 2001 (max)

Table 5-5 Estimated Groundwater Budget for the WRFS

^aThis is the minimum inflow reported by Rush (1968); Thomas et al. (1996) report a range of 5,500 to 9,000 afy

difference between the two is assumed to be interbasin outflow to southern White River Valley and northern Pahroc Valley, which is estimated to be 4,000 and 9,400 afy, respectively.

For Dry Lake Valley, interbasin flow occurs in the northern portion of the valley as inflow from northern Pahroc Valley in some amount less than 2,000 afy. Recharge is estimated to be about 15,700 afy. The sum of the inflow and recharge, about 17,700 afy, constitutes the total outflow from Dry Lake Valley to Delamar Valley as there is no other significant source of groundwater discharge elsewhere within the valley.

For Delamar Valley, the recharge is estimated to be about 6,400 afy. Interbasin inflow is equated to the volume of outflow from Dry Lake Valley, or about 17,700 afy. Like Dry Lake Valley, Delamar Valley has no other significant source of groundwater discharge, so the combined estimates of inflow and recharge constitute the total outflow from Delamar Valley to northern Coyote Spring Valley, or about 24,100 afy.





Figure 5-7 Estimated Groundwater Budgets for Basins within the WRFS

Section 5.0 SE ROA 11777

6.0 UNCERTAINTY ANALYSES

Uncertainty analyses were conducted to evaluate the estimated budget components of the WRFS and the three project basins described in Section 5.0. Descriptions of the objectives, approach, are results of the uncertainty analyses are provided, followed by a summary. Analysis details are provided in Appendix E.

6.1 Objective

The objective of the uncertainty analyses is to develop an understanding of the uncertainties associated with estimated variables, including groundwater budget components and recharge efficiencies, and their effect on the calculated recharge efficiencies and final groundwater budgets.

6.2 Approach

The approach consists of considering alternative scenarios to evaluate the effect of the uncertainty associated with the target value and the constraints placed on the most uncertain variables of the base-case solution. The base-case solution in this uncertainty analysis is the solution described in Section 5.0. An additional special scenario was also considered to evaluate the effect of incorporating the BARCASS ET estimates, where available, into the calculations.

For all scenarios, except the last one, the specific steps are as follows:

- 1. Identify the constraints with the largest uncertainty based on the range of estimated values.
- 2. Starting with the Excel base-case solution file, create an uncertainty file by replacing the appropriate variable with the low or high value.
- 3. Execute the solver to derive a new solution and basin groundwater budgets.
- 4. Compare the results of the new solution to the results of the base-case solution.

Comparisons between the base-case and new solutions were made to determine the effects of the uncertainty with respect to the following parameters: (1) recharge efficiencies, (2) the WRFS budget components, and (3) the budget components for Cave, Dry Lake, and Delamar valleys. The effects on selected interbasin flow volumes were also examined in some cases.

Constrained variables involved in estimating the recharge distribution for the WRFS are as follows:

Groundwater ET estimates for each basin in the WRFS

Section 6.0

SE ROA 11778



Southern Nevada Water Authority - Water Resources Division

- Total inflow to the WRFS from the Meadow Valley Wash flow system
- Total outflow from the WRFS to the Colorado River
- Maximum recharge efficiency

Based on the groundwater ET estimates presented in Section 4.0 and summarized in Table 4-5, most of the groundwater ET estimates independently derived by SNWA are very similar to previously reported values. Because the differences between these estimates are small, it is assumed that uncertainty associated with the estimates is also small. However, there are relatively large differences between the estimates for White River Valley and the Muddy River Springs Area. Therefore, it is assumed that the ET estimates for these two basins are the most uncertain. Another constrained variable that plays a major role in the distribution of recharge is the maximum recharge efficiency.

The effect of the uncertainty associated with these variables (groundwater ET, WRFS inflow, WRFS outflow, and maximum recharge efficiency) was evaluated with respect to their affect on the recharge distribution and associated basin groundwater budgets. A set of ten alternate scenarios was implemented separately for the low and high values of each variable (Table 6-1). The derived solutions and the comparisons to the base-case solution are presented and discussed in the next section.

Variable	Low Value	High Value	Sources
White River Valley Groundwater ET (afy)	37,000ª	79,560	Eakin, 1966; LVVWD, 2001
Muddy River Springs Area ET (afy)	2,300	5,989	Eakin, 1966; this study
Total Inflow to the WRFS (afy)	7,000	32,000	Rush, 1968; LVVWD, 2001
Total Outflow from the WRFS (afy)	11,100	49,000	Rush, 1968; LVVWD, 2001
Maximum Recharge Efficiency (percent of precipitation)	0.25	1	Maxey and Eakin, 1949

Table 6-1 Range of Uncertainty for Selected Variables

^aBased on spring discharge estimates (Eakin, 1966, p. 261; Scott et al., 1971, p. 50).

The special scenario, referred to as the "SNWA/BARCASS" scenario, was designed to evaluate the effects of using the groundwater ET estimates derived by BARCASS for Long, Jakes, White River, and Cave valleys, instead of those derived by this study. The principal elements of the BARCASS interpretation were incorporated in the SNWA solution file (base-case) as follows:

- The groundwater ET estimates derived by this study for the four subject basins were replaced with the BARCASS estimates (see Table 4-5 for the BARCASS estimates)
- Groundwater flow routing from Cave Valley was changed to the BARCASS interpretation, in which all flow is routed to White River Valley (Welch and Bright, 2007, p. 6), instead of White River and Pahroc valleys.

6.3 Analysis and Results

Descriptions of the analysis and results for the scenarios tested for each of the variables listed in Table 6-1 are provided in Table 6-2 and discussed in the following subsections. Analysis details are provided in Appendix E.

6.3.1 Groundwater ET Estimates

The effects of the uncertainty associated with the groundwater ET estimates of White River Valley and the Muddy River Springs Area are described below.

White River Valley Groundwater ET

Estimated volumes of groundwater ET from White River Valley range between 37,000 afy (Eakin, 1966; Scott et al., 1971) and 79,560 afy (LVVWD, 2001). The effects of these two values on the recharge efficiencies and groundwater budgets are presented in Table 6-2.

The effects on the recharge efficiencies are relatively small, especially for the case of the high ET value. However, the effects on recharge are significant for both cases. As can be seen in Table 6-2, a decrease in the groundwater ET of White River Valley causes a decrease in recharge. Inversely, an increase in the groundwater ET of White River Valley causes an increase in recharge. The resulting total recharge volumes for the WRFS are 124,882 and 184,897 afy, as compared to the base case volume of 155,224 afy. The resulting recharge volumes for the three project basins are 28,721 and 44,584 afy, as compared to the base case value of 36,727 afy.

In the case where the ET value was decreased to the low value of 37,000 afy (Eakin, 1966; Scott et al., 1971), the total outflow from the WRFS remained the same. In the case where the ET value was increased to the high value of 79,560 afy (LVVWD, 2001), the total outflow from the flow system increases from 25,000 to 40,063 afy. The effect of the uncertainty of this variable is rather large.

Muddy River Springs Area Groundwater ET

Estimated volumes of groundwater ET from the Muddy River Springs Area range between 2,300 afy (Eakin, 1966, p. 261) and 5,989 afy (this study). Solutions were derived for each of these values.

In the case of the low value of 2,300 afy (Eakin, 1966), the target value of total ET was adjusted to 135,735 afy to reflect the lower ET value for the Muddy River Springs Area. Since the largest groundwater ET value was derived as part of this study, the high value case is identical to the base case. The results are presented in Table 6-2. The case of the low value causes a small decrease in the total recharge from 155,224 to 154,705 afy. The total WRFS outflow increases from 25,000 to 28,169 afy. The effect of the uncertainty of this variable is small.

6.3.2 Total Inflow from the Lower Meadow Valley Wash Flow System

As described in Section 4.0, the inflow from Lower Meadow Valley Wash has been estimated by several authors. The lowest volume was estimated by Rush (1968, p. 26, 27) at 7,000 afy, and the

Section 6.0

Table 6-2	esults of Uncertainty Analysis for Scenarios Listed in Table 6-1
	Results of Uncertainty Ar

6-4

	Base	Recharge	Efficiency	White Riv Groundv	er Valley vater ET	Muddy Riv Area Grour	er Springs Idwater ET	WRFS	inflow	WRFS	Outflow
Variable	Case	Low Value (0.25)	High Value (1.0)	Low Value (37,000 afy)	High Value (79,560 afy)	Low Value (2,300 afy)	High Value (5,989 afy)	Low Value (7,000 afy)	High Value (32,000 afy)	Low Value (11,100 afy)	High Value (49,000 afy)
			Recha	Irge Efficiency	as a Percent	age of Precip	itation				
ecipitation Zone: 8 to 12 in. (% of precipitation)	0.38	1.10	0.22	0.24	0.56	0.38	0.38	0.39	0.38	0.20	0.52
cipitation Zone: 12 to 15 in. (% of precipitation)	3.64	4.98	2.98	2.67	4.61	3.62	3.64	3.71	3.62	2.72	4.42
cipitation Zone: 15 to 20 in. (% of precipitation)	14.79	11.12	16.53	12.64	16.69	14.75	14.79	14.94	14.75	15.05	16.34
Precipitation Zone: >20 in. (% of precipitation)	44.52	20.55	63.77	42.96	45.78	44.50	44.52	44.62	44.50	58.06	45.56
				WRFS	Budget Comp	onents					
Recharge (afy)	155,224	168,241	155,224	124,882	184,897	154,705	155,224	157,424	154,690	141,324	179,224
Inflow (afy)	9,200	9,200	9,200	9,200	9,200	9,200	9,200	1,000	32,000	9,200	9,200
ET (afv)	139,424	139,424	139,424	109,082	151,642	135,735	139,424	139,424	139,424	139,424	139,424
Outflow (afy)	25,000	38,018	25,000	25,000	42,455	28,169	25,000	25,000	47,266	11,100	49,000
					Cave Valley						
Recharge (afy)	14,659	14,360	14,944	11,932	17,250	14,613	14,659	14,854	14,612	13,606	16,760
Inflow (afy)	1	1	1	1	}	ŀ	Ť	1	1	1	}
ET (afy)	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285
Outflow (afy)	13,374	13,074	13,658	10,647	15,965	13,328	13,374	13,568	13,326	12,320	15,475
					Dry Lake Valle	, A					
Recharge (afy)	15,667	20,017	13,962	11,945	19,380	15,603	15,667	15,940	15,601	12,712	18,666
inflow (afy)	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000
ET (afy)	1	1	ł	1	1	1	ł	ł	1	-	ł
Outflow (afy)	17,667	22,017	15,962	13,945	21,380	17,603	17,667	17,940	17,601	14,712	20,666
					Delamar Valle	ĥ					
Recharge (afy)	6,401	8,294	5,610	4,844	7,954	6,374	6,401	6,515	6,372	5,108	7,65
Inflow (afy)	17,667	22,017	15,962	13,945	21,380	17,603	17,667	17,940	17,601	14,712	20,66
ET (afy)	ł	1	1	1	1	1	1	1	;	ł	1
Outflow (afv)	24.068	3 30,310	21,572	18,789	29.334	23,977	24,068	24,456	3 23,974	19,819	9 28.32

Section 6.0 SE ROA 11781

Southern Nevada Water Authority - Water Resources Division

largest by the LVVWD (2001, p. 6-2, 6-3) at 32,000 afy. These two values were substituted to the estimate of 9,200 afy used in the base case. The flow volume was subdivided between the Muddy River Springs Area and the California Wash in the same proportions as in the base case; a fraction of 0.43 is routed to the Muddy River Springs Area, the remaining fraction, 0.57, is routed to California Wash.

The effects on the recharge efficiencies and groundwater budgets are presented in Table 6-2. The results indicate that, despite the large relative changes in the inflow amounts, the recharge volumes changed very little. The resulting volumes for the WRFS recharge for the low and high values of inflow are 157,424 and 154,690 afy, respectively, as compared to a base case value of 155,224 afy. The resulting recharge volumes for the project basins are 37,309 and 36,586 afy, respectively, as compared to the base case value of 36,727 afy. In the case of the high value of inflow, the total outflow increases by a relatively large amount, from 25,000 to 47,266 afy.

Total Outflow to the Colorado River 6.3.3

As described in Section 4.0, outflow from the WRFS to the Colorado River occurs as Muddy River streamflow (of groundwater origins), spring discharge from Rogers and Blue Point springs, and subsurface outflow from Lower Moapa Valley. The low and high values for the reported range of outflow are 11,100 afy (Rush, 1968; Wells, 1954, p. 570) and 49,000 afy (LVVWD, 2001, p. 6-3). The two values were tested in the solution by changing the constraint on the outflow to the Colorado River from being greater than or equal to the estimated 25,000 afy, to being equal to each of these two values (Table 6-2). The constraint for total outflow was set equal to each of these two values one at the time.

The case where the outflow is constrained to equal 11,100 afy does not yield a solution, unless other constraints are changed. The constraints placed on outflow from White River Valley and the total inflow to the Muddy River Springs Area were changed to allow the solver to converge to a solution. The lower bound for the White River Valley outflow was changed from greater than or equal to 6,300 afy to a positive number (i.e., >0). The total inflow to the Muddy River Springs Area was changed from greater than or equal to 40,000 afy to a positive number. Interbasin flow volumes calculated by the solver are 2,085 afy for the outflow from White River Valley, and 28,022 afy for the total inflow to the Muddy River Springs Area. The constraint imposed on the maximum recharge efficiency was increased from 0.63 to the absolute maximum of 1. The derived solution exhibits notable changes in the recharge efficiencies and the water budgets (Table 6-2). The maximum calculated recharge efficiency increased from 0.63 in the base case to 0.88. The WRFS recharge decreased from 155,224 afy in the base case to 141,324 afy. The total recharge for the three project basins decreased from 36,727 afy in the base case to 31,426 afy. Although the underflow out of White River Valley is believed to be significant, the volume derived for this case is very low (2,085 afy). Hydrologic conditions prevailing along the boundary between White River Valley and Pahroc Valley dictate that at least 6,300 afy (Maxey and Eakin, 1949, p. 45; Welch and Bright, 2007 p. 6) of outflow occurs through this boundary. This volume may be higher and has previously been estimated to be about 40,000 afy (Eakin, 1966, p. 265). Also, the calculated volume of total inflow to the Muddy River Springs Area (28,022 afy) is much lower than the absolute minimum of about 34,000 afy corresponding to the spring discharge of the Muddy Springs (Eakin, 1964, p. 24; Wells, 1954, p. 566). This case is, therefore, highly unlikely, if not impossible.



The case where the total outflow is constrained to equal the maximum reported value of 49,000 afy yields a solution without any changes to the constraints. The effects on the recharge efficiencies are relatively small (Table 6-2). A notable increase occurs in the recharge efficiencies in the lower part of the precipitation levels (more than 20 percent for the 12-to-15-inch zone). The effects on the recharge estimates and interbasin flow volumes are also large (Table 6-2). The total recharge increased from 155,224 to 179,224 afy, reflecting the increase in the outflow. The total recharge to the three project basins increased from 36,727 to 43,082 afy. The outflow from White River Valley increased from 6,595 to 17,078 afy. The inflow to the Muddy River Springs Area increased from 41,828 to 65,757 afy. Although this volume seems large, it is within the realm of possibilities.

6.3.4 Maximum Recharge Efficiency

The maximum recharge efficiency of 0.25 reported by Maxey and Eakin (1949) was used as the low value to be tested in this uncertainty analysis. The value of 0.25 is associated with a large precipitation interval and does not, therefore, represent a maximum recharge efficiency for narrow 1-in. intervals such as the ones used in this analysis. Thus, the value of 0.25 most probably represents the lowest possible value for a maximum recharge efficiency. The absolute maximum recharge efficiency is 1, which represents a situation where all precipitation becomes recharge. Given the same altitude and climatic conditions, the maximum efficiency depends on the geology, specifically, the vertical hydraulic conductivity of the receiving medium. In highly fractured media such as karstic carbonate-rock aquifers, the absolute maximum efficiency may be reached at the local scale. However, at the regional and basin scales, it is highly unlikely that this maximum is reached. Although both of these values represent the extremes, they were used in two alternate scenarios to gage how the uncertainty associated with this variable might effect the solution.

The results of the two tests are presented in Table 6-2. Although the effects on the recharge efficiencies seem drastic, the recharge volumes for the WRFS as a whole and the individual basins do not change appreciably. The total flow system recharge is 168,241 afy for the low value as compared to the base case value of 155,224 afy. To balance the larger recharge of 168,241 afy in the low value case, the total outflow had to increase. The total recharge corresponding to the high efficiency value remained unchanged. The recharge for the three project basins was, however, effected by both values. The recharge is 42,671 afy for the low value and 34,516 afy for the high value, as compared to the base case value of 36,727 afy.

6.3.5 SNWA/BARCASS Scenario

The effects of using the BARCASS groundwater ET estimates and their interpretation of flow routing out of Cave Valley are presented in Table 6-3. The calculated recharge efficiencies are very close to the base-case efficiencies. The total WRFS recharge for this case increased from 155,224 to 163,591 afy. The difference is the result of the increase in the groundwater ET estimate of the four basins that are common to the two studies. Most of it is due to the larger groundwater ET estimates for White River Valley, 76,701 afy as compared to the estimate derived by this study, 67,342 afy. The total recharge for Cave, Dry Lake, and Delamar valleys increases from 36,727 to 38,941 afy; whereas, the total ET for these three basins is 1,285 afy for this study and 1,550 afy for BARCASS

Variable	Base Case	SNWA/BARCASS Scenario
Recharge Efficiency as	a Percent of Prec	pitation
Precipitation Zone: 8 to 12 in.	0.38	0.43
Precipitation Zone: 12 to 15 in.	3.64	3.91
Precipitation Zone: 15 to 20 in.	14.79	15.34
Precipitation Zone: >20 in.	44.52	44.90
White River Va	alley Flow System	
Recharge (afy)	155,224	163,591
Inflow (afy)	9,200	9,200
ET (afy)	139,424	147,79
Outflow (afy)	25,000	25,000
Cav	ve Valley	
Recharge (afy)	14,659	15,39
Inflow (afy)	0	
ET (afy)	1,285	1,55
Outflow (afy)	13,374	13,84
Dry L	ake Valley	
Recharge (afy)	15,667	16,70
Inflow (afy)	2,000	2,00
ET (afy)	0	10-10-10 A. A. A. A.
Outflow (afy)	17,667	18,70
Dela	mar Valley	
Recharge (afy)	6,401	6,83
Inflow (afy)	17,667	18,70
ET (afy)	0)
Outflow (afy)	24,068	3 25,54

Table 6-3 Results of SNWA/BARCASS Scenario

(Welch and Bright, 2007; Appendix A, Excel file named "ofr20071156_appendixa.xls" - Sheet named "Discharge").

6.4 Summary

A summary of the results for all scenarios used in the uncertainty analysis is presented in Table 6-4. The table lists the minimum and maximum effects of changing the specified variable in the solution in terms of the difference between the base-case and the new solution. There is a relatively large range for the recharge efficiencies, with the efficiency associated with the >20 in. precipitation zone ranging from about 20 and 64 percent of precipitation. Although this range appears to be significant, when expressed in terms of recharge volume, the differences are less dramatic. The range of



uncertainty for the WRFS recharge ranges from 124,882 to 184,897 afy as compared to the base-case value of 155,224 afy. This represents a maximum uncertainty of 20 percent. For Cave, Dry Lake, and Delamar valleys, the uncertainty for recharge ranges from 28,721 to 44,584 afy as compared to the base-case value of 36,727 afy. This also represents an uncertainty range of about 20 percent. The maximum uncertainty associated with the recharge volume estimates is caused by the uncertainty in the estimate of groundwater ET for White River Valley.

Variable	Base Case	Minimum Value	Maximum Value
Recharge Efficiency as	a Percentage	of Precipitatio	on
Precipitation Zone: 8 to 12 in.	0.38	0.20	1.10
Precipitation Zone: 12 to 15 in.	3.64	2.67	4.98
Precipitation Zone: 15 to 20 in.	14.79	11.12	16.53
Precipitation Zone: >20 in.	44.52	20.55	63.77
White River	Valley Flow Sy	stem	
Recharge (afv)	155,224	124,882	184,897
Inflow (afy)	9,200	7,000	32,000
ET (afv)	139,424	109,082	151,642
Outflow (afv)	25,000	11,100	49,000
C	ave Valley		
Recharge (afy)	14,659	11,932	17,250
Inflow (afy)			
ET (afy)	1,285	1,285	1,285
Outflow (afy)	13,374	10,647	15,965
Dŋ	Lake Valley		
Recharge (afy)	15,667	11,945	20,017
Inflow (afy)	2,000	2,000	2,000
ET (afy)			
Outflow (afy)	17,667	13,945	22,017
De	lamar Valley		
Recharge (afy)	6,401	4,844	8,294
Inflow (afy)	17,667	13,945	22,017
ET (afy)			
Outflow (afv)	24,068	18,789	30,310

Table 6-4 Summary of Uncertainty Analysis Results

7.0 SUMMARY AND CONCLUSIONS

This report presents a regional assessment of the precipitation, natural recharge, groundwater discharge, and interbasin flow of the WRFS. This assessment and was completed in support of water-right hearings related to SNWA applications 53987 through 53992, inclusive, in Cave, Dry Lake, and Delamar valleys.

The objectives of this assessment were to estimate the groundwater recharge distribution for the WRFS and derive predevelopment groundwater budgets for the flow system and for the individual basins, specifically Cave, Dry Lake, and Delamar valleys. To achieve these objectives, new recharge efficiencies were derived for the WRFS using the water-balance method, improved estimates of groundwater ET and precipitation distribution, and other hydrologic information collected since the Reconnaissance Series Reports were published. The water-balance method was implemented using the Excel Solver. These efficiencies were then applied to the precipitation distribution to generate a distribution of recharge from which basin recharge estimates were then computed and used to develop basin groundwater budgets. Finally, an uncertainty analysis was conducted to evaluate the effect of the most uncertain variables on the solution. The major findings of this study related to the precipitation distribution, the groundwater ET and the recharge distribution of the WRFS and its basins and are summarized below.

Precipitation Distribution

The PRISM grid (800-m 1971 to 2000 normals - Version 2; May 3, 2007) provides a good representation of the long-term mean annual precipitation distribution for the WRFS. The PRISM grid represents, within ten percent, the period of record mean annual value for most stations within the WRFS and vicinity (Figure 4-2). The PRISM precipitation distribution appears to overestimate precipitation at the station locations, but with little spatial correlation (i.e., overestimates regardless of location and altitude). If in fact PRISM overestimates the "true" values of precipitation, then the "true" values of groundwater ET are underestimated and, therefore, the "true" value of the total recharge of the WRFS is underestimated. This is because the groundwater ET was computed by subtracting precipitation from the total ET estimate; therefore, a larger value of precipitation would yield a smaller value of groundwater ET. Because groundwater ET is used as the target value in the water-balance method (as applied here), a smaller value would yield a corresponding smaller value of recharge.

The PRISM precipitation grid provides a much better spatial distribution of precipitation than the Hardman precipitation maps (1936, 1965), particularly at the higher precipitation intervals, where the resolution of the PRISM grid is much better than the Hardman maps. This is because (1) PRISM incorporates more data (e.g., high-altitude precipitation stations), (2) PRISM has greater topographical resolution (800 meter versus 1:250,000 scale), and (3) PRISM incorporates natural phenomena affecting the spatial precipitation distributions such as rain shadows and coastal effects.



Therefore, it was concluded that for the purposes of this regional-scale evaluation, the PRISM precipitation distribution best represents the precipitation distribution for the WRFS.

For Cave, Dry Lake, and Delamar valleys, Eakin (1962; 1963a) had assumed through personal communication with Hardman, that the elevation contours from an improved topographical base correlated to the precipitation zones defined by Hardman (1936). Based on this assumption, Eakin (1962; 1963a) equated the 6,000 to 7,000 ft elevation interval to the 8 to 12 in. precipitation zone, the 7,000 to 8,000 ft interval to the 12 to 15 in. zone, the 8,000 to 9,000 interval to the 15 to 20 in. zone, and for altitudes greater than 9,000 ft, to the >20 in. zone. Below altitudes of 6,000 ft, less than eight inches of precipitation was assumed. This assumption was tested for Cave, Dry Lake, and Delamar valleys by comparing a digitized version of the Hardman (1962) map and superposing elevation contours that were generated from 30 m DEM (USGS, 2001). The results of this comparison revealed that the assumptions were used in the derivation of the recharge estimates for these basins using the standard Maxey-Eakin recharge efficiencies. The comparison indicates that the Hardman precipitation map yields about 130,000 afy more precipitation for the three project basins for precipitation zones greater than 8-in. (Table 5-4).

Groundwater ET

Predevelopment groundwater ET for the WRFS is estimated to be 139,424 afy, which falls within the literature range of 100,600 (Eakin, 1966; Rush, 1968) and 176,305 (LVVWD, 2001) (Table 4-5). Estimates of groundwater ET for the three project basins are as follows: 1,285 afy for Cave Valley, and zero for the other two. The estimated basin groundwater ET volumes compare reasonably well with previous estimates, except for two basins: White River Valley and the Muddy River Springs Area.

For White River, more recent estimates, including this study, reflect newer mapping and utilize data from ET measurement sites. These estimates all fall within a range of about 10,000 afy (Table 4-5) of each other but are at least 30,000 afy greater than the early estimate of Eakin (1966, p. 261). The disparity between these estimates and Eakin's (1966) can be traced to the total acreage delineated for the groundwater discharge area (Table B.1-1). Only 36,000 acres were delineated for the area used in the Eakin (1966) estimate (Maxey and Eakin, 1949), while the other studies delineated more than four times this acreage. Because of the similarity between the more recent measurements and the difference in the acreage delineated for the groundwater discharge area, it is concluded that the groundwater estimate reported by Eakin (1966) is low and not representative of the long-term mean annual conditions.

For the Muddy River Springs Area, the groundwater ET estimate of 5,989 afy derived by this study represents the highest reported value, and is about 3,700 afy greater than the lowest estimate derived by Eakin (1966), which is 2,300 afy. The acreage delineating this groundwater discharge area was estimated by this study to be 1,951 acres. The area was not delineated or reported by Eakin (1966). While the percent difference between the estimates is seemingly large, the difference in terms of volume is insignificant with respect to its influence on the outcome of the analyses described in this report.

Natural Recharge

The Excel solver was used to derive a recharge solution using the groundwater balance approach. The solution is an estimate of the total groundwater discharge and recharge distribution for the WRFS. The recharge distribution was used in conjunction with the groundwater ET estimates to construct the groundwater budget for the WRFS and its individual basins. The results of this analysis are summarized below:

- The total groundwater recharge for the WRFS under predevelopment steady-state conditions was calculated to be about 155,000 afy (rounded). The total recharge falls within the reported range of values (104,650 to 201,500 afy).
- Recharge efficiencies were derived for 1-in precipitation intervals, and are summarized by the precipitation zones defined by Maxey and Eakin (1949) for comparison purposes. They are as follows: 45 percent for >20 in. zone; 15 percent for 15 to 20 in. zone; 4 percent for 12 to 15 in. zone, and 0.4 percent for 8 to 12 in. zone (Equation 5-5). The relationship between recharge and precipitation is somewhat different from that of Maxey and Eakin (1949) because the precipitation distributions and groundwater ET estimates are significantly different. However, the method by which the relationships were derived are essentially the same. The relationship derived by this analysis is very comparable to more recent relationships developed by others (Figure 5-4).
- The recharge efficiencies were applied to the PRISM precipitation distribution to derive a
 spatial distribution of recharge, yielding basin recharge volumes that compare well with the
 recharge estimates reported in the literature for most basins. The exceptions are Garden
 Valley and Dry Lake Valley which are higher based on this analysis, and Coyote Spring Valley
 which is lower (Table 5-2).
- As discussed previously, the altitude-precipitation assumptions of Eakin (1962; 1963a) yield much different estimates of precipitation than the Hardman (1962) precipitation map. These differences also translate into significantly different estimates of recharge for Cave, Dry Lake, and Delamar valleys (Table 5-4). Applying the standard Maxey-Eakin recharge efficiencies to the Hardman (1962) precipitation map yields over 13,000 afy more recharge in these basins than Eakin (1962; 1963a). For Cave Valley the difference is about 7,500 afy more, in Dry Lake Valley about 4,000 afy more, and in Delamar Valley about 1,700 afy more. The recharge was estimated to be about 21,500, 9,000, and 2,900 afy, respectively, or 33,400 afy in total. This is only about 3,300 afy less than the total for the three basins estimated by this study, although the distribution is somewhat different.

Groundwater Budgets

A groundwater budget for the WRFS and its individual basins was constructed using the recharge, groundwater ET, and boundary flow estimates derived by this analysis. All of these components fall within the literature range of values. The budget components for Cave, Dry Lake, and Delamar valleys are listed in Table 7-1.



SE ROA 11788



Basin Name	Recharge	Inflow	Groundwater ET	Outflow
Cave Valley	14,659	0	1,285	13,374
Dry Lake Valley	15,667	2,000	0	17,667
Delamar Valley	6,401	17,667	0	24,068
TOTAL	36,727		1,285	

Table 7-1	and the second state of the second state of the
Groundwater Budget Components for Cave,	Dry Lake, and Delamar Valleys

Uncertainty Analysis

An uncertainty analysis was conducted to evaluate the effect of the most uncertain variables on the calculated recharge efficiencies and final groundwater budgets:

- The uncertainty on the constraint placed on the maximum recharge efficiency is significant for the low value (0.25), and not significant for the high value (1.0). For the case of the low value, the total WRFS recharge increases from 155,224 afy to 168,241 afy, and remains unchanged for the case of the high value (1.0). The total recharge for the three project basins ranges from 34,516 afy (high value) to 42,671 (low value).
- The largest uncertainty is associated with the White River Valley groundwater ET based on the range of estimated values (37,000 to 79,560 afy). For the case of the low value, the total WRFS recharge decrease from 155,224 afy to 124,882 afy, and increases to 184,897 afy for the high value. The total recharge for the three project basins ranges from 28,721 (low value) to 44,584 afy (high value).

8.0 REFERENCES

- Anderson, T.W., Freethey, G.W., and Tucci, P., 1992, Geohydrology and water resources of alluvial basins in south-central Arizona and parts of adjacent states: U.S. Geological Survey Professional Paper 1406-B, p. B1-B67.
- Avon, L., and Durbin, T.J., 1994, Evaluation of the Maxey-Eakin method for estimating recharge to ground-water basins in Nevada: American Water Resources Association, Water Resources Bulletin, Vol. 30, No. 1, p. 99–111.
- Bechtel SAIC Company, LLC, 2004, Technical Basis Document No. 1—Climate and Infiltration, Rev. 1: U.S. Department of Energy, Office of Civilian Radioactive Waste Management, DE-AC28-01RW12101, 206 p.
- Belcher, W.R., Elliott, P.E., and Geldon, A.L., 2001, Hydraulic-property estimates for use with a transient ground-water flow model for the Death Valley Regional Ground-Water Flow System, Nevada and California (revised August 1, 2002): U.S. Geological Survey Water-Resources Investigations Report 2001-4210, 31 p.
- Berger, D.L., 2000, Water budgets for Pine Valley, Carico Lake Valley, and Upper Reese River Valley hydrographic areas, middle Humboldt River Basin, north-central Nevada—Methods for estimation and results: U.S. Geological Survey Water Resources Investigations Report 99-4272, 40 p.
- Brothers, K., Buqo, T.S., Tracy, J.V., Stock, M., Bentley, C., Zdon, A., and Kepper, J., 1993, Hydrology and steady state groundwater model of Cave Valley, Lincoln and White Pine counties, Nevada: Las Vegas Valley Water District, Cooperative Water Project, Report No. 11, 78 p.
- Brothers, K., Katzer, T., and Johnson, M., 1996, Hydrology and steady state groundwater model of Dry Lake and Delamar valleys, Lincoln County, Nevada: Las Vegas Valley Water District, Cooperative Water Project, Report No. 16, 48 p.

BSC, see Bechtel SAIC Company.

- Bunch, R.L., Harrill, J.R., 1984, Completion of selected hydrologic data from the MX missile-siting investigation, east-central Nevada and western Utah: U.S. Geological Survey Open-File Report 84-702, 123 p.
- Buqo, T.S., 2004, Allometric responses in coupled groundwater-surface water-reservoir systems: Journal of the Nevada Water Resources Association, Vol. 1, No. 1, p. 40-57.

Section 8.0

SE ROA 11790



- Contor, B.A., 2004, Recharge on non-irrigated lands: University of Idaho, Idaho Water Resource Research Institute, Technical Report 04-006, 19 p.
- D'Agnese, F.A., Faunt, C.C., Turner, A.K., and Hill, M.C., 1997, Hydrogeologic evaluation and numerical simulation of the Death Valley Regional Ground-Water Flow System, Nevada, and California: U.S. Geological Survey Water-Resources Investigations Report 96-4300, 124 p.
- Daly, C., Neilson, R.P., and Phillips, D.L., 1994, A statistical-topographic model for mapping climatological precipitation over mountainous terrain: Journal of Applied Meteorology, Vol. 33, No. 2, p. 140–158.
- Daly, C., Taylor, G.H., and Gibson, W.P., 1997, The PRISM approach to mapping precipitation and temperature: American Meteorological Society, Proceedings of the 10th AMS Conference on Applied Climatology, Reno, Nevada, October 20–23, p. 10–12.
- Daly, C., Taylor, G.H., Gibson, W.P., and Parzybok, T., 1998, Development of high-quality spatial climate datasets for the United States: Proceedings of the First International Conference on Geospatial Information in Agriculture and Forestry, Lake Buena Vista, Florida, June 1–3, 1998, p. I-512–I-519.
- Davisson, M.L., and Rose, T.P., 2000, Maxey-Eakin methods for estimating groundwater recharge in the Fenner Watershed, southeastern California: U.S. Department of Energy, Lawrence Livermore National Laboratory, UCRL-ID-139027, 13 p.
- Dettinger, M.D., 1989, Reconnaissance estimates of natural recharge to desert basins in Nevada, U.S.A., by using chloride-balance calculations: Journal of Hydrology, Vol. 106, p. 55-78.
- Dixon, G.L., and Katzer, T.C., 2002, Geology and hydrology of the lower Virgin River Valley in Nevada, Arizona, and Utah: Virgin Valley Water District, Mesquite, Nevada, Report No. VVWD-01, 126 p.
- Dixon, G.L., Rowley, P.D., Burns, A.G., Watrus, J.M., and Ekren, E.B., 2007, Geology of White Pine and Lincoln counties and adjacent areas, Nevada and Utah—The geologic framework of regional groundwater flow systems: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. HAM-ED-0001.
- Donovan, D.J., and Katzer, T., 2000, Hydrologic implications of greater ground-water recharge to Las Vegas Valley, Nevada: Journal of the American Water Resources Association, Vol. 36, No. 5, p. 1133–1148.
- Eakin, T.E., 1961, Ground-water appraisal of Long Valley, White Pine and Elko counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 3, 35 p.

- Eakin, T.E., 1962, Ground-water appraisal of Cave Valley in Lincoln and White Pine counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 13, 19 p.
- Eakin, T.E., 1963a, Ground-water appraisal of Dry Lake and Delamar valleys, Lincoln County, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 16, 25 p.
- Eakin, T.E., 1963b, Ground-water appraisal of Pahranagat and Pahroc valleys, Lincoln and Nye counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 21, 36 p.
- Eakin, T.E., 1963c, Ground-water appraisal of Garden and Coal Valleys, Lincoln and Nye Counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 18, 29 p.
- Eakin, T.E., 1964, Ground-water appraisal of Coyote Spring and Kane Spring valleys and Muddy River Springs Area, Lincoln and Clark counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 25, 39 p.
- Eakin, T.E., 1966, A regional interbasin groundwater system in the White River area, southeastern Nevada: Water Resources Research, Vol. 2, No. 2, p. 251–271.
- Eakin, T.E., Maxey, G.B., Robinson, T.W., Fredericks, J.C., and Loeltz, O.J., 1951, Contributions to the hydrology of eastern Nevada: State of Nevada, Office of the State Engineer, Water Resources Bulletin No. 12, 171 p.
- Ertec Western, Inc., 1981a, MX Siting Investigation—Water Resources Program—Results of Regional Carbonate Aquifer Testing, Coyote Spring Valley, Nevada: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, Report E-TR-57, 65 p.
- Ertec Western, Inc., 1981b, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume I, Report E-TR-52-I: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 77 p.
- Ertec Western, Inc., 1981c, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume II, Report E-TR-52-II: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 286 p.
- Ertec Western, Inc., 1981d, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume IIA, Report E-TR-52-II: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 41 p.
- Ertec Western, Inc., 1981e, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume IIB, Report E-TR-52-II: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 131 p.

Section 8.0

SE ROA 11792



- Faybishenko, B., 2007, Climatic forecasting of net infiltration at Yucca Mountain using analogue meteorological data: Vadose Zone Journal, Vol. 6, p. 77–92.
- Flint, A.L., and Flint, L.E., 2007, Application of the basin characterization model to estimate in-place recharge and runoff potential in the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah: U.S. Geological Survey Scientific Investigations Report 2007-5099, 20 p.
- Flint, A.L., Flint, L.E., Hevesi, J.A., and Blainey, J.M., 2004, Fundamental concepts of recharge in the Desert Southwest—A regional modeling perspective, p. 159–184, in Hogan, J.F., Phillips, F.M., and Scanlon, B.R., eds., Ground-water recharge in a desert environment—The southwestern United States: American Geophysical Union, Water Science and Applications Series, Vol. 9, 294 p.
- Flint, A.L., Flint, L.E., Kwicklis, E.M., Fabryka-Martin, J.T., and Bodvarsson, G.S., 2002, Estimating recharge at Yucca Mountain, Nevada, U.S.A.—Comparison of methods: Hydrogeology Journal, Vol. 10, p. 180–204.
- Hardman, G., 1936, Nevada precipitation and acreages of land by rainfall zones: University of Nevada-Reno, Agricultural Experiment Station, mimeographed report, 10 p.
- Hardman, G., 1962, Precipitation map of Nevada, in Lamke, R.D., and Moore, D.O., 1965, Interim inventory of surface water resources of Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Bulletin No. 30, p. 6.
- Hardman, G., 1965, Nevada Precipitation Map, adapted by George Hardman, July 1965, from map prepared by George Hardman, Victor Kral, and Victor Haffey, and others, 1936: University of Nevada-Reno, Agricultural Experiment Station (Plate).
- Hardman, G., and Mason, H.G., 1949, Irrigated lands of Nevada: University of Nevada-Reno, Agricultural Experiment Station, Bulletin No. 183, 57 p.
- Hargreaves, G.H., and Samani, Z.A., 1985, Reference crop evapotranspiration from temperature: American Society of Agricultural Engineers, Applied Engineering in Agriculture, Vol. 1, No. 2, p. 96–99.
- Harrill, J.R., Gates, J.S., and Thomas, J.M., 1988, Major ground-water flow systems in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Investigations Atlas HA-694-C, 4 p., scale 1:1,000,000, two sheets.
- Harrill, J.R., and Prudic, D.E., 1998, Aquifer systems in the Great Basin region of Nevada, Utah, and adjacent states—Summary report: U.S. Geological Survey Professional Paper 1409-A, 66 p.
- Hess, J.W., and Mifflin, M.D., 1978, A feasibility study of water production from deep carbonate aquifers in Nevada: Water Resources Center, Desert Research Institute Publication No. 41054, 125 p.



- Hevesi, J.A., Flint, A.L., and Flint, L.E., 2002, Preliminary estimates of spatially distributed net infiltration and recharge for the Death Valley region, Nevada-California: U.S. Geological Survey Water-Resources Investigations Report 02-4010, 36 p.
- Katzer T., and Donovan, D.J., 2003, Surface-water resources and basin water budget for Spring Valley, White Pine and Lincoln counties, Nevada: Las Vegas Valley Water District, Las Vegas, Nevada, 70 p.
- Kirk, S.T., and Campana, M.E., 1990, A deuterium-calibrated groundwater flow model of a regional carbonate-alluvial system: Journal of Hydrology, Vol. 119, p. 357–388.
- Kumar, C.P., and Seethapathi, P.V., 2002, Assessment of natural ground water recharge in upper Ganga canal command area: Association of Hydrologists of India, Journal of Applied Hydrology, Vol. XV, No. 4, p. 13–20.
- Las Vegas Valley Water District, 2001, Water resources and ground-water modeling in the White River and Meadow Valley flow systems, White Pine, Lincoln, Nye, and Clark counties, Nevada, 285 p.
- Las Vegas Valley Water District, Woodward-Clyde Consultants, and Dames & Moore Consultants, 1994, Environmental report covering selected hydrographic basins in Clark, Lincoln, Nye, and White Pine counties, Nevada: Las Vegas Valley Water District, Cooperative Water Project, Report No. 14, 199 p.
- Lichty, R.W., and McKinley, P.W., 1995, Estimates of ground-water recharge rates for two small basins in central Nevada: U.S. Geological Survey Water-Resources Investigations Report 94-4104, 31 p.
- LVVWD, see Las Vegas Valley Water District.
- Maurer, D.K., and Berger, D.L., 1997, Subsurface flow and water yield from watersheds tributary to Eagle Valley hydrographic area, west-central Nevada: U.S. Geological Survey Water-Resources Investigations Report 97-4191, 56 p.
- Maurer, D.K., and Halford, K.J., 2004, Update estimates of the distribution of average annual precipitation in Carson Valley, 1971–2000, Douglas County, Nevada, and Alpine County, California: Journal of the Nevada Water Resources Association, Vol. 1, No. 1, p. 20–39.
- Maxey, G.B., and Eakin, T.E., 1949, Groundwater in the White River Valley, White Pine, Nye, and Lincoln counties, Nevada: State of Nevada, Office of the State Engineer, Water Resources Bulletin No. 8, 59 p.
- McCurdy, G., and Albright, W., 2004, Estimates of potential evapotranspiration at selected sites in eastern Nevada from meteorological station data: Desert Research Institute Publication No. 41198, 113 p.

Section 8.0

SE ROA 11794



- Mizell, S.A., Russell, C.E., and Kluesner, T.L., 2007, Reconnaissance estimation of groundwater recharge to selected hydrographic basins of eastern Nevada and western Utah using the chloride mass-balance method: Division of Hydrologic Sciences, Desert Research Institute Publication No. 41232, 29 p.
- Moreo, M.T., Laczniak, R.J., and Stannard, D.I., 2007, Evapotranspiration rate measurements of vegetation typical of ground-water discharge areas in the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah, September 2005–August 2006: U.S. Geological Survey Scientific Investigations Report 2007-5078, 37 p.
- Multi-Resolution Land Characteristics Consortium, 1992, National Land Cover Data 1992, accessed at website http://landcover.usgs.gov/natllandcover.asp or http://www.mrlc.gov/mrlc2k_nlcd.asp or http://epa.gov/mrlc/ncld.html.
- Nevada State Engineer (The office of the State Engineer of The State Nevada), 2007a, The ruling (#5712) In the matter of applications 72218, 72219, 72220 and 72221 filed to appropriate the underground waters of the Kane Springs Valley hydrographic basin (206), Lincoln County, Nevada.
- Nevada State Engineer (The office of the State Engineer of The State Nevada), 2007b, The ruling (#5726) In the matter of applications 54003 through 54021, inclusive, filed to appropriate the underground waters of the Spring Valley hydrographic basin (184), White Pine County, Nevada.
- Nichols, W.D, 2000, Determining ground-water evapotranspiration from phreatophyte shrubs and grasses as a function of plant cover or depth to ground water, Great Basin, Nevada and eastern California p. A1–A14, in Nichols, W.D., Regional ground-water evaporation and ground-water budgets, Great Basin, Nevada: U.S. Geological Survey Professional Paper 1628, 108 p.
- NLCD, see Multi-Resolution Land Characteristics Consortium.
- Plume, R.W., and Carlton, S.M., 1988, Hydrogeology of the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Atlas HA-694-A (plate).
- PRISM Group, 2006, PRISM (1971 to 2000) precipitation distribution was developed using data from 1971 to 2000: Oregon State University, Corvallis, Oregon, accessed on August 3, 2007, at website http://www.prism.oregonstate.edu/products/matrix.phtml.
- Prudic, D.E., Harrill, J.R., and Burbey, T.J., 1995, Conceptual evaluation of regional ground-water flow in the carbonate-rock province of the Great Basin, Nevada, Utah, and adjacent states— Regional aquifer-system analysis—Great Basin, Nevada-Utah: U.S. Geological Survey Professional Paper 1409-D, 102 p.
- Quiring, R.F., 1965, Annual precipitation amount as a function of elevation in Nevada south of 38 ½ degrees latitude: U.S. Weather Bureau Research Station, Las Vegas, Nevada, 14 p.
- Reiner, S.R., Laczniak, R.J., DeMeo, G.A., Smith, J.L., Elliott, P.E., Nylund, W.E., and Fridrich, C.J., 2002, Ground-water discharge determined from measurements of evapotranspiration, other available hydrologic components, and shallow water-level changes, Oasis Valley, Nye County, Nevada: U.S. Geological Survey Water-Resources Investigations Report 01-4239, 65 p.
- Rich, L.R., 1951, Consumptive use of water by forest and range vegetation: Proceedings—American Society of Civil Engineers, Vol. 77, separate No. 90, 14 p.
- Rush, F.E., 1968, Water-resources appraisal of the Lower Moapa-Lake Mead area, Clark County, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 50, 67 p.
- Russell, C.E., and Minor, T., 2002, Reconnaissance estimates of recharge based on an elevation-dependent chloride mass-balance approach: U.S. Department of Energy, National Nuclear Security Administration, DOE/NV/11508-37 (Desert Research Institute Publication No. 45164), 138 p.
- San Juan, C.A., Belcher, W.R., Laczniak, R.J., and Putnam, H.M., 2004, Hydrologic components for model development, in Belcher, W.R., ed., Death Valley Regional Ground-Water Flow System, Nevada and California—Hydrogeologic framework and transient ground-water flow model: U.S. Geological Survey Scientific Investigations Report 2004-5205, p. 99–136.
- Scanlon, B.R., Keese, K.E., Flint, A.L., Flint, L.E., Gaye, C.B., Edmunds, M., and Simmers, I., 2006, Global synthesis of groundwater recharge in semiarid and arid regions: Hydrological Processes, Vol. 20, p. 3335–3370.
- Scott, B.R., Smales, T.J., Rush, F.E., and Van Denburgh, A.S., 1971, Water for Nevada—Report 3– Nevada's Water Resources: Nevada Department of Conservation and Natural Resources, 87 p.
- Smith, J.L., Laczniak, R.J., Moreo, M.T., and Welborn, T.L., 2007, Mapping evapotranspiration units in the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah: U.S. Geological Survey Scientific Investigations Report 2007-5087, 21 p.
- SNWA, see Southern Nevada Water Authority.
- Southern Nevada Water Authority, 2006, Water resources assessment for Spring Valley (presentation to the Office of the Nevada State Engineer): Las Vegas, Nevada.
- Thomas, J.M., Calhoun, S.C., and Apambire, W.B., 2001, A deuterium mass-balance interpretation of groundwater sources and flows in southeastern Nevada: Desert Research Institute Publication No. 41169, 46 p.
- Thomas, J.M., Mason, J.L., and Crabtree, J.D., 1986, Ground-water levels in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Atlas HA-694-B, scale 1:1,000,000, two sheets.

Section 8.0

SE ROA 11796

8-7



- Thomas, J.M., Welch, A.H., and Dettinger, M.D., 1996, Geochemistry and isotope hydrology of representative aquifers in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Professional Paper 1409-C, 99 p.
- Tumbusch, M.L., and Schaefer, D.H., 1996, Selected hydrologic data for and location of MX wells in east-central and southern Nevada, January 1980 through May 1996: U.S. Geological Survey Open-File Report 96-469, 37 p.
- U.S. Geological Survey, 2001, Digital elevation model (DEM), Earth Resources Observation and Science (EROS), U.S. Geological Survey, accessed at http://edc.usgs.gov/guides/dem.html.
- U.S. Geological Survey, 2004, Southwest Regional Gap Analysis Project, accessed at website http:// fws-nmcfwru.nmsu.edu/swregap/default.htm.

Walkenbach, J., 2003, Microsoft Office Excel 2003 Bible: Wiley Publishing, Inc., 936 p.

- Watt, J.T., and Ponce, D.A., 2007, Geophysical framework investigations influencing ground-water resources in east-central Nevada and west-central Utah: U.S. Geological Survey Open-File Report 2007-1163, 43 p.
- Welborn, T.L., and Moreo, M.T., 2007, Irrigated acreage within the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah: U.S. Geological Survey Data Series 273, 19 p.
- Welch, A.H., and Bright, D.J., eds., 2007, Water resources of the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah—Draft Report: U.S. Geological Survey Open-File Report 2007-1156, 103 p.
- Wells, J.V.B., 1954, Compilation of records of surface waters of the United States through September 1950—Part 9, Colorado River Basin: U.S. Geological Survey Water-Supply Paper 1313, 749 p.
- Westenburg, C.L., DeMeo, G.A., and Tanko, D.J., 2006, Evaporation from Lake Mead, Arizona and Nevada, 1997–99: U.S. Geological Survey Scientific Investigations Report 2006-5252, 24 p.
- White, W.N., 1932, A method of estimating ground-water supplies based on discharge by plants and evaporation from soil—Results of investigations in Escalante Valley, Utah: U.S. Geological Survey Water-Supply Paper 659-A, 209 p.
- Wilson, J.L., and Guan, H., 2004, Mountain-block hydrology and mountain-front recharge, p. 113– 137, in Hogan, J.F., Phillips, F.M., and Scanlon, B.R., eds., Groundwater recharge in a desert environment—The southwestern United States: American Geophysical Union, Water Science and Applications Series, Vol. 9, 294 p.
- Wilson, J.W., 2007, Water-level surface maps of the carbonate-rock and basin-fill aquifers in the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent

areas in Nevada and Utah: U.S. Geological Survey Scientific Investigations Report 2007-5089, 11 p., two plates.

Wright, J.L., 1982, New evapotranspiration crop coefficients: American Society of Civil Engineers, Journal of Irrigation and Drainage Division, Vol. 108, p. 57–74.

SE ROA 11798

8-9



This Page Intentionally Left Blank

Section 8.0 SE ROA 11799

Appendix A

Precipitation Station Data

SE ROA 11800

A.1.0 PRECIPITATION STATION DATA

Precipitation station data were compiled and the period of record mean annual value was derived for each station. Data were compiled from the following sources:

- Nevada Division of Water Resources (NDWR)
- USGS
- Western Regional Climate Center (WRCC)
- National Climatic Data Center (NCDC)
- NRCS SNOwpack TELemetry (SNOTEL).

Precipitation stations selected from these sources included stations qualified as climate normals by the NCDC (Note: the average annual precipitation as reported by WRCC was used for these stations). Other stations, except for the "Sheep Peak" and "Hayford Peak" stations, have more than 20 "non-zero" years of reported annual precipitation (i.e., years in which the reported annual precipitation was greater than zero).

Summary data for the stations located within and near the WRFS are listed in Table A.1-1. The fields in the table are as follows:

- Station Number
- Station Name
- UTM Easting
- UTM Northing
- · Altitude (ft-amsl)
- Annual Precipitation, Mean (inches)
- Annual Precipitation, Minimum (inches)
- Annual Precipitation, Maximum (inches)
- Period of Record
- Period of Record Duration
- NCDC Normal
- Source

Appendix A

A-1



Southern Nevada Water Authority - Water Resources Division

VCDC Source	lormal	No NDWR	No NDWR	No NDWR	KI- NICIAID	NO NUMAN	No NDWR	No NRCS (SNOTE	No NRCS (SNOTE	No USGS	No USGS	No USGS	No USOS	No USGS	No USGS	No USGS	No WRCC	30 yr WRCC	30 yr WRCC	10 yr WRCC	No WRCC	10 yr WRCC	10 yr WRCC	No WRCC	20 VIL WRCC
Period of Record	Number of N Years)	52	52	52		79	52	25	25	22	22	18	22	22	18	22	55	80	60	19	60	19	57	111	
Period of	Record	1954 to 2005	1954 to 2005	1954 to 2005	2007 00 1001	1954 to 2005	1954 to 2005	1981 to 2005	1981 to 2005	1984 to 2005	1984 to 2005	1986 to 2003	1984 to 2005	1984 to 2005	1986 to 2003	1984 to 2005	1928 to 1982	1928 to 2007	1948 to 2007	1989 to 2007	1948 to 2007	1989 to 2007	1951 to 2007	1897 to 2007	Ford is and
tion	Max (in.)	26.30	24.49	02 3U	00.12	26.80	33.00	40.80	38.80	32.16	26.25	29.00	46.00	47.00	28.25	28.50	23.64	18.73	10.59	11.85	14.78	12.96	24.98	16.16	1
recipita	Min (in.)	2.90	6.00	220	0.00	0.00	6.50	19.00	13.30	12.00	9.00	7.25	12.00	0.00	3.00	0.00	4.42	1.84	0.94	3.18	0.69	3.35	3.72	4.22	
Annual F	Mean (in.)	14.72	13.74	1106	CU.41	13.67	17.51	27.35	24.22	20.41	15.67	15.75	26.13	19.68	15.10	16.72	12.73	9.10	5.80	5.37	4.51	5.73	12.23	9.56	
Altituda	(ft-amsl)	7.732	£ 000	000 1	1,800	7,100	7,200	9,100	9,200	10,650	9,700	9,840	10,440	9,200	9,600	9,300	6,250	4,400	4,330	1,270	2,920	1,250	3,420	6,250	
UTM	Northing ^b (m)	4 323.835	001 000 1	001 '007't	4,365,136	4,409,348	4,254,601	4,354,622	4,333,562	4,337,547	4,443,655	3,985,798	4,309,377	4,236,086	3,985,794	4,355,941	4,219,700	4,166,179	4,421,801	4,002,962	4,038,794	4,022,206	4,136,549	4,350,044	
UTM	Easting ^b (m)	703 315		043,840	665,201	702,800	730,436	705,390	677,114	706,108	680,594	680,192	732,764	728,118	680,192	737,695	624,110	719,105	780,345	703,844	646,235	733,343	718,409	685,372	
	Station Name		Connois rass	Current Creek	Robinson Summit	Schellbourne Pass	Wilson Creek Summit	Berry Creek	Ward Mountain	Cave Mountain	Cherry Creek Range	Hayford Peak	Mt. Washingon	Mt. Wilson	Sheep Peak	Unnamed Peak NW of Mt. Moriah	Adaven	Caliente	Callao	Callville Bay	Desert Game Range	Echo Bav	Elain	Elv WBO	LIJ WEG
	station No.ª		1	2	e	4	2		7	60	σ	10	11	12	13	14	15	16	17	18	19	20	21	00	22

Table A.1-1 Precipitation Station Information (Pare 1 of 2)

Appendix A SE ROA 11802

A-2

	NCDC	Norma	30 yr	10 yı	No
	Period of Record	(Number of Years)	48	40	55
	Derind of	Record	1960 to 2007	1951 to 1990	1948 to 2002
tion	ation	Max (in.)	12.64	14.69	15.49
forma	Precipit	Min (in.)	3.89	4.59	1.65
A.1-1 tion In 2 of 2)	Annual	Mean (in.)	7.89	7.61	8.67
Table tion Sta (Page		Altitude (ft-amsl)	4,340	5,260	6,020
Precipita	UTM	Northing ^b (m)	4,415,084	4.313,196	4.282.815
	UTM	Easting ^b (m)	808.023	757.384	706 113
	-		+	Ť	T

n Fish Spri Garrison Geyser I Great Bi Hiko Key Pitt Lages	Station	Eactingb	dar the the	Altitude							200106
Fish Spri Garrison Geyser I Great B: Hiko Key Pitt Lages	Name	(u)	(m)	(ft-amsI)	(in.)	Min (in.)	Max (in.)	Record	(Number of Years)	Normal	
risn sprinsen Garrison Geyser I Great Bi Hiko Key Pitt Lages	The Define	R08 023	4.415.084	4,340	7.89	3.89	12.64	1960 to 2007	48	30 yr	WRCC
Garrison Geyser I Great Bi Hiko Key Pitt Key Pitt Lages	IIIgs reluge		204 040 1	A DED	761	4.59	14.69	1951 to 1990	40	10 yr	WRCC
Geyser I Great Bi Hiko Key Pitt Lages		151,384	4,010,130	0000	10.0	1 66	15.40	1948 to 2002	55	No	WRCC
Great Ba Hiko Key Pitt Lages	Range	706,113	4,282,815	0,020	10.0			2000 -+ 0101	CB	10 VF	WRCC
Hiko Key Pitt Lages	asin National Park	740,673	4,320,446	6,830	13.27	7.37	21.20	1948 10 2001	00	10.01	MIPCC
Key Pitt Lages		657,455	4,157,377	3,940	7.08	1.45	13.68	1989 to 2007	2	if ni	
Lages	man WMA.	657,315	4,164,774	3,950	7.94	4.64	11.94	1964 to 1989	26	No	MACU
10001		702,945	4,436,011	5,960	8.20	4.83	13.20	1984 to 2007	24	10 yr	WRCC
A LUNCH MAN	Itev Steward	705,365	4,243,927	6,350	15.69	9.39	28.29	1971 to 1998	28	No	WRCC
Locand	ale ale	726,628	4,049,785	1,410	5.14	3.23	9.81	1968 to 1992	25	No	WRCC
N Port		673.483	4,302,024	5,570	10.20	4.99	18.83	1957 to 2007	51	30 yr	WRCC
רחומי		201 044	A 363 530	6300	8.81	3.76	16.21	1914 to 2007	94	30 yr	WRCC
Mcgill		1+6'160	noninnn's			i d		1040 to 2001	en en	No	WRCC
Overtor		728,168	4,047,975	1,290	4.40	1/70	0.34	1940 10 2001	3		JUDIN
Dahran	anat Mildlife Refude	666,918	4,126,112	3,400	6.28	2.23	11.54	1964 to 2007	44	su yr	ANNO
		767 708	4 391 336	4.780	6.68	2.03	12.21	1950 to 2007	58	30 yr	WRCC
Partou	-	2011/101		007.0	VGGY	2 01	27.20	1948 to 2006	59	30 yr	WRCC
3 Pioche		724,042	4,201,109	0,160	07.01	10.0	10.10	1050 44 20007	ED.	10 VL	WRCC
9 Ruth		673,938	4,350,154	6,840	12.04	6.68	19.40	1002 01 0081	8	if an	UDGIVI
Choch	nne 5 N	725,336	4,311,105	5,930	9.65	5.48	14.56	1988 to 2007	20	IN	
	Viellou Stata Bark	747.440	4.212.891	5,950	12.17	5.05	23.48	1974 to 2007	34	10 yr	WRCC
bunde t	Valley oldre r ann	672 777	4.254.266	5,300	9.38	5.73	17.11	1948 to 2007	09 /	30 yr	WRCC
Kuune Z	side	722 534	4.034.875	2,000	6.55	1.66	16.90	1972 to 2007	7 36	30 yr	WRCC

Appendix A

A-3 SE ROA 11803

^aUsed in Figures 4-2 and 4-3. ^bNorth American Datum of 1983, Zone 11



This Page Intentionally Left Blank

Appendix A SE ROA 11804

Appendix B

Groundwater Discharge Areas and Evapotranspiration Rates

SE ROA 11805

JA_4567

B.1.0 INTRODUCTION

This appendix contains maps of the groundwater discharge areas for basins comprising the WRFS, and ET rates reported in the literature and used in this study to derive estimates of predevelopment groundwater discharge. A list of references used is also provided.

B.1.1 Groundwater Discharge Areas

Groundwater discharge occurs as ET and as spring flow from springs located on the valley floor, usually within or near the groundwater ET areas.

Maps depicting groundwater ET distribution under predevelopment conditions and selected spring locations were developed for the following basins:

- Long Valley (HA 175)
- Jakes Valley (HA 174)
- White River Valley (HA 207) and four inset maps for added detail
- Cave Valley (HA 180)
- Garden Valley (HA 172)
- Pahranagat Valley (HA 209)
- Muddy River Springs Area (Upper Moapa Valley) (HA 219)
- California Wash (HA 218)
- Lower Moapa Valley (HA 220)
- Black Mountains Area (HA 215)

The total acreage delineated for the groundwater discharge areas for each of these basins is listed in Table B.1-1, including the estimates derived as part of this study and those of previous investigations. The discharge areas and the distribution of ET classes within the areas are depicted in Figures B.1-1 through B.1-11. It should be noted that the discharge area of White River Valley is presented in five maps. Figure B.1-3 shows the whole valley and the locations of four subareas for which more details are provided in the next four maps (Figures B.1-4 through B.1-7).

B.1.2 ET Rates

A literature survey for annual ET rates measured in the WRFS region was conducted and the rates were compiled into the dataset presented in Table B.1-2. Descriptions of the fields contained in the table are as follows:

Report: reference to report containing information. Consists of author(s) and date. Full
citation is in the reference list.

Appendix B

SE ROA 11806



- Location: Name of valley where ET rate was measured
- ET Unit Description: Description of vegetation present in area where ET rate was measured.
- DTW: Depth to water in area where ET rate was measured.
- Total ET: Total ET rate measured. Includes groundwater ET and precipitation.
- Precipitation: Precipitation rate in area where ET rate was measured.
- Groundwater ET: Measured groundwater ET rate.

HA Name	SNWA	Smith et al.	LVVWD (2001) ^a	Nichols (2000)	USGS Reco Series Repor Resource	nnaissance ts and Water Bulletins
	(2007)	(2007)	(2001)	1.000	Estimate	Author
Leas Valley	17 594	18.283	21,822	21,882	11,000	Eakin (1961)
	971	1.224	416	416	NR ^b	Eakin (1966)
Vhite River Valley	144,692	178,096	163,922		36,000	Maxey and Eakin (1949)
Cave Valley	16.648	13,348	10,293		NR ^b	Eakin (1962)
Cardon Vallov	950		6,144		2,000 to 3,000	Eakin (1963c)
Baltien valley	8.658	-	8,976		9,000	Eakin (1963b)
Panranagat valley	1 951	-	1,016		NR ^b	Eakin (1964)
Muddy River Springs Area	4.070		1 152		1,700 ^c	Rush (1968)
California Wash	1,379		5.004	-	5.600°	Rush (1968)
Lower Moapa Valley	7,533		5,301	_	5,000	Duch (1069)
Black Mountains Area	464		NR		200	Rush (1966)

Table B.1-1 Total Acreage of Groundwater Discharge Areas

*Area includes irrigated lands within the groundwater discharge area.

^bPhreatophyte area was not delineated.

*Does not include 1,000 and 3,400 acres of irrigated land in California Wash and Lower Moapa Valley, respectively.



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Figure B.1-1 Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Long Valley

Appendix B



Southern Nevada Water Authority - Water Resources Division



Note: Illipah Creek Discharge reported in USGS, 2006.

B-4

Figure B.1-2

Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Jakes Valley

> Appendix B SE ROA 11809



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Note: Also shown are four subareas for which more details are provided in Figures B.1-4 through B.1-7

Figure B.1-3

Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in White River Valley

B-5

SE ROA 11810





Figure B.1-4 White River Valley Discharge Area - Inset 1



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Figure B.1-5 White River Valley Discharge Area - Inset 2

Appendix B

SE ROA 11812





Figure B.1-6 White River Valley Discharge Area - Inset 3



Figure B.1-7 White River Valley Discharge Area - Inset 4

Appendix B

SE ROA 11814





Figure B.1-8 Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Cave Valley



Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Garden Valley

Appendix B





Figure B.1-10

Distribution of Predevelopment Groundwater Evapotranspiration and Location of Selected Springs in Pahranagat Valley

B-12

JA_4579



Figure B.1-11 Distribution of Predevelopment Groundwater ET and Location of Selected Springs in Muddy River Springs Area, California Wash, Lower Moapa Valley, and Black Mountains Area

Appendix B



Southern Nevada Water Authority - Water Resources Division

Lee (1912) p. 56, and 119-122 Owen: White (1932) p. 99 and 100 Escal: Maxey and Eakin (1949) White	Location	ET Unit Description	Depth to Water (#)	Total ET (ft/yr)	Precipitation (ft/yr)	Groundwater ET (ft/yr)
Lee (1912) Independent p. 56, and 119-122 Owen: White (1932) Escala p. 99 and 100 Escala Maxey and Eakin (1949) White				43	{	:
Des (1912) Owens p. 56, and 119-122 Owens White (1932) Escalt p. 99 and 100 Escalt Maxey and Eakin (1949) White p. 44 White	andanna area of	Alfalfa	67th 60	333	1	1
P. 30, and 113-124 Owens White (1932) Escalt p. 99 and 100 Maxey and Eakin (1949) White p. 44	e Valley CA	Bare Soil (Tank 1)	0.1 10 0.9	000 TO 00 V		-
White (1932) p. 99 and 100 Maxey and Eakin (1949) White p. 44	s valiey, on	Saltgrass (Tanks 2-7)	0.1 to 6.0	0.66 to 3.59		
White (1932) Escala p. 99 and 100 Maxey and Eakin (1949) White p. 44 White		Greasewood	1.3 to 3.3	0.99 to 2.1 ^a	0.13 to 0.37°	ł
p. 99 and 100 Maxey and Eakin (1949) White p. 44	ante Vallev. UT	Alfalfa	2.3 to 3.3	1.92 to 2.16 ^a	0.13 to 0.43 ^a	ł
Maxey and Eakin (1949) White		Saltorass	0.92 to 2.3	0.76 to 1.88 ^a	0.01 to 0.37 ^a	1
Maxey and Eakin (1949) White p. 44		Native phreatophytes	1	1		0.8
p. 44	River Valley	Cutivitied crone	1	1	1	1.25
		Colucator depo	0.26 to 0.65	3.3 to 8.6	0.0 to 0.22	4
Gatewood et al. (1900) Catewood	of Mellow, A7	Cattonucod	0.42 to 0.57	3.4 to 8.7	0.0 to 0.22	ł
p. 123 -128	To valiey, AZ	Dombrid	0.14 to 0.51	2.9 to 7.1	0,0 to 0.22	1
		Microd sublithereh and graacowood	10 to 40	1	1	0.2
Eakin (1961) p. 23 Long	valley			-	1	1
Eakin (1962) p. 9 Cave	Valley	Main drainage channel			1	1
Eakin (1963a) p. 1 and 18 Dry L	ake/Delamar valleys	No phreatophytes, discharge from springs	1			
Eakin (1963c) p. 19 Garde	en/Coal valleys	Along stream channels and spring areas	1	1	1	01.101
Eakin (1983b) n 19 20 Pahra	anagat/Pahroc vallevs	Phreatophytes, agriculture, and lakes	1		1	1.0 10 4.0
דמאוני (12000) אי יבי בי		Saltorass and wetland meadow	\$	}	1	1.5
		Saltorace and mod wet meadow	\$	1	ł	1.25
Rush and Eakin (1963)	Vellev	Saltorass and dry meadow	5 to 20	1		0.5
p. 13	(mana)	Greasewood	20 to 25	ł		0.25
		Greasewood and rabbitbrush mixed with big sage	25 to 50	1	ł	0.1
		Drv meadow	5 to 10	1	ł	0.5
Clove	er Valley	Rabbithursh	10 to 25	1	1	0.2
1 ind	Allow.	Pabhithrush and big sade	15 to 40	1	ł	0.1
(in)	valicy	Dri meadaw	;	l	1	0.5
Eagle	e Valley	Dobbithoush and his care	10 to 15	1	1	0.3
		Nauvitariusi anv vig vege	20 to 60	Ĩ	1	0.1
		Saturusii	20 to 50	1	1	0.1
Rush (1964) p. 22 Lowe	er Meadow Valley	Greasewood and rapping and amondo high	1	1	1	0.1
		Greasewoou, raubilulusit, ariu decoste doni	0 to 5	1	1	3,0
		Cottonwood, Willow, and sanceual	40 to 20	1	1	0.1
c	the Mallant	Greasewood and rabbitbrush	101020			00
	aca valley	Rabbitbrush	15 to 25	1	1	1.1
Patte	erson Valley	Greasewood, rabbitbrush	20 to 30	1	1	1.0
Ros	a Mallev	Rabbitbrush and big sage	20 to 25	1	1	0.1

B-14

Appendix B SE ROA 11819

le B.1-2	ted in the Literature (e 2 of 6)
Т	tes Rep (F
	ET Ra

Report	Location	ET Unit Description	Depth to Water (ft)	Total ET (ft/yr)	Precipitation (ft/yr)	Groundwater ET (ft/yr)
		Dathitheret	5 to 25	f		0.1
Rush (1964) p. 22	Spring Valley	Methand meadow (very wet meadow and dry meadow)	0 to 10	1	1	0.5 to 1.5
cont.	Contro Control	Not velipeated	1	int.	1	1
Takin (1964))	Coyote aprilig	Not delineated	1	-	1	ł
5. 24 and 25	Kane spinig		1	1	1	1
	Muddy River Springs	Springs area	0 to 5	1	1	1.5
		Wetland meadow and sailgrass	0 to 10	1	1	1.0
		Saligrass, rabolitorusri, and moust area, wermeadow	5 to 15	1	ł	0.5
Rush and Kazmi (1965)	Soring Valley	Greasewoou, saiglass, incador glaco, and comp	10 to 50	1	ł	0.2
p. 23			5 to 15	1	1	0.1
		Bare soil and sparse vegerauon	0 to 5	1	1	2.0
		Cottonwood, willow, and wildrose	0.10.5	1	1	1,75
		Wetland meadow	01010	1	-	0.5
10071 1 01		Meadow grass and rabbitorush	10 10 50	1	1	0.2
Hood and Kush (1905)	Snake Valley	Greasewood and rabbitbrush	01010		1	0.75
c7.d		Playas (flooded part of year)	00.410		1	0.1
		Playas (rarely flooded, but shallow water table)	U TO 3U	1		
Enlin (1066) n 361	Jakes Vallev	Not delineated	1	1	ľ	
Eduit (1900) p. 201	four cours	Metland meadow and saltgrass, including lowland spring areas	0	ł	f	61
		Salidrass. rabbitbrush, greasewood, and Goshute Lake Playa	<10	-	1	0.5
Eakin, Hughes and Moore	Steptoe Valley	Createswood rabbithrish and saltdrass	<20	-	ł	0.3
(1961) p. 25		Polymore whithereb and greasemend	<12	ł		0.1
		Daligiass, lauditulusit, and gicastrood	1	1	ţ	3.5
		Open water	01040	1.11	1	0.5
	C Butto Vollow	Saltgrass	101010		1	0.2
Glancy (1968) p. 28	o. putte valley	Rabbitbrush and greasewood	00 01 01			10
		Greasewood	30 to 50+	1	1	110
	Hidden Vallev	None	1	ł	ł	
	Carnet Valley	None	1	1	ł	1
	Californio Mech Area	Salifyinsh saltorass saltoedar, mesquite, and cottonwood	2 to 50	ł	1	1.0
Rush (1966) p. 55	California wash Area	Solthinsh coltonass saltredar mesouite. cottonwood, and tules	2 to 50	1	ţ	2.0
	Lower moapa valiey	Tulos mossings (IV Mash and Roners Spring)	0 to 5	1	1	6.0
	Black Mountains Area		5.0 to 7.8	1.14 to 1.81 ^a	0.14 to 0.44 ^a	1
		Dethihtich	5.0 to 6.2	1.07 to 2.19ª	0.14 to 0.44 ^a	1
Robinson (1970)	Humboldt River Valley, NV	Rabbind usit	1.9 to 4.0	0.36 to 1.00 ^a	0.14 to 0.44 ^a	1
p. D28		Date Soll	35 to 5.8	2.07 to 3.94ª	0.14 to 0.44 ^a	ł

Appendix B

SE ROA 11820

Southern Nevada Water Authority - Water Resources Division

			Parts to			Groundwater
Report	Location	ET Unit Description	Depth to Water (ft)	Total ET (ft/yr)	Precipitation (ft/yr)	ET (ft/yr)
10 - 10 10 10 10 10 10 10 10 10 10 10 10 10	Transf Vollavi	No phreatophytes	>50	1	1	ł
Hamil (1971) p. 21	Lipher varies	Alkali meadow (sacaton and thistle)	10 to 15	2.71		1
		Alkali meadow (saltorass, sacaton, and rabbitbrush)	7 to 9	2.05	ł	1
		Alkali meadow (NV saltbush, sacaton, and rabbitbrush)	5 to 7	2.69	-	1
Duck! (4000)		Desert sink scrub (rabbitbrush, sacaton, and ephedra)	10 to 11	1.99	I	i.
Duell (1990) D. E26	Owens Valley, CA	Desert sink scrub (saltgrass and greasewood)	8 to 9	1.26	0.52	
		Rabbitbrush meadow (saltgrass, rabbitbrush, sacaton, and	10 to 11	1.55	0.49	ľ
		Rush and sedge meadow (saltorass, sacaton, and rush)	54	3.24	0.26	ł
101210	C 11 C 11 1 1 1 1 1	Social also (greasewood)	25 to 30	0.59	0.49 to 0.66	
Carman (1993)	Smith Creek valley and	Smith Creak (rehhithnish)	10 to 15	1.05	0.66 to 0.98	-
p. z and to		Smoke Creek desert (creasewood, salt brush, and sagebrush)	8.9	ľ	1	0.82 ^a
Nichols (1994)	NV dead Bank NV	Cmith Creek Valley (nreasewood and rabbitbrush)	5.9	1	1	1.2ª
Trom Nichols (2000)	NOTIFICITION OF CALIBRATINO	Raitmad Valley (greasewood, salt brush, and sagebrush)	5.9	ļ	1	0.2ª
Printed - (4007)		Ash Meadows 1 (salt grass)	1.6	-	-	2.45
from Nichols (2000)	Ash Meadows, NV	Ash Meadows 2 (salt grass and wiregrass)	0	ł	1	2.52
Devitt et al. (1998)	Virgin River, NV	Riparian (mostly tamarix)	ł	2.5 to 4.8	ł	1
p. 2,40/ and 2,410		Open water body	ł	8.6	1	61
		Submerged aquatic vegetation (shallow part of open water areas, includes sparse emergent vegetation)	1	8.6	į	1
100001 10 10 10 10 10 10		Dense grassland vegetation (grasses, short rushes, and trees)	1	3.5	1	ł
Laczniak et al. (1999)	Ash Meadows, NV	Sparse grassland vegetation (grasses)	1	1.3	ł	1
		Moist bare soil (grasses)	-	2.6	1	1
		Dense wetland vegetation (reedy and rushy marsh plants)	ţ	3.9	1	1
		Dense meadow vegetation (trees, grasses, and shrubs)	1	3.4	1	1
		Open water	ł	1	1	1
		Plava / hara soil	ł	1	-	0.15
		<10 percent plant cover	1	1		0.29 to 0.41
Nichols (2000)	16 valleys in Great Basin,	10 to <20 nercent plant cover	1	1	ł	1.28 to 1.35
p. C16	NN	10 to < 35 hercent plant cover	1	ł	1	2.14 to 2.15
		36 to < 50 percent plant cover	1	1	1	2.50
				1	-	2.58

Appendix B SE ROA 11821

Table B.1-2	T Rates Reported in the Literature	(Page 4 of 6)
	ETR	

	Location	ET Unit Description	Water (ft)	Total ET (ft/yr)	Precipitation (ft/yr)	ET (ft/yr)
		Onon water	0	8.4 to 8.8	1	
		Submerged aquatic vegetation (shallow part of open water areas, Submerged aquatic vegetation (shallow part of open water areas, includes enveroemt vegetation)	0	8.1 to 8.5	1	1
		Shares are stand veneration (grasses and very low density shrubs)	few to 12	0,6 to 2.3	ł	
		Sharse woodland vedetation (mesquite)	10 to 40	0.7 to 1.8	1	;
itte	Allow Decional Flow	Dense meadow / forested vegetation (trees, grasses, and shrubs)	few to 20	3.0 to 4.0	1	-
aczniak et al. (2001) ^b System . 13 and 29	valley regional row i, primarily Ash we and Dacis Valley	Sparse to moderately dense shrubland vegetation (greasewood, rab- hithrush wolfberry, and seepweed)	5 to 20	0.7 to 2.5	1	1
		Dense wetland vedetation (reedy and rushy marsh plants)	0	3.7 to 4.3	4	1
		Dense to moderately dense grassland vegetation (saltgrass and/or chord nucless with occasional tree or shrub)	< 5 <	2.5 to 3.7	1	ł
		Maint Dara Cail (varu enarce prasses)	<5	2.2 to 3.0	;	1
		Moist Bare don (rei) operational and a second	5 to 40	0.1 to 0.7	1	1
		Open riaya (uale son)	1	5.31 ^a	0.65ª	ł
		Open water (submerged aquaito vegetation)	17	1.33ª	0.65 ^a	0.68ª
		Greasewoou, radonu usit, wild fyc, organization (providerate)	5	1,33 ^a	0.65ª	0,68ª
		Raubiulusi, wilding, groadmood, and samphish (moderate)	10	1.33 ^a	0.65ª	0.68ª
		Kabolitorush, wildhye, greasewood, and age and moderate)	\$5	1.33 ^a	0.65ª	0.68ª
Berger et al. (2001) Ruhv I	Lake National Wildlife	Saliglass, laboliorush, microst, microst accelerate and rabhithrush)	>80	0.99 ^a	0.65ª	0.34ª
o. 8, 16, 23, and 24 see Refuge	e Area, Ruby Valley	Desert still up uptartu (intoderate degeo and catalla)	1 10 3	4.195	0.65ª	3.54 ^a
footnates		Buindsh matsh (moderate to derise buildsh and drasses)	2	3.19ª	0.65 ^a	2.54 ^a
		Meadow (dense mixed serges, rusines, and graded)	\$	1.32ª	0.65 ^a	0.67ª
		Saltgrass, rabbitorusn, wildrye, and greasewood (moderate)	1	2.36ª	0.65ª	1.71ª
		Grasslands	1	0.80ª	0,65ª	0.15ª
		Playarbare soil Onen water body	1	8.6	0.50	8.1
		Submerged and sparse aquatic vegetation (shallow part of open Submerged and sparse aquatic vegetation)	0	8.6	0.50	8.1
		Water areas, incurso operation (Reedy and rushy marsh plants)	0	3,9	0.50	3.4
		Dense meadow and woodland vegetation (trees, grasses, shrubs)	0 to 20	3.3	0.50	2.8
Reiner et al. (2002) Oasis p. 17 and 30	s Valley, NV	Moderately dense to dense grassland (grasses, short rushes, occa- moderately dense to dense grassland (grasses, short rushes, occa- dional eratement trees.)	<10	3.2	0.50	2.7
		Source to moderately dense grassland (grasses)	few to 10	2.0	0:50	1.5
		Apartse to inconstancy across second as the soil (arasses)	S	2.6	0.50	2.1
		Sparse to moderately dense shrubs (greasewood, rabbitbrush, and wolfberry)	5 to 20	1.2	0.50	0.7

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Appendix B

1.2	10
1	1111
0	14
	-

Southern Nevada Water Authority - Water Resources Division

		(Page 5 of 6)				
Report	Location	ET Unit Description	Depth to Water (ft)	Total ET (ft/yr)	Precipitation (ft/yr)	Groundwater ET (ft/yr)
		1	13 to 14	0.59	0.35	0.25
		Nev. saltbush scrup	>16	0.31	0.11	0.19
		Kapoliprush scrub	6.9 to 7.9	0.71 to 1.06	0.10 to 0.30	0.61 to 0.74
Harrington & Steinwand	Owens Valley, CA	Nev. saltoush meadow	>13	0.49 to 0.80	0.10	0.31 to 0.38
(2003, 2004)		Desert sink scrup	8.5 to 11	1.68	0.23	1.44
		Rabbitbrush meadow	6.6 to 11	1.37 to 2.25	0.11 to 0.99	0.87 to 2.18
		Alkali meadow (sacatori, sanglass, and morty)	\$10	0.21 to 0.37	0.06 to 0.35	0.15 to 0.01
		bare to soil playa	5 to 20	0,60	0.06	0.54
		Moderate density vegetation (grasses, arrowweed, mesquite, and	2 to 20	2.0	0.23	1.6
DeMeo et al. (2003)	Death Valley, CA	pickleweed)	00>	2.9	0.16	2.7
b. 0' 20' allo 24		High density vegetation (grasses and mesquices)	002	3.9	0.25	3.6
		High density vegetation (grasses and mesquires)	~	01710020	0.04 to 0.28	-1.40 to 0.13
		Salt to encrusted playa	55	100,000		1
		Rabbitbrush and greasewood	000	n		1
		Flood imigated alfalfa	3 to 6	3.1	I	
		Flond initiated pasture	2 to 5	3.2	1	1
and the second se	Carson Valley and Douglas	Flood irrigated pasture	3 to 4	2.8	1	1
Maurer et al. (2006)	County, NV, and Alpine	Flond irritated alfalfa	40	3.0	1	1
b. a' 10' alla 77	County, CA	Non invite asture	6 to 7	1.7		1
		Riterbraish and sade	60	1.5	ł	:
		Flood initiated basture	0 to 2	4.4	1	1
		High density saltcedar (saltcedar)	8	3.53	1	+
Westenburg et al. (2006)	Topock Wash, Havasu area,	Medium density, mixed vegetation (mesquite, saltoedar, salt grass, ammuned harcharis, rush)	4 to 8.	2.39	1	-
p. 10 and 21	Lower Lotorado River, 22	I mutto modium density arrowneed (Arrowneed)	80	2.35	1	ł
		Low to internation density another of the second se	3.0	1.34	1	0.74
Cooper et al. (2006)	San Luis Valley, CO	Greasewood and rabbithrish	8.2	0.91	Ì	0.28
p. 21 Laczniak et al (2006)	Contraction CA	saligrass, grass, yerba mansa, arrowweed, desert baccharis, mes-	0 to 15	2.28 to 2.68	0.05 to 0.49	2.1 to 2.3
p. 12 and 30	Deatri valiey, on	quite	1	1.28	0.51	0.77
Devitt et. al (2006 data.	Spring Valley	Bare soil/low vegetation (greasewood and sage)		77.8	0.87	0.90
Innoublished)	White River Valley	Medium vegetation (greasewood)		144		

Appendix B SE ROA 11823

	iterature	
B.1-2	id in the L	: 6 of 6)
Table	Reporte	(Page
	ET Rates	

Report	Location	ET Unit Description	Depth to Water (ft)	Total ET (ft/yr)	Precipitation (ft/yr)	Groundwater ET (ft/yr)
		Month (sother indirection introductions (bambool)	1	5.93		
		Marsh (caual, pundar, pundarmooter.	1	1.34	1	-
		Desert Vegetation		1.66	1	
		Barren (510% vegetation)	1	4.36	1	
		Saltoedar (11 to 60%), arrowneed (>23%)	1	5.12	1	1
		Saltcedar (61 to 100%), anowweed (250%)	1	5.45	1	
		Saficedar (11 to bu%), intesquire (11 to busy, another of the comparison of the comp		5,29	1	1
LCRAS (BOR, 1995-	I ower Colorado River, AZ	Satteedar (15 to 45%), mesquite (15 to 45%), arrowweed (20 to	1	5,43	1	ł
2005) Fort Mohave		40%)	1	3.99	1	ł
Reservation		Sciewocalitioties incognic (1) to 300% arrowieed (25%)	1	4,99		ł
		SCIEWOEarIntoliey Intesquite (01.10.100.00) anticedar (<20%)	1	4.76		1
		Mesquire (21 to 30 /6), another (21 /00 / 20 / 20 / 20 / 20 / 20 / 20 / 2	1	4.69	-	-
		Arrowweed (31.10 100%), uces (510%)	1	4.96	1	1
		Cottonwood and willow trees (or 10 100 /o)		4 29	1	1
		Low Veg (phreatophyte vegetation >10% to \$30%)		000		1
		Moist Soil Unit (flooded in winter and irrigated in summer)		5.03		
		Reasonal Methand (flooded in winter and not irrigated in summer)	1	3.65	ŀ	1
		Seasonal vound (nease thribland (neasewood)	17.16	0.836	0.52	0.32
		Moderately delise descriptionary (wrassewrood and rahhitbrush)	9.78	0.835	0.72	0.12
		Sparse deset structure (greater shrinkand (greasewood and rabbitbrush)	7.24	1.01	0.76	0.24
Moreo et al. (2007)	Snake, Spring, and White	Modelately derive deservation and analogical and analogical and and analogical and an	3.89	2.25	0.66	1.58
p. 13, 15, and 20	River valleys		32.39	1.06	0.74	0.32
		Dense desert strubiario (greasewouu)	23.58	1.02	0.95	0.06

^aRates reflect growing season only or less. ^bLaczniak et al. (2001) relies on rates from previously published literature.

Southern Nevada Water Authority - Water Resources Division

B.2.0 REFERENCES

- Berger, D.L., Johnson, M.J., Tumbusch, J.L., Mackay, J., 2001, Estimates of evapotranspiration from the Ruby Lake National Wildlife Refuge area, Ruby Valley, northeastern Nevada, May 1999–October 2000: U.S. Geological Survey Water-Resources Investigations Report 01-4234, 38 p.
- Carman, R.L., 1993, Measurement of evapotranspiration in phreatophyte areas, Smith Creek Valley and Carson Desert, west-central Nevada, 1983: U.S. Geological Survey Water-Resources Investigations Report 89-4118, 18 p.
- Cooper, D.J., Sanderson, J.S., Stannard, D.I, and Groeneveld, D.P., 2006, Effects of long-term water table drawdown on evapotranspiration and vegetation in an arid region phreatophyte community: Journal of Hydrology, Vol. 325, p. 21–34.
- DeMeo, G.A., Laczniak, R.J., Boyd, R.A., Smith, J.L., and Nylund, W.E., 2003, Estimated ground-water discharge by evapotranspiration from Death Valley, California, 1997–2001: U.S. Geological Survey Water-Resources Investigations Report 2003-4254, 27 p.
- Devitt, D.A., 2007, Personal communication to R. Shanahan (Southern Nevada Water Authority) regarding evapotranspiration in Spring and White River valleys, 29 and 30 August.
- Devitt, D.A., Sala, A., Smith, S.D., Cleverly, J., Shaulis, L.K., and Hammet, R., 1998, Bowen ratio estimates of evapotranspiration for *Tamarix Ramosissima* stands on the Virgin River in southern Nevada: Water Resources Research, Vol. 34, No. 9, p. 2407–2414.
- Duell, L.F.W., Jr., 1990, Estimates of evapotranspiration in alkaline scrub and meadow communities of Owens Valley, California, using the Bowen-ratio, eddy-correlation, and Penman-combination methods: U.S. Geological Survey Water-Supply Paper 2370-E, 49 p.
- Eakin, T.E., 1961, Ground-water appraisal of Long Valley, White Pine and Elko counties, Nevada: Nevada Department of Conservation and Natural Resources Ground-Water Resources Reconnaissance Series Report 3, 35 p.
- Eakin, T.E., 1962, Ground-water appraisal of Cave Valley in Lincoln and White Pine counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 13, 19 p.
- Eakin, T.E., 1963a, Ground-water appraisal of Dry Lake and Delamar valleys, Lincoln County, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 16, 25 p.



- Eakin, T.E., 1963b, Ground-water appraisal of Pahranagat and Pahroc valleys, Lincoln and Nye counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 21, 36 p.
- Eakin, T.E., 1963c, Ground-water appraisal of Garden and Coal valleys, Lincoln and Nye counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 18, 29 p.
- Eakin, T.E., 1964, Ground-water appraisal of Coyote Spring and Kane Spring valleys and Muddy River Springs Area, Lincoln and Clark counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 25, 39 p.
- Eakin, T.E., 1966, A regional interbasin ground-water system in the White River area, southeastern Nevada: Water Resources Research, Vol. 2, No. 2, p. 251–271.
- Eakin, T.E., Hughes, J.L., and Moore, D.O., 1967, Water-resources appraisal of Steptoe Valley, White Pine and Elko counties, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 42, 45 p.
- Gatewood, J.S., Robinson, T.W., Colby, B.R., Hem, J.D., and Halpenny, L.C., 1950, Use of water by bottom-land vegetation in lower Safford Valley, Arizona: U.S. Geological Survey Water-Supply Paper 1103, 209 p.
- Glancy, P.A., 1968, Water-resources appraisal of Butte Valley, Elko and White Pine counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 49, 50 p.
- Harrill, J.R., 1971, Water-resources appraisal of the Pilot Creek Valley area, Elko and White Pine counties, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 56, 46 p.
- Harrington, R., and Steinwand, A., 2003, Development of hydrologic and vadose models to improve groundwater management in the Owens Valley: Inyo County (California) Water Department, Bishop, California, 217 p.
- Harrington, R., and Steinwand, A., 2004a, (Addendum to) Development of hydrologic and vadose models to improve groundwater management in the Owens Valley: Inyo County (California) Water Department, Bishop, California, 49 p.
- Harrington, R., and Steinwand, A., 2004b, Personal communication to R. Shanahan (Southern Nevada Water Authority) and T. Katzer regarding evapotranspiration in Owens Valley, 3 November.
- Hood, J.W., and Rush, F.E., 1965, Water-resources appraisal of the Snake Valley area, Utah and Nevada: Nevada Department of Conservation and Natural Resources, Water-Resources Reconnaissance Series Report 34, 40 p.

Appendix B

SE ROA 11826



- Laczniak, R.J., DeMeo, Reiner, S.R., Smith, J.L., and Nylund, W.E., 1999, Estimates of ground-water discharge as determined from measurements of evapotranspiration, Ash Meadows area, Nye County, Nevada: U.S. Geological Survey Water-Resources Investigations Report 99-4079, 70 p.
- Laczniak, R.J., Smith, J.L., and DeMeo, G.A., 2006, Annual ground-water discharge by evapotranspiration from areas of spring-fed riparian vegetation along the eastern margin of Death Valley, 2000–02: U.S. Geological Survey Scientific Investigations Report 2006-5145, 46 p.
- Laczniak, R.J., Smith, J.L., Elliot, P.E., DeMeo, G.A., Chatigny, M.A., Roemer, G.J., 2001, Ground-water discharge determined from estimates of evapotranspiration, Death Valley Regional Flow System, Nevada and California: U.S. Geological Survey Water-Resources Investigations Report 01-4195, 51 p.
- Lee, C.H., 1912, An intensive study of the water resources of a part of Owens Valley, California: U.S. Geological Survey Water-Supply Paper 294, 135 p.
- Maurer, D.K., Berger, D.L., Tumbusch, M.L., Johnson, M.J., 2006, Rates of evapotranspiration, recharge from precipitation beneath selected areas of native vegetation, and streamflow gain and loss in Carson Valley, Douglas County, Nevada, and Alpine County, California: U.S. Geological Survey Scientific Investigations Report 2005-5288, 70 p.
- Maxey, G.B., and Eakin, T.E., 1949, Groundwater in the White River Valley, White Pine, Nye, and Lincoln counties, Nevada: State of Nevada, Office of the State Engineer, Water Resources Bulletin No. 8, 59 p.
- Moreo, M.T., Laczniak, R.J., and Stannard, D.I., 2007, Evapotranspiration rate measurements of vegetation typical of ground-water discharge areas in the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah, September 2005–August 2006: U.S. Geological Survey Scientific Investigations Report 2007-5078, 37 p.
- Nichols, W.D., 1994, Groundwater discharge by phreatophyte shrubs in the Great Basin as related to depth to groundwater: Water Resources Research, Vol. 30, No. 12, p. 3265–3274.
- Nichols, W.D., 2000, Determining ground-water evapotranspiration from phreatophyte shrubs and grasses as a function of plant cover and depth to ground water, Great Basin, Nevada and eastern California, p. A1–A14, *in* Nichols, W.D., Regional ground-water evaporation and ground-water budgets, Great Basin, Nevada: U.S. Geological Survey Professional Paper 1628, 108 p.
- Nichols, W.D., Laczniak, R.J., DeMeo, G.A., and Rapp, T.R., 1997, Estimated ground-water discharge by evapotranspiration, Ash Meadows area, Nye County, Nevada, 1994: U.S. Geological Survey Water-Resources Investigations Report 97-4025, 13 p.
- Reiner, S.R., Laczniak, R.J., DeMeo, G.A., Smith, J.L., Elliot, P.E., Nylund, W.E., and Fridrich, C.J., 2002, Ground-water discharge determined from measurements of evapotranspiration, other available hydrological components, and shallow water-level changes, Oasis Valley, Nye County, Nevada: U.S. Geological Survey Water-Resources Investigations Report 01-4239, 65 p.

- Robinson, T.W., 1970, Evapotranspiration by woody phreatophytes in the Humboldt River Valley near Winnemucca, Nevada: U.S. Geological Survey Professional Paper 491-D, 41 p.
- Rush, E.F., 1964, Ground-water appraisal of the Meadow Valley area, Lincoln and Clark counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 27, 43 p.
- Rush, E.F., 1968, Water-resources appraisal of the Lower Moapa-Lake Mead area, Clark County, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 50, 67 p.
- Rush, E.F., and Eakin, T.E., 1963, Ground-water appraisal of Lake Valley in Lincoln and White Pine counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 24, 43 p.
- Rush, F.E., and Kazmi, S.A.T., 1965, Water Resources appraisal of Spring Valley, White Pine and Lincoln counties, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 33, 36 p.
- U.S. Bureau of Reclamation, 1995 to 2005, Lower Colorado River Accounting System evapotranspiration calculations calendar years 1995 to 2005: Bureau of Reclamation, Lower Colorado River Region, Boulder City, Nevada.
- U.S. Geological Survey, 2006, Water Resources Data Nevada Water Year 2006, U.S. Geological Survey Water-Data Reports NV-06-1.
- Westenburg, C.L., Harper, D.P., DeMeo, G.A., 2006, Evapotranspiration by phreatophytes along the lower Colorado River at Havasu National Wildlife Refuge, Arizona: U.S. Geological Survey Scientific Investigations Report 2006-5043, 44 p.
- White, W.N., 1932, A method of estimating ground-water supplies based on discharge by plants and evaporation from soil—Results of investigations in Escalante Valley, Utah: U.S. Geological Survey Water-Supply Paper 659-A, 209 p.





This Page Intentionally Left Blank

B-24

Appendix B SE ROA 11829

Appendix C

Spring Data

SE ROA 11830

C.1.0 SPRING DISCHARGE

For groundwater discharge areas where there are gaged springs, estimates of groundwater discharge based on the spring discharge records can be used to estimate the lower bound of groundwater discharge from the area.

Table C.1-1 lists the historical averages of groundwater discharge for gaged springs in White River Valley, Pahranagat Valley, and the Muddy River Springs Area.

Table C.1-2 lists the individual discharge measurements of these springs for their periods of record.

Table C.1-3 lists the annual average discharge of these springs based on the available data.

Note that spring discharge measurements for the Muddy River Springs Area are actually reported as stream flow measurements on the Muddy River.



C-1
						1	ade i vi zj				-	
s Ber	g Name	Data Collection Frequency	HA	UTM Easting	UTM Northing	Elevation (ft-ams1) ^a	Local Number	Mean Average Annual Discharge (cfs)	Period of Record (WY)	Period of Record Duration (No. of Years) ^b	No. of Discharge Measurements Used in Analysis ^c	Remarks
						Wh	ite River Valley					
to1 Forrest Ho	me Creek st Home, NV	Miscellaneous	207	643,629	4,247,921	6,213	207 N06 E59 21BC	1.79 ^d	2003	-	ł	
501 Hot Creek Sunnyside	Spring near , Nevada	Continuous	207	661,573	4,249,541	5,229	207 ND6 E61 18AADA	11.30	1935-2006	72	41	
501 Arnoldson	Spring	Miscellaneous	207	667,919	4,308,473	5,625	207 N12 E61 12DCCD	3.66	1911-2006	26	64	
701 Cold Sprin	0	Miscellaneous	207	66,7609	4,309,454	5,653	207 N12 E61 12BDAD	1.30	1911-2006	25	57	
901 Preston Bi	ig Spring	Continuous	207	666,296	4,311,153	5,732	207 N12 E61 02ACAB	7.86 ^e	1983-2006	60	1	
001 Lund Sprir	δu	Miscellaneous	207	673,266	4,302,019	5,608	207 N11 E62 04AABA	7.94	1911-2006	26	47	
101 Moorman	Spring	Miscellaneous	207	662,053	4,273,440	5,299	20 N09 E61 32DABC	0.500	1935-2006	72	38	
201 Shingle Si	pring	Miscellaneous	207	679,925	4,267,716	6434	207 N08 E63 19AA	0.003	1979-2004	26	2	
301 Flag Sprin	ngs 3	Miscellaneous	207	672,579	4,254,416	5,294	207 N07 E62 33BCCC	2.16	1949-2006	58	39	
302 Flag Sprir	ngs 2	Miscellaneous	207	672,576	4,254,570	5,285	20 N07 E62 33BCCB	2.94	1982-2006	25	34	
303 Flag Sprir	ngs 1	Miscellaneous	207	672,719	4,254,697	5,294	207 ND7 E62 33BCAB	2.29	1982-2006	25	35	
401 Butterfield	d Spring	Miscellaneous	207	673,530	4,256,472	5,324	207 N07 E62 28ABDC	2.73	1949-2006	89	39	
501 Hardy Sp	stings	Miscellaneous	207	667,553	4,278,196	5,354	207 N09 E61 13CB	0.450	1967-2004	39	2	
502 Hardy Sp	vring NW	Miscellaneous	207	667,352	4,278,196	5,349	207 N09 E61 13CB	0.010	2004	٠	1	
601 Nicholas	Spring.	Miscellaneous	\$ 207	668,104	4,308,847	5,635	207 N12 E61 12DBDD	2.71	1911-2006	26 5	99	
1801 Williams	Hot Spring	Miscellaneous	\$ 207	653,089	4,312,874	6,300	207 N13 E60 33AB	0,16	1935-200	11	e	
1901 Moon Riv	ver Spring	Miscellaneous	\$ 207	658,908	4,246,394	5,223	207 N06 E60 25BDAD	4.01	1979-1991	16	80	
-	Contract	Minanglana	207	669 895	4.276.841	5,480	207 N09 E62 19DB	1.63	1985-199-	4 10	14	

Southern Nevada Water Authority - Water Resources Division

Appendix C SE ROA 11832

JA_4594

Site umber	Spring Name	Data Collection Frequency	АН	UTM Easting	UTM Northing	Elevation (ft-ams1) ^a	Local Number	Mean Average Annual Discharge (cfs)	Period of Record (WY)	Period of Record Duration (No. of Years) ^b	No. of Discharge Measurements Used in Analysis ^c	Remarks
						Pa	hranagat Valley					
10101	Hiko Springs	Miscellaneous	209	657,549	4,162,744	3,878	209 S04 E60 14DBAB	5.76	1913-2004	98	26	
190201	Solar Panel Spring	Miscellaneous	209	667,262	4,123,643	3,238	209 S08 E61 23BABD	0.001	2004	Ł	1	
090401	Crystal Springs	Continuous	209	656,165	4,155,348	3,803	209 S05 E60 10AD	12.7*	2005-2006	2	ł	Based on Crystal Sprin and Diversion gages (2005-2006)
090501	Ash Springs	Cantinuous	209	659,684	4,147,460	3,622	209 S06 E61 01AD	16.5°	2005-2006	6	1	Based on Ash Springs and Diversion gages (2005-2006)
107060	Brownie Spring	Miscellaneous	209	658,088	4,149,897	3,695	209 S05 E60 26DAD 1	0.50	1963	+	۲	Estimate from Eakin, 1963 0.5-1.0 cfs
						Muddy	River Springs Area					
150101	Muddy River near St. Thomas, NV	Continuous	215	737,700	4,037,477	1,100	220 S17 E68 13C	19.31	1913-1916	Ŧ	÷.	Water Year 1914 is on complete water year.
9416000	Muddy River near Moapa, NV	Continuous	219	705,855	4,065,298	1,712	219 S14 E65 15DD	41.0 ^e	1913-2006	8	ſ	
9419000	Muddy River near Glendale, NV	Continuous	220	719,946	4,058,093	1,460	220 S15 E67 07CA	42.6ª	1950-2006	55	1	

Appendix C

C-3 SE ROA 11833



Southern Nevada Water Authority - Water Resources Division

Appendix C SE ROA 11834

Table C.1-2	neous Discharge Measurements of Selected Springs in the WRFS	(Page 2 of 19)
	Miscellaneous I	

Used in this Study Analysis	Yes	No	No	Yes													
Remarks		Discharge appears low.	Discharge appears low.	1	1	ł	ł	l	ł	1	ł	ł	-	1	ł	1	ł
Data Source	USGS, 1993	USGS-NWIS, 2004	USGS, 1994	USGS, 1995	USGS-NWIS, 2004												
Measurement Rating ^a (E, G, F, P)	1	1	ł	1	l	ł	Ì	1	1	l	ţ	1	ł	ł	1	ł	ł
Discharge (cfs)	8.80	1.10	2.80	12.0	14.1	12.1	10.2	10.4	10.6	15.4	7.75	7.84	10.8	10.7	11.0	11.0	9.85
Discharge (gpm)	3,950	494	1,257	5,386	6,330	5,430	4,578	4,670	4,760	6,910	3,480	3,520	4,850	4,800	4,940	4,940	4,420
Date	5/3/1993	10/19/1993	3/29/1994	10/19/1994	4/17/1997	9/25/1997	4/29/1998	9/22/1998	4/7/1999	9/13/1999	4/20/2000	9/14/2000	4/17/2001	9/13/2001	4/16/2002	9/17/2002	4/24/2003
Station Name	Hot Creek Spring near Sunnyside, Nevada	Hot Creek Spring near Sunnvside, Nevada															
Station Number	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501
НА	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207

Appendix C

SE ROA 11835

 Table C.1-2

 Miscellaneous Discharge Measurements of Selected Springs in the WRFS

 (Page 3 of 19)

Used in this Study Analysis	Yes	Ver	Tes	3	Yes	Tes N	Yes	162															
Remarks	-	ł	ł.	ł	1	1	Ì	1	ŀ	1	1	ł	ł	1	ł	L F	1	ł	ł	1	;	1	ł
Data Source	This Study	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS, 2004	USGS, 2005	USGS, 2005	USGS, 2006	USGS, 2006	USGS, 2006	USGS, 2006	Maxey and Eakin, 1949												
Measurement Rating ^a (E, G, F, P)	۵.	ł	Ì	ľ	ł	ł	ł	1	ľ	1	1	1	1	-		-		1	1		1	1	1
Discharge (cfs)	13.7	10.6	10.4	10.2	9.46	10.8	14.0	14.3	13.6	14.0	3.14	3.66	3.52	3.25	3.86	3.85	3.85	3.83	3.80	3.80	3.82	3.82	3.82
Discharge (gpm)	6,150	4,760	4,670	4,580	4,246	4,847	6,284	6,418	6,104	6,284	1,409	1,643	1,580	1,459	1,732	1,728	1,728	1,719	1,706	1,706	1,715	1,715	1,715
Date	8/5/2003	9/11/2003	4/23/2004	9/24/2004	6/30/2005	9/22/2005	4/28/2006	8/2/2006	8/15/2006	9/14/2006	10/27/1910	6/15/1913	6/15/1922	5/7/1935	3/6/1936	3/29/1936	3/30/1936	4/7/1936	4/29/1936	5/5/1936	5/7/1936	5/12/1936	5/16/1936
Station Name	Hot Creek Spring near Sunnyside, Nevada	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring									
Station Number	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070501	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601
НА	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207

Southern Nevada Water Authority - Water Resources Division

Appendix C

Table C.1-2 scellaneous Discharge Measurements of Selected Springs in the WRFS (Page 4 of 19)

Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No	No	Yes	Yes																	
Remarks	-	ł	-		1	-	1	1		1	ł			L	-		1	ł					1	1	ł	(FIL)	ł	Discharge appears low.	Discharge appears low.	ł	-
Data Source	Maxey and Eakin, 1949	USGS-NWIS, 2004	Hess and Mifflin, 1978	USGS, 1982	USGS-NWIS, 2004	USGS, 1985	USGS, 1986	USGS, 1988	USGS, 1989	USGS, 1990	USGS, 1991	USGS, 1991	USGS, 1992	USGS, 1992																	
Measurement Rating ^a (E, G, F, P)	1	1	1	1	1	i	1		1	-	1	1	-	ļ	ł	ł	1	1	Ĭ	1	1	-	1	1	1	-	1	1	-	ł	1
Discharge (cfs)	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.82	3.12	3.07	4.02	3.34	4.06	3.52	4.01	3.34	3.34	1.11	0.89	3.98	4.13
Discharge (gpm)	1,715	1,715	1,715	1,715	1.715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,715	1,400	1,380	1,804	1,499	1,822	1,580	1,800	1,499	1,499	498	399	1,786	1,854
Date	5/19/1936	5/23/1936	5/26/1936	5/30/1936	6/2/1936	6/5/1936	6/9/1936	6/16/1936	6/19/1936	6/23/1936	6/27/1936	7/7/1936	7/12/1936	7/19/1936	7/26/1936	8/4/1936	8/15/1936	8/25/1936	5/9/1947	11/13/1966	1/19/1982	7/30/1984	1/21/1985	2/1/1986	2/23/1988	3/14/1989	4/4/1990	11/6/1990	3/3/1991	10/24/1991	3/19/1992
Station Name	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Soring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Amoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson Spring	Arnoldson Spring
Station Number	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601
НА	207	207	207	207	207	207	202	207	202	207	207	207	207	207	207	207	207	207	202	207	202	202	202	207	207	207	207	207	202	207	207

Appendix C

SE ROA 11837



Southern Nevada Water Authority - Water Resources Division

	Measurement Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	22
	Remarks	ł			1	-	1	ł	ł	ł		1	L	ł	ł	}	1	ł	1	ł	1	4		ł	1	1	;		1	1		
i an in shinida nar	Data Source	USGS, 1993	USGS, 1993	USGS, 1994	USGS, 1994	USGS, 1995	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	LISGS-NWIS 2004	11SGS-NVVIS, 2004	11SGS-NWIS, 2004	LISGS-NWIS. 2004	USGS-NWIS, 2004	USGS, 2006	USGS, 2006	Maxev and Eakin, 1949	Maxev and Eakin, 1949	Maxev and Eakin, 1949	Maxev and Fakin 1949	Maxwy and Eakin 1949	Maxey and Eakin 1040	Maxey and Lakin, 1979	Maxey and Eakin, 1976	A DESCRIPTION OF A DESC
of 19)	Measurement Rating ^a (E, G, F, P)	1	1	1	1	ł	1	ł	1		1	1	1	1	1	1		1		;	-		1	1	-	1		l	1	}	1	
ISUTEMEI (Page 5	Discharge (cfs)	4.20	3.50	3.20	3.50	3.80	3.32	3.63	5.37	4.01	4.08	3.43	3.56	3.03	3.23	352	2000	NO C	40.2	4.12	51.4	330	3.45	1 13	131	121	107 1	1.40	1.40	14.1	1.41	
arge Mea	Discharge (gpm)	1,885	1,571	1,436	1,571	1,706	1,490	1,630	2,410	1.800	1.830	1 540	1 600	1 360	1 450	1 580	000'1	1,000	1,320	1,100	1000	1 481	1 548	Car	104	000	100	979	628	633	633	
ous Discha	Date	10/15/1992	5/4/1993	10/19/1993	3/30/1994	10/20/1994	5/20/1997	9/24/1997	4/30/1998	9/23/1998	4/8/1999	0/1/1000	0000/01/19	DODOLONIC	PUDGIGEIR	1002/01/1	1002/21/6	4/11/2002	2002/81/6	4/23/2003	5002/0L/6	4/2/2/2/04	2000/01/21/2	0107/0101	10/2//1910	0001110	3/0/1930	3/29/1936	3/30/1936	4/7/1936	4/29/1936	
Miscellane	Station Name	Amoldson Spring	Amoldson Shrind	Amoldson Spring	Amoldson Spring	Amoldson Spring	Arnoldson Spring	Arnoldson Shrind	Amoldson Spring	Arnoldeon Spring	Annoldson Spring	Alillolason Second	Arnoidson Spring	Rillide liospiourie	Arnoldson Spring	Amolason spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Arnoldson Spring	Amoldson spining	Arnoldson spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	
	Station Number	2070601	1000102	1000102	1000/07	2070601	2070601	2070601	1000102	1000102	1000/02	20/001	20/0601	20/0601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	2070601	20/0601	2070601	2070701	2070701	2070701	2070701	2070701	2070701	2070701	
	HA	207	102	102	102	102	102	202	107	107	102	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	

Appendix C SE ROA 11838

Measurement	Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vec	Vac	100	Tes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes												
	Remarks		1	ľ	ł	1	ł			È.	P.	ł	ł	P.	ł	ł	1	1	1.0		1	ł	1	1	11	1	1	ł	ł	Į,	1	:	-
	Data Source	Maxey and Eakin, 1949	Maxey and Eakin, 1949	Maxey and Eakin, 1949	Maxev and Eakin, 1949	Mavey and Fakin 1949	Mayay and Fakin 1949		Maxey and Eakin, 1949	Hess and Mifflin, 1978	USGS, 1982	USGS-NWIS, 2004	USGS, 1985	USGS-NWIS, 2004	USGS, 1986	USGS-NWIS, 2004	USGS, 1989																
	Measurement Rating ^a (E, G, F, P)	1	;		1		1		1	ł	-		11	1	1	ł	-	ł	1	ł	1	1		1	1	ł	1	;	1	1	1	1	
	Discharge (cfs)	1.39	1.44	1.39	1 38	00.1	10.1	40.1	1.49	1.40	1.40	1.40	1.38	1.41	1.37	1.38	1.38	1.38	1.40	1.38	1.39	1.39	1.40	1.74	1.74	1.71	0.98	0.90	0.98	1.11	2.23	3.34	
	Discharge (gpm)	624	646	624	610	010	263	109	699	628	628	628	619	633	615	619	619	619	628	619	624	624	628	781	780	768	440	404	440	498	1.000	1.499	
	Date	5/12/1936	5/16/1036	5/10/1036	2001/00/2	0021/02/0	5/26/1936	5/30/1936	6/2/1936	6/5/1936	6/9/1936	6/16/1936	6/19/1936	6/23/1936	6/27/1936	7/7/1936	7/12/1936	7/18/1936	7/19/1936	7/26/1936	8/4/1936	8/15/1936	8/75/1936	5/0/1047	11/13/1966	1/10/1082	7/30/1984	1/17/1985	1/21/1985	2/1/1986	2/23/1988	3/14/1989	
	Station Name	Cold Spring	Cold Opinia	Cold optility	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Spring	Cold Opting	Cold Coring	Cold Spring	Cold Coring	Cold Spring	Cold Opting	Cold Opting	Cold Spring	Cold Soring	Cold Spring	Funde plot
	Station Number	2070704	1010102	10/0/07	10/0/07	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	2070701	1010102	1010102	1010102	INJUINZ	10/0/07	10/0/07	10/0/07	1010102	1010102	10/0/02	10/0/02	10/0/02	2070704	1010102	ZU/U/U
	HA	200	107	107	207	207	207	207	207	207	207	207	207	207	207	207	102	107	107	107	107	102	102	102	107	102	107	102	202	107	102	102	201

Table C.1-2 Miscellaneous Discharge Measurements of Selected Springs in the WRFS

Appendix C

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

C-9 SE ROA 11839

1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Selected Springs in the WRFS	
Table C.1-2	rge Measurements of	(Page 7 of 19)
	Miscellaneous Dischar	

Data Source 3S, 1991	ement 19ª F, P)	scharge messurement Rating ³ (cfs) (E, G, F, P) 0.85 USC	Discharge Discharge Measurement (gpm) (cfs) (E, G, F, P) (E, G, F, P) 0.85 USC
GS, 1991	SU .	1.34 US	601 1.34 US
SGS, 1992	5	2.24 U	1,005 2.24 U
SGS, 1992	n	1.78 U	799 1.78 U
SGS, 1993	5	0.33 U	148 0.33 U
SGS, 1993	<u>n</u>	0.04 U	18 0.04 U
SGS, 1994	0	0.23 U	103 0.23 U
JSGS, 1994	-	0.30 1	135 0.30 1
JSGS, 1995	-	0.24 1	108 0.24 1
JSGS-NWIS, 2004	-	0.85 1	380 0.85 1
USGS-NWIS, 2004	1	0.78	350 0.78
USGS-NWIS, 2004	1	1.52	680 1.52
USGS-NWIS, 2004		0.58	260 0.58
USGS, 2005	1	1.41	633 1.41
USGS, 2005	1	1.47	660 1.47
USGS, 2006		1.12	503 1.12
USGS, 2006	1	62.0	355 0.79
Maxey and Eakin, 1	Ę.	5.36	2,406 5.36
Carpenter, 1915	1	5.36	2,406 5.36
Maxey and Eakin, 1	i	10.2	4,574 10.2
Maxey and Eakin, 1	i	9.34	4,192 9.34
Maxey and Eakin, *	1	6.39	2,868 6.39
Maxey and Eakin,	1	7.22	3,241 7.22
Maxey and Eakin, '	1	8.14	3,653 8.14
Maxey and Eakin,	1	9.49	4,259 9.49
USGS-NWIS, 2004	1	9.58	4.300 9.58
Hess and Mifflin. 1	1	R 24	¥24
110.00 1089		0.00	
1000, 1302		6.99	3,137 6.99
USGS, 1985		11.8	5,296 11.8
USGS, 1986		5.51	2,473 5.51
USGS, 1987		11.0	4,937 11.0

Southern Nevada Water Authority - Water Resources Division

Appendix C SE ROA 11840

Measurement Reating ^a Data Source Remarks this Study (E, G, F, P) Analysis	VSGS, 1988 Yes	Yes	0000 1303	USGS, 1990 Yes	USGS, 1991 Yes	11CCS 1001 Yes	USGS, 1991 153	USGS, 1992 155	USGS, 1992 Yes	USGS, 1993 Yes	USGS, 1993 Yes	USGS, 1994 Yes	USGS, 1994 Yes	USGS, 1995 Yes	USGS-NWIS, 2004 Yes	G This Study Yes	G This Study Yes	USGS-NWIS, 2004 Yes	USGS-NWIS, 2004 Yes	Vac	3												
P)	· USGS, 1988	11555 1080	0000, 1303	USGS, 1990	USGS, 1991	115/25 1001	- USGS, 1991	- USGS, 1992	- USGS, 1992	- USGS, 1993	- USGS, 1993	- USGS, 1994	- USGS, 1994	- USGS, 1995	- USGS-NWIS, 2004	USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	- USGS-NWIS, 2004	This Study	This Study	- USGS-NWIS, 2004	- USGS-NWIS, 2004		
Discharge Ratin (cfs) (E, G, F	5.57	37.8	4.40	4.68	7.13	000	9.80	7.34	6.32	10.1	7.80	8.60	8.60	6.40	6.66	5.75	7.22	29.6	8.53	09.6	8.93	10.9	7.31	9.18	8.18	8.53 -	6.84 -	6.25 G	6.25 G	7.04	- W2	6.25 G	
Discharge (gpm)	2,500	0000	2,002	2,101	3,200	000	4,399	3,294	2,837	4,533	3,501	3,860	3,860	2,873	2,990	2,580	3,240	4,340	3,830	4,310	4,010	4,890	3,280	4,120	3,670	3,830	3,070	2,805	2,805	3,160	3,190	2,805	
Date	2/23/1988	Dis 1 1000	3/14/1989	3/22/1990	11/9/1990	Pior Poor	3/3/1991	10/24/1991	3/19/1992	10/14/1992	5/4/1993	10/19/1993	3/30/1994	10/19/1994	5/20/1997	9/24/1997	4/29/1998	9/23/1998	4/8/1999	9/14/1999	4/19/2000	9/13/2000	4/18/2001	9/13/2001	4/17/2002	9/18/2002	4/23/2003	8/6/2003	8/8/2003	9/10/2003	4/22/2004	6/24/2004	
Station Name	Lund Soring	coird Chining	Lund Spring	Lund Spring	Lund Spring		Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Lund Spring	Runda auna								
Station Number	2071001	1001107	2071001	2071001	2071001	1001107	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	2071001	
AH	207	107	207	207	207	102	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	104

traments of Selected Springs in the WRFS Table C.1-2

Appendix C

C-11 SE ROA 11841

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

 Table C.1-2

 Miscellaneous Discharge Measurements of Selected Springs in the WRFS

 (Page 9 of 19)

Used in this Study Analysis	Yes	Yes	Yes	No	No		oN	No	Yes	Yes	Yes	Yes	Vac	Vec	100	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vec	Vac	res	Yes	Yes	Yes	Yes	Yes	Yes
Remarks		1.1		Spring orifice lowered	Possibly only half the flow	Monthrament is nrohably	Measurement is provavry .422 cfs	This is probably the wrong discharge, and is the same measurement as 6/15/1949.	1	-		I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.		-	1		-	1	-	1	1		1	ł		ł	1			-	1
Data Source	USGS, 2005	USGS, 2006	USGS. 2006	Mayav and Fakin 1949	Millor of all 1062	Willich et al., 1900	Stearns et al., 1937	Hess and Mifflin, 1978	Mifflin, 1968	USGS. 1982	11SGS 1985	LICCS NIMIS 2004	10007 'CIANI-0000	USGS, 1987	USGS, 1988	USGS, 1989	USGS, 1990	USGS, 1991	USGS, 1991	11SGS 1992	11000 1000	USGS, 1332	USGS, 1993	USGS, 1993	USGS, 1994	USGS, 1994	USGS, 1995	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004
Measurement Rating ^a (E, G, F, P)	1	•	;		:	1	ł	I	1	1					1	1	1	1	1			-	1	ł	ł	}	1	1	}	1	ł
Discharge (cfs)	11.6	9.47	101	1.71	7.01	0.22	4.23	4.23	0.50	0.50	0.67	10.0	0.53	0.61	0.56	0.67	0.51	0.67	0.60	240	0.45	0.58	0.43	0.38	0.43	0.46	0.47	0.38	0.52	0.57	0.55
Discharge (gpm)	5,206	4 250	F X 24	104'0	4/0/4	100	1,900	1,900	205	Dec	200	007	240	274	251	301	229	301	040	010	202	260	193	171	193	206	211	170	234	255	248
Date	9/21/2005	ADDOLTON	DODUCTION OF	8/13/2000	3/16/1935	9/15/1945	6/15/1949	11/15/1966	1111068	CONTRACT	1123/1302	C2861/1/1/1	2/1/1986	2/11/1987	2/23/1988	3/14/1989	3/22/1990	11/8/1990	Distant	I BBL/C/P	10/24/1991	3/19/1992	10/15/1992	5/4/1993	10/19/1993	3/29/1994	10/19/1994	4/17/1997	Q174/1997	4/29/1998	4/8/1999
Station Name	Lund Chrind		Fund spiring	Lund Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	and the second se	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Soring	Mouthan Caring	Builde upuilloom	Moorman Spring	Moorman Soring	Moormon Coring	Moorman Spring	Moorman Spring							
Station Number	2074004	1001/07	1001/02	2071001	2071101	2071101	2071101	2071101		20/1101	2071101	2071101	2071101	2071101	2071101	2071101	0024404	1011/02	LOLL/02	2071101	2071101	2071101	2071101	2071101	2071101	2071101	2011101	2071101	2011/02	1011102	2071101
HA	200	102	207	207	207	207	207	207		207	207	207	207	207	207	207	100	102	207	207	207	207	207	207	207	202	202	102	102	107	207

Southern Nevada Water Authority - Water Resources Division

Table C.1-2	ellaneous Discharge Measurements of Selected Springs in the WRFS	(Page 10 of 19)
	Miscellane	

ed in Study Ilysis	(es	res	res	(es	/PC		Les .	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Us this And																										-						
Remarks	i,	ł	ł	-	-	1	{	ŧ.	I.	ł	1	1	ł		1	1	}		1	ł	Reported as 7/62-32D1	ł	1	ł	1	ł	1	ł	+	;	1	-
Data Source	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	LISCS NIVIS 2004		USGS-NWIS, 2004	SNWA	USGS, 2004	USGS-NWIS, 2007	USGS, 2005	USGS-NWIS, 2007	USGS, 2005	USGS, 2006	USGS, 2006	Bunch and Harrill, 1984	SNWA	Maxey and Eakin, 1949	USGS, 1982	USGS, 1985	USGS, 1986	USGS, 1987	USGS, 1988	USGS. 1989	11SGS 1990	USGS. 1991	1JSGS. 1991	USGS. 1992	USGS, 1992				
Measurement Rating ^a (E, G, F, P)	1	l t	1		1	1	1	1	1	1	ď	ł		1	1	1	1	1	1	ш	1	1	1	3	}	1	1	1	1	1	1	1
Discharge (cfs)	0.39	0.51	0.49	0.40	0.40	0.35	0.49	0.47	0.49	0.58	0.51	0.47	0.43	0.43	0.42	0.42	0.53	0.46	0.004	0.001	2.50	2.33	2.19	1.69	2.02	2.01	201	00 6	1 78	2.04	1 00	1.68
Discharge (gpm)	175	230	000	202	207	156	221	211	220	260	231	211	192	192	189	189	238	206	2	0.35	1 122	1 046	983	759	an7	GUD	GUO	200	200	000	206	754
Date	9/15/1999	4/20/2000	0000/01/0	0007/01/2	4/17/2001	9/13/2001	4/18/2002	4/24/2003	9/10/2003	4/22/2004	6/23/2004	9/22/2004	6/30/2005	6/30/2005	9/21/2005	9/21/2005	4/27/2006	9/13/2006	R/1/1979	Q/14/2004	1111040	CROMACIT	1/16/1085	SIGNOR	28011112	8401/20/0	214314000	00011000	3/22/1390	DAU 10/11	1001001	2/18/1002
Station Name	Moorman Shrind	Moorman Coring	Rinder aprilia	Moorman spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Spring	Moorman Snring	Moorman Spring	Moorman Shrind	Moorman Spring	Moorman Soring	Moorman Spring	Moorman Snring	Moorman Spring	Chinele Carine	Chinele Corine	Sunde abund	riag opinige o	Flat Springs o	Lied opinies o	Flag Springs o	Flag Springs 3	Fiag springs 3	Flag Springs 3	Flag Springs 3	Flag Springs 3	Hag Springs 3	Flag Springs 3
Station Number	9074404	1011/07	1011/02	2071101	2071101	2071101	2071101	2071101	2071101	2071101	2071101	2071101	2074104	1011102	1011102	2014404	1011102	2074404	1011107	1021102	1021/02	20/1301	20/1301	20/1301	20/1301	20/1301	20/1301	2071301	2071301	2071301	2071301	2071301
НА	202	107	201	207	207	207	207	202	207	207	10.7	107	102	102	107	107	107	107	107	107	707	201	207	102	207	207	207	207	207	207	207	207

Appendix C

SE ROA 11843

Southern Nevada Water Authority - Water Resources Division

Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Vác	CD1	TES V	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vac	Voc	C L L L L L L L L L L L L L L L L L L L	IES	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yec	Vac	201
Remarks	1	1		1			ł	ł	£	ŧ	-	ł		ł	ł	1	;	1 STA		1	}	1	:	1	1	1	ł	4	1	1	1		1	ł
Data Source	USGS, 1993	USGS, 1993	USGS, 1994	11SGS 1994	00001 1001	USGS, 1990	USGS-NWIS, 2004	LISGS-NWIS, 2004	LISCS-NIMIS 2004	1000 SIMIL SOON	1000-10100-1000	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS, 2004	USGS, 2004	USGS, 2005	USGS. 2005	11505 2008	110/00 4000	USGS, 1302	USUS, 1803	USGS, 1986	USGS, 1987							
Measurement Rating ^a (E, G, F, P)	1	1	1			ł		1	1.1	1	t	1	1	1	-			1	-	1	ł	1	1	1	1	1	ł	1		1	1	ł	ł	l
Discharge (cfs)	1.60	1 80	100	1.50	ne.1	1.70	3.25	2.18	2.27	2.54	2.81	1.68	2.63	2.63	2.65	000	2.43	2.94	2.23	1.98	2.90	2.07	1.78	1.84	1.80	1.75	2.47	030	010	R1.7	2.57	2.86	2.68	3.53
Discharge (gpm)	718	ana	000	900	6/3	763	1,460	978	1,020	1,140	1,260	754	1.180	1480	1,100	040'1	1,000	1,320	1,000	890	1,300	930	800	825	810	785	1 109	020 F	1,013	983	1,153	1,284	1,203	1,584
Date	10/14/1992	-1014000	CRELICIC	CRALIALIOL	3/29/1994	10/19/1994	4/17/1997	5/21/1997	9/29/1997	4/29/1998	9/23/1998	4/8/1000	9/13/1999		4/4/2000	8/14/Z000	4/17/2001	9/13/2001	4/16/2002	5/30/2002	9/19/2002	4/24/2003	9/11/2003	4/23/2004	9/11/2004	AUNCIACIO	RI20/2005	OLDOLDODC	CUUZ/22/8	4/28/2006	7/24/1982	1/16/1985	2/4/1986	2/11/1987
Station Name	Elan Carinae 2	riag opilities o	Flag Springs 3	Flad Sprinds 3	Flad Sorings 3	Flag Springs 3	Elan Shrinns 3	riay opinios o	Flag optings o	Field optimits a	Flag Springs 3	Flag Springs 3	Flag Springs 3	Flag Springs 3	Flag Springs 3	Flad Springs 3	Flad Sprinds 3	Flan Shrinds 3	Flan Springs 3	Flad Springs 3	Elan Springe 3	o control for	Fiag Springs o	Flag springs 3	Flag Springs 3	Flag Springs 3	Flag Springs 2	Flag Springs 2	Flag Springs 2	Flag Springs 2				
Station Number	1001000	1021702	2071301	2071301	2071301	2071301	2071301	2071301	2071301	2071301	1001102	1001202	1021702	20/1301	2071301	2071301	2071301	2071301	2071301	2071301	2071301	2071201	2071204	2071201	1001107	1001102	20/1301	2071301	2071301	2071301	2071302	2071302	2071302	2071302
АН	tee	20/	207	207	207	207	207	207	207	207	107	102	207	207	207	207	207	207	207	207	207	202	107	102	102	102	207	207	207	207	207	207	207	207

Table C.1-2 Miscellaneous Discharge Measurements of Selected Springs in the WRFS

Appendix C SE ROA 11844

surement lsed in	is Study nalysis	Yes	Yes	No	Yes	Yes	Vac	No.	3	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Mea	₽ 4				-	+		-		-	-	-	-	-													-						
Damarke	Kemarks	Ŀ	L L	Discharge appears low.	1		-	-	I	-	1	-	1	1	-	1	1	i.	1	1	1	1	1	1	1	1	1	ł	1	-	ł	1	4
	Data Source	USGS, 1988	USGS, 1989	11SGS 1990	1000 1000	USGS, 1991	USGS, 1991	USGS, 1992	USGS, 1992	USGS, 1993	USGS, 1993	USGS, 1994	USGS, 1994	USGS, 1995	USGS-NWIS, 2004	USGS, 2004	USGS, 2005	USGS, 2005	11000 2006														
Measurement	Rating ^a (E, G, F, P)	1		1		1	1	ł	1	1	-	1	ł	-	1	1	1	1	1	-	1	1	;	1	1	ł	1	}	1	1	1	ł	
Discharge	uiscriarge (cfs)	3.56	000	0.40	0.49	2.23	2.23	2.80	2.97	2.90	3.20	2.90	3.10	2.90	3.25	3.21	3.50	2.27	2.85	3.19	3.12	3.39	3.21	3.07	3.10	2.79	3.07	2.94	2.44	3.12	2.70	2.68	
Dischargo	(gpm)	1.598	1 302	200'1	720	1,001	1,001	1,257	1,333	1,302	1,436	1,302	1,391	1.302	1,460	1.440	1.570	1.020	1.280	1 430	1 400	1 520	1.440	1 380	1.390	1 250	1 380	1 320	1 095	1.400	1.212	1.203	
	Date	2123/10RB	00011000	3/14/1303	3/22/1990	11/8/1990	3/4/1991	10/23/1991	3/18/1992	10/14/1992	5/3/1993	10/19/1993	3/29/1994	10/19/1994	4/17/1997	9/29/1997	4/20/1998	9/23/1998	4/8/1000	0/13/1000	NAPOON D	000CIVIO	100012112	1112/2004	4/16/2002	0/12/2010	2002/01/8	0/44/2003	402/2004	9/24/2004	6/30/2005	9/22/2005	
	Station Name	Flat Cadada 0	Flag Springs 4	Flag Springs 2	Flag Springs 2	Flag Springs 2	Flag Springs 2	Flag Springs 2	Flag Springs 2	Flad Sprinds 2	Flag Springs 2	Flan Springs 2	Flag Springs 2	Flas Carings ?	Elar Shrinke 2	Clar Chrings 2	Flat Corings 2	Flag opinies 4	Fiay opings 2	Frag opings 2	Fiag optings 4		Fiag Springs 2	Lidy opinity 2	Firs Cariago 2	riag opinigs 2	Flag Springs 4	riag opinitys 4	Field Springs 4	Flag optings 4	Fire Corings 2	Flag Opinigo 2	FIBD opinido 6811
	Station	0001000	20/1302	2071302	2071302	2071302	2071302	2071302	2071302	2011202	2071302	0004200	2001102	2001102	2001/02	2001/02	2001202	20011302	2021202	2021202	20/1302	2051/02	20/1302	2001/02	20/1302	20/1302	2071302	20/1302	2021/02	2021202	2001102	2001102	20/1302
-	HA		207	207	207	207	207	207	207	102	102	104	102	102	102	107	107	202	102	207	207	207	207	207	207	207	207	207	207	207	102	202	201

the M	
Measurements of Selected Springs in	(Page 12 of 19)
eous Discharge	
	neous Discharge Measurements of Selected Springs in the M

Appendix C

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

C-15 SE ROA 11845

Table C.1-2 Miscellaneous Discharge Measurements of Selected Springs in the WRFS (Page 13 of 19)

Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes									
Remarks		11				1	Discharge appears low.	1								1	1		4	1	1		4		-	-	-		1	1	1
Data Source	USGS, 1982	USGS, 1985	USGS, 1986	USGS, 1987	USGS, 1988	USGS, 1989	USGS, 1990	USGS, 1991	USGS, 1991	USGS, 1992	USGS, 1992	USGS, 1993	USGS, 1993	USGS, 1994	USGS, 1994	USGS, 1995.	USGS-NWIS, 2004														
Measurement Rating ^a (E, G, F, P)	1	ł	-	1	ł	1	ł	1	-	1	1	ł	ł	ł	ł	1	1	1	ł	ļ	1	-	1	1	1	}	1	ł	1	1	i.
Discharge (cfs)	2.24	2.36	1.91	2.32	2.01	3.12	1.54	2.23	2.90	2.02	2.22	2.00	2.00	2.40	2.10	1.90	2.38	2.43	2.12	3.5	2.13	2.25	2.58	2.63	1.84	2.41	2.16	2.14	3.10	1.94	2.04
Discharge (gpm)	1,005	1,059	857	1.041	902	1,400	691	1,001	1,302	205	966	898	898	1,077	943	853	1,070	1.090	952	1.570	956	1,010	1,160	1,180	826	1,080	970	960	1,390	871	915
Date	7/25/1982	1/16/1985	2/4/1986	2/11/1987	2/23/1988	3/14/1989	3/22/1990	11/9/1990	3/4/1991	10/23/1991	3/18/1992	10/14/1992	5/3/1993	10/19/1993	3/29/1994	10/19/1994	4/17/1997	9/25/1997	4/29/1998	9/23/1998	4/8/1999	9/15/1999	4/4/2000	9/14/2000	4/17/2001	9/13/2001	4/16/2002	4/16/2002	9/19/2002	4/24/2003	9/11/2003
Station Name	Flag Springs 1	Flan Springs 1	Flan Shrinds 1	Flan Snrings 1	Elan Springe 1	Flan Springs 1	Flad Springs 1	Flag Springs 1	Flag Springs 1	Flag Springs 1	Flan Springs 1	Flad Sorinds 1	Flan Snrings 1	Flad Springs 1	Flan Snrinns 1	Flan Springs 1	Flad Springs 1	Flad Shrinds 1	Flag Chrings 1	Elan Shrinde 1	Flad Springs 1	Flad Springs 1	Flao Sorinos 1	Flag Springs 1	Flad Springs 1	Flag Springs 1					
Station Number	2071303	2071303	2071203	2071203	0001107	2071203	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2074303	2071303	2071303	2071303	CUC1107	20171202	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2071303	2071303
НА	207	207	202	102	102	102	207	207	207	207	202	202	202	207	207	207	207	202	102	102	102	202	207	207	202	207	202	207	207	207	207

Southern Nevada Water Authority - Water Resources Division

Measurement Used in this Study	Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vac	Vac	Vac	Voc	Voc	C I I I	res	Yes	SS	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	res
Remarks		1		ł		1	Rennted as 7/62-28B1	1	1	1			;		1	1		ł	11	1	-		:	1	ti	11	-	Ť.	1	1		-	-
Data Source		USGS-NWIS, 2004	USGS, 2004	USGS. 2005	115/25 2005	1000 2008	USGS, 2000	LOCO MINIO SUDA	1002 4000 4000	1000, 1302	USGS, 1803	USGS, 1800	1000 1001	1907 1901	USGS, 1988	USGS, 1989	USGS, 1990	USGS, 1991	USGS, 1991	USGS, 1992	USGS, 1992	USGS, 1993	USGS, 1993	USGS, 1994	USGS, 1994	USGS, 1995	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004
Measurement Ratino ^a	(E, G, F, P)	1		1		1	1	t	1	ł	1		l	1	1	1	1	ł	1	t 1	ł	1	1	1	1	1	1	1	1	1	1	1	
Discharge	(cfs)	2.12	212	340	000	2.02	2.40	2.50	2.01	2.51	3.17	3.29	2.29	4.17	2.23	3.12	2.01	2.45	2.45	3.38	3.05	2,80	2.80	3.30	3.30	2.60	2.23	2.90	3.41	3.34	2.76	2.70	2.03
Discharge	(mdb)	950	DED	000	COR	106	1,077	1,122	006	1,127	1,423	1,477	1,028	1,872	1,001	1,400	902	1,100	1,100	1,517	1,369	1.257	1,257	1,481	1,481	1.167	1.001	1300	1.530	1,500	1.240	1,210	A AED
4	Date	4/02/2004	*UDDIFUID	4007/47/R	6/30/2005	9/22/2005	4/28/2006	1/1/1949	3/9/1966	7/25/1982	1/16/1985	2/4/1986	2/11/1987	8/12/1987	2/23/1988	3/14/1989	3/22/1990	11/8/1990	3/4/1991	10/23/1991	3/18/1992	10/14/1992	5/3/1993	10/19/1993	3/29/1994	10/19/1994	4/17/1997	012511997	4/29/1998	9/23/1998	4/8/1999	9/15/1999	ODOCIDOR .
	Station Name	Pinton A	riag spinigs i	Flag Springs 1	Flag Springs 1	Flag Springs 1	Flag Springs 1	Butterfield Spring	Rutterfield Soring	Rutterfield Spring	Butterfield Spring	Dutterfield Shring	Butterfield Spring	Dutterfield Shrind	Dutterfield Spring	Duttertield Opting	Dutterield Opting	Dutter terta Contro	Butteriteta Spring	Butterfield Spring	Butterfield Spring	Dutterfield Shring											
Station	Number		20/1303	2071303	2071303	2071303	2071303	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	1011102	1041107	1041107	1041102	2011401	1041/07	20/1401	20/14/12	20/1401	1041/07	2011401	1041/07	1041107
1	HA		207	207	207	207	207	207	207	207	207	207	207	207	207	207	207	202	202	102	107	102	107	102	102	102	201	20/	207	107	102	107	102

Table C.1-2

Appendix C

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

C-17 SE ROA 11847

Southern Nevada Water Authority - Water Resources Division

Measurement Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yec	Vac	Vec	sa :	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	3	Yes	Ves	102
Remarks	1	11		1	;	1		1		1.1	1	1	ł	11	Reported as West Immigrant Spring	Discharge is confluence of five springs	200 yds. West of Hardy Springs	-	i		-	1	ł	-	1	1	1
Data Source	USGS-NWIS, 2004	USGS-NWIS, 2004	LISGS-NWIS, 2004	TISCS-NWNS 2004	LIECE NIMIE 2004	TODO SUMIC SOOM	USGS-NWIS, 2004	USGS, 2003	USGS-NWIS, 2004	USGS, 2004	USGS, 2005	USGS, 2005	USGS, 2006	USGS. 2006	Hess and Mifflin, 1978	SNWA	SNWA	Maxey and Eakin, 1949									
Measurement Rating ^a (E, G, F, P)	1	1				ł	1	1	I	l	ł		-	1	Į	u.	ω	ł	ł	1	-	1	1	1	ł	1	1
Discharge (cfs)	3.32	563	30.0	08.7	2.43	2.79	2.18	2.18	2.27	2.10	1.90	2.23	2.84	969	0.45	0.45	0.01	2.28	2.63	2.68	2.65	2.65	2.75	2.74	2.70	2.70	2.70
Discharge (gpm)	1 490	1 180	0001	1,330	1,090	1,250	978	978	1,020	942	853	1 001	1 275	4 476	200	200	ŝ	1.023	1,180	1,203	1.189	1,189	1,234	1,230	1,212	1,212	1.212
Date	0112/2000	110000	4/1//2001	10/2/2001	4/16/2002	9/19/2002	4/24/2003	9/11/2003	4/23/2004	9/24/2004	6/30/2005	2000/00/0	200012014	200211714	9/14/2000	9/14/2004	9/14/2004	10/27/1910	5/7/1935	3/6/1936	3/29/1936	3/30/1936	4/7/1936	4/29/1936	5/5/1936	5/7/1936	5/10/1036
Station Name	o 11 Calas	Butterneid opring	Butterfield Spring	Rutterfield Spring	Dutterfield Shrind	Duttorfield Chrind	Builde niailiaina	Bunde planta	Butterneld spring	Butterfield Spring	Hardy Springs	Hardy Spring NW	Nicholas Spring	Nicholas Spring	Altaholae Carina	Nicitoles Opinia	Nicholas Spring	Nicholas Spring	Micholae Soring	Nicholas Spring	Nicholas Spring	Michalan Carlan					
Station Number		2071401	2071401	2071401	2071401	2071401	2071401	2071401	2071401	1011107	1041107	20/1401	2071401	2071401	2071401	2011001	2071502	2071601	2071601	POSTCO4	1001702	1001107	1001/07	1001102	2071601	2071601	
HA		207	207	207	207	207	207	207	207	102	102	207	207	207	207	207	102	207	207	too	107	102	107	102	102	202	5

C-18

Appendix C SE ROA 11848

	ted Springs in the WRFS		
Table C.1-2	Miscellaneous Discharge Measurements of Selec	(Page 16 of 19)	

Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Vac	Con Nor	Vor	Vac	Tes Vie	Yes	Tes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No	Yes	Vac	Yes	Yes	221							
Remarks	11	1	1				1	1	ł	1	;	ł	Ł	ŧ	1	F.	1	1		ł	1	1	1	I	1	1	11	Discharne annears low					
Data Source	Maxey and Eakin, 1949	Maxey and Eakin, 1949	Maxev and Eakin, 1949	Movey and Eakin 1049	makey and Eakin 1040	Maxey and Eakin, 1949	USGS-NWIS, 2004	Hess and Mifflin, 1978	USGS, 1982	USGS-NWIS. 2004	USGS. 1985	USGS, 1986	115.55 1088	1000, 1900	1000, 1909	USGS, 1990	USGS, 1991	USGS, 1991	USGS, 1992														
Measurement Rating ^a (E, G, F, P)	ł	1	11		1	1	1	1	1				1		ł	į.	ł	1	1	1	-	1	-	;	1	1		1	1	1	1	1	1
Discharge (cfs)	2.70	2.70	010	010	5.10	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.70	2.50	2.51	2.38	267	2.45	CY C	272	3.56	19:0	2.45	2.90	3.34	2.60
Discharge (gpm)	1.212	1 212	6161	2121	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1,212	1 212	1 212	1 100	1 125	1 DAR	000 -	DEF	1 000	00011	1,598	301	1,100	1,302	1,499	1,167
Date	5/19/1936	5192H036	016011000	0/20/1830	5/30/1936	6/2/1936	6/5/1936	6/9/1936	6/16/1936	6/19/1936	6/23/1936	6/27/1936	7/7/1936	7/12/1936	7/18/1936	7/19/1936	7/26/1936	8/4/1936	8/15/1036	8/75/1036	21014047	1101000	111011080	7001/01/1	1/30/1304	0021/17/1	0061/1/7	2/23/1988	3/14/1989	4/4/1990	11/6/1990	3/3/1991	10/24/1991
Station Name	Michelee Coring	Billiolas opring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Sprind	Nicholas Soring	Nicholas Spring	Michalas Sarina	Number Come	NICHOLAS Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring
Station Number	PODACOO	1001/07	20/1601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	1001107	1001/07	20/1001	2071501	20/1601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601
HA	-	107	207	207	207	207	202	207	207	207	207	207	207	207	202	202	107	107	107	207	207	207	207	207	207	207	207	207	207	207	207	207	207

Appendix C

SE ROA 11849



Southern Nevada Water Authority - Water Resources Division

Weasurement Used in this Study Analysis	No	Yes	Yes	Yes	Yes	Vac	Vac	Noc	Ver	Vac	102	Yes	Yes	Yes	Vac	SP1	Vec	TES	Yes	Yes	Vac	Sa :	Yes	Yes	Yes						
Remarks	Discharge appears low	1	1				E	ł	ł	-	1	-	1	+	1	1	1	1				1	1	1	1	1		ł		No date given, listed as un-published record, by Stearns et al., 1935	1
Data Source	USGS, 1992 [USGS, 1993	USGS, 1993	1964	1994	USGS, 1994	USGS, 1995	USGS-NWIS, 2004	USGS-NWIS: 2004	LISCS MIMIS 2004		USGS-NVVIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS-NWIS, 2004	USGS, 2004	USGS, 2005	115.05 2005	000, 5000	USGS, 2006	USGS, 2006	Steams et al., 1937	Maxey and Eakin, 1949								
Measurement Rating ^a (E, G, F, P)	1		1		1	1	1	-	-		-	1	1	1	1	-			1	ł		1	-	1	1	-	1	1	1	ł	1
Discharge (cfs)	0.24	3.00	300	000	2.80	2.00	2.90	2.65	2.94	3.03	2.74	2.92	2.52	2.81	3.45	212	376	21.2	2.63	2.72	2.81	2.50	2.67	2.64	2.69	0.00	507	2.55	2.89	0.11	0.30
Discharge (gpm)	108	AAG 1	340 1	040'1	1,257	898	1,302	1,190	1,320	1,360	1,230	1.310	1,130	1.260	1 550	049	1000	1,240	1,270	1,220	1,260	1,120	1,200	1.185	1 207	1021	1,20/	1,145	1,297	50	135
Date	3/19/1992	1014514000	70010101	0/4/1880	10/19/1993	3/30/1994	10/20/1994	5/20/1997	9/24/1997	4/30/1998	9/23/1998	4/8/1999	9/14/1999	4/19/2000	0/12/2000	PUDCION I	10/2/01/4	1002/21/6	4/17/2002	9/18/2002	4/23/2003	9/10/2003	4/22/2004	0/03/2004	SUDOI NIT	CD07/1//	9/21/2005	4/27/2006	9/13/2006	1/1/1935	12/16/1947
Station Name	Mishalan Carina	Niciolas opinig	Nicholas Spring	Nicholas Shring	Micholae Coning	Nicholae Soring	Nicholas Opinia	Nicholds Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Spring	Nicholas Sprind	Nicholas Sprind	Michalon Corine	Nicholas Sping	Nicholas Spring	Nicholas Spring	Nicholas Spring	Michalae Chrinn	Williams Hot Spring	Williams Hot Spring							
Station Number	1001200	20/1001	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	2071601	1001102	1001107	1001/07	1091/07	2071601	2071601	2071601	2071601	2071601	2071601	2074601	2071601	1001104	20/1601	2071601	2071601	2071601	1001200	2071801	2071801
HA		20/	207	207	207	207	207	202	104	207	202	107	201	202	207	207	207	207	207	207	202	LUC	207		207	207	207	207	100	207	200

Table C.1-2 Miscellaneous Discharge Measurements of Selected Springs in the WRFS

Appendix C SE ROA 11850

Table C.1-2	us Discharge Measurements of Selected Springs in the WRFS	(Page 18 of 19)
	Miscellaneous Discha	

Used in this Study Analysis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vac	Vac	Vac	NIA	Vac	103	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Voc	SE A	168	0N	No
Remarks	Measured Q at 6-in. ABS pipe below the reservoir outside the fence.		-	1			+		-	1		Discharge appears nign	1	-	-		1		-		1					1	Discharge could be confused with Crystal Springs	Discharge could be confused with Crystal Springs
Data Source	SNWA	Ertec, 1981	11SGS 1985	1000 1000	0000, 1000	USGS, 198/	USGS, 1987	USGS, 1988	USGS, 1989	USGS, 1990	USGS, 1985	USGS, 1986	USGS, 1987	USGS, 1987	USGS, 1988	USGS, 1989	11SGS 1990	110/00 1001	115/25 1001	115/25 1002		USGS, 1992	USGS, 1893	USGS, 1993	USGS, 1994	USGS, 1994	Carpenter, 1915	Hardman and Miller, 1934
Measurement Rating ^a (E, G, F, P)	u.	1		-	1	1	{	ł	ł	1	P.	1	ł	-	1	1			1		1	1	1	1	ļ	1	1	t
Discharge (cfs)	0.07	160	POT	4.01	4.13	3.96	4.1	4.9	5.12	4.23	2.25	3.11	1.77	1.88	1.78	1 34	00 1	76.1	00.1	00.7	1.65	1.84	0.75	2.10	1.30	1.10	00.6	12.0
Discharge (gpm)	30	200	000	1,800	1,854	1777.1	1,840	2,199	2,298	1,899	1,752	2,422	1,379	1.464	1 386	4 DAA	101 1	1,485	611	000C'L	1,449	1,433	584	1,636	1,012	857	4,039	5,368
Date	7/26/2005	01414070	0/11/9/8	1/16/1985	2/3/1986	2/11/1987	8/11/1987	2/23/1988	3/14/1989	3/22/1990	1/17/1985	2/1/1986	3/26/1987	8/12/1987	2/22/1088	014 4 14 000	3/14/1303	3/22/1990	11/8/1990	3/8/1991	10/24/1991	3/18/1992	10/14/1992	5/4/1993	10/19/1993	3/29/1994	11/15/1912	1/1/1931
Station Name	Williams Hot Spring		Moon River Springs	Emiorant Springs	Emiorant Springs	Emigrant Springs	Cilligram Coringo	Emigran opinius	Emigrant spirities	Emigrant Springs	Emiorant Springs	Eminrant Shrinds	Emicrant Shrinds	Hiko Springs	Hiko Springs													
Station Number	2071801		2071901	2071901	2071901	2071901	2071901	2071901	2071901	2071901	2072001	2072001	FUNCTOR	1002002	20/2001	20/2001	2072001	2072001	2072001	2072001	2072001	2072001	2072001	2072001	1002102	FUNCTOR	2090101	2090101
HA	207		207	207	207	207	207	207	207	202	202	102	107	201	207	207	207	207	207	207	207	207	207	202	102	102	209	209

Appendix C

C-21 SE ROA 11851

	he WRFS	
Table C.1-2	ellaneous Discharge Measurements of Selected Springs in th	(Page 19 of 19)
	Nis	

arks Used in this Study Analysis	Yes	Yes	Yes	Yes	Yac	3 X	- res	- Yes	Yes	- Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Vec		162	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No.	Sol Sol	
Source		2	-		2	55	65	95	99	82	85	86	87 -		- 06	- 161	101		765				994	994	- 966	- 266		100			963
nt Data (Smith, 193	Smith. 194.	Cmith 104	10001	0262, 130	USGS, 196	USGS, 196	USGS, 19(USGS, 19(USGS, 19(USGS, 19(USGS, 19	USGS, 19	USGS, 19	USGS, 19	USGS. 19	11555 10	01 0000 10	USGS, 13	USGS, 19	USGS, 19	USGS, 19	USGS, 15	USGS, 19	USGS, 19	USGS, 19	11955 10	10000	This Stud	SNWA	Eakin, 19
Measuremer Rating ^a (E, G, F, P)	ł	1		1	1	1	1	1	1	1	1	1	1	1	1	1			-	1	1	1	1	1	1	1			٩.	۵.	
Discharge (cfs)	6.57	6 ED	10.0	0.40	5.36	6.41	6.43	6.58	6.31	6.54	6.76	6.08	5.77	6.24	4.30	6.68	00.	4.80	4.24	5.28	6.40	4.40	4.00	4 60	6.00	479	000	0.08	6.00	0.001	0.5
Discharge (gpm)	2 949	200 0	2,320	2,8/3	2,406	2,877	2,886	2,953	2,832	2,935	3,034	2,729	2,590	2.801	1 930	2008	00010	2,199	1,903	2,370	2,873	1.975	1.795	2 085	2 603	0.4KD	2,100	2,730	2,693	0.45	224
Date	1/1/1934		1/1/1241	1/1/1943	6/15/1963	2/7/1965	5/19/1965	7/13/1965	10/12/1965	7/29/1982	1/21/1985	1/28/1986	3/25/1987	2/12/1988	3/14/1090	20001-1-1-000	11/0/1320	4/3/1991	11/4/1991	3/25/1992	10/14/1992	4/20/1993	10/19/1993	NOOMOOIC	1001/01/01/01	2001/01/01	4/10/123/	9/23/1997	7/19/2004	5/24/2004	6/1/1963
Station Name	113ta Casiman	Hiko opinide	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Shrinds	Hiko Springs	Hiko Springs	Hiko Springe	Hiko Shrinds	Liko Sorinde	Hillon Sortings	Like Coringe	epilide office	HIKO Spilligs	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Sorings	Litto Corinde	Liko Optinga	chillide ovilu	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Springs	Hiko Springs	Solar Panel Spring	Brownie Spring
Station Number	1010000	1010602	2090101	2090101	2090101	2090101	2000101	2000101	101000C	1010002	2000101	1010007	1010enz	2020101	1010602	1010602	2090101	2090101	2090101	2090101	2000101	1010000	ININGNZ	1010602	2090101	2090101	2090101	2090101	2090101	2090201	2090701
HA		209	209	209	209	000	000	202	200	200	RU2	502	202	ROZ	ROZ	209	209	209	209	209	000	202	502	807	209	209	209	209	209	209	000

Southern Nevada Water Authority - Water Resources Division

Appendix C SE ROA 11852

Station Number	Station Name	НА	Year	Discharge (cfs)	Data Source
09415510	Preston Big Spring near Preston, NV	207	1984	7.24	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	1985	7.98	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2001	7.7	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2002	7.45	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2003	7.61	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2004	7.92	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2005	7.61	USGS-NWIS, 2007
09415510	Preston Big Spring near Preston, NV	207	2006	9.38	USGS-NWIS, 2007
2150101ª	Muddy River near St. Thomas, NV	215	1914	19.3	Wells, 1954
09415589	Crystal Springs Diversion near Hiko, NV	209	2005	1.7	USGS, 2005
09415589	Crystal Springs Diversion near Hiko, NV	209	2006	1.28	USGS, 2006
09415590	Crystal Spring near Hiko, NV	209	2005	11.2	USGS, 2005
09415590	Crystal Spring near Hiko, NV	209	2006	11.3	USGS, 2006
09415639	Ash Springs Diversion at Ash Springs, NV	209	2005	3.03	USGS, 2005
09415639	Ash Springs Diversion at Ash Springs, NV	209	2006	3.88	USGS, 2006
09415640	Ash Springs Creek below Hwy. 93 at Ash Springs, NV	209	2005	13,9	USGS, 2005
09415640	Ash Springs Creek below Hwy. 93 at Ash Springs, NV	209	2006	12.1	USGS, 2006
09416000	Muddy River near Moapa, NV	219	1914	47.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1915	47.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1917	46.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1918	47	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1945	45.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1946	46.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1947	47.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1948	46.3	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1949	47	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1950	46.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1951	46.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1952	46.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1953	46	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1954	46	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1955	47.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1956	45.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1957	47.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1958	49.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1959	48.9	USGS-NWIS, 2007

Table C.1-3Annual Discharge of Spring Gaging Stations in the WRFS(Page 1 of 4)

Appendix C

SE ROA 11853



Station Number	Station Name	НА	Year	Discharge (cfs)	Data Source
09416000	Muddy River near Moapa, NV	219	1960	47.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1961	46.3	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1962	44.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1963	44.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1964	44.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1965	43.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1966	41.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1967	46	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1968	40.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1969	42.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1970	40.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1971	38.1	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1972	43.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1973	45.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1974	40.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1975	39.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1976	41.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1977	37.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1978	36.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1979	39.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1980	39.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1981	37.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1982	37.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1983	39.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1984	39.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1985	38.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1986	36.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1987	37.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1988	39.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1989	33.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1990	36.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1991	36.1	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1992	36.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1993	39.6	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1994	39.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1995	35.6	USGS-NWIS, 2007

Table C.1-3 Annual Discharge of Spring Gaging Stations in the WRFS (Page 2 of 4)

G-24

Appendix C SE ROA 11854

Station Number	Station Name	HA	Year	Discharge (cfs)	Data Source
09416000	Muddy River near Moapa, NV	219	1996	33.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1997	32.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1998	34.9	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	1999	34.8	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2000	34.5	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2001	32.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2002	31.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2003	31.7	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2004	30.4	USGS-NWIS, 2007
09416000	Muddy River near Moapa. NV	219	2005	33.2	USGS-NWIS, 2007
09416000	Muddy River near Moapa, NV	219	2006	33.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1951	44.1	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1952	53.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1953	46	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1954	43.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1955	54.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1956	43.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1957	48.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1958	48.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1959	45	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1960	44.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1961	60.7	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1962	44.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1963	40.1	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1964	40	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1965	44.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1966	41.9	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1967	45.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1968	43.9	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1969	54.5	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1970	45.1	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1971	42.5	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1972	41.9	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1973	44.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1974	38.5	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1975	40.6	USGS-NWIS, 2007

Table C.1-3Annual Discharge of Spring Gaging Stations in the WRFS(Page 3 of 4)

Appendix C

SE ROA 11855



Table C.1-3
Annual Discharge of Spring Gaging Stations in the WRFS
(Page 4 of 4)

Station Number	Station Name	НА	Year	Discharge (cfs)	Data Source
09419000	Muddy River near Glendale, NV	220	1976	41.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1977	38.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1978	55.4	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1979	42.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1980	46.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1981	44.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1982	37.1	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1983	55	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1985	36.7	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1986	36.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1987	37.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1988	38.4	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1989	30.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1990	41.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1991	35.9	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1992	36.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1993	54.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1994	34,7	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1995	32.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1996	31.2	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1997	30.4	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1998	54.9	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	1999	37.3	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	2000	39.8	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	2001	31.6	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	2002	31.7	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	2003	31.7	USGS-NWIS, 2007
09419000	Muddy River near Glendale, NV	220	2004	32	USGS-NWIS, 200
09419000	Muddy River near Glendale, NV	220	2005	72.2	USGS-NWIS, 200
09419000	Muddy River near Glendale, NV	220	2006	34.6	USGS-NWIS, 200

^aSNWA Station Number

÷.

C.2.0 REFERENCES

- Carpenter, E., 1915, Ground water in southeastern Nevada: U.S. Geological Survey Water-Supply Paper 365, 86 p.
- Eakin, T.E., 1963, Ground-water appraisal of Pahranagat and Pahroc valleys, Lincoln and Nye counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 21, 36 p.
- Ertec Western, Inc., 1981, MX Siting Investigation—Water Resources Program—Results of Regional Carbonate Aquifer Testing, Coyote Spring Valley, Nevada: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, Report E-TR-57, 65 p.
- Hardman, G., and Miller, M.R., 1934, The quality of the waters of southeastern Nevada—Drainage basins and water resources: University of Nevada-Reno, Agricultural Experiment Station, Bulletin No. 136, 62 p.
- Hess, J.W., and Mifflin, M.D., 1978, A feasibility study of water production from deep carbonate aquifers in Nevada: Water Resources Center, Desert Research Institute Publication No. 41054, 125 p.
- Maxey, G.B., and Eakin, T.E., 1949, Groundwater in the White River Valley, White Pine, Nye, and Lincoln counties, Nevada: State of Nevada, Office of the State Engineer, Water Resources Bulletin No. 8, 59 p.
- Mifflin, M.D., 1968, Delineation of ground-water flow systems in Nevada: Desert Research Institute, Water Resources Center Technical Report Series H-W, Publication No. 4, 115 p.
- Miller, M.R., Hardman, G., and Mason, H.G., 1953, Irrigation waters of Nevada: University of Nevada-Reno, Agricultural Experiment Station, Bulletin No. 187, 63 p.
- Smith, A.M., 1938, Biennial Report of the State Engineer—For the period July 1, 1936, to June 30, 1938: State of Nevada, Office of the State Engineer, 197 p.
- Smith, A.M., 1942, Biennial Report of the State Engineer—For the period of July 1, 1940, to June 30, 1942, Inclusive: State of Nevada, Office of the State Engineer, 151 p.
- Smith, A.M., 1944, Report of the State Engineer of Nevada—For the fiscal years ending June 30, 1943–1944: State of Nevada, Office of the State Engineer, 135 p.

SNWA, see Southern Nevada Water Authority.

Appendix C

SE ROA 11857



- Southern Nevada Water Authority, 2006, Physical Settings of Selected Streams in Clark, Lincoln, and White Pine Counties: Groundwater Development Project, Volume 2. (In Review).
- Stearns, N.D., Stearns, H.T., and Waring, G.A., 1937, Thermal springs in the United States: U.S. Geological Survey Water-Supply Paper 679-B, 206 p.
- U.S. Geological Survey, 1962 to 2006, Water Resources Data, Nevada, Water Years 1962-06: U.S. Geological Survey Water-Data Reports NV-62 to NV-06-1 (published annually).
- U.S. Geological Survey, 2004, National Water Information System (NWIS Web) data accessed on June 10, 2004, at website http://waterdata.usgs.gov/nwis/.
- U.S. Geological Survey, 2007, National Water Information System (NWIS Web) data accessed on August, 2007 at website http://waterdata.usgs.gov/nwis/.

USGS-NWIS, see U.S. Geological Survey, National Water Information System.

Wells, J.V.B., 1954, Compilation of records of surface waters of the United States through September 1950—Part 9, Colorado River Basin: U.S. Geological Survey Water-Supply Paper 1313, 749 p.

Appendix D

Interbasin Groundwater Flow through Selected Boundaries

SE ROA 11859

JA_4621

D.1.0 INTRODUCTION

This appendix presents the assumptions, data, and calculations used to approximate interbasin groundwater flow through selected boundaries of the WRFS. These estimates were used as the basis for some of the Excel Solver constraints discussed in Section 5.0 and to partition groundwater outflow for selected basins.

D.2.0 INTERBASIN FLOW WITHIN THE WRFS

This section includes estimates of the groundwater outflow from Coyote Spring Valley to Hidden Valley and the interbasin flow for Cave, Dry Lake, and Delamar valleys.

D.2.1 Outflow From Coyote Spring Valley to Hidden Valley

Outflow from Coyote Spring Valley to Hidden Valley is likely to occur through the carbonate-rock aquifer. The outflow volume may be estimated using the available water-level and hydraulic property data, and Darcy's Law expressed in the form of the following equation:

$$Q = TIW \times \left(\frac{365}{43,560}\right)$$
 (Eq. D.2-1)

where,

Q =the groundwater flow (afy)

 $T = the transmissivity (ft^2/day)$

I = the hydraulic gradient (ft/ft)

W = the estimated width of the section of flow (ft)

Although most of the groundwater from Coyote Spring Valley flows to the southeast into the Muddy Springs River Area (Eakin, 1966; Harrill et al., 1988), the hydrogeology of the valley supports a portion of this groundwater flowing south to Hidden Valley. No carbonate-rock wells have been drilled in Hidden Valley to date. However, carbonate-rock wells for which hydraulic data are available exist in Coyote Spring Valley and California Wash, the two valleys located upgradient and downgradient from Hidden Valley, respectively (Figure D.2-1 and Table D.2-1). An estimate of this groundwater flux may be derived using Equation D.2-1 and water levels and transmissivities measured in these carbonate-rock wells.

Appendix D

D-1

SE ROA 11860





Note: For an index of springs names and well IDs, see Table D.2-1.

Figure D.2-1 Interbasin Outflow and Water-Level Elevations of Selected Wells in Southern Coyote Spring Valley and Vicinity

			Loc	ation	20.00		Then.		I accelerated and	Last	fairs I antelli
ap-ID	Station Number	Station Name	UTM Easting ² (m)	UTM Northing ⁵ (m)	Elev ^b (ft-amsl)	Site Type	Depth (ft-bgs)	Open Interval (ft-bgs)	Measurement Date	to Water (ft-bgs) ^c	Elevation ^d (ft-amsl)
0-01	382223114502501	180 ND5 E64 18CC 1 Sidehill Pass Well	688,678	4,249,221	5,975	Well-Basin Fill	1	1	7/25/2005	158	5,817
0-02	180 N07 E63 13DB 1	180 N07 E63 13DB 1	687,817	4,259,662	6,014	Well-Basin Fill	250	200 to 240	12/15/1995	180	5,834
0-03	382822114515301	180 ND7 E63 14AB 1 USGS-MX	685,674	4,260,043	6,012	Well-Basin Fill	273	200 to 263	3/15/1994	223	5,789
0-04	382822114515302	160 ND7 E63 14AB 2 USGS-MX	685,674	4,280,043	6,012	Well-Carbonate	422	380 to 422	10/25/1980	231	5,781
9-05	382810114521501	180 N07 E63 14BADB1 USAF	685,674	4,260,043	6,013	Well-Basin Fill	269	50 to 269	5/29/2007	219	5,794
90-0	382810114521502	180 NO7 E63 14BADB2 USAF	685,674	4,260,043	6,013	Well-Carbonate	418	50 to 418	7/15/2002	221	5,792
20-02	382807114521001	180 N07 E63 14BADD1 USGS-MX (Cave Valley)	585,797	4,259,954	6,012	Well-Carbonate	450	210 to 250/ 375 to 435	5/29/2007	220	5,792
80-0	382747114525701	180 N07 E63 15DBAD1 USBLM	684,672	4,259,311	6,026	Well-Basin Fill	1	i	8/4/2003	\$L	6,011
60-0	180 N07 E63 27CD 1	180 N07 E63 27CD 1	684,288	4,255,971	5,988	Well-Basin Fill	245	200 to 240	5/15/1998	157	5,831
0-10	180 N07 E63 27DD 1	180 N07 E63 27DD 1	685,088	4,255,990	5,988	Well-Basin Fill	290	240 to 280	5/15/1998	168	5,820
0-11	180 N07 E63 270D 2	180 NO7 E63 27DD 2	685,088	4,255,990	5,988	Well-Basin Fill	320	260 to 300	7/15/1998	183	5,805
0-12	180 N07 E63 33D 1	180 ND7 E63 33D 1	683,325	4,254,561	5,982	Well-Basin Fill	300	198 to 300	1/15/2000	192	5,790
0-13	382640114492801	180 N07 E64 19 1 Gulf Oil Corp.	787,689	4,257,364	6,005	Well-Basin Fill	265	240 to 265	3/15/1980	215	5,790
0-14	383458114473601	180 N08 E64 04ABDD1 USBLM	692,206	4,272,813	5,224	Well-Basin Fill	200	160 to 200	2/21/2007	131	6,093
0-15	383307114471001	180 NDS E64 15BCBC1 USBLM	692,846	4,269,374	6,163	Well-Basin Fill	i	3	5/29/2007	262	5,901
0-16	383056114501501	180 NDB E64 30CDBC1 USBLM	688,462	4,265,229	6,087	Well-Basin Fill	352	50 to 352	6661/11/2	327	5,760
2-17	383501114504801	180 ND9 E63 01A 1	887,559	4,283,005	6,535	Well-Basin Fill	1		10/16/1962	2	6,534
0-18	383828114474501	180 N09 E64 16ACB 1 Cave Spring	591,761	4,279,249	6,488°	Spring	ï	ł	4	4	6,488
0-19	383632114465801	180 N09 E64 27BCDD1 USBLM	692,983	4,275,700	6,414	Well-Basin Fill	315	277 to 315	5/29/2007	221	6,193
0-20	384207114505601	180 N10 E63 25A 1	686,984	4,285,891	6,604	Well-Basin Fill	50		9/22/2006	14	6.590
0-21	384534114495301	180 N10 E64 068DA 1 Robbers Roost Well	588,337	4,292,295	6,848	Well-Basin Fill		•	7/26/2005	140	6.708
220	180 N11 E83 250D 1	180 N11 E63 250D 1	687,670	4,294,728	6,987	Well-Basin Fill	140	100 to 140	4/15/1998	5	6,896
-23	382458114474301	180 S07 E64 33 1 Sidehill Spring	692,408	4,254,280	6,531*	Spring	1	1	1	•	6,531
9-24	180MSD1M	180W501M	688,048	4,273,716	6,457	Well-Carbonate	1,212	788 to 1,192	9/7/2007	1,051	5,406
0-25	180W902M	180W902M	689,805	4,248,363	5,987	Well-Carbonate	503	196 to 882	9/26/2007	138	5,849
0-26	382738114525601	180 N07 E63 15C 1	684,703	4,259,034	6,004	Well-Basin Fill	1	4	1/1/1900	300	5,704
0-27	381624114540302	180 N05 E63 34BBCC2 USBLM	683,551	4,238,220	6,229	Well-Basin Fill	1		8/25/2003	8	6,221
1-01	375624114444501	181 N01 E62 24ABB 1 USBLM	698,010	4,201,548	4,695	Well-Basin Fill	515	400 to 515	11/13/2006	48	4,648
1-02	181 NO1 E65 02AA 1	181 ND1 E65 02AA 1	706,529	4,206,411	5,663	Well-Volcanic	į	1	B/15/1960	10	5,653
1-03	181 N02 E63 13CA 1	181 N02 E63 13CA 1 Coyoté Spring	687,693	4,211,513	5,224*	Spring	ŗ				5,224
1-04	380336114473501	181 N02 E64 03B 1 USBLM	683,124	4,215,351	4,972	Well-Basin Fill	742	702 to 742	5/9/2006	658	4,314
1-05	380531114534201	181 ND3 E63 27CAA 1 USGS-MX (N. Dry Lake)	684,519	4,218,103	5,391	Well-Carbonate	2,395	No perforations	7/25/2007	846	4,545
1-06	380616114494101	181 N03 E64 20BD 1 USBLM - Coyote Well	590,485	4,220,368	5,071	Well-Basin Fill	380	50 to 380	5/29/2007	269	4,801
20-11	380550114412301	181 N03 E65 21D 1 Bristol Well	702,610	4,219,921	5,464	Well-Basin Fill	80	20 to 80	5/29/2007	20	5,444
1-08	181 ND3 E65 22 1	181 ND3 E65 22 1	703,867	4,220,163	5,593	Well-Basin Fill	240	60 to 240	1/15/1968	en	5,590
60-14	381256114500701	181 N04 E64 07DC 1 USGS-MX (Muleshoe Valley)	689,481	4,232,095	5,534	Well-Basin Fill	1,190	1,050 to 1,190	5/29/2007	253	5,280
1-10	381256114500702	181 N04 E64 07DC 2 USGS-MX	689,526	4,232,282	5,539	Well-Basin Fill	672	50 to 672	B/27/2003	269	5,270
11-11	381256114500703	181 N04 E84 07DC 3 USGS-MX	689,526	4,232,282	5,539	Well-Basin Fill	1,134	50 to 1,134	3/10/1990	253	5,285

Table D.2-1 Water Level Data for Selected Wells and Springs (Page 1 of 4)

Appendix D

SE ROA 11862

D-3

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

			Loca	tion	10 70		Well		Last Water Level	Depth	Water Level
Map-ID	Station Number	Station Name	UTM Easting ^a (m)	UTM Northing ^a (m)	Elev ^b (ft-amsl)	Site Type	Depth (ft-bgs)	Open Interval (ft-bgs)	Measurement Date	to Water (ft-bgs) ^c	Elevation ^d (ft-amsI)
		And the recounds a little Stafe Spring	701 112	4.233.949	6.150°	Spring	1	1			6,150
81-12	381358114412201	181. NU4 EGO U4UDU I LUNETIEM OFURA	705.540	4 227 429	6.319	Well-Volcanic	81	60 to 80	10/15/1954	20	6,268
81-13	181 NU4 E65 25UU 1		ROO DAD	4 227 795	6.090 ⁿ	Spring	1	1	į	1	6,090
81-14	381029114430701	181 NU4 E65 29UB 1 Spring	200 889	HOC BEC F	6 17B ^a	Spring	i		1	9	6,178
81-15	381508114421801	181 ND5 E65 32AU 1 Spring	100,000	A DOR RAM	E EAO	Well-Mellemic	38	10 to 28	5/1/1972	10	6,539
81-15	181 N05 E65 34DC 1	181 ND5 E65 34DC 1	103,004	HO'CO7'#	0000	Mail Malancia	1 50	10 m 03	7/15/1972	12	6,630
21-18	181 N05 E65 35BA 1	181 ND5 E65 358A 1	704,885	4,236,789	2,042	Mell-Volcanic	120	ED IN 452	APR/1004	40	4,623
81-18	374536114443001	181 S02 E65 19CA 1	698,859	4,181,583	4,672	Mell-Basin Fill	001	act of no	TOOLOGIT	304	4 249
81-18	374215114453101	181 S03 E64 12AC 1 USGS-MX(S. Dry Lake Well)	697,515	4,175,351	4,643	Well-Basin Fill	1,000	0/0 00 00 000	TIDDIDOUT	000	CHC Y
81-20	374215114453102	181 S03 E64 12AC 2 USGS-MX	697,515	4,175,351	4,643	Well-Basin Fill	1,300	1,2/0101,280	Indiana	200	A 264
81-21	374215114453103	181 S03 E64 12AC 3 USGS-MX	697,515	4,175,351	4,643	Well-Basin Fill	86/	/08 10 /80	00100100	700	POOR P
81.27	181M-1	181M-1	688,537	4,198,181	4,968	Well-Carbonate	1,472	765 to 1.471	9/4/200/	0/0	1876
201.00	MONOMAN + C	TRIMODAM	698,688	4,174,479	4,804	Well-Basin Fill	1,260	637 to 1,240	9/4/2007	487	4,307
10 10	101 VISUOM 1	184 SN3 F63 224B 1	684,620	4,172,308	5,313	Well-Basin Fill	6	1	3/15/1968	6	5,310
47-19	101 202 202 202 101	tor cod Eck 07AC 1 lamoh 7B	689.759	4,165,424	4,810	Well-Basin Fill	1	50 to 1,000	5/26/2003	515	4,295
07-18	181 SUG ED4 U/HU 1	101 304 ED4 UPO 1 Gracev Spring	696.124	4,157,193	5,786°	Spring	•			1	5,786
10-79	182 SUD E04 UZUE 1	162 500 E63 12AD 1 USGS-MX (Delamar Well)	688,422	4,146,273	4,713	Well-Basin Fill	1,195	920 to 980/ 1.040 to 1.180	7/30/2007	863	3,850
00.00	275630114520000	182 SOB E63 12ADBD2 USGS-MX	688,422	4,146,273	4,713	Well-Basin Fill	981	540-530/816-847/ 877-940/950-971	4/1/1981	292	3,846
00-30			RAD 874	4 135 306	4.582	Well-Volcanic	1,321	1,000 to 1,300	9(4/2007	827	3,755
82-04	182M-1	100/04-1	500 M78	A 133 200	4 802	Well-Volcanic	1.702	1.274 to 1.677	9/4/2007	1,316	3,485
82-05	182W906M	182/WSUDA	200,000	A 250 505	6 364	Well-Basin Fill	529	50 to 229	8/15/1963	212	6,152
83-01	183 ND7 E65 17D 1	183 NOV E65 1/17	TOD ODA	A 259 RUD	5344	Well-Basin Fill	264	50 to 264	6/15/1966	200	6,144
83-02	183 N07 E65 17DA 1	183 NO7 E65 17/DA 1	toping t	noninny't	6.230	Mall.Rasin Fill	230	50 to 230	7/25/2005	177	6,143
83-03	382757114414301	183 N07 E65 17DAA 1 USBLM	TOU, UN	A 26.4 208	RALT	Well-Basin Fill	220	147 to 210	6/15/1967	147	5,900
83-04	1 183 N07 E65 11CC 1	183 N07 E65 11CC 1	100'10/	1000 100 1	UNC 3	Molt Drecin Eill	UU:	50 to 100	7/15/1979	5	5,236
10-10	207 N07 E61 07DD 1	207 N07 E61 070D 1	ILA'NOG	101,002,9	242'0	Weindown an	3	50 to 6.305	6/15/1981	18	5,191
07-02	207 N07 E61 10B 1	207 N07 E61 10B 1	50/'t-00	1000001t	Contra I	Africa Control and)	1986	8	5,211
207-03	3 207 ND7 E61 11AC1	207 N07 E61 11AC1	D00,023	nte:007't	147'0	Mull Darin Elli	22	30 to 55	10/15/2001	8	5,292
207-04	1 207 ND7 E62 21AC 1	207 ND7 E62 21AC 1	D/3,421	001/207/5	0'070	Well-Doolo Lal	404	S0 to 101	8/4/1991	49	5,199
207-05	5 362718115094901	207 N07 E61 19BDDC1 USGS-MX	660, 169	4,251,458	R67'C	Well-Device Ett	ARG	FOID 456	61211211T	8	5,185
207-06	3 207 NO6 E61 06BB 1	207 N06 E61 05BB 1	669,866	4,253,191	577'C	Wei-Baset Fill	004	50 to 110	1/15/2003	9	5,171
207-07	7 207 N07 E61 36CC 1	207 N07 E61 36CC 1	667,912	4,253,734	/81'C	In the part of the	20	50 to 70	7/45/1979	6	5,165
207-06	8 207 N07 E61 36CCD 1	207 ND7 E61 36CCD 1	568,003	4,203,623	0,164	Waling and the second state	004	SO IN THO	8/15/1970	m	5,196
207-06	9. 207 N07 E61 36DD 1	207 N07 E61 36DD 1	669,105	4,203,/03	407'C	Well-Desiti Fill	3	6 In 513	7/15/1966	8	5,186
207-10	0 207 NDS E61 D6CC 1	207 N06 E61 06CC 1	669,812	4,252,240	ALZ'G	Well-Dasin Fill			1	1	5,229
207-11	1 382259115090801	207 N06 E61 18AADA1 NDW - Hot Creek Spring	067'190	078'867'6	1040 S	Intell David Tall	ADD	50 to 400	7/15/1979	ŝ	5,214
207-15	2 207 ND6 E61 09CCB 1	207 N06 E61 09CCB 1	110,500	110'007'5	21710	Mole Contraction of the			1986	200	5,084
207-1	3 207 N06 E62 07AB1	207 ND6 E62 07AB1	971'0/9	056'107'5	+07'0	VEIT-COLORINA	117	40 to 117	5/15/1968	25	5,259
207-14	4 207 N06 E62 07CD 1	207 N06 E62 07CD 1	6/0,054	040'007'5	+07'C	Wein-Desite Internate		50 to 3 980	6/22/1966	142	5,143
1-70c	5 207 NOS E62 07CD 2	207 NO6 E62 07CD 2	669,947	4,200,410	00710	WER-Cervinen	-	in the second			

Table D.2-1 Water Level Data for Selected Wells and Springs

Appendix D SE ROA 11863

Southern Nevada Water Authority - Water Resources Division

			Loc	ttion	10 10		114-11	R	Lact Matael avai	Last	Water aval
01-de	Station Number	Station Name	UTM Easting ² (m)	UTM Northing ² (m)	Elev ^b (ft-ams!)	Site Type	Depth (ft-bgs)	Open Interval (ft-bgs)	Measurement Date	to Water (ft-bgs) ^c	Elevation ^d (ft-amsI)
7-16	207 N06 E61 31AA 1	207 N06 E61 31AA.1	661,137	4,245,452	5,143	Well-Basin Fill	152	70 to 152	3/15/2001	27	5,116
7-17	207 N06 E61 32BA 1	207 ND6 E61 32BA 1	662,019	4,245,179	5,149	Well-Basin Fill	8	•	3/15/1979	18	5,131
7-18	207 ND6 E61 33DC 1	207 ND6 E61 33DC 1	664,040	4,243,999	5,208	Well-Basin Fill	,	1	10/14/1968	9	5,202
81-7	207 N06 E61 33D 1	207 N06 E61 33D 1	664,251	4,244,203	5,207	Well-Basin Fill	200	50 to 200	B/15/1979	100	5,107
7-20	207 N06 E61 28DA1	207 N06 E61 28DA1	664,334	4,245,971	5,195	Well-Carbonate	ŗ	1	1986	8	5,142
7-21	382111115055901	207 N06 E61 27AADC1 USGS-MX	666.042	4,246,692	5,214	Well-Basin Fill	8	50 to 96	9/21/1594	20	5,144
7-22	207 NO6 E61 27DD 1	207 N06 E61 27DD 1	566,042	4,245,663	5,204	Well-Basin Fill	250	50 to 250	6/15/1970	8	5,106
7-23	207 N06 E62 31AD 1	207 NO5 E62 31AD 1	670,827	4,244,903	5,434	Well-Basin Fill	250	50 to 250	7/15/1979	145	5,289
7-24	382005115023701	207 NDB E62 31ADD 1	570,967	4,244,822	5,454	Well-Basin Fill	300	50 to 300	3/9/1990	123	5,331
7-25	207 NO5 E61 31CB 1	207 N05 E61 31CB 1 Murphy Meadows 2	660,306	4,234,215	5,104	Well-Basin Fill	100	50 to 100	7/25/2005	11	5,093
7-26	373614115083701	207 ND4 E61 16D 1	663,950	4,230.069	5,140	Well-Basin Fill	į	1	5/10/1963	84	5,056
8-01	208 N05 E63 34BBCC 1	208 N05 EB3 34BBCC 1 Griswald Well	683,428	4,236,136	6,255	Well-Basin Fill	,	4	7/12/1998	3	6,252
8-02	380914115044101	208 ND4 E61 36C 1 USBLM	668,322	4,224,628	5,044	Well-Basin Fill	i	1	7/23/2000	69	4,975
8-03	374218115031501	208 N03 E62 08C 1 USBLM	671,468	4,222,073	5,063	Well-Basin Fill	1	r	5/1/1963	217	4,847
8-04	208 N03 E62 17AB 1	206 N03 E62 17AB 1	672,041	4,221,499	5,053	Well-Basin Fill	350	150 to 350	3/15/1974	252	4,801
8-05	208 N03 E62 27 1	208 N03 E62 27 1	674,858	4,217,787	4,976	Well-Basin Fill	357	320 to 357	8/15/1958	260	4,716
8-06	380505114593501	208 N03 E62 35888 1 USBLM	675,936	4,217,111	4,960	Well-Basin Fill	315	50 to 315	3/9/1985	251	4,710
20-8	208 N03 E62 35888 2	208 N03 E62 368BB 2 Pahroc MX	675,845	4,217,088	4,965	Well-Basin Fill	3	j.	7/26/2004	268	4,698
8-08	380450114594201	208 N03 E62 35B 1 USBLM	675,775	4,216,845	4,979	Well-Basin Fill	270	50 to 270	9/21/1994	250	4,729
8-09	373924115003101	208 N03 E62 35B 2 USBLM	676,045	4,216,558	4,956	Well-Basin Fill	â		5/8/1963	252	4,704
10-6	374058115113501	209 S03 E60 13DACD1	659,265	4,172,123	4,060	Well-Carbonate	478	213 to 479	8/25/2003	209	3,851
9-02	209 S03 E61 22BB 1	209 S03 E61 2288 1	664,606	4,171,854	5,005	Well-Carbonate	1,427	35 to 1427	6/15/1965	860	4,145
80-03	209 S03 E62 25AB 1	209 S03 E62 25AB 1 Pahroc Spring	678,091	4,170,481	5,403*	Spring	ł	1	4	1	5,403
10-B	209M-1	209M-1	677,377	4,168,166	5,123	Well-Carbonate	1,616	1,273 to 1,595	9/4/2007	1,200	3,923
90-60	373554115125201	209 S04 E60 14DBAB1 Hiko Spring	657,549	4,162,744	3,878°	Spring-Regional	1	1	4	í	3,878
90-60	373158115141601	209 \$05 E60 10ABCC1	855,626	4,155,431	3,841	Well-Basin Fill	140	50 to 140	1/1/1988	30	3,811
10-60	09415590	209 S05 E60 10 1 Crystel Spring near Hiko, NV	856,168	4,155,349	3,803 ^e	Spring-Regional	ł.		1	1	3,803
80-60	372857115124001	209 S05 E60 26DAD 1 Brownie Spring	658,088	4,149,897	3,696*	Spring	1	-		1	3,695
80-80	372749115113401	209 S06 E61 06BBBB1 Ash Springs	659,848	4,147,834	3,622°	Spring-Regional	ì	1	÷	1	3,622
09-10	209 S07 E61 21DA 1	209 S07 E61 21DA 1 Grove Spring	664,633	4,132,301	3,395°	Spring	ì	(1	3,395
11-00	209 S08 E61 238ABD 1	209 S08 E61 23BABD 1 Solar Panel Spring	667.261	4,123,643	3,238*	Spring	ï	1	1	ų	3,238
09-12	209 S07 E62 21AC 1	209 S07 E62 21AC 1	674,019	4,133,040	4,123	Well-Basin Fill	ì	1	8/15/2001	31	4,092
09-13	209 S08 E61 32BD 1	209 S08 E61 32BD 1	662,530	4,119,798	4,525	Well-Basin Fill	1	t	5/15/1976	25	4,500
08-14	Maynard Spring	Maynard Spring	674,523	4,117,711,4	3,098*	Spring	1	1	t	*	3,133
10-01	CSI-2	210 S13 E63 14CD 1 CS1-2	687,085	4,075,781	2,209	Well-Carbonate	1,015	523.8 to 644/ 684.1 to 844.4/ 884.5 to 1,004.7	6/15/2006	388	1,820
10-02	CSI-1	210 S13 E63 22DC 1 CSI-1	686,043	4,074,459	2,266	Well-Carbonate	920	520 to 600/640 to 760/800 to 880	5/30/2005	439	1,827
and the second	Anterest sectors	ALACTA TOS JONDOCT ISCO MY OF DT 4	200 000	CEV FLV F	22.7.0	Mail Concernation	030	ED IN DED	Thinnin T	95.4	1821

Table D.2-1 Water Level Data for Selected Wells and Springs (Page 3 of 4)

Appendix D

D-5 SE ROA 11864

1	2
11	11
	1111
	CHA.
1	
	-

Table D.2-1 er Level Data for Selected Wells and Springs (Page 4 of 4)	
Water L	

Npbplic Station Number	ap-ID SI 10-04 210 S	100 million (100 m				at a statut					-	
TODO TODO <th< th=""><th>10-04 210 S</th><th>station Number</th><th>Station Name</th><th>Easting²</th><th>UTIM Northing[*] (m)</th><th>Elev.^b (ft-amsl)</th><th>Site Type</th><th>Depth (ft-bgs)</th><th>(B-bgs)</th><th>Measurement Date</th><th>Water (ft-bgs)^c</th><th>(famsi)</th></th<>	10-04 210 S	station Number	Station Name	Easting ²	UTIM Northing [*] (m)	Elev. ^b (ft-amsl)	Site Type	Depth (ft-bgs)	(B-bgs)	Measurement Date	Water (ft-bgs) ^c	(famsi)
TUDOR STITCOR	10-04 210 S			100 000	EUL 640 1	1910	Well-Carbonate	1.040	320 to 1,020	7/1/10/2007	340	1,821
2006 3047111653M01 2108 (3810 MM) 2108 (3810 MM) <td>10-05 36474</td> <td>S13 E63 25AD 1</td> <td>210 S13 E63 25AD 1 CSVM-1</td> <td>200,990</td> <td></td> <td>101.12</td> <td>Well Production</td> <td>BCB</td> <td>121 to 628</td> <td>7/10/2007</td> <td>352</td> <td>1,821</td>	10-05 36474	S13 E63 25AD 1	210 S13 E63 25AD 1 CSVM-1	200,990		101.12	Well Production	BCB	121 to 628	7/10/2007	352	1,821
2006 2016 <th< td=""><td>0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>41114532801</td><td>210 S13 E63 26AAAA1 USGS-MX CE-DT-5</td><td>688,084</td><td>4,0/4,219</td><td>2112</td><td>Whether all the lot of the lot of the</td><td>740</td><td>AED to 700</td><td>9/20/2006</td><td>361</td><td>1,819</td></th<>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	41114532801	210 S13 E63 26AAAA1 USGS-MX CE-DT-5	688,084	4,0/4,219	2112	Whether all the lot of the lot of the	740	AED to 700	9/20/2006	361	1,819
3100 210 54 4 ER0 OMA 1 210 54 0 ER0 MB 1 210 56 0 ER0 MB 1		S13 E63 26AB 1	210 S13 E63 26AB 1 CSV-RW2	687,852	4,074,082	2,200	Weil-Larbonate	Not a	TEN TEN	7000012	1085	2,046
NUM ADDRESS AD	010 200	STA ERD DIAA 1	210 S14 E62 01AA 1 CSVM-5	680,295	4,068,774	3,131	Well-Carbonate	1/180	1,0201 1010201	ADDRACHT	TAR	1 824
TUD TUD <td>10-012</td> <td>1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td> <td>210 STE FRA DARR 1 CSVM-2</td> <td>685,625</td> <td>4,059,370</td> <td>2,573</td> <td>Well-Carbonate</td> <td>1,400</td> <td>720 to 1,360</td> <td>Innzisi/</td> <td>OHI</td> <td>4 010</td>	10-012	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	210 STE FRA DARR 1 CSVM-2	685,625	4,059,370	2,573	Well-Carbonate	1,400	720 to 1,360	Innzisi/	OHI	4 010
210:00 210:01<	210-08 210	510 E03 U300 1	210 010 E00 0000 1 0000 8	686.453	4,078,333	2.252	Well-Carbonate	1,160	420 to 1,160	7/10/2007	433	210'1
210-10 210-10<	210-09 210	S13 E03 118C 1		870 310	4 102 600	2,651	Well-Carbonate	1,200	380 to 1,180	7/9/2007	443	2,201
216-01 S264811446501 2165 NEST FERA TOCOT 2165 NEST FERA TOCO 2165 NEST FERA TOCO 2165 NEST FERA TOCOT	210-10 210 3	S10 E62 25AD 1	210 S10 E62 23AU 1 COVIN-3	SALAND	A N30 284	2,069	Well-Carbonate	565	510 to 560	9/20/2006	257	1,812
216.00 2 16 5 T E4 4 100C1 2 16 7 10 100 T 100C1 2 16 7 10 100C1 2 16 7 1 100C1 2 16	216-01 3628-	146114495501	216 S17 E64 09DD 1 CRYSTAL WELL 2	041 400		5.079	Midl Carbonate	767	442 to 492	9/20/2006	260	1,813
2010 2 (15) (1 (25) (16 / 1764) 2 (15) (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (16 / 1764) 2 (17 / 16 / 176) 2 (17 / 176	216-02 216	S17 E64 10CC 1	216 S17 E64 10CC 1 Crystal Well 1	D34,303	a) /'Acn't	0.000	Mhill Carbonate	833	553 to 833	6/20/2006	257	1,813
216.04 36272114.305401 216 S17 E64 210.801 663, 140 1.00 1.00 1.301 1.112001 800 1.31 216.04 36773114.305401 216 S17 E64 210.801 623, 43 2.691 1.00 1.140 10.1801 1112001 422 1.8 216.05 Paulte-M3 691, 15 681, 107 2.288 Well-Carbonale 570 1.301 1.801 1112001 422 1.8 216.05 Paulte-M3 281, 15 2.288 Well-Carbonale 2.00 1.3401 1.800 1.167000 422 1.8 216.05 GV-PUNE-W35 281, 16 2.288 Well-Carbonale 2.00 1.3401 1.800 1.16 1.431 1.8 1.	216-03 216	S17 E64 21CB 1	216 S17 E64 21CB 1 GV-RW1	692,928	4,000,040	2,003	Mail Contronate	576	510 to 575	6/12/1958	260	1,812
16-00 CM-1 CM-1 <t< td=""><td>716-04 3627</td><td>23114505401</td><td>216 S17 E64 21CBBD1</td><td>692,740</td><td>4,038,717</td><td>2'012</td><td>AVEII-LIGITICOLIGICAL</td><td>200</td><td>× AAD to 5 0001</td><td></td><td></td><td></td></t<>	716-04 3627	23114505401	216 S17 E64 21CBBD1	692,740	4,038,717	2'012	AVEII-LIGITICOLIGICAL	200	× AAD to 5 0001			
Faulte-MS Pauke-MS Each of To S201 6,70 S20 16,70 S20 16,70 <t< td=""><td>216-05 GV-1</td><td></td><td>GV-1</td><td>682,983</td><td>4,034,143</td><td>2,691</td><td>Well-Carbonale</td><td>1,400</td><td>1,140 to 1,380/ 1,140 to 1,180/ 1,240 to 1,280/ 1,340 to 1,380</td><td>7/11/2007</td><td>880</td><td>1,811</td></t<>	216-05 GV-1		GV-1	682,983	4,034,143	2,691	Well-Carbonale	1,400	1,140 to 1,380/ 1,140 to 1,180/ 1,240 to 1,280/ 1,340 to 1,380	7/11/2007	880	1,811
216-05 Paulue-M3 Paulue-M3 Enulue-M3 Enulue-M3 <thenulue-m3< th=""> <thenulue-m3< th=""> <thenul< td=""><td>1</td><td></td><td></td><td>204 505</td><td>CUE PPU P</td><td>2 738</td><td>Well-Carbonate</td><td>670</td><td>630 to 670</td><td>10/15/2000</td><td>422</td><td>1,816</td></thenul<></thenulue-m3<></thenulue-m3<>	1			204 505	CUE PPU P	2 738	Well-Carbonate	670	630 to 670	10/15/2000	422	1,816
ZF6-07 GV-PWWS1 Z16 S18 E63 0GDA 1 GV-PWWS1 E66.187 VL01/GA Z.2460 Well-Carbonate 6.55 557 to E65 711/2007 435 13 Z16-07 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 2450 to 145 2746 Mell-Carbonate 1,145 700 lb 1,145 21516-00 711/2007 431 1,18 Z16-10 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 69.016 1,000 711/2007 431 1,18 Z16-10 GV-VERR GV-PUKE-WS2 GV-PUKE-WS2 69.016 1,000 711/2007 731 1,18 Z16-11 GV-VERR GV-PUKE-WS2 69.016 1,000 711/2007 712 1,18 Z16-11 GV-PUKE-WS2 GV-PUKE-WS2 69.016 1,00 711/2007 712 1,18 Z16-11 GV-PUKE-WS2 GV-PUKE-WS2 69.016 1,00 711/2007 712 1,18 Z16-11 GV-PUKE-WS2 GV-PUKE-WS2 69.016 1,00 25.25 VMIC-Carbonate 1,50 910101,200 712 1,18	216-06 Paiul	ites-M3	Paiutes-M5	000'100	and the state	001 0	(Moll Prehonata	0000	1 240 to 1,980	7/15/2002	684	1,814
Z16-06 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS2 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS2 GV-DUKE-WS1 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS1 GV-DUKE-WS1 GV-DUKE-WS2 GV-DUKE-WS2 GV-DUKE-WS2 <t< td=""><td>216-07 GV-F</td><td>PW4WS1</td><td>216 S18 E63 05DA 1 GV-PW-WS1</td><td>683,007</td><td>4,031,434</td><td>0547</td><td>Well-velocitation</td><td>205</td><td>537 to 685</td><td>7/11/2007</td><td>435</td><td>1,813</td></t<>	216-07 GV-F	PW4WS1	216 S18 E63 05DA 1 GV-PW-WS1	683,007	4,031,434	0547	Well-velocitation	205	537 to 685	7/11/2007	435	1,813
ALOND GV-DIVE-MS2 GV-DUKE-MS2 GS-180 4,028,393 2,405 Weil-Carbonate 1,445 2/11/2007 7/12 1/13 216-11 GV-PW-MIWIY GV-PW-MIWY GV-PW-MIWY GV-PW-MIWY GV-PW-MIWY GV-PW-MIWY 216-11 200 to 1,100 7/11/2007 7/12 1/13 216-12 GV-PW-MIWY GV-PW-MIWY GV-PW-MIWY GV-PW-MIWY 682,400 4,001,468 2,552 Mell-Carbonate 1,450 7/11/2007 7/12 1/13 216-13 S2507/1457201 216-14 682,416 2,001,488 2,630 7/11/2007 7/12 1/18 216-13 S25101 217-5101 216-14 882,114 1/18 1/16 1/16 1/16 1/16 1/16 1/16 1/16 1/16 1/16 1/16 1/16	TAL OF CAL	DrikEJAS1	GV-DUKE-WS1	686,197	4,029,178	2,248	Well-Carboniald	200	and the second	COOCIENT.	431	1818
ZPE-UR GUNCRET CUT Control Contro Control Control	210 00-017	CONTRACTOR OF	CALINE MS2	686,185	4,029,177	2,249	Well-Carbonate	1,965	8// 10 1944	10121111	100	100 4
216-10 GWAKERK COVART Vencum 683,460 4,031,750 2,502 Well-Carbonate 1,500 901 1,500 7112 1/2 1/2 216-11 GWAWIW GWAWIW GWAWIW 682,452 4,001,488 2,555 Well-Carbonate 1,500 901 1,500 7112 712 1,8 216-12 GWAWIWC GWAWIWC 683,115 4,002,318 2,555 Well-Carbonate 1,500 940 1,500 7112 1,8 1	-10 AU-912	TON-SUP		663.838	4,028,991	2,405	Well-Carbonate	1,145	700 to 1,145	ORSE/GL/Z	0/0	120'1
Z16-11 GV-PW-MWY GV-PW-MWY GV-PW-MWY GV-PW-MWY T/2 T/3 T	216-10 GV+	KERR	CV-KEHR	COD ARM	A na4 730	2502	Well-Carbonate	1,500	900 to 1,500	7/11/2007	688	1,814
15-12 GVPWAIWZ CVPWAIWZ Sextract Non-tool Non-tool <t< td=""><td>216-11 GV-</td><td>PWWW.</td><td>GV-PW-WW</td><td>010 000</td><td>001 1001</td><td>2626</td><td>Well-Carbonate</td><td>1,500</td><td>940 to 1,500</td><td>7/11/2007</td><td>712</td><td>1,813</td></t<>	216-11 GV-	PWWW.	GV-PW-WW	010 000	001 1001	2626	Well-Carbonate	1,500	940 to 1,500	7/11/2007	712	1,813
216-13 362507114572/01 216 518 EES 05A0DB1 686.11b 4,006.585 2,000 vvmmounteener 1,434 1,424 0,140.0 616/2000 882 1,8 217-01 217 517 E63 21DC1 217 517 E63 21DC1 217 517 E63 21DC1 867,158 4,065,585 2,241 Well-Carbonate 1,6 50 to 1,200 821,2000 556 1,8 218-01 353427114472301 218 516 E64 155ADD1 ECP-2 586,723 4,045,916 2,243 Well-Carbonate 1,9 139 to 1,100 821/2000 354 1,6 218-02 35314711447601 218 516 E64 155ADD1 ECP-2 586,723 4,045,560 2,234 Well-Carbonate 139 139 to 1,100 821/2000 354 1,6 218-03 35314711447601 218 516 E64 155ADD1 ECP-2 586,723 4,044,569 2,168 Well-Carbonate 1,75 50 to 1,100 821/2000 414 1,6 218-03 35164 4,046,560 2,234 Well-Carbonate 1,125 710 to 1,126 821/2000 414 1,6	216-12 GV-	PWLMW2	GV-PW4MV2	700'790	4,004,100	D 660	Well Carbonate	1.979	1.197 to 1.979	6/13/2007	754	1,814
217-01 217 S17 E63 21DC1 217 S17 E63 21DC1 217 S17 E63 21DC1 218 S16 E64 00ABC01 TH-2 684,334 4,030,356 2,446 Mmccanone 10 50 to 1,200 8/21 S2000 5/26 1/8 218-07 363427114472301 218 S16 E64 00ABC01 TH-2 697,584 4,046,916 2,341 Well-Carbonate 10 50 to 1,100 8/21/2000 4/6 1/6 218-07 363427114472601 218 S16 E64 00ABC01 TH-2 697,723 4,046,540 2,238 Well-Carbonate 10 50 to 1,100 8/21/2000 3/6 1/6 218-03 363147114474601 218 S16 E64 00ABC01 TH-2 697,723 4,046,560 2,234 Well-Carbonate 10 50 to 1,100 8/21/2000 3/6 1/6 218-03 36314711447501 218 S16 E64 00ABC01 TH-2 696,732 4,046,560 2,234 Well-Carbonate 1,125 8/21/2000 8/14 1/6 218-03 PAIUTES-ECP1 PAIUTES-ECP1 PAIUTES-ECP1 A/040,616 2,243 Well-Carbonate 1,125 7/4 1/16 1/1	216-13 3625	507114572701	216 S18 E63 05AADB1	683,115	4,032,318	00007	Well-Certonate	1 434	1,434 to 2,480	6/15/2000	882	1,846
218-01 353427114472301 218516.E64 02ABCD1 TH-2 897,684 4,049,916 2,341 wmen-canoniese 9 100,01,028 8121/2000 416 118 218-01 353427114472301 218516.E64 15A/DD1 ECP-2 696,733 4,044,569 2,158 Well-Carbonate 19 100,01,028 8121/2000 354 1,6 218-03 3534714474601 218516.E64 15A/DD1 ECP-2 696,723 4,044,569 2,158 Well-Carbonate 19 100,0100 8121/2000 354 1,6 218-03 35347114474601 218516.E62 2,044,569 2,158 Well-Carbonate 19 101,01,125 711/2000 414 1,6 218-04 PalUTES-ECP1 696,732 4,046,560 2,243 Well-Carbonate 1,7 716,000 821,2000 429 1,6 218-04 74 4,045,560 2,243 Well-Carbonate 74 74,1015000 429 1,6 218-05 714.01560 2,243 Well-Carbonate 74 74,1015000 429 <t< td=""><td>217-01 217</td><td>1 S17 E63 21DC 1</td><td>217 S17 E63 21DC 1</td><td>684,324</td><td>080'000'5</td><td>2,140</td><td>VICIL-COLOCION</td><td></td><td>S0 to 1 200</td><td>8/21/2000</td><td>526</td><td>1,815</td></t<>	217-01 217	1 S17 E63 21DC 1	217 S17 E63 21DC 1	684,324	080'000'5	2,140	VICIL-COLOCION		S0 to 1 200	8/21/2000	526	1,815
218-02 365/45114480501 218 S16 E64 15AADD1 ECP-2 596,733 4,046,742 2,228 www.centoniate 0 50,01,100 8/21/2000 354 1,6 218-03 353/47114474501 218 S16 E64 15AADD1 ECP-2 696,733 4,046,560 2,168 Well-carbonate 0 50,01,100 8/21/2000 354 1,6 218-03 353/47114474501 218 S16 E64 238D881 TH-1 697,234 4,044,569 2,168 Well-carbonate 1,715 7/15/2000 4/14 1,6 218-05 PAIUTES-ECP1 B96,714 4,046,560 2,243 Well-Carbonate 74 74 to 1500 10/15/2000 4/29 1,6 218-05 PAIUTES-ECP3 B96,714 4,046,560 2,243 Well-Carbonate 74 74 to 1500 10/15/2000 4/29 1,6 218-05 PAIUTES-ECP3 B96,745 2,040,576 2,108 Well-Carbonate 74 74 to 1500 10/15/2000 4/29 1,6 218-04 364,016,576 2,108 Well-Carbonate 74 <t< td=""><td>218-01 3634</td><td>427114472301</td><td>218 S16 E64 02ABCD1 TH-2</td><td>897,684</td><td>4,049,915</td><td>2,341</td><td>Well-Carbonate</td><td>430</td><td>130 to 1 228</td><td>8/21/2000</td><td>416</td><td>1,812</td></t<>	218-01 3634	427114472301	218 S16 E64 02ABCD1 TH-2	897,684	4,049,915	2,341	Well-Carbonate	430	130 to 1 228	8/21/2000	416	1,812
218-03 353147114/14501 218 S15 E64 238DB81 TH-1 697.234 4,044,959 2,168 Wein-Garoonate U <thu< th=""> <thu< th=""> U</thu<></thu<>	218-02 363	1245114480501	218 S16 E64 15AADD1 ECP-2	696,723	4,046,/42	277'7	AVEI-LOGICAL AND		50 to 1 100	8/21/2000	354	1,813
218-04 PAIUTES-ECP1 PAIUTES-ECP1 Del Centronale 1,120 701 to 1,125 11,1200 4.29 1,11,120 218-05 PAIUTES-ECP3 PAIUTES-ECP3 B68,714 4,046,964 2,243 Well-Carbonate 74 74 to 1500 10/15/2000 429 1/1 218-05 PAIUTES-ECP3 PAIUTES-MC 696,714 4,046,964 2,243 Well-Carbonate 74 74 to 1500 10/15/2000 429 1/1 218-05 PAIUTES-MC 696,806 4,040,816 2,108 Well-Carbonate 680 640 to 680 10/15/2000 297 1/1 218-05 PAIUTES-MC 687,482 4,071,381 2,278 Well-Carbonate 680 640 to 680 10/15/2000 297 1/1	218-03 363	1147114474601	218 S16 E64 23BDBB1 TH-1	697,234	4,044,959	2,105	anarion ion-taxy		600 to 701/	UUUGISHL	414	1,820
T8-05 PAIUTES-ECP3 PAIUTES-ECP3 656,714 4,046,564 2,243 Well-Carbonate 74 74 74 74 74 74 710,00 10,0200 227 11,02 218-05 PAIUTES-M2 PAIUTES-M2 E66,606 4,040,876 2,109 Well-Carbonate 680 640,1660 10,162000 297 11,1 218-05 PAIUTES-M2 E66,606 4,040,876 2,109 Well-Carbonate 680 640,1660 10,17600 297 11,1 218-01 364604114471301 219 216 Well-Carbonate 680 640,1660 11/1/2002 456 1,1 218-01 364604114471301 219 837,481 2,278 Well-Carbonate 630 640,1660 21/1/1/2002 456 1,1 218-01 364,016 2,070,248 2,070,248 2,070,248 2,070 245 1,1 218-01 1,1/160 7/1/10/2007 245 2,070,248 2,062 Well-Carbonate 1,760 960.1	218-04 PAIL	UTES-ECP1	PAIUTES-ECP1	696,729	4,046,590	2,234	Well-Carbonate	1,120	.701 to 1,125	UUUCISTON	DCV	1814
Z16-03 FMULCE-Curd FMULCE-Curd E65(86) 4,040,876 2,108 Well-Carbonate 680 640 to 580 1,075,000 Zar Loc 218-06 74,040,876 2,778 V/ell-Carbonate 680 640 to 580 1,075,000 Zar 1,075,000 Zar 1,075,000 2,56 1,1 218-01 364,612 4,071,381 2,278 V/ell-Carbonate 325 325 to 997 1,1/1/2002 456 1,1 216-01 364,912 219 2,070,248 2,062 V/ell-Carbonate 1,780 960 to 1,760 7/10/2007 245 1,1 216.07 1,100 1,780 960 to 1,760 7/10/2007 245 1,1		ILLER CODS	DANITES.FCP3	696,714	4,046,984	2,243	Well-Carbonate	74	74 to 1500	OUDZIGTUTI	EVer	C18 1
218-06 PAULTES-M2 PAULTES-M2 PAULTES-M2 PAULTES-M2 245 1/1/12002 456 1/1 219-01 364604114471301 219<513	144 CO-812	In Learence		695.836	4,040,876	2,109	Well-Carbonate	680	640 to 680	10/15/2000	R	21011
219-01 364604114471301 219 513 E64 35AC/ANT USOS-MAX US-U-I-0 UN-1 UN-1 245 10 2062 Well-Carbonate 1,780 960 to 1,760 7/10/2007 245 1, 1, 2, 2, 2, 2, 1, 2, 2, 1, 2, 2, 2, 1, 2, 2, 2, 1, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2,	218-06 PAI	IUTES-M2	PAIULES-MZ	C92 708	4 071 381	2.278	Well-Carbonate	325	325 to 937	11/1/2002	456	1,822
1.10000 PTTT 2002 1 040,200 PB4,300 PB4,200 PTTT 2002 1 040,000 PTTTT 2002 PTTTT 2002 PTTTTT 2002 PTTTTTT 2002 PTTTTTTTTTT	219-01 364	4604114471301	219 S13 E64 35ACAA1 USGS-MA UC-UT-0		010 020 1	Canic	Well-Carbonate	1.780	960 to 1,760	7/10/2007	245	1,817
	219-02 UM	TVM-1	DMVM-1	DUD, AND	4'010'5	Z'UNK						

Appendix D SE ROA 11865

Carbonate wells located in the area of interest and where aquifer tests have been conducted include: test wells MX-4 (Map-ID 210-03) and MX-5 (Map ID 210-05), RW-1 (Map ID 216-03) and its observation well, Harvey Well, Well ECP-2 (Map ID 218-02), and Well TH-2 (Map ID 218-01). Tests wells MX-4 and MX-5 are located in Coyote Spring Valley within about 300 ft of each other. Transmissivities at test wells MX-4 and MX-5 have been reported by several authors (Bunch and Harrill, 1984, p. 119; IT Corporation, 1996, Appendix A; Dettinger et al., 1995, Table 5, p. 33). The average T values are 290,516 and 400,055 ft²/day for MX-4 and MX-5, respectively. At the RW-1 test location, SRK Consulting, Inc., (2001, tbl 4) reports the following transmissivities: 326,900 ft²/ day (pumping) and 411,400 ft²/day (recovery) using the observation data collected from Harvey Well; and 64,090 ft²/day using data collected at the pumping well, RW-1. The mean transmissivity at this location is 267,456 ft²/day. Transmissivities calculated for Well ECP-2 and Well TH-2 are 109,500 and 53,820 ft²/day, respectively (Mifflin and Associates, 2001, Appendix A, p. 3-5). The average transmissivity for the area based on the five test sites is 224,269 ft²/day.

A hydraulic gradient was calculated between monitor well CSVM-2 (Map ID 210-08) located in southern Coyote Spring Valley and monitor well GV-1 (Map ID 216-05), located in northern Garnet Valley. The wells have average potentiometric heads of 1,823.82 and 1,810.85 ft, respectively; based on more than 40 depth-to-water measurements each. The distance between the two wells is about 15.76 mi or 83,213 ft. Based on this information, a hydraulic gradient of 0.00016 ft/ft was calculated.

Outflow is assumed to occur across the southern boundary of Coyote Spring Valley. This boundary is crossed by several range-front faults, but the exact locations where flow occurs may not be identified with the available information. The length of the flow section was measured along a line that is approximately perpendicular to the north-to-south direction of flow hydraulic. Its length is estimated to be about 9.5 mi or 50,160 ft.

The groundwater outflow volume was calculated as the product of transmissivity, the hydraulic gradient and the flow width. The resulting estimated outflow volume is about 15,000 afy.

D.2.2 Interbasin Flow among Cave, Dry Lake, and Delamar Valleys

The majority of the interbasin flow for Cave, Dry Lake, and Delamar valleys occurs as groundwater outflow from Cave Valley to the southern third of White River Valley and northern Pahroc Valley, and from southern Delamar Valley to northern Coyote Spring Valley and perhaps also to the very southern part of Pahranagat Valley. Also, a small amount of groundwater inflow to northern Dry Lake Valley likely occurs from northern Pahroc Valley. Elsewhere among these basins, interbasin flow is limited by geologic structure and lithology. The following sections describe the interbasin flow among these areas.

D.2.2.1 Cave Valley

Groundwater outflow from Cave Valley is thought to occur on the western side of Cave Valley through Shingle Pass to the southern third of White River Valley, and through the southern portion of the valley to northern Pahroc Valley (Figure D.2-2). Total outflow from Cave valley, which for this analysis is estimated to be about 13,400 afy, is estimated as the difference between natural recharge

Appendix D

SE ROA 11866

D-7



Southern Nevada Water Authority - Water Resources Division



Note: For an index of springs names and well IDs, see Table D.2-1.

Figure D.2-2 Interbasin Flows and Water-Level Elevations of Selected Wells and Springs in Cave Valley and Vicinity

D-8
and groundwater ET. The following sections describe the assumptions, data, and method by which this volume was partitioned to the interbasin flow locations.

D.2.2.1.1 Outflow to White River Valley through Shingle Pass

Outflow from Cave Valley to the southern third of White River Valley occurs through fractured carbonate rock associated with the Shingle Pass Fault. The volume of outflow was approximated by equating it to the downgradient spring discharge minus the recharge from the contributing watersheds in White River Valley. The springs selected to derive this downgradient spring discharge volume are listed in Table D.2-2 and are depicted in Figure D.2-3. These springs were selected because they are cold-water springs (i.e., local springs) and have similar isotopic composition to springs in the Egan Range and Cave Valley. These are indications that the sources of their respective discharges are derived from local recharge. The total discharge from these springs is about 7,330 afy.

	Loc	ation	1.1.1.1.1.1	1.120.201	Mean	-
Spring Name	UTM Easting ^a (m)	UTM Northing ^a (m)	Elevation (ft-amsl)	No. of Measurements ^b	Annual Discharge (cfs) ^b	۵D (‰) ^c
Flag Springs 3	672,579	4,254,416	5,294	39	2.16	-105
Flag Springs 2	672,576	4,254,570	5,285	34	2.94	NA
Flag Springs 1	672,719	4,254,697	5,294	35	2.29	NA
Butterfield Spring	673,530	4,256,472	5,324	39	2.73	-105
Shingle Spring	679,925	4,267,716	6,434	2	0.003	-104
oningio opinig		1		Total	10.12 (7,327 afy)	

Table D.2-2 Discharge Data for Selected Springs in White River Valley

^aNorth American Datum of 1983, Zone 11

^bData from Table C.1-1

°USGS, 2004

Based on the recharge analysis presented in Section 5.0, a recharge grid was derived and used as the basis to calculate the total recharge of the contributing watersheds in White River Valley. The recharge was calculated to be about 3,240 afy for the watersheds depicted in Figure D.2-3. It is assumed that the spatial distribution of recharge derived as part of the recharge analysis is sufficiently accurate for the purpose for which it is used here. However, it is acknowledged that the actual value for this area could be more or less, depending on the accuracy of the precipitation-recharge relationship defined in Section 5.0. The outflow from Cave Valley to White River Valley through Shingle Pass, calculated as the difference of the spring discharge and the recharge from the contributing watersheds, is estimated to be about 4,000 afy.

To check the validity of this approximation, the transmissivity value of the rocks comprising the presumed flow path was calculated by re-arranging Equation D.2-1 in the form:

Appendix D

D-9

SE ROA 11868





Figure D.2-3 Watershed for Selected Springs in White River Valley

D-10	
the second se	

Appendix D SE ROA 11869

$$T = \frac{Q}{IW} \times \frac{43,560}{365}$$
(Eq. D.2-2)

where,

Q =the groundwater flow (afy)

 $T = the transmissivity (ft^2/day)$

I = the hydraulic gradient (ft/ft)

W = the estimated width of the section of flow (ft)

The hydraulic gradient was calculated using the following data: (1) the water-level elevation (5,406 ft-amsl) of Monitor Well 180W501M, recently drilled by SNWA and completed in carbonate bedrock east of Shingle Pass, (2) the elevation of Butterfield Spring (5,324 ft-amsl), and (3) the distance between the two locations of about 14 mi or about 74,000 ft (Figure D.2-3). The hydraulic gradient between these two locations was calculated to be 0.00111. The estimated length of the cross section of flow measured along the west entrance to Shingle Pass in White River Valley (Figure D.2-3) was estimated to be about one mile or 5,280 ft. Using these values, a transmissivity of 81,451 ft²/day was calculated. This transmissivity value falls within the range of reported values. Dettinger et al. (1995) had reported a range of 10 to 250,000 ft²/d for the carbonate wells of the region. Larger transmissivity values, as high as 410,000 ft²/d (SRK Consulting, Inc., 2001, tbl 4), have been reported since, making the range wider. The calculated transmissivity calculated for the Shingle Pass is, however, on the high side of the range, which suggests that the estimated interbasin flow of 4,000 afy may also be on the high side.

D.2.2.1.2 Outflow to Northern Pahroc Valley

Outflow from southern Cave Valley to the northeastern portion of Pahroc Valley occurs through fractured carbonate rock and along fault zones associated with the west range-front fault of the southern Schell Creek Range. With about 4,000 afy of the total 13,400 afy of Cave Valley outflow occurring through Shingle Pass, the outflow from southern Cave Valley is estimated to be the difference, or 9,400 afy.

D.2.3 Dry Lake and Delamar Valleys

For Dry Lake Valley, interbasin flow occurs in the northern portion of the valley as inflow from northern Pahroc Valley and as outflow to Delamar Valley to the south (Figure D.2-4). Inflow from Pahroc Valley is estimated to be no more than 2,000 afy, while the outflow to Delamar Valley is estimated as the total of the inflow and natural recharge, or about 17,700 afy.

For Delamar Valley, interbasin flow occurs as inflow from Dry Lake Valley and as outflow to northern Coyote Spring Valley and possibly the very southern part of southern Pahranagat Valley (Figure D.2-4). The outflow is estimated as the sum of the inflow from Dry Lake Valley and the natural recharge derived from within the hydrographic area boundaries of both basins, or about 24,100 afy. Interbasin flow elsewhere among these basins is precluded by geologic structure and lithology, principally the Caliente and Kane Springs Wash caldera complexes to the east and

SE ROA 11870





Note: For an index of springs names and well IDs, see Table D.2-1.

Figure D.2-4

Interbasin Flows and Water-Level Elevations of Selected Wells and Regional Springs for Dry Lake and Delamar Valleys and Vicinity

D-12 Appendix D SE ROA 11871 southeast and the North and South Pahroc ranges and associated range-front faults. The following sections describe the interbasin flow approximations in greater detail.

D.2.3.1 Inflow from Pahroc Valley

The approximate location of inflow from Pahroc Valley to Dry Lake Valley is depicted in Figure D.2-5 and is coincident with a series of northwest-trending, right-lateral faults that form the boundary between southern Cave Valley, northern Pahroc Valley, and northern Dry Lake Valley. These faults extend from southeastern White River Valley through northeastern Pahroc Valley and into Dry Lake Valley, where they transition into the east range-front fault of the North Pahroc Range.

Water-level data are sparse in this area and are limited to shallow basin-fill wells in northern Pahroc Valley and the North Dry Lake MX Well (Table D.2-1). Based on water-level elevation data for the northern Pahroc Valley alluvial wells (about 4,800 ft-amsl) and the North Dry Lake MX well (about 4,540 ft-amsl) (Ertec Western, Inc., 1981), there is a difference in water-level elevation of about 200 to 300 ft between the two areas and groundwater systems. No carbonate-rock well data exists for this area of Pahroc Valley; however, nearby Well 207 N6E61 28DA1 in southeastern White River Valley (Map ID 207-20) is completed in the carbonate-rock aquifer and has a reported water-level elevation of 5,142 ft-amsl based on shut-in pressures from drill-stem testing (Thomas, 1986). This water-level elevation and that of the North Dry Lake MX well are indicative of a gradient in the carbonate-rock aquifer from southeastern White River Valley to northern Pahroc and Dry Lake valleys. However, groundwater flow is complicated by the local right-lateral fault structures, and given the lack of data, it is difficult to estimate how much groundwater might flow into northern Dry Lake Valley. Because there is a significant hydraulic gradient and accommodating structures, it is assumed that about 2,000 afy of groundwater flows from Pahroc Valley to northern Dry Lake Valley (Figure D.2-5).

D.2.3.2 Outflow from Dry Lake Valley to Delamar Valley

Outflow from Dry Lake Valley to northern Delamar Valley occurs through fractured carbonate rocks and valley-fill sediments. These two hydrographic areas are topographically separated by a low alluvial divide but are connected geologically and hydrologically, making them one contiguous basin. Based on water-level elevations from wells located within Dry Lake and Delamar Valleys, there is a north-south gradient, thus indicating outflow from Dry Lake Valley to Delamar Valley. Because there is no significant groundwater ET in Dry Lake Valley, the outflow is calculated as the sum of the inflow from Pahroc Valley and the locally derived natural recharge. This value is estimated to be about 17,700 afy. As in Dry Lake Valley, no significant groundwater ET occurs in Delamar Valley, and the outflow is calculated as the sum of the inflow from Dry Lake Valley, 17,700 afy, and the locally-derived natural recharge, about 6,400 afy, for a total of about 24,100 afy.

D.2.3.3 Outflow from Delamar Valley

Groundwater outflow from Delamar Valley is controlled by the Kane Springs Wash caldera complex underlying the Delamar Mountains to the southeast, the South Pahroc Range and western Delamar Mountains and associated range-front faults, and the southwest trending Pahranagat Shear Zone (Figure D.2-6). Regional flow through the caldera complex is unlikely, as the majority of flow occurs





Note: For an index of springs names and well IDs, see Table D.2-1.

D-14

Figure D.2-5 Inflow from Pahroc Valley to Northern Dry Lake Valley

Appendix D

SE ROA 11873



Figure D.2-6 Delamar and Pahranagat Valleys and the Pahranagat Shear Zone

Appendix D

SE ROA 11874



along the range-front faults and the northeast extension of the Pahranagat Shear Zone to areas of lower potential in northern Coyote Spring Valley and possibly the very southern part of Pahranagat Valley. The range-front faults transition into right-lateral shear zones as they pass through southern Pahranagat Valley and become normal faults on the eastern side of the Sheep Range. Water-level elevation data from the few wells and springs in Delamar Valley and the adjacent areas of southern Pahranagat and northern Coyote Spring valleys indicate a hydraulic gradient to the southwest and south along and across the Pahranagat Shear Zone and into northern Coyote Spring Valley.

SNWA monitor wells 182M-1 and 182W906M, recently drilled to depths of 1,345 and 1,735 ft-bgs, respectively, in southern Delamar Valley, have water-level elevations of 3,755 and 3,486 ft-amsl, respectively (Figure D.2-6). Both wells are completed in volcanic rocks (Hiko Tuff) comprising the valley fill. Monitor Well 182M-1 is located north of the Pahranagat Shear Zone on the southwest side of the valley, while Monitor Well 182W906M is located near the west range-front fault of the Delamar Mountains and within the trace of the Pahranagat Shear Zone. Water levels in both wells are most likely representative of the local groundwater system rather than the regional potential of the carbonate-rock aquifer. A comparison of their water-level elevations with elevations of the springs in southern Pahranagat Valley and wells in Coyote Spring Valley completed in the carbonate-rock aquifer, reveals a significant hydraulic potential between these areas. Spring elevations in southern Pahranagat Valley range in elevation from about 3,240 ft-amsl at Solar Panel Spring to about 3,133 ft-amsl at Maynard Spring. This represents a hydraulic potential of at least 350 ft from southern Delamar Valley to southern Pahranagat Valley.

SNWA's monitor well CSVM-3 in northern Coyote Spring Valley is completed to 1,230 ft-bgs in the carbonate-rock aquifer and has a water-level elevation of 2,207 ft-amsl (Figure D.2-6). The nearest location reflecting the potentiometric level the carbonate-rock aquifer is the elevation of Ash Springs, in central Pahranagat Valley, which emerges from carbonate rocks (Sevy Dolomite) coincident with the north-trending range front. Ash Springs is about 30 mi from Monitor Well CSVM-3 at an elevation of 3,622 ft-amsl. This represents a potential in the carbonate-rock aquifer of about 1,400 ft from central Pahranagat Valley to northern Coyote Spring Valley. When compared to the SNWA monitor wells in Delamar Valley, the hydraulic potential is even greater, ranging from about 1,550 to 1,280 ft.

Based on the conceptualization of flow and the supporting hydraulic head data, groundwater flow is to the south and southwest from Delamar Valley along the strike of the Pahranagat Shear Zone. The large hydraulic head differences between Delamar and Pahranagat valleys and Coyote Spring Valley suggest that there is potential for significant flow from these areas and across the shear zone to Coyote Spring Valley. The flow is controlled principally by the shear zone, for which hydraulic properties are unavailable making an estimate of the flow using Darcy's Law unrealistic. However, a transmissivity value representing the hydraulic properties across the shear zone was calculated using Equation D.2-2 and the estimated outflow to Coyote Spring Valley derived in Section 5.0. This value was compared to the range of transmissivity values compiled for the carbonate-rock aquifer to determine where within the range the value falls. Values used in the calculation are as follows: (1) 46,500 afy of total outflow from Pahranagat and Delamar valleys to Coyote Spring Valley, (2) a hydraulic gradient of 0.00893 ft/ft based on the water-level elevations for Ash Springs and Monitor Well CSVM-3, and (3) a cross section of flow of about 125,000 ft (23.7 mi), equal to the length of the shear zone from the northeast Sheep Range to the west range-front fault of the Delamar Mountains.

The calculated transmissivity value is $4,972 \text{ ft}^2/\text{day}$, or about $5,000 \text{ ft}^2/\text{day}$. As expected, this value falls at the low end of the range (10 to $419,000 \text{ ft}^2/\text{day}$ [Dettinger, 1995, p 15-17; SRK Consulting Inc., 2001, tbl. 4]), and represents a reasonable average transmissivity for flow across the shear zone. It is likely that transmissivity is locally substantially higher or lower given the geologic variability within the shear zone.

D.3.0 WRFS BOUNDARY FLOW

The majority of the WRFS boundary inflow and outflow occurs in the southern portion of the flow system as inflow from Lower Meadow Valley Wash and as outflow to the Colorado River. Minor amounts of outflow from southwestern Long Valley to Newark Valley (Harrill et al., 1988; Thomas et al., 1986) and from Garden Valley to Penoyer Valley might occur based on hydraulic head potential (San Juan et al., 2004; Chapter C, p. 119). Elsewhere, boundary flow is limited by geologic structure and lithology. The boundary flow in the southern portion of the flow system is discussed in the following sections.

D.3.1 Inflow from Lower Meadow Valley Wash

Groundwater inflow from Lower Meadow Valley Wash occurs through the underlying valley-fill sediments and carbonate rocks. The magnitude of this flow has been estimated by Rush (1968, p. 24), Kirk and Campana (1990, p. 372; Thomas et al., 1996, p. C36), Prudic et al. (1995, p. D71), and Thomas et al. (1996, p. C36) at 7,000 afy, 5,500 to 9,000 afy, 13,000 afy, and 8,000 afy respectively. For this analysis, it was assumed that about 9,200 afy of groundwater inflow from Lower Meadow Valley Wash enters California Wash and the Muddy River Springs Area. This value represents the average of the estimates.

D.3.1.1 Inflow to California Wash

Gage records and miscellaneous discharge measurements for the Muddy River support groundwater inflow from Lower Meadow Valley Wash. Two synoptic discharge measurement studies indicate an apparent 2 to 2.5 cfs increase in streamflow along the river reach extending from the White Narrows to Jackman Narrows (Figure D.3-1). Rush (1968, p. 13 and 15) reports measurements from a February 5, 1968, study in which the gage flow above White Narrows was 46.6 cfs, while a measurement made minutes later above Jackman Narrows near Glendale was 48.3 cfs. On February 6, 1968, measurements were made at three sites: near Glendale, at Jackman Narrows, and at a site about 1 mi below Jackman Narrows. Flow rates of 48, 54, and 47.8 cfs, respectively, were recorded at these sites. Based on this study, the net gain from above White Narrows to Jackman Narrows is about 7.4 cfs, or 5,350 afy. Rush (1968) suggests that the observed increase in flow is likely due to inflow from alluvial sediments and/or underlying consolidated rocks, namely the carbonate-rock aquifer.

A second and more recent study corroborates the findings of Rush (1968). Beck and Wilson (2006) describe the results of a synoptic discharge study for the same river reaches described by Rush (1968). This study was conducted during February 7, 2001 and involved many of the same





Figure D.3-1 Two Sets of Synoptic Discharge Measurements along Muddy River

Appendix D

measurement sections. For essentially the same reach (from White Narrows to Jackman Narrows near Glendale), the river gained about 2.5 cfs based on the maximum values of the gage record for the Muddy River near Moapa gage (09416000) and the measurement section at Muddy River near Lewis Ranch near Glendale (09418900). Maximum flow rates at these sites were measured at 37.1 cfs and 39.6 cfs, respectively (Beck and Wilson, 2006 p. 9, Tbl. 3). From station 09418900 to the Muddy River below Anderson Wash near Logandale gage (09419490), an apparent increase in flow of 3.7 cfs was observed. In total, from the Muddy River near Moapa gage to the Muddy River below Anderson Wash near Logandale gage, an apparent increase in flow of 6.2 cfs was observed, or about 4,500 afy.

Based on these studies and because of the location of the gaining reaches, the increases in flow observed in the river are, in part, a result of groundwater outflow from Lower Meadow Valley Wash to California Wash. For this study, this flow is estimated to be about 5,200 afy.

D.3.1.2 Inflow to Muddy River Springs Area

The remaining outflow from Lower Meadow Valley Wash of about 4,000 afy flows to the Muddy River Springs Area through fractured carbonate rocks and accommodating structures associated with the east range-front fault of the Meadow Valley Mountains. This water is mostly consumed by local ET in the Muddy River Springs Area. Total groundwater discharge (spring discharge, groundwater ET, and underflow) from this area during predevelopment conditions is estimated to be at least 40,000 afy based on (1) the average annual flow of the Muddy River (adjusted for precipitation runoff events) of 33,700 afy at the Muddy River near Moapa gage (09416000) (Eakin, 1964, p. 24), and (2) an estimate of predevelopment groundwater ET of 6,000 afy.

D.3.2 Outflow to Colorado River

Outflow from the WRFS to the Colorado River occurs as spring discharge from the Muddy Springs, Rogers and Blue Point springs, and subsurface outflow from Lower Moapa Valley. The outflow to the Colorado River from Lower Moapa Valley has been estimated to be between 11,100 afy (Rush, 1968, p. 24 tbl. 5, p. 26 tbl 7) and 49,000 afy (LVVWD, 2001). The estimate of Rush (1968) includes an estimate of Muddy River streamflow that represents the total flow, including some unquantified contribution of surface runoff from precipitation events. Estimates for this study are based on gage records of Muddy River discharge, including miscellaneous discharge measurements and the Muddy River near St. Thomas, Nevada, gage, and an estimate of subsurface outflow from Lower Moapa Valley. For this analysis, it is estimated that groundwater outflow from the WRFS to the Colorado River, including the groundwater component of the measured flows at the Muddy River near St. Thomas gage, is less than 25,000 afy.

D.3.2.1 Black Mountains Area/Rogers and Blue Point Springs

The Black Mountains Area is bounded by the Colorado River (Lake Mead) in the east, the Black Mountains in the southwest, and the Muddy Mountains in the northwest. The most dominant features are the Muddy Mountains, Bitter Spring Valley, and the large washes of Gypsum, Callville, Echo, and Valley of Fire. Rogers and Blue Point springs are located in the eastern portion of the hydrographic area and discharge to the Colorado River (Figure D.3-2). From water year (WY) 1986 to WY2006,





Figure D.3-2 Measurement Sites along the Muddy River in Lower Moapa Valley

the mean annual discharge of Rogers Spring was 1.66 cfs (USGS, 2006). The mean annual discharge reported for Blue Point Springs for the period of WY 2000 to WY 2006 was 0.55 cfs. The combined average annual discharge for the two springs is about 2.21 cfs, or about 1,600 afy. An additional 400 afy of groundwater outflow to the Colorado River from the Black Mountains Area is assumed to account for any underflow or additional spring discharge in the area. Therefore, the total groundwater discharge from the Black Mountains Area is estimated to be about 2,000 afy.

D.3.2.2 Outflow from Lower Moapa Valley

Muddy River at St. Thomas, Nevada (1913 to 1916)

The Muddy River near St. Thomas gaging station was located just upstream from the confluence of the Muddy and Virgin Rivers. Because of its early record and location with respect to the Colorado River, the gage records for this station were most representative of predevelopment conditions of flow from the Muddy River to the Colorado River. The flow at this location during the period of record had flow contributions from (1) surface-water runoff related to precipitation events, (2) groundwater discharge from the Muddy River Springs Area, and (3) inflow from California Wash.

The gage records began in June of WY 1913 and ended in September of WY 1916, during which time there was a 7-month period of missing records (June of WY 1915 to December of WY 1916) (Wells, 1954, p. 570). The mean-annual flow for the only complete year (WY 1914) was 19.3 cfs, or about 14,000 afy (Wells, 1954, p. 570). The period of record mean annual flow was calculated to be 19.6 cfs based on the mean monthly values; however, this value reflects large flood events during February of WY 1914 (136 cfs). If this value is excluded, the period of record mean annual discharge is 17.5 cfs, or almost 13,000 afy. Rush (1968, p. 24, tbl. 5) estimated this flow to be 10,000 afy, but qualified the estimate as a rough approximation based on few data gathered in 1967. This flow most likely represented agricultural return flows.

At the confluence with the Virgin River, discharge from the Muddy River was observed during this period; however, it is impossible to determine the magnitude of the groundwater component given the limitations of the gage records. For this analysis, it is estimated that about half of the flow was groundwater discharge, or 7,000 afy.

Subsurface Outflow from Lower Moapa Valley to the Colorado River

Groundwater outflow via the subsurface from Lower Moapa Valley to the Colorado River has the greatest uncertainty of the outflow components due to the lack of hydraulic information related to the basin-fill aquifer(s) underlying the valley floor. Rush (1968) estimated an outflow of 1,100 afy from the alluvial basin but did not report an estimate for the consolidated rocks. Specific capacity data for wells within the valley indicate that the underlying materials are transmissive. Moapa Valley Water District (MVWD) constructed a production well (Well No. 1) completed in bedrock (sandy limestone) to 154 ft-bgs near Logandale, Nevada. This well had a specific capacity of 105 gpm/ft (Rush, 1968, Table 19). A second MVWD well (Well No. 2) was constructed and completed in bedrock (porous limestone) to a depth of 154 ft-bgs and had a specific capacity of 24 gpm/ft (Figure D.3-2). An estimate of groundwater outflow to the Colorado River was derived using Equation D.2-1.

Appendix D

SE ROA 11880



The transmissivity value was approximated by multiplying the specific capacity by 2,000 (Driscoll, 1986). Using the average for the two MVWD wells, a transmissivity value of 129,000 gpd/ft, or 17,000 ft²/day was calculated. This value falls within the reported range of 0.02 to 64,600 ft²/day for the basin-fill aquifer in the region (Belcher et al., 2001, Appendix A).

An average hydraulic gradient between Lower Moapa Valley and the Colorado River was estimated using the available water-level data. Well data selected for use in this calculation are presented in Table D.3-1.

		Land Surface Elevation (ft-amsl)	Depth-to-V	rements	Mean Head	
Site ID	Well Name		Date	Depth (ft-bgs)	Mean Depth (ft-bgs)	Elevation (ft-amsl)
	A STREET STREET STREET	1.02302	3/14/1985	21.20	21.6	1 480 6
363832114323801	220 S15 E67 07CBD 1	1,502.2	3/28/1990	21.90	21.0	1,480.6
363715114292901	220 S15 E67 22BBCB1 MVWD	1,412.1	4/10/1967	22.00	22.0	1,390.1
		1 057 4	12/11/1966	6.00	72	1.349.9
363531114284801	220 S15 E67 34AAB 1	1,357.1	3/14/1985	8.36	1.12	0.5 1.515
363419114271901			11/09/1949	9.00		
	220 S16 E67 01CBBB1	1244	5/11/1950	7.82	80	1,304.1
		1,312.1	11/10/1967	5.70	- 8.0	
			3/14/1985	9,26		
1	Contract Contractor		1/01/1950	9.00	00	1 272.2
363315114260501	220 S16 E68 07CBC 1	1,282.1	3/14/1985	10.82	9.9	1,272.2
100 C	Constra de Californi	1.7.5.1	2/08/1948	23.00	21.4	1 230 7
363102114254801	220 S16 E68 30BDAB1	1,252.1	3/14/1985	19.88	21.4	1,200.1
St. Thomas Well - Shallow	St. Thomas Well - Shallow	1,174.0	1915	30.00	30.0	1,144.0
St. Thomas Well - Deep	St. Thomas Well - Deep	1,174.0	1915	284.00	284.0	890.0
St. Thomas Well - Average	St. Thomas Well - Average	1,174.0	1915	157.00	157.0	1,017.0

Table D.3-1 Selected Wells Located in Lower Moapa Valley and Black Mountains Area near the Muddy River

Sources: Carpenter (1915, p. 63); NDWR Driller's Logs; USGS NWIS Database

Water-level data available for the Lower Moapa Valley and the Black Mountains Area near the confluence of the Muddy River with the Colorado River were compiled and evaluated. Data for 49 wells are available for Lower Moapa Valley but only 6 of the 49 wells have more than one depth-to-water measurement. The six wells with at least two measurements were extracted from the

dataset and sorted from north to south along the main direction of flow. Two of the six wells had exactly the same coordinates and virtually the same water level, so one (220 S16E6830BDAB1) of these two wells was eliminated from the hydraulic gradient computation. The well used to estimate the transmissivity (MVWD Well No. 1) was added to the short list of selected wells in Lower Moapa Valley.

A well near St. Thomas was drilled by the San Pedro, Los Angeles and Salt Lake Railroad to a total depth of 805 ft (Carpenter, 1915, p. 63). Two measurements were reported this well. During drilling, the first water was encountered at 30 ft-bgs. At this point, the hole was cased off and the well was drilled to 805 ft-bgs. At the time, the depth to water was 284 ft bgs. The mean depth-to-water was used to represent this well in the analysis.

The wells were ordered from north to south in the main direction of flow, and hydraulic gradients were calculated for each pair of wells starting from the northernmost well. The calculations are shown in Table D.3-2. An average hydraulic gradient was calculated using wells located in the lowlands of Lower Moapa Valley along the main direction of flow. The average water level for the St. Thomas well was used in the analysis. The resulting average hydraulic gradient in the valley-fill aquifer along the main axis of Lower Moapa Valley is about 0.00432.

The section of flow was identified on a hydrolgeologic map and projected on a line perpendicular to the direction of flow. The width of the projected section of flow was estimated to be 26,500 ft, and is approximately depicted on Figure D.3-2. The volume of subsurface outflow to the Colorado River was estimated using a transmissivity of 17,000 ft²/day, a hydraulic gradient 0.00432, and a flow section width of about 26,500 ft. The estimated volume of outflow is about 16,000 afy.

Site ID	Well Name	Distance between Wells (ft)	Mean Depth to Water (ft)	Mean Head Elevation (ft-amsl)	Head Difference (ft)	Hydraulic Gradient (ft/ft)
363832114323801	220 S15 E67 07CBD 1	- Her	21.55	1,480.6		99 99
363715114292901	220 S15 E67 22BBCB1 Moapa Valley Water District	17,266	22.00	1,390.1	90.473	0.00524
262531114284801	220 S15 E67 34AAB 1	11,039	7.18	1,349.9	40.190	0.00364
363531114204001	220 S16 E67 01CBBB1	10.284	7.95	1,304.2	45.778	0.00445
363419114271901	220 310 207 0108051	0.052	0.01	1 272 2	31,956	0.00361
363315114260501	220 S16 E68 07CBC 1	0,000	5.51	1,27,20		0.00007
363102114254801	220 S16 E68 30BDAB1	13,525	21.44	1,230.7	41.539	0.00307
St. Thomas Well - Average	St. Thomas Well - Average	36,032	157.00	1,017.0	213.67	0,00593

	Table D.3-2	nti ta UrM
Hydraulic Gradient Calculations	in Lower Moapa Valley and Black Mou	ntains Area

Southern Nevada Water Authority - Water Resources Division

D.4.0 REFERENCES

- Beck, D.A., and Wilson, J.W., 2006, Synoptic discharge, water-property, and pH measurements for Muddy River Springs Area and Muddy River, Nevada, February 7, 2001: U.S. Geological Survey Scientific Investigations Report 2006-5237, 12 p.
- Belcher, W.R., Elliott, P.E., and Geldon, A.L., 2001, Hydraulic-property estimates for use with a transient ground-water flow model of the Death Valley Regional Ground-water Flow System, Nevada and California (revised August 1, 2002): U.S. Geological Survey Water-Resources Investigations Report 2001-4210, 31 p.
- Berger, D.L., Kilroy, K.C., and Schaefer, D.H., 1988, Geophysical logs and hydrologic data for eight wells in the Coyote Spring Valley area, Clark and Lincoln counties, Nevada: U.S. Geological Survey Open-File Report 87-679, 59 p.
- Bunch, R.L., Harrill, J.R., 1984, Completion of Selected Hydrologic Data from the MX Missile-Siting Investigation, East-Central Nevada and Western Utah: U.S. Geological Survey Open-File Report 84-702, 128 p.
- Carpenter, E., 1915, Ground water in southeastern Nevada: U.S. Geological Survey Water-Supply Paper 365, 86 p.
- Dettinger, M.D., Harrill, J.R., Schmidt, D.L., and Hess, J.W., 1995, Distribution of carbonate-rock aquifers and the potential for their development, southern Nevada and adjacent parts of California, Arizona, and Utah: U.S. Geological Survey Water-Resources Investigations Report 91-4146, 100 p.
- Driscoll, F.G., 1986, Groundwater and wells, second edition: Johnson Filtration Systems, Inc., St. Paul, Minnesota.
- Eakin, T.E., 1964, Ground-water appraisal of Coyote Spring and Kane Spring Valleys and Muddy River Springs Area, Lincoln and Clark counties, Nevada: Nevada Department of Conservation and Natural Resources, Ground-Water Resources Reconnaissance Series Report 25, 39 p.
- Eakin, T.E., 1966, A regional interbasin groundwater system in the White River Area, southeastern Nevada: Water Resources Research, Vol. 2, No. 2, p. 251–271.
- Ertec Western, Inc., 1981, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume II, Report E-TR-52-II: U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 286 p.

- Harrill, J.R., Gates, J.S., and Thomas, J.M., 1988, Major ground-water flow systems in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Investigations Atlas HA-694-C, 4 p., scale 1:1,000,000, two sheets.
- IT Corporation, 1996, Underground test area subproject, Phase I Data Analysis Task—Volume IV, hydrologic parameter data documentation package: IT Corporation, Las Vegas, Nevada, Report ITLV/10972-181, 78 p.
- Johnson, M.J., 2002, Drilling and development of Well WS-2 for Duke Energy Moapa, LLC, Garnet Valley, Clark County, Nevada: Duke Energy Moapa, LLC, Las Vegas, Nevada, 127 p.
- Johnson, M.E., 2005, Drilling and Development of CSI Well No. 1 for Coyote Springs Investment LLC, Coyote Spring Valley, Clark County, Nevada, 49 p.
- Kirk, S.T., and Campana, M.E., 1990, A deuterium-calibrated groundwater flow model of a regional carbonate-alluvial system: Journal of Hydrology, Vol. 119, p. 357–388.
- Las Vegas Valley Water District, 2001, Water resources and ground-water modeling in the White River and Meadow Valley flow system, Clark, Lincoln, Nye, and White Pine counties, Nevada, 285 p.
- Luzier, J.E., 2003, Dry Lake and Delamar Valleys, Nevada Groundwater Hydrology and Development Potential: Luzier Hydrosciences, 32 p.
- McKay, W.A. and Kepper, J., 1988, Estimating Hydraulic Parameters Using Wildcat Oil and Gas Data: A Feasibility Study in East-Central Nevada: Water Resources Center, Desert Research Institute, Publication No. 41117, 48 p.
- Mifflin & Associates, Inc., 2001, Hydrologic and groundwater modeling analyses for the Moapa Paiute Energy Center: Mifflin & Associates, Inc., Las Vegas, Nevada, 219 p.
- Nevada Division of Water Resources, 2004, Well log database available on March, 2004 at website http://water.nv.gov/Engineering/wlog/wlog.cfm.
- Prudic, D.E., Harrill, J.R., and Burbey, T.J., 1995, Conceptual evaluation of regional ground-water flow in the carbonate-rock province of the Great Basin, Nevada, Utah, and adjacent states— Regional aquifer-system analysis–Great Basin, Nevada-Utah: U.S. Geological Survey Professional Paper 1409-D, 102 p.
- Rush, F.E., 1968, Water-resources appraisal of the Lower Moapa-Lake Mead area, Clark County, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 50, 67 p.
- San Juan, C.A., Belcher, W.R., Laczniak, R.J. and Putnam, H.M., 2004, Hydrologic components for model development, *in* Belcher, W.R., ed., Death Valley Regional Ground-Water Flow System,

Appendix D

SE ROA 11884



Nevada and California—Hydrogeologic framework and transient ground-water flow model: U.S. Geological Survey Scientific Investigations Report 2004-5205, p. 99–136.

Schmidt, D.L. and Dixon, G.L., 1995, Geology and aquifer system of the Coyote Spring Valley area, southeastern Nevada: U.S. Geological Survey Open-File Report 95-579, 47 p.

SNWA, see Southern Nevada Water Authority.

- Southern Nevada Water Authority, 2005, Analysis of potential water-related effects of Southern Nevada Water Authority's proposed groundwater development project in Three Lakes Valley South, Clark County, Nevada, 118 p.
- SRK Consulting, Inc., 2001, Installation report-Harvey Well replacement, Well RW-1, Nevada Power Company, Las Vegas, Nevada, 84 p.
- Thomas, J.M., Welch., A.H., and Dettinger, M.D., 1996, Geochemistry and isotope hydrology of representative aquifers in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Professional Paper 1409-C, 99 p.
- Thomas, J.M., Mason, J.L., U.S. Geological Survey, and Crabtree, J.D., Ertec Western, Inc., 1986. Ground-Water Levels in the Great Basin Region of Nevada, Utah, and Adjacent States, 2 p.
- Tumbusch, M.L., and Schaefer, D.H., 1996, Selected hydrologic data for and location of MX wells in east-central and southern Nevada, January 1980 through May 1996: U.S. Geological Survey Open-File Report 96-469, 37 p.
- U.S. Geological Survey, 2001, Digital elevation model (DEM), Earth Resources Observation and Science (EROS), U.S. Geological Survey, accessed at http://edc.usgs.gov/guides/dem.html.
- U.S. Geological Survey, 2004, National Water Information System (NWIS Web) data accessed at website http://waterdata.usgs.gov/nwis/.
- U.S. Geological Survey, 2006, Water Data Report 2006, 09419550 Rogers Spring Near Overton Beach, NV. 1.66 cfs, 3 p.
- U.S. Geological Survey, 2007, National Water Information System (NWIS Web) data accessed at website http://waterdata.usgs.gov/nwis/.
- Wells, J.V.B., 1954, Compilation of records of surface waters of the United States through September 1950-Part 9. Colorado River Basin: U.S. Geological Survey Water-Supply Paper 1313, 749 p.

Appendix D SE ROA 11885

Appendix E

Description of Excel Analysis Files (CD-ROM Contents)

SE ROA 11886

JA_4648

E.1.0 INTRODUCTION

The purpose of this appendix is to describe in detail the calculations conducted in support of the data analyses related to the water resource assessment of the WRFS. The objectives of the calculations were to: (1) derive a spatial distribution of precipitation recharge for the WRFS using the water-balance method, (2) derive basin groundwater budgets, and (3) evaluate the solution through uncertainty analysis. The descriptions are organized based on the Excel files containing the calculations and provided with the report. These files consist of one file containing the solution and several files containing the uncertainty analyses. For hard copies of the report, the Excel files are included on a CD enclosed with the document. In the electronic version of the document, the files are in a companion zip file named "SNWA-WRFS-Excel-Analysis-Files.zip."

The files are provided for the sole purpose of documentation of the analyses. In essence, the solution file contains a calibrated groundwater budget model for the WRFS. Any changes to the target value, the initial estimates of the parameters, or the constraints may immediately change the water budgets, and may yield a different solution or no solution at all if the solver is executed. Similarly, the uncertainty analysis files contain solutions to scenarios where a specific variable was selected and carefully altered to test its effect on the solution. Consequently, the reader should not alter the contents of the files unless they thoroughly understand their setup and have good knowledge about the WRFS. The uncertainty files constitute good examples of how the solution file may be altered and still yield a meaningful solution, yet perhaps unrealistic in some cases.

E.2.0 SOLUTION FILE

The Excel file containing the recharge solution and associated groundwater budget is named: "SNWA-WRFS-Groundwater-Budget-Solution." A description of its contents and the solution process are provided in the following text.

E.2.1 File Contents

The solution file includes four worksheets named as follows:

- Worksheet 1: "1-Explanation"
- Worksheet 2: "2-Groundwater-ET-Estimates"
- Worksheet 3: "3-Precipitation-Recharge-Details"
- Worksheet 4: "4-Solution"

Descriptions of each of the four worksheets follows.





E.2.1.1 Worksheet 1 - "1-Explanation"

The "explanation" worksheet contains a brief description of the file and its contents.

E.2.1.2 Worksheet 2 - "2-Groundwater-ET-Estimates"

This worksheet contains a table listing the groundwater ET volumes for each of the HAs of the WRFS, as estimated by this study. The sum of the groundwater ET volumes for all HAs in the flow system is used as the target in the solver window that can be activated from the solution worksheet (see solution process in Section E.2.2). The table contains the following fields:

- HA: Hydrographic Area Number used by Nevada State Engineer HA is surrogate for groundwater basin in this report.
- SNWA GW ET: Groundwater ET estimate for HA in acre-feet per year derived as part of this study and forms the main basis for the solution. Details are provided in Section 4.0 of this report.
- SNWA/BARCASS GW ET: Groundwater ET estimate for HA in acre-feet per year. Consists
 of BARCASS (Welch and Bright, 2007) estimates for the northern basins (Long, Jakes, White
 River, and Cave valleys), and SNWA estimates (SNWA GW ET) for the southern basins
 derived as part of this study.
- Recon GW ET*: Groundwater ET estimate reported for HA in Reconnaissance reports. Star (*) refers to footnote which provides sources of estimates. Estimates are provided here for comparison purposes only.
- LVVWD (2001): Groundwater ET estimate reported for HA in report published by LVVWD in 2001 in support of SNWA's Water-Right Hearing for Spring Valley, Nevada. Estimates are provided here for comparison purposes only.

E.2.1.3 Worksheet 3 - "3-Precipitation-Recharge-Details"

This worksheet fulfills two roles: (1) it contains the precipitation data which serves as input data to the solution, and (2) contains the recharge calculations performed by the solver during the solution process. All calculations are performed by built-in formulas using the solution derived in worksheet 4. This worksheet should, therefore, not be altered by the user. A description of the table contents is provided, followed by an explanation of the process used to generate them.

E.2.1.3.1 Table Contents

Each row in the table contains information for the mid-interval precipitation value of 1-in. precipitation bands or zones, sorted by HA. The table contains the following information:

HA Name: This is the name of the hydrographic area used by the Nevada State Engineer.

- HA Number: This is the number of the hydrographic area used by the Nevada State Engineer.
- Precipitation Rate: This is the middle precipitation value of 1-in. precipitation bands in inches. The generation process is described in Section E.2.1.3.2.
- Precipitation Band Area: This is the area (in acres) of a 1-in. precipitation band for a given basin. The generation process is described later in this section.
- Precipitation Volume: This field contains the volume of precipitation for a 1-in. band in afy
 within a given basin. It is obtained by multiplying the values in the previous two columns.
- Recharge Efficiency: The values in this field are extracted from the recharge efficiency table calculated during the solution process using the power function and the selected precipitation "values." The recharge efficiency table is located in the "solution worksheet" and is described later in the text.
- Recharge in afy: This field contains the recharge volume of a 1-in. precipitation band. It is the product of multiplying the "Precipitation Volume" by the "Recharge Efficiency."

E.2.1.3.2 Generation of Precipitation Data

The precipitation data serves as the basis for the solver calculations and the derivation of the spatial distribution of recharge. The precipitation values in the table were generated using ArcMap 9.2 as follows:

- 1. The process was initiated from the 800-m PRISM grid (Figure E.2-1A).
- The 800-m PRISM grid was contoured to generate 1-in. contour lines (Figure E.2-1B). Two
 consecutive contour lines represent a 1-in. precipitation band.
- 3. The 1-in. precipitation bands were converted to polygons (Figure E.2-1C).
- Additional contour lines were added midway between each two consecutive contour lines (Figure E.2-1D). These middle-of-the-band contour lines are used as "values" representing the bands.
- 5. "Areas of no recharge," which include the union of the valley floors, the less-than-8-inch-precipitation areas, and the groundwater ET areas, were then removed from the resulting precipitation map.
- 6. A table of precipitation "values" was generated starting from a minimum precipitation of 3 in. to the maximum precipitation occurring within the flow system.
- 7. The surface areas of each precipitation interval in each HA were calculated using the operation "Calculate Geometry" in ArcMap 9.2.

SE ROA 11889





Generation Process of One-Inch Precipitation Bands

8. The table was then exported to Excel and added to the solution file in Worksheet "3-Precipitation-Recharge-Details." The precipitation rate and the precipitation band area are in columns 3 and 4 of this table.

E.2.1.4 Worksheet 4 - "4-Solution"

Worksheet 4 is the core of the Solver. Worksheet 4 requires the table in Worksheet 3, described in Section E.2.1.3. Without the precipitation data and the recharge calculations in Worksheet 3, the Solver will not function. This worksheet is organized in three principal areas: (1) Area 1 - Main Solver Area, (2) Area 2 - Recharge Efficiency Calculations, (3) Area 3 - Basin Groundwater Budget Calculations. Each of these areas is described in the following text.

E.2.1.4.1 Area 1 - Main Solver Area

A description of the main solver area and the process followed to derive initial estimates for the coefficients of the power function are provided in the following text.

E.2.1.4.1.1 Area 1 Description

This area contains the target cell, parameter cell, and a list of the constraints used. Constraints were imposed on the parameters, but also on selected flow terms which are listed in this area as additional supporting information. The target, parameters and constraints are color-coded. The color codes are provided within this area. Cell R4 is the target cell and is color-coded as blue. Cells R6 to R9 contain the parameters and are color-coded as green. The appropriate initial estimates for the four parameters are provided in cells T6 to T9. These values are to be copied onto the parameter cells before executing the solver to ensure consistency of the solution. Cells R20, R21, V7, V16, and Q48 are the constraints and are color-coded as pink. Cell Q48 is located in the recharge efficiency calculation area described in the next section. Cells highlighted in yellow signify that the value in that cell or the cell next to it is constant.

The target is represented by the estimated value of total groundwater ET for the WRFS. The target cell contains a formula relating groundwater ET to the other components of the budgets; this formula is as follows:

$$ET_{ww} = R_T + Inflow - Outflow$$
 (Eq. E.2-1)

where,

ET_{gw} = total groundwater ET for the WRFS (afy) R_T = total natural recharge for the WRFS (afy) Inflow = total groundwater inflow to the WRFS from the Meadow Valley Flow System (afy) Outflow = total groundwater outflow from the WRFS to the Colorado River (afy)

Appendix E

SE ROA 11891



The target value for " ET_{gw} " and the estimate of "Inflow" are 139,400 and 9,200 afy, respectively. Due to the large number of unknown parameters, no unique solution can be derived by the solver. The solution included here is the best based on initial estimates of parameters and defensible estimates of the constraints used. If the solver starts from different initial estimates or constraints, different solutions, and possibly no solution, may be produced. Thus, appropriate initial estimates for the parameters were estimated and are provided in cells T6 to T9. They are as follows:

- Power function constant: $a = 8.0 \times 10^{-5}$ (see Figure E.2-2)
- Power function exponent: b = 3.62 (see Figure E.2-2)
- Outflow from Coyote Spring Valley to Hidden Valley = 15,000 afy (Appendix D)
- Total outflow = 25,000 afy (Appendix D)

E.2.1.4.1.2 Initial Estimates for Coefficients of Power Function

The initial estimates for the two coefficients of the power function, a and b, represent the constant and exponent of another power function derived from the step-function defined by the standard Maxey-Eakin coefficients. The precipitation ranges used in the Maxey-Eakin method were split into the two values defining the range (Table E.2-1). For the Maxey-Eakin range specified as "greater than 20 in.," the maximum value was set to match the mid-value of the maximum precipitation band identified for the WRFS in this analysis (25.5 in./yr). A mean precipitation value was then calculated for each precipitation range, and a mean recharge value was calculated by multiplying it by the corresponding Maxey-Eakin recharge efficiency (Table E.2-1). The resulting point data were fit with a power function shown in Figure E.2-2. The independent variable, X, represents precipitation in inches per year; and the dependent variable, Y, represents recharge in inches per year. The coefficients of this power function (Figure E.2-2) serve as the initial estimates for the recharge versus precipitation power function derived for this study.

Precipitation Zone Used in Maxey-Eakin Method (in./yr)	Minimum Precipitation Rate (in./yr)	Maximum Precipitation Rate (in./yr)	Mean Precipitation (in./yr)	Standard Maxey-Eakin Efficiency	Mean Recharge (in./yr)
8 to 12	8.00	12.00	10.00	0.03	0.30
12 to 15	12.00	15.00	13.50	0.07	0.945
15 to 20	15.00	20.00	17.50	0.15	2.625
Greater than 20	20.00	25.50	22.75	0.25	5,6875

Table E.2-1 Data for Maxey-Eakin Power Function

E.2.1.4.2 Area 2 - Recharge Efficiency Calculations

Area 2 of the solution sheet contains two tables and two graphs. The first table contains the 1-in. precipitation bands and the corresponding recharge efficiencies calculated using the power function



Figure E.2-2 Power Function Fit to Maxey-Eakin Precipitation-Recharge Points (Table E.2-1)

coefficients listed as parameters in the main Excel Solver area described above. For each precipitation band, the recharge efficiency is calculated as the recharge rate expressed as the power function divided by precipitation as follows:

$$Eff = \frac{[a(P-8)^{b}]}{P}$$
(Eq. E.2-2)

where,

Eff = recharge efficiency or R/P as a fraction

a = power function constant

b = power function exponent

P = Precipitation (in./yr)

P-8 = Effective precipitation (in./yr)

These efficiencies utilize whatever values of a and b are stored in the appropriate cells: initial estimates or the final values derived by the solver. A constraint is imposed on the maximum recharge efficiency (cell Q48). The maximum recharge efficiency cannot be greater than 1, but the exact efficiency for the WRFS is unknown and most likely much less than 1. Therefore, a literature review

Appendix E



was conducted to identify the available reported values, which are listed in Table E.2-2 and discussed in the subsequent text.

Authors	Recharge Estimation Method and Location	Bedrock Aquifer	Maximum Recharge Efficiency (Fraction of Precipitation)
Maxey and Eakin (1949)	Water balance WRFS	Carbonate-rock aquifer	0.25
Lichty and McKinley (1995)	Precipitation runoff modeling, index-site extrapolation and chloride mass balance methods	Volcanic tuffs	0.50
Maurer and Berger (1997)	Application of Darcy's Law and chloride mass balance in Kings Canyon of Carson Basin, Nevada	Metamorphic rocks	0.35
Nichols (2000)	Water balance, eastern Nevada	Mainly limestone and dolomite, with some volcanic and metamorphic rocks	0.63
Davisson and Rose (2000)	Hydrologic mass balance calibrated groundwater age dates in Fenner Basin California	Granitic, metamorphic and volcanic rocks	0.48
Faybishenko (2007)	Potential evapotranspiration at meteorological stations in western U.S.	N/A	0.41

Table E.2-2 Maximum Recharge Efficiency Reported

Maxey and Eakin (1949) derived a maximum recharge efficiency of 0.25 for the precipitation zone greater than 20 in. based on the Hardman (1936) precipitation map. This value represents a maximum for a large interval, but not the maximum for the much smaller 1-in. bands used in this study. Lichty and McKinley (1995) reported a maximum recharge efficiency of 0.50 for two small basins located in central Nevada. They derived their recharge efficiencies based on the application of a precipitation-runoff model, the PET index-site extrapolation method, and the chloride mass balance to volcanic tuffs. Maurer and Berger (1997) estimated a maximum recharge efficiency of 0.35 for metamorphic rocks in Kings Canyon of Carson Basin by application of Darcy's Law and the chloride mass balance. Nichols (2000) reported a maximum recharge efficiency of 0.63 percent based on new ET estimates and a preliminary version of the 4-km 1971-to-2000 PRISM precipitation map for basins in Nevada dominated by carbonate rocks. Davisson and Rose (2000) calibrated the Maxey-Eakin recharge model to the local conditions of the Fenner Basin in California, using local recharge estimates estimated with field information. Faybishenko (2007) calculated a maximum infiltration efficiency of 0.412 based on data from several states including Nevada. The largest reported recharge maximum efficiency is 0.63 percent of precipitation. This value was used to constrain the upper limit of the recharge efficiency in the solver. It is deemed appropriate not only because it is the largest value, but also because it was derived for a nearby area with similar climatologic and geologic conditions.

The solution's recharge efficiencies are displayed on a scatter plot showing the calculated recharge efficiencies versus precipitation rates (first graph) in Area 2 of the solution worksheet. The other table and graph show the same results averaged over the wider precipitation zones defined by Maxey and Eakin (1949) for comparison.

E.2.1.4.3 Area 3 - Groundwater Budget Calculations

This area contains the groundwater budget calculations for each of the basins of the WRFS in the form of a chart. Two additional estimates of interbasin flow volumes were added to support the calculations and are provided at the top of the area. In the lower-left corner of this area, the budget for the entire flow system is shown for comparison with the overall groundwater budget resulting from the individual basin calculations for checking purposes. The cells in the chart are linked to the appropriate cells in the solver area, the ET worksheet, and the detailed precipitation-recharge worksheet. The chart and the WRFS budget are automatically updated when the solver is executed.

E.2.2 Solution Process

The solution process includes two major steps: (1) execution of the Excel Solver to derive a recharge distribution, and (2) derivation of the basin budgets using the recharge distribution.

E.2.2.1 Executing the Solver

The solver is activated by clicking "Solver" under Excel menu item "Tools" from the solution sheet (Worksheet 4). At this point, the "Solver Parameters" window pops up (Figure E.2-3). In its upper part, the window displays information about the objective of the solver. This information includes the specification of the target cell (Cell R4) and an option for the objective function specifying whether the solver is to maximize, minimize, or equate the objective function to a target value. When the equality option ("Value of") is selected as it is in this case, a target value must be specified. In this case, a value of "139,424" is specified, representing the total GW ET in afy for the WRFS calculated in Worksheet 2.

Under the objective function information, the cells where the parameters are located in the worksheet are provided under "By Changing Cells." In this case the parameter cells range from R6 to R9. During the solution process, the solver iteratively changes the values in these cells until the objective is achieved, in this case, until the value in the target cell equals the specified target value of 139,424 afy.

Under the parameter information, a list of the constraints is provided to the solver to constrain the solution under the title "Subject to the Constraints." Each row in the list consists of a cell address in the solution worksheet, a logical operator, and a value. The logical operator may be set to "Equal," "Less than or equal", or "Greater than or equal." The constraints on the solution allow the solver to narrow the domain of feasible solutions for problems with many unknowns. The constraints used in this case are as follows:

Power Function Constant, a is positive

Appendix E

SE ROA 11895





Figure E.2-3 Excel Solver Window

- Power Function Exponent, b is positive
- Outflow from White River Valley is between 6,300 and 40,000 afy (Maxey and Eakin, 1949 and LVVWD, 2001)
- Outflow from Coyote Spring Valley to Hidden Valley is less than or equal to 15,000 afy (Appendix D)
- Total inflow to Muddy River Springs Area is greater or equal to 40,000 afy (Section 4.0)
- Total outflow is greater or equal to 25,000 afy (Appendix D)
- Maximum recharge efficiency is less or equal to 63 percent (specified as 0.63) (Nichols, 2000)

In the right-hand side of the solver window are a "Solver" button to execute the solver, a "close" button to close the window, an "options" button to specify the solution method and convergence criteria, a "Reset" button, and a "Help" button. Once the "Solve" button is activated, the user is presented with various options for keeping or resetting the solution, saving scenarios, and reporting results. The novice user should use the default options.

E.2.2.2 Deriving Basin Groundwater Budgets

The groundwater budgets in the appropriate area of the solution worksheet (Area 3) are updated automatically every time the solver is executed. Although these calculations are conducted in Excel, they are performed outside of the solver, but using the results of the solver (recharge and selected

> Appendix E SE ROA 11896

interbasin flow volumes). The solver provides a recharge distribution for all basin in the WRFS in Worksheet 3. Assuming that each basin is under predevelopment steady-state conditions, a groundwater budget is derived for each basin. The process starts from the most upgradient basin of the WRFS (Long Valley), and ends at the southernmost basins (Lower Moapa Valley and Black Mountains Area). For each basin, any inflow to the basin is added to the estimate of basin recharge, and the basin's groundwater ET value is then subtracted. The remainder is the outflow is routed to the next contiguous basin(s) located downgradient. For basins having more than one outflow boundary, independent estimates of selected outflow boundaries were made. Such is the case for two basins: Cave Valley and Pahroc Valley. Outflow from Cave Valley is to White River Valley and Pahroc Valley. Outflow from Pahroc Valley is to Pahranagat Valley and Dry Lake Valley. Portions of these outflows are specified in this area of the top of this worksheet (Area 3).

E.3.0 UNCERTAINTY ANALYSIS EXCEL FILES

All Excel files containing the eleven uncertainty cases are organized exactly the same as the solution file. The only difference in the first ten cases (Table E.3-1) is that each file contains an alternate scenario where a single variable was assigned a different value to test the solution. Each of these ten scenarios corresponds to the low and high values of the four most uncertain variables. The corresponding filenames and related information are listed in Table E.3-1. The eleventh Excel file is named: "SNWA-BARCASS-WRFS-Groundwater-Budget-Solution," and contains the special scenario conducted to evaluate the BARCASS interpretation for the northern portion of the WRFS. The contents of this file are identical to the contents of the file containing the main solution, except for the following:

- The groundwater ET values under the heading "SNWA/BARCASS GW ET" in Worksheet 2 were used in the calculations, instead of those under the "SNWA GW ET" heading.
 - The target value in the solver window was changed from 139,424 to 147,791 afy.
 - The ET values for the four northern valleys were replaced with the BARCASS estimates in the basin groundwater budget chart located in the solution sheet (Worksheet 4 - Area 3).
- In the groundwater budget chart (Worksheet 4 Area 3), all groundwater outflow from Cave Valley was routed to White River Valley, instead of White River and Pahroc valleys.

The solver in all eleven uncertainty files is executed in the same manner as in the solution file.



Table E.3-1 Range of Uncertainty for Selected Budget Components

Excel Filename	Variable	Low Value	High Value	Sources
SNVVA-VVRFS-Groundwater-Budget-Unc-V/RV-G/VET-Low swiv/a_V/RFS-Groundwater-Budget-Unc-V/RV-G/VET-High	White River Valley Groundwater ET (afy)	37,000	79,560	Eakin, 1966; LVVMD, 2001
SNVVA-WRFS-Groundwater-Budget-Unc-MRSA-GWET-Low SNVVA-WRFS-Groundwater-Budget-Unc-MRSA-GWET-High	Muddy River Springs Area ET (afy)	2,300	5,989	Eakin, 1966; this study in Section 4.0
SNWA-WRFS-Groundwater-Budget-Unc-Inflow-Low SNMA-WRFS-Groundwater-Budget-Unc-Inflow-High	Total Inflow to WRFS (afy)	7,000	32,000	Rush, 1968; LVVWD, 2001
SNWA-WRFS-Groundwater-Budget-Unc-Outflow-Low SNMA-WRFS-Groundwater-Budget-Unc-Outflow-High	Total Outflow from WRFS (afy)	11,100	49,000	Rush, 1968; LVVWD, 2001
SNWA-WRFS-Groundwater-Budget-Unc-Max-Eff-Low snum, MiDFS-Groundwater-Budget-Unc-Max-Eff-High	Maximum Recharge Efficiency (percent of precipitation)	0.25	1	Maxey and Eakin, 1949

Part A - Natural Recharge, Discharge, and Interbasin Flow for White River Flow System

Appendix E

E.4.0 REFERENCES

- Davisson, M.L., and Rose, T.P., 2000, Maxey-Eakin methods for estimating groundwater recharge in the Fenner Watershed, southeastern California: U.S Department of Energy, Lawrence Livermore National Laboratory, UCRL-ID-139027, 13 p.
- Eakin, T.E., 1966, A regional interbasin groundwater system in the White River area, southeastern Nevada: Water Resources Research, Vol. 2, No. 2, p. 251–271.
- Faybishenko B., 2007, Climatic forecasting of net infiltration at Yucca Mountain using analogue meteorological data: Vadose Zone Journal Vol. 6, p. 77–92.
- Hardman, G., 1936, Nevada precipitation and acreages of land by rainfall zones: University of Nevada-Reno, Agricultural Experiment Station, mimeographed report, 10 p.
- Lichty R.W., and McKinley, P.W., 1995, Estimates of ground-water recharge rates for two small basins in central nevada: U.S. Geological Survey Water-Resources Investigations Report 94-4104, 31 p.
- Las Vegas Valley Water District, 2001, Water resources and ground-water modeling in the White River and Meadow Valley flow systems—Clark, Lincoln, Nye and White Pine counties, Nevada, 285 p.
- LVVWD, see Las Vegas Valley Water District.
- Maurer, D.K., and Berger, D.L., 1997, Subsurface flow and water yield from watersheds tributary to Eagle Valley hydrographic area, west-central Nevada: U.S. Geological Survey Water-Resources Investigations Report 97-4191, 56 p.
- Maxey, G.B., and Eakin, T.E., 1949, Groundwater in White River Valley, White Pine, Nye, and Lincoln counties, Nevada: State of Nevada, Office of the State Engineer, Water Resources Bulletin No. 8, 59 p.
- Nichols, W.D., 2000, Regional ground-water evaporation and ground-water budgets, Great Basin, Nevada: U.S. Geological Survey Professional Paper 1628, 108 p.
- Rush, E.F., 1968, Water-resources appraisal of the Lower Moapa–Lake Mead area, Clark County, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 50, 67 p.

Walkenbach, 2003, Microsoft Office Excel 2003 Bible, Wiley Publishing, Inc., 936 p.

Appendix E

SE ROA 11899



Welch, A.H., and Bright, D.J., eds., 2007, Water resources of the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah—Draft Report: U.S. Geological Survey Open-File Report 2007-1156, 103 p.



Part B

Hydrogeology of Cave, Dry Lake, and Delamar Valleys

Prepared by: Andrew G. Burns, James M. Watrus, and Gary L. Dixon

November 2007

SOUTHERN NEVADA WATER AUTHORITY Groundwater Resources Department Water Resources Division snwa.com

SE ROA 11901

Part B - Hydrogeology of Cave, Dry Lake, and Delamar Valleys

CONTENTS

List of	Figures	
List of	Plates .	
List of	Tables.	······································
List of	f Acrony	ms and Abbreviations IX
1.0	Introdu	action
1.0	marous	D
	1.1	2-1
2.0	Geolo	gic Framework
	2.1	Cave Valley. 2-1 2.1.1 Geology 2-1 2.1.2 Geophysics 2-2
		2.1.2.1 Gravity 2-2 2.1.2.2 Audiomagnetotellurics 2-2 2.1.2.3 Seismic Studies 2-2 2.4 2-4
	2.2	Dry Lake Valley 2-4 2.2.1 Geology 2-4 2.2.2 Geophysics 2-6 2.2.2.1 Gravity 2-6 2.2.2.2 Seismic Reflection 2-8
	2.3	Delamar Valley 2-8 2.3.1 Geology 2-8 2.3.2 Geophysics 2-8 2.3.2.1 Gravity 2-8
3.0	Grou	ndwater Occurrence and Movement
	3.1 3.2 3.3	Groundwater-Site Inventory
4.0	Sumi	nary and Conclusions 4-1
	4.1 4.2	Cave Valley
5.0	Refe	rences

Appendix A - Hydrographs for Wells in Cave, Dry Lake, and Delamar Valleys

SE ROA 11902



This Page Intentionally Left Blank



SE ROA 11903

JA_4665
FIGUR Number	RES R TITLE	PAGE
2-1	Location of Gravity Stations and Depth-to-Basement Map for Cave Va Northern Dry Lake Valley	Iley and 2-3
2-2	AMT Model along Profile E across Central Cave Valley, Nevada	2-4
2-3	(a) ECN-01 Seismic Reflection Section Displayed in Time(b) Results of Gravity Depth-to-Basement.	2-5
2-4	Location of Gravity Stations and Depth-to-Basement Map for Dry Lak and Delamar Valleys	e 2-7
3-1	Cave Valley Water-Level Elevation Map	3-5
3-2	Historical Water-Level Elevations at Cave Valley MX Well (Map ID	180-07) 3-6
3-3	Dry Lake and Delamar Valleys Water-Level Elevation Map	3-8
3-4	Historical Water-Level Elevations at Dry Lake MX Well (Map ID 18)	1-05) 3-9
A.1-1	Historical Water-Level Elevations at 180 N07 E63 14BADB1 USAF (Map ID 180-05).	A-1
A.1-2	Historical Water-Level Elevations at 180 N07 E63 14BADD1 USGS-MX (Cave Valley) (Map ID 180-07)	A-1
A.1-3	Historical Water-Level Elevations at 180 N07 E63 15DBAD1 USBL (Map ID 180-08).	M A-2
A.1-4	Historical Water-Level Elevations at 180 N08 E64 04ABDD1 USBL (Map ID 180-14).	M A-2
A.1-5	Historical Water-Level Elevations at 180 N08 E64 15BCBC1 USBL (Map ID 180-15)	M A-3
A.1-6	Historical Water-Level Elevations at 180 N08 E64 30CDBC1 USBL (Map ID 180-16).	M A-3
A.1-7	Historical Water-Level Elevations at 180 N09 E64 27BCDD1 USBL (Map ID 180-19)	.M A-4
A.1-8	Historical Water-Level Elevations at 180 N10 E63 25A 1 (Map ID 1	80-20) A-4

iii SE ROA 11904



FIGURES (CONTINUED) PAGE TITLE NUMBER Historical Water-Level Elevations at 180 N10 E64 06BDA 1 A.1-9 A.1-10 Historical Water-Level Elevations at 180W501M (Map ID 180-24) A-5 A.1-11 Historical Water-Level Elevations at 180W902M (Map ID 180-25) A-6 A.1-12 Historical Water-Level Elevations at 181 N01 E62 24ABB 1 USBLM A.1-13 Historical Water-Level Elevations at 181 N03 E63 27CAA 1 USGS-MX (N. Dry Lake) (Map ID 181-05) A-7 A.1-14 Historical Water-Level Elevations at 181 N03 E64 20BD 1 USBLM - Coyote Well (Map ID 181-06)A-7 A.1-15 Historical Water-Level Elevations at 181 N03 E65 21D 1 Bristol Well A.1-16 Historical Water-Level Elevations at 181 N04 E64 07DC 1 USGS-MX (Muleshoe Valley) (Map ID 181-09) A-8 A.1-17 Historical Water-Level Elevations at 181 S03 E64 12AC 1 USGS-MX (S. Dry Lake Well) (Map ID 181-19) A-9 A.1-18 Historical Water-Level Elevations at 181 S03 E64 12AC 2 USGS-MX A.1-19 Historical Water-Level Elevations at 181M-1 (Map ID 181-22)..... A-10 A.1-20 Historical Water-Level Elevations at 181W909M (Map ID 181-23) A-10 Historical Water-Level Elevations at 182 S06 E63 12AD 1 A.1-21 USGS-MX (Delamar Well) (Map ID 182-02) A-11 A.1-22 Historical Water-Level Elevations at 182M-1 (Map ID 182-04)..... A-11

A.1-23 Historical Water-Level Elevations at 182W906M (Map ID 182-05) A-12

PLA NUM	ER TITLE
1	Geology and Geologic Cross Sections of Cave, Dry Lake, and Delamar ValleysPocket





SE ROA 11907

vi

JA_4669

TAE	BLES BER TITLE	PAGE
3-1	Water Level Data for Selected Wells and Springs in Cave, Dry Lake, and Delamar Valleys	3-2

SE ROA 11908

JA_4670



viii

ACRONYMS

AMT	audiomagnetotelluric
BLM	Bureau of Land Management
NDWR	Nevada Division of Water Resources
SNWA	Southern Nevada Water Authority
USAF	U.S. Air Force
USGS	U.S. Geologic Survey
UTM	Universal Transverse Mercator
WRFS	White River Flow System

ABBREVIATIONS

afy	acre-feet per year
amsl	above mean sea level
bgs	below ground surface
ft	foot
km	kilometer
m	meter
mGal	milligalileo
mi	mi





SE ROA 11911

JA_4673

1.0 INTRODUCTION

Part B of this report describes the geologic framework and the occurrence and movement of groundwater within Cave, Dry Lake, and Delamar Valleys. The purpose and scope of this description is presented in the following section.

1.1 Purpose and Scope

The purpose of describing the geologic framework of Cave, Dry Lake, and Delamar Valleys is to provide the basis from which an understanding of the hydrogeology of the basins can be developed, particularly the geologic features controlling local and regional groundwater flow. The hydrogeology is further defined through an evaluation of the hydraulic data that describes the occurrence and movement of groundwater.

The geologic framework is presented as a surface geologic map and a series of geologic cross sections (Plate 1), depth-to-basement maps, and profiles interpreting seismic and audiomagnetotelluric (AMT) data. Hydraulic data from wells and springs are used in conjunction with the geologic framework to describe the occurrence and movement of groundwater within the basins and vicinity. The hydrogeologic description is relied upon in Part C of this report, which evaluates the potential water-related effects of developing the Southern Nevada Water Authority (SNWA) applications in these basins.

1-1



Section 1.0 SE ROA 11913

2.0 GEOLOGIC FRAMEWORK

The SNWA has described the stratigraphy, structural geology, and hydrogeology of the study area in a report titled "Geology of White Pine and Lincoln Counties and Adjacent Areas, Nevada and Utah: The Geologic Framework of Regional Groundwater Flow Systems" (Dixon et al., 2007). This section provides an overview of the geologic information for Cave, Dry Lake, and Delamar valleys presented in that report.

2.1 Cave Valley

2.1.1 Geology

Cave Valley is part of the White River Flow System (WRFS) and consists of two distinct but connected areas, separated by an oblique-slip fault at Shingle Pass (Plate 1). One of these areas, northern Cave Valley, is a narrow graben with mostly east-dipping Cambrian rocks at shallow depth and containing relatively thin basin-fill sediments. The southern area, south of Shingle Pass, generally contains less than 3,000 ft of basin-fill sediments and volcanic rocks but in a narrow, central, north-trending axial part, these Cenozoic rocks are 6,000 ft or more thick (Plate 1, Cross Sections U—U', R—R', and Q—Q').

Along the western side of Cave Valley, the Egan Range is a complexly faulted horst of east-dipping Cambrian to Permian rocks, overlain by Tertiary volcanic rocks (Plate 1). The Egan Range separates Cave Valley from White River Valley to the west. Halfway southward down Cave Valley a northeast-striking oblique-slip fault passes through the Egan Range at Shingle Pass. This fault has partitioned the valley into two sub-basins by displacing a section of the Egan Range and forming an east-dipping fault block that extends northeast across and beneath Cave Valley where it terminates against the range-front fault of the Schell Creek Range. Based on oil test well drilling and gravity surveys, this block contains the Mississippian Chainman Shale (Hess, 2004; Mankinen et al., 2007; Scheirer, 2005; Eastman, 2007a and b).

Farther south, the Egan Range remains an east-tilted horst of Cambrian through Tertiary rocks, then bends southeast to join the southern end of the Schell Creek Range. Here Cave Valley terminates where the Egan and Schell Creek ranges join each other in a complex of north-northeast- and north-northwest-striking normal and oblique-slip faults. To the east, the Schell Creek Range separates Cave Valley from Lake and northern Dry Lake (Muleshoe) Valleys. This section of the Schell Creek Range contains a narrow, heavily faulted sequence of Precambrian through Tertiary rocks that dips east. Here the dominant fault is on the western flank of the range. In the northeast part of the valley, the Schell Creek Range is cored by Precambrian to Cambrian quartzite. West of the Geyser Ranch, the rocks are mostly Late Proterozoic and Cambrian quartzite (Van Loenen, 1987), but farther south the rocks are dropped down along an east-trending fault at Patterson Pass and are mostly



of middle to upper Paleozoic and Tertiary age. In the south, where Cave Valley terminates at the intersection of the Schell Creek Range and the Egan Range, a Tertiary pluton has mineralized adjacent carbonate rocks at the Silver King Mine.

2.1.2 Geophysics

2.1.2.1 Gravity

The isostatic gravity anomaly of Cave Valley is characterized by 20 to 30 mGal lows centered on the valleys relative to the isostatic values in the surrounding ranges. There are significant gravity fluctuations within the ranges, reflecting the lithologic variation within them. Larger and more continuous maximum gradient picks are present in southern Cave Valley relative to northern Cave Valley (Scheirer, 2005). In the south, two main lines of maximum gradients are found paralleling the eastern and western margins of the valley.

The depth-to-basement separates the isostatic gravity anomaly into portions that arise from the Cenozoic deposits (basin-fill) from those of pre-Cenozoic rocks (basement), and the resulting basin gravity anomaly is illustrated in Figure 2-1. Basin depths estimated from gravity extend to approximately 6.0 km, or to almost 20,000 ft-bgs in Cave Valley. In northern Cave Valley, typical basement depths are hundreds of meters, and no site has an estimated depth greater than 2 km. In southern Cave Valley, the basin has depth-to-basement estimates greater than 1 km for more than half of its length; its deepest inferred depth is just east of the valley's axis.

2.1.2.2 Audiomagnetotellurics

In conjunction with gravity studies of Scheirer (2005) the AMT method was applied to Cave Valley in an effort to delineate structure and stratigraphy in the upper 1 km, in particular, faults and stratigraphy within the valleys, as well as estimates of depth to basement rocks.

An abrupt contact between the resistive limestone basement rock on the east side of Cave Valley in the Sidehill Pass area (Figure 2-2) and the more conductive valley fill agrees with the sharp gravity gradient observed by Scheirer (2005), who calculated a steep eastern basin margin which is bounded by a range-front fault (Figure 2-2). Drill hole 180W902M provided additional evidence of the reliability of the method.

2.1.2.3 Seismic Studies

Subsurface structure of Cave Valley and northern Dry Lake Valley (Muleshoe) is provided by a portion of the industry-shot ECN-01 seismic reflection line (Figure 2-3). The seismic reflection image illustrates the asymmetric character of Cave Valley, with steeper eastern side where the range-front fault of the Schell Creek Range lies and a less-steep, but still fault controlled, western side along the eastern side of the Egan Range.



JA_4678





Source: Dixon et al., 2007

Figure 2-2 AMT Model along Profile E across Central Cave Valley, Nevada

2.2 Dry Lake Valley

2.2.1 Geology

Dry Lake Valley is a deep graben that contains in most places 3,000 to 5,000 ft of basin-fill sediments (Mankinen et al., 2007) (Plate 1), but locally along the axis of the graben as much as 7,000 ft of sediments and underlying down faulted volcanic and carbonate rocks are present (Scheirer, 2005; modified by Mankinen, et al., 2007) (Plate 1, Cross Sections Q—Q', T—T', P—P', and S—S').

Along the western side of Dry Lake Valley at the junction with the southern Egan and Schell Creek ranges, the North Pahroc Range extends south for about 40 mi separating the valley from Pahroc Valley to the West (Plate 1). The North Pahroc Range consists of upper Paleozoic rocks overlain by Tertiary volcanic rocks (Plate 1, Cross Section T—T'). These rocks dip west off major faults along the eastern side of the range. The North Pahroc Range is separated from the smaller South Pahroc Range at the southern end of the valley by an east-trending belt of faulted rocks of low relief formed by the east-striking Timpahute transverse zone (Plate 1). The belt of faulted rocks are the oldest structural features in the valley, and help form the topographic divide between Dry Lake Valley and Delamar Valley to the south. This zone is of little consequence with respect to the hydraulic connectivity between the two basins. The Seaman and the North Pahroc ranges join together at their southern ends, and the Hiko Range continues south of this intersection.

On the east side and from north to south, the Fairview, Bristol, Highland, and Chief Ranges are a 60-mi-long group of north-trending, heavily-faulted ranges of mostly east-dipping rocks that separate Dry Lake Valley from Lake, Patterson and Panaca Valleys to the east. The northern portion of Dry



Source: after Scheirer, 2005

Figure 2-3 (a) ECN-01 Seismic Reflection Section Displayed in Time (b) Results of Gravity Depth-to-Basement

Section 2.0



Lake Valley (Muleshoe Valley) is bounded on the east by the Fairview Range (Plate 1). The Fairview Range is a horst made up of Devonian to Pennsylvanian rocks at both the northern and southern ends of the range (Plate 1, Cross Section Q—Q'). The central part of the range consists of the western lobe of the Indian Peak caldera complex. To the south, the valley is bounded on the east by the Bristol, Highland, and Chief ranges. A low pass between the Fairview Range and the Bristol Range is cut by numerous east-striking faults of the Blue Ribbon transverse zone, which crosses the entire Great Basin at about this latitude (Rowley, 1998; Rowley and Dixon, 2001). The small horsts due west of this area are named the West, Ely Springs, Black Canyon, and Burnt Spring Ranges.

The Bristol Range is a horst that consists mostly of an east-dipping sequence of Cambrian carbonate rocks. The range is cored by a Tertiary pluton on the northern end that is associated with silver deposits of the Jackrabbit and Bristol districts. A low angle, west-dipping detachment or gravity-slide fault that placed Devonian rocks on Cambrian rocks is exposed in the northwestern part of the range (Page and Ekren, 1995). The Highland Range, the southward continuation of the Bristol Range, consists of east-dipping Cambrian carbonate rocks, underlain by Precambrian and Cambrian quartzite. A west-dipping, west-verging, moderately dipping fault on the western side of the range, the breakaway part of the Highland detachment fault, placed the younger carbonate rocks on the older quartzite. The Chief Range, south of the Highland Range, is made up of east-dipping Precambrian and Cambrian quartzite that is unconformably overlain by Tertiary volcanic rocks and cut by a Tertiary pluton that controls the small Chief gold district. The faults that lift the range on the western side consist of an oblique-slip fault (right lateral and normal) and the west-dipping Highland detachment fault (Rowley et al., 1994).

The small West Range consists of Devonian sedimentary rocks and Tertiary volcanic rocks on which are Devonian rocks emplaced by a low-angle fault that can be interpreted as either a detachment fault or a gravity-slide plane (Page and Ekren, 1995). The Ely Springs Range consists of Cambrian through Silurian rocks, overlain by Tertiary volcanic rocks. The Black Canyon Range consists of Cambrian sedimentary rocks and Tertiary volcanic rocks. The Burnt Springs Range consists of Cambrian sedimentary rocks unconformably overlain by Tertiary volcanic rocks.

2.2.2 Geophysics

2.2.2.1 Gravity

Dry Lake Valley is bounded by the North Pahroc Range in the west and Highland and Bristol ranges in the east, and the valley is marked by low saddles to the north (adjoining Muleshoe Valley) and south (adjoining Delamar Valley). The isostatic gravity anomaly has a greater than 30 mGal negative value in the center of the valley (Figure 2-4).

The depth-to-basement solution for Dry Lake Valley exhibits a normal structural graben on the margins of the basin and also a central graben down the axis of the valley. The depth-to-basement in the margins of the valley are approximately 1 km, whereas in the central graben most depths are greater than 3 km. Because the ranges surrounding Dry Lake Valley are composed predominately of volcanic rocks, the sedimentary fill might have a lower density than most other basins, in which case the density contrasts with bedrock would be larger and inverted basins would be shallower.





2.2.2.2 Seismic Reflection

A continuation of the industry-shot ECN-01 seismic line extends from Cave Valley to Northern Dry Lake Valley (a.k.a. Muleshoe Valley) (Figure 2-3). The seismic profile (reflectors) indicate that the shallow portions of Muleshoe Valley are weak to absent, but in its deeper section they exhibit characteristics similar to those of the Cave Valley reflectors and agree with the gravity surveys.

2.3 Delamar Valley

2.3.1 Geology

Delamar Valley, just south of Dry Lake Valley, is a graben that deepens to the south with a general maximum thickness of more than 5,000 ft of basin-fill sediments east of the South Pahroc Range (Plate 1) (Mankinen et al., 2007). Locally as much as 7,000 ft of sediments and underlying downfaulted volcanic and carbonate rocks are present (Plate 1, Cross Sections O—O', M—M', N—N') (Scheirer, 2005; modified by Mankinen et al., 2007).

Along the western side of the valley (Plate 1), the South Pahroc Range extends southward from the North Pahroc Range, separating the valley from Pahranagat Valley to the west. The South Pahroc Range is a series of west-titled blocks of volcanic rocks; the main faults are on the eastern side of the range. The South Pahroc Range terminates against the east-northeast-trending Pahranagat shear zone, which also terminates Pahranagat and Delamar Valleys at their southern extent.

The eastern side of the valley is bounded by the Delamar Mountains, which extend southward for 40 mi from the Burnt Springs Range in southern Dry Lake Valley (Plate 1). The boundary between the two ranges can be placed at the northern caldera wall of the Caliente caldera complex, here controlled by the east-trending Timpahute transverse zone (Ekren et al., 1976; Rowley, 1998; Swadley and Rowley, 1994). The Delamar Mountains consists of east-dipping Late Proterozoic to Cambrian rocks and Tertiary volcanic rocks. The range, however, is dominated by Tertiary caldera complexes. The western end of the Caliente caldera complex is in the northern part of the range, and the Kane Springs Wash caldera complex is in the central part of the range (Rowley et al., 1995; Scott et al., 1995 and 1996). The main bounding fault of the Delamar Mountains is the down-to-the-west normal fault on the western side, and this is joined from the southwest by several splays of the left-lateral Pahranagat shear zone (Ekren et al., 1977).

2.3.2 Geophysics

2.3.2.1 Gravity

Delamar Valley is surrounded by volcanic ranges to west, south, and east that are highly faulted. The isostatic gravity anomaly is similar to those in the other valleys in the area, but the maximum horizontal gradients are only sporadically clustered along some sections of the South Pahroc and Delamar ranges. The basin gravity anomaly (Figure 2-4) has a minimum restricted to the southern half of Delamar Valley, which leads to the bowl-shaped basin inferred from the gravity inversion

(Figure 2-4). The maximum depth is almost 2.0 km, and it is located west of the center of the southern portion of Delamar Valley.





2-10

Section 2.0

3.0 GROUNDWATER OCCURRENCE AND MOVEMENT

Understanding the groundwater occurrence and movement within Cave, Dry Lake, and Delamar valleys is an important consideration when developing a conceptualization of the underlying groundwater systems. This section describes the occurrence and movement of groundwater within Cave, Dry Lake, and Delamar valleys based on an inventory of groundwater sites and an assessment of the current and recent water-level conditions.

3.1 Groundwater-Site Inventory

Groundwater sites within Cave, Dry Lake, and Delamar valleys include wells and springs which can provide an indication of groundwater conditions, including hydraulic gradients, potentiometric surfaces and flow directions within the basin-fill and carbonate-rock aquifers. Groundwater-site data were compiled from the Nevada Division of Water Resources (NDWR) Well Log database, U.S. Geological Survey (USGS) National Water Information System/Groundwater Site-Inventory database, published and unpublished reports, and internal SNWA databases. The compiled data include location data, reference elevations, site types, well-construction data, lithologic descriptions, and depth-to-water measurements. Table 3-1 lists a summary of these data for sites located within Cave, Dry Lake, and Delamar valleys.

3.2 Cave Valley

Twenty-five wells in Cave Valley were identified through the site inventory; 20 are completed in the basin fill and 5 are completed in carbonate rocks. Most of the basin-fill wells were drilled for stock watering where depths to water are shallow, and range from 2 to 327 ft-bgs. The remaining wells were drilled as part of groundwater exploration and monitoring programs involving the U.S. Air Force (USAF) MX-Missile Program or SNWA. NDWR driller's logs for wells in Cave Valley suggest that the basin fill is composed of mostly sand and gravel with significant cemented strata in the northern portion of the valley and interbedded sands and clay in the southern portion of the valley. Petroleum exploration drilling in Cave Valley provides additional geologic information to depths of several thousand feet. These data provide some insight as to the horizontal gradients of the basin-fill and carbonate aquifers comprising the groundwater system(s) of the basin, and are described in greater detail in the following sections.

Basin-Fill Aquifer

Depths to water in Cave Valley range from near ground surface in parts of northern Cave Valley (i.e., near Cave Spring) to greater than 200 ft-bgs in the southern portion of the valley. Depths to water in the vicinity of the south playa are in excess of 150 ft-bgs, suggesting that the stand of phreatophytes in this area subsists mainly on a perched groundwater system and/or precipitation. Water-level elevations in Cave Valley range from approximately 7,000 ft-amsl in the far northern

Section 3.0

SE ROA 11924

3-2

Section 3.0

SE ROA	11925
--------	-------

	Valleys	
	id Delamar	
	ry Lake, ar	
	in Cave, D	2)
Table 3-1	d Springs	(Page 1 of
	d Wells an	
	or Selecter	
	evel Data f	
	Water Le	

IIII

			Loc	ation	20.00		TALAN		I vetWater	Last	Weter Lavel
01-de	Station Number	Station Name	UTM Easting ^a (m)	UTM Northing ² (m)	Elev. ^b (ft-amsi)	Site Type	Depth (ft-bgs)	Open Interval (ft-bgs)	Measurement Date	to Water ⁶ (ft-bgs)	Elevation ^d (ft-amsl)
10-08	38223114502501	180 N06 E64 18CC 1 Sidehill Pass Well	688,678	4,249,221	5,975	Well-Basin Fill	1	,	7/25/2005	158	5,817
80-02	180 N07 E63 13DB 1	180 N07 E63 13DB 1	687,817	4,259,662	6,014	Well-Basin Fill	250	200 to 240	12/15/1995	180	5,834
80-03	382822114515301	180 N07 E63 14AB 1 USGS-MX	685,674	4,260,043	6,012	Well-Basin Fill	273	200 to 263	3/15/1994	223	5,789
80-04	382822114515302	180 N07 E63 14AB 2 USGS-MX	685,674	4,260,043	6.012	Well-Carbonate	422	380 to 422	10/25/1980	231	5,781
80-08	382810114521501	180 N07 E63 14BADB1 USAF	685,674	4,260,043	6.013	Weil-Basin Fill	269	50 to 269	5/29/2007	219	5,794
80-08	382810114521502	180 N07 E63 14BADB2 USAF	685,674	4,260,043	6,013	Well-Carbonate	418	50 to 418	7/15/2002	221	5,792
180-07	382807114621001	180 N07 E63 14BADD1 USGS-MX (Cave Valley)	685,797	4,259,954	6,012	Well-Carbonate	460	210 to 250/ 375 to 435	5/29/2007	220	5,792
180-081	382747114525701	180 N07 E63 15DBAD1 USBLM	684,672	4,259,311	6,026	Well-Basin Fill	1	i	8/4/2003	15	6,011
180-09	180 N07 E63 27CD 1	180 N07 E63 27CD 1	684,288	4,255,971	5,988	Well-Basin Fill	245	200 to 240	5/15/1998	157	5,831
180-10	180 N07 E63 27DD 1	180 NO7 E63 27DD 1	685,088	4,255,990	5,968	Well-Basin Fill	290	240 to 280	5/15/1998	168	5,820
180-11	180 NO7 E63 27DD 2	180 NO7 E63 27DD 2	685,088	4,255,990	5,988	Well-Basin Fill	320	260 to 300	7/15/1998	183	5,805
180-12	180 N07 E63 33D 1	180 NO7 E63 33D 1	683,325	4,254,561	5,982	Well-Basin Fill	300	198 to 300	1/15/2000	192	5,790
180-13	382640114492801	180 N07 E64 19 1 Gulf Oil Corp	689,787	4,257,364	6,005	Well-Basin Fill	265	240 to 265	3/15/1980	215	5,790
180-14	383458114473601	180 N08 E64 04ABDD1 USBLM	692,206	4,272,813	6,224	Well-Basin Fill	200	160 to 200	2/21/2007	131	6,093
180-15	383307114471001	180 N08 E64 15BCBC1 USBLM	692,846	4,269,374	6,163	Well-Basin Fill)	đ	5/29/2007	262	5,901
180-16	383056114501501	1B0 NDB E64 30CDBC1 USBLM	688,462	4,265,229	6,087	Well-Basin Fill	352	50 to 352	7/11/1999	327	5,760
180-17	383501114504801	180 N09 E63 01A 1	687,559	4,283,005	6,536	Well-Basin Fill	Ā	1	10/16/1962	N	6,534
180-18	383828114474501	180 N09 E64 16ACB 1 Cave Spring	691,761	4,279,249	6,488*	Spring	ŝ	ŗ	(1	6,488
180-19	383632114465801	180 N09 E64 278CDD1 USBLM	692,983	4,275,700	6,414	Well-Basin Fill	315	277 to 315	5/29/2007	221	6,193.
180-20	384207114505601	180 N10 E63 25A 1	686,984	4,285,891	6,604	Well-Basin Fill	20	i	9/22/2006	14	6,590
180-21	384534114495301	180 N10 E64 06BDA 1 Robbers Roost Well	688,337	4,292,295	6,848	Well-Basin Fill	Į	0.0-0-0	7/26/2005	140	6,708
180-22	180 N11 E63 25DD 1	180 N11 E63 250D 1	687,670	4,294,728	6,987	Well-Basin Fill	140	100 to 140	4/15/1998	64	6,696
180-23	382458114474301	180 S07 E64 33 1 Sidehill Spring	692,408	4,254,280	6,531°	Spring	í	,	1	1	6,531
180-24	180M501M	180W501M	588,048	4,273,716	6,457	Well-Carbonate	1,212	788 to 1,192	9/7/2007	1,051	5,406
180-25	180W902M	180W902M	689,805	4,248,353	5,987	Well-Carbonate	606	196 to 882	9/26/2007	138	5,849
180-26	382738114525601	180 N07 E63 15C 1	684,703	4,259,034	6,004	Well-Basin Fill	1	,	1/1/1900	300	5,704
180-27	381624114540302	180 N05 E63 348BCC2 USBLM	683,551	4,238,220	6,229	Well-Basin Fill	1	1	8/25/2003	80	6,221
181-01	375624114444501	181 N01 E62 24ABB 1 USBLM	698,010	4,201,548	4,695	Well-Basin Fill	515	400 to 515	11/13/2006	48	4,648
181-02	181 NO1 E65 02AA 1	181 NO1 E65 02AA 1	706,529	4,206,411	5,663	Well-Volcanic	ı.	r	6/15/1960	10	5,653
181-03	181 ND2 E63 13CA 1	181 N02 E63 13CA 1 Coyote Spring	687,693	4,211,513	5,224*	Spring	t	£	,	1	5,224
181-04	380336114473501	181 N02 E64 03B 1 USBLM	693,124	4,215,351	4,972	Well-Basin Fill	742	702 to 742	5/9/2006	658	4,314
181-05	380531114534201	181 N03 E63 27CAA 1 USGS-MX (N. Dry Lake)	684,519	4,218,103	5,391	Well-Carbonate	2,395	No perforations	7/25/2007	846	4,545

Southern Nevada Water Authority - Water Resources Division

Section 3.0

3-3

Table 3-1 Vells and Springs in Cave, Dry Lake, and Delamar Valleys (Page 2 of 2) Location Ref. PL
Table 3-1 Vells and Springs in Cave, Dry Lake, and Delamar Valley. (Page 2 of 2) Location Ref. PL
Table 3-1 Vells and Springs in Cave, Dry Lake, and Dela (Page 2 of 2) Location
Table 3-1 Vells and Springs in Cave, Dry Lak (Page 2 of 2) Location
Table 3-1 Vells and Springs in Cave, (Page 2 of 2) Location
Table 3-1 Vells and Springs i (Page 2 of 2 Location
Vells and Sp (Page Location

ip-ID Station 31-06 380516114 31-07 380550114 31-08 181 NOS Ef	n Mumber				Det. PL						TTUTE TOTAL
31-06 380616114 31-07 380550114 31-08 181 N03 EF		Station Name	UTM Easting ^a (m)	UTM Northing ^a (m)	Elev, ^b (ft-amsl)	Stte Type	Depth (ft-bgs)	Open Interval (ft-bgs)	Measurement Date	to Water (ft-bgs)	Elevation ^d (ft-ams!)
11-07 380550114 31-08 181 N03 EF	494101	181 ND3 E64 20BD 1 USBLM - Coyote Well	690,486	4,220,368	5.071	Well-Basin Fill	380	50 to 380	5/29/2007	269	4,801
51-08 181 N03 EP	412301	181 N03 E65 21D 1 Eristol Mell	702,610	4,219,921	5,464	Well-Basin Fill	80	20 to 80	5/29/2007	20	5,444
	85 22 1	181 N03 E65 22 1	703,867	4,220,163	5,593	Well-Basin Fill	240	60 to 240	1/15/1966	ŋ	5,590
31-09 381256114	500701	181 N04 E64 07DC 1 USGS-MX (Muleshoe Valley)	689,481	4,232,096	5,534	Well-Basin Fill	1,190	1,050 to 1,190	5/29/2007	253	5,280
31-10 381256114	500702	181 N04 E64 07DC 2 USGS-MX	689,526	4,232,282	5,539	Well-Basin Fill	672	50 to 672	8/27/2003	269	5,270
81-11 381256114	200703	181 N04 E64 07DC 3 USGS-MX	689,526	4,232,282	5,539	Well-Basin Fill	1,134	50 to 1,134	3/10/1990	253	5,285
31-12 381358114	412201	181 N04 E65 04DBD 1 Little Field Spring	701,112	4,233,949	6,150°	Spring	1	1	1	1	6,150
31-13 181 NO4 E6	65 26DC 1	181 N04 E65 26DC 1	705,540	4,227,429	6,319	Well-Volcanic	81	60 to 80	10/15/1954	50	6,268
31-14 381029114	130701	181 N04 E65 29CB 1 Spring	090'669	4,227,795	°090'9	Spring	ı	q	1	ł	6,090
31-15 381506114	1421801	181 ND5 E65 32AD 1 Spring	700,888	4,236,201	6,178°	Spring	ij	9	,	î	6,178
31-16 181 N05 E6	65 34DC 1	181 N05 E65 34DC 1	703,684	4,235,544	6,549	Well-Volcanic	28	10 to 28	2791/132	OE-	6,539
31-17 181 NO5 E6	65 35BA 1	181 NO5 E65 358A, 1	704,865	4,236,789	6,642	Well-Volcanic	22	12 to 22	7/15/1972	12	6,630
81-18 374536114	1443001	181 S02 E65 19CA 1	698,859	4,181,583	4,672	Well-Basin Fill	156	50 to 156	4/28/1994	48	4,623
81-19 374215114	1453101	181 S03 E64 12AC 1 USGS-MX (S. Dry Lake Well)	697,515	4,175,351	4,643	Well-Basin Fill	1,000	600 to 970	7/30/2007	394	4,249
374215114	1453102	161 S03 E64 12AC 2 USGS-MX	697,515	4,175,351	4,643	Welt-Basin Fill	1,300	1,270 to 1,290	7/30/2007	382	4,262
81-21 374215114	1453103	181 S03 E64 12AC 3 USGS-MX	697,515	4,175,351	4,643	Well-Basin Fill	798	768 to 788	3/10/1990	382	4,261
81-22 181M-1		181M-1	688,537	4,198,181	4,966	Well-Carbonate	1.472	765 to 1.471	9/4/2007	675	4,291
81-23 181W909M	V	Me0ew181	668,688	4,174,479	4,804	Well-Basin Fill	1,260	637 to 1,240	9/4/2007	497	4,307
B1-24 161 S03 Et	63 22AB 1	181 S03 E63 22AB 1	684,620	4,172,306	5,313	Well-Basin Fill	•		3/15/1966	e	5,310
B1-25 181 S04 Et	64 07AC 1	181 S04 E64 07AC 1 Jacob 7B	669,759	4,165,424	4,810	Well-Basin Fill	ŀ	50 to 1,000	5/26/2003	515	4,295
82-01 182 S05 Et	64 02CB 1	182 S05 E64 02CB 1 Grassy Spring	695,124	4,157,193	5.786*	Spring	ł		1	i	5,786
82-02 372639114	1520901	182 S06 E63 12AD 1 USGS-MX (Delamar Well)	688,422	4,146,273	4.713	Well-Basin Fill	1,195	920 to 980/ 1,040 to 1,180	7/30/2007	853	3,850
82-03 372639114	1520902	182 S06 E63 12ADBD2 USGS-MX	688.422	4,148,273	4,713	Well-Basin Fill	585	540 to 630/ 816 to 847/ 877 to 940/ 950 to 971	4/1/1981	867	3,846
82-04 182M-1		182M-1	680,874	4,135,306	4,582	Well-Volcanic	1,321	1,000 to 1,300	9/4/2007	827	3,755
82-05 182W906M		182W906M	690,078	4,133,299	4,802	Well-Volcanic	1,702	1,274 to 1,677	9/4/2007	1,316	3,485
09-04 209M-1		209M-1	677,377	4,168,166	5,123	Well-Carbonete.	1,616	1,273 to 1,595	9/4/2007	1,200	3,923

Part B - Hydrogeology of Cave, Dry Lake, and Delamar Valleys



portion of the valley to approximately 5,800 ft-amsl in the south (Figure 3-1). These water-level elevations indicate a north-to-south hydraulic gradient of approximately 0.009.

Carbonate Aquifer

As described in Section 2.0, Cave Valley is effectively partitioned into two sub-basins by the northeast-striking oblique-slip Shingle Pass Fault that has displaced a part of the Egan Range, forming an east-dipping tilt block that extends northeast across and underneath Cave Valley where it terminates against the range-front fault of the Schell Creek Range. This partitioning is reflected in the limited carbonate water-level data.

In the northern sub-basin, a well constructed by SNWA and completed in carbonate rocks to a depth of 1,212 ft-bgs, has a depth-to-water of about 1,051 ft-bgs and water-level elevation of about 5,406 ft-amsl (Figure 3-1). Groundwater production from this well was very limited during air-lift development and subsequent pumping to purge the well before water-chemistry sampling, suggesting that the hydraulic connection between the well and the regional carbonate aquifer is also very limited. The water-level elevation most likely represents the hydraulic head of the nearby Shingle Pass Fault through which a small amount of inter-basin flow from Cave Valley to White River Valley is presumed to occur (Appendix D of Part A). This well is about 80 ft higher than the elevation of nearby local springs in adjacent White River Valley.

In the southern sub-basin, there are two groundwater sites indicative of hydraulic heads in the carbonate aquifer. The first is a well cluster located on the western margin of the basin drilled as part of the USAF MX-Missile Program (Figure 3-1). Depth-to-water at this location is about 220 ft-bgs with a water-level elevation of about 5,790 ft-amsl. The second location is a SNWA monitor well located at the western range-front fault of the Schell Creek Range, near Sidehill Pass. Depth-to-water at this location is about 140 ft-bgs, with a water-level elevation of about 5,850 ft-amsl. Water-level elevations ranging from about 5,850 to 5,790 ft-amsl in the southern sub-basin are almost 450 ft higher than the well site in the northern sub-basin.

Water-Level Trends

The Cave Valley MX well (Map ID 180-07), located in the south-central portion of the valley, has shown a subtly rising water-level trend since the early 1980s with about a 10 ft increase in the overall water-level elevation for the well (Figure 3-2). This well was identified as penetrating the carbonate rocks. Inspection of the other hydrographs for Cave Valley found in Appendix A show a similar type of increasing trend. The other wells for which hydrographs were constructed were identified as penetrating basin-fill materials.

Interbasin Flows

As described in Appendix D of Part A of this report, groundwater outflow from Cave Valley is thought to occur on the western side of Cave Valley through Shingle Pass to the southern third of White River Valley, and through the southern portion of the valley to northern Pahroc Valley. Outflow from Cave Valley to White River Valley occurs through fractured bedrock associated with the Shingle Pass Fault and was estimated to be less than 4,000 afy (Appendix D of Part A). The outflow from southern Cave Valley to the northeastern portion of Pahroc Valley occurs through fractured carbonate bedrock and along fault zones associated with the west range-front fault of the

3-4

Section 3.0



Figure 3-1 Cave Valley Water-Level Elevation Map

Section 3.0

3-5







southern Schell Creek Range. Outflow at this location was estimated to be 9,400 afy (Appendix D of Part A, p. D-6).

3.3 Dry Lake Valley and Delamar Valley

As described in the geology and geophysical discussion of Section 2.0, Dry Lake and Delamar Valleys are essentially one basin from a geologic framework perspective. This conclusion is supported by the available water-level data compiled and evaluated for the two basins.

Twenty-five groundwater sites were compiled for Dry Lake and Delamar Valleys; 17 are completed in the basin fill, 6 are completed in volcanic rocks, and 2 are completed in carbonate rocks. Like Cave Valley, most of the basin-fill wells were drilled for stock watering where depths to water are relatively shallow. As with Cave Valley, the remaining wells were drilled as part of groundwater exploration and monitoring programs involving the USAF MX-Missile Program or SNWA.

SNWA constructed two monitor wells in Dry Lake Valley and two monitor wells in Delamar Valley in 2005. In Dry Lake Valley, one well was drilled to 1,260 ft-bgs and completed in the basin-fill (181W909M), while the other well was drilled to 1,472 ft-bgs and completed in carbonate rocks (181M-1). SNWA monitor wells constructed in Delamar Valley (182M-1 and 182W906M) were drilled to 1,321 and 1,702 ft-bgs, respectively, both completed in volcanic rocks. SNWA constructed a fifth well (209-M1) in Pahranagat Valley at the northwest margin of Delamar Valley, which was drilled to 1,616 ft-bgs and completed in carbonate rocks.

3-6

Section 3.0

Basin-Fill Aquifer

The NDWR driller's logs indicate the basin fill of Dry Lake and Delamar Valleys consists primarily of sands and gravels, with increasing clay content and thicker sequences of volcanic tuff to the south. The water-level data indicate that the depth to water in northern Dry Lake Valley is relatively shallow, ranging from 10 to about 270 ft-bgs. The depth to water deepens to the south, where it ranges from 3 to about 658 ft-bgs in the central and southern portion of Dry Lake Valley, and exceeds 1,300 ft-bgs in southeastern Delamar Valley. Water-level elevations range from 6,540 to 5,270 ft-amsl in the northern portion of Dry Lake Valley, and from 5,653 to 4,249 ft-amsl in the central and southern portion of the valley. Water-level elevations in Delamar Valley range from 3,850 ft-amsl in the central portion of the valley to about 3,490 in the southeastern portion of the valley. The systematic decrease in water-level elevations from northern Dry Lake Valley to southern Delamar Valley are indicative of a north-south gradient and direction of groundwater flow. A hydraulic gradient from the central portion of Dry Lake Valley to the central portion of Delamar Valley was calculated to be 0.0025.

Carbonate Aquifer

Two wells in the compiled data set were identified as penetrating the carbonate-rock aquifer. Both wells are located on the west side of Dry Lake Valley (Figure 3-3), with water-level elevations ranging from 4,545 to 4,291 ft-amsl. These water-level elevations are approximately 1,300 to 1,500 ft lower in elevation than carbonate-rock wells in Cave Valley to the north. An SNWA monitor well in Pahranagat Valley on the northwestern margin of Delamar Valley has an elevation of 3,923 ft-amsl, which is 370 to 620 ft lower than the water levels in the carbonate wells north of this location in Dry Lake Valley, and about 70 to 440 ft higher than the basin-fill wells in southern Delamar Valley.

Water-Level Trends

Water-level fluctuations in Dry Lake and Delamar valleys appear to be minor with a slight upward trend over the past 25 years similar to the trend observed for Cave Valley. The water-level variations can likely be attributed to elimatic variability, as there is little to no groundwater development in these two basins. Figure 3-4 shows that water-level elevations for the USGS-MX (N. Dry Lake Well) have increased approximately five feet from 1986 to the present. This well was identified as penetrating the carbonate-rock aquifer. The other wells for which hydrographs were constructed were identified as penetrating the basin-fill aquifer system. The other hydrographs are provided in Appendix A.

Interbasin Flows

As described in Appendix D of Part A of this report, interbasin flow for Dry Lake Valley occurs as a minor amount of inflow from northern Pahroc Valley, and as outflow to Delamar Valley in the south. The location of inflow is coincident with a series of northwest-trending right-lateral faults that form the boundary between southern Cave Valley, northern Pahroc Valley, and northern Dry Lake Valley. The inflow at this location was estimated to be less than 2,000 afy (Appendix D of Part A). The outflow from Dry Lake Valley to northern Delamar Valley occurs through fractured carbonate rocks and the basin fill. These two hydrographic areas are topographically separated by a low alluvial divide, but are connected geologically and hydrologically making them one contiguous basin.

Section 3.0

SE ROA 11930





Figure 3-3 Dry Lake and Delamar Valleys Water-Level Elevation Map

3-8	Section 3.0
	SF ROA 11931



Historical Water-Level Elevations at Dry Lake MX Well (Map ID 181-05)

Groundwater outflow from Delamar Valley is controlled by the Kane Springs Wash caldera complex underlying the Delamar Mountains to the southeast, the South Pahroc Range and western Delamar Mountains and associated range-front faults to the west, and the northeast/southwest trending Pahranagat shear zone in the south. Regional flow through the caldera complex is unlikely, as the majority of flow occurs along the north-south trending range-front faults and the northeast extension of the Pahranagat Shear Zone to areas of lower potential in northern Coyote Spring Valley and possibly the very southern part of Pahranagat Valley. The range-front faults transition into right-lateral shear zones as they pass through southern Pahranagat Valley and become normal faults on the eastern side of the Sheep Range. Water-level elevation data from the few wells and springs in Delamar Valley and the adjacent areas of southern Pahranagat and northern Coyote Spring valleys indicate a hydraulic gradient to the southwest and south along the shear zone and into northern Coyote Spring Valley.

Section 3.0

SE ROA 11932



3-10

Section 3.0

4.0 SUMMARY AND CONCLUSIONS

Geologic and geophysical data and information was assembled and used to describe the geologic framework of Cave, Dry Lake, and Delamar valleys. The resultant framework is presented on Plate 1 of this report as a surficial geology map and a series of geologic cross sections. A groundwater-site inventory was completed to identify groundwater sites within the three basins, and included compiling relevant hydrologic data for the purpose of developing a conceptualization of groundwater flow within the basins. The results and conclusions of this effort is summarized for each basin in the following sections:

4.1 Cave Valley

Geology

- In the east, Cave Valley is bounded by the southern extension of the Schell Creek Range which is comprised of an east-dipping and heavily faulted sequence of Precambrian to Tertiary rocks. The dominant fault is on the western flank of the range, and in the northeast portion of the valley, the range is cored by Precambrian to Cambrian quartzite.
- In the west, Cave Valley is bounded by the Egan Range, which is a complexly faulted, east-dipping horst comprised of Cambrian to Permian rocks overlain by Tertiary volcanic rocks.
- Cave Valley has been effectively partitioned into two sub-basins by the northeast-striking
 oblique-slip Shingle Pass Fault, which has displaced a part of the Egan Range, forming an
 east-dipping fault block that extends northeast across and underneath the valley where it
 terminates against the west range-front fault of the Schell Creek Range. This fault block
 contains the Mississippian Chainman Shale formation which, in this area, is considered a
 confining unit because of its thickness and hydraulic properties.

Hydrogeology

- The northern sub-basin is relatively shallow as the depth to basement is typically no greater than 3,300 ft-bgs. The basin fill is comprised of mostly sands and gravels, and occasional cemented strata near the surface. Depth to water in the basin fill for this area ranges from 2 to 221 ft-bgs, with water-level elevations ranging from 6,896 to 6,119 ft-amsl. A single well drilled by SNWA to a depth of 1,212 ft-bgs has a depth to water of 1,051 ft-bgs and water-level elevation of 5,406 ft-amsl. This carbonate water-level is most likely reflective of the hydraulic head within the Shingle Pass Fault.
- The southern sub-basin is significantly deeper, with the depth to basement extending to almost 20,000 ft-bgs. The basin fill is comprised of mostly sands and gravels, but with greater clay

Section 4.0

SE ROA 11934



content than the northern sub-basin. Depth to water in the basin fill ranges from 8 to 327 ft-bgs, with water-level elevations ranging from 6,221 to 5,704 ft-amsl. Two wells are completed in carbonate rocks on the east and west side of the sub-basin. Depth to water in these wells ranges from 140 to 220 ft-bgs, with water level elevations ranging from 5,850 to 5,790 ft-amsl.

- Within the basin fill, there is a north to south hydraulic gradient and direction of groundwater flow. The hydraulic heads in the carbonate rocks appear to be structurally controlled by the fault block associated with the Shingle Pass Fault, as reflected by the water-level elevations north and south of the block. Flow through this fault block is unlikely because of the presence of the Mississippian Chainman Shale formation.
- Water-level data indicate a slight increasing trend since the early 1980s of about 10 ft during the period of record. Because there is virtually no groundwater development within the basin, this trend is most likely reflective of the natural variability of the groundwater system.

Interbasin Flow

- Inter-basin flow associated with Cave Valley is limited to a small amount of outflow through the Shingle Pass Fault zone to the southern third of White River Valley, and outflow from the southern portion of the valley to northern Pahroc Valley. These outflows are estimated to be about 4,000 afy and 9,400 afy, respectively.
- Inflow from southern Steptoe Valley is unlikely due to the lithology and structure of the geologic framework underlying northern Cave Valley. It is also likely that a groundwater divide between northern Cave Valley and southern Steptoe Valley exists because groundwater recharge occurs in this area. However, the available water-level data are insufficient to verify this conclusion.

4.2 Dry Lake and Delamar Valleys

Dry Lake and Delamar valleys can be considered one basin from the geologic and hydrologic perspective. This is supported by the geologic and geophysical data compiled to develop the geologic framework, and hydraulic data used to conceptualize groundwater flow within the basins.

Geology

4-2

- To the west, Dry Lake and Delamar valleys are bounded by the North Pahroc Range and the South Pahroc Range, respectively. The North Pahroc Range is a west-dipping horst consisting of upper Paleozoic rocks overlain by Tertiary volcanic rocks. The South Pahroc Range consists of a series of west-dipping volcanic rocks, which extend southward to southern Delamar Valley where they are terminated by the east-northeast trending Pahranagat Shear Zone.
- To the east, Dry Lake and Delamar valleys are bounded from north to south by the Fairview, Bristol, Highland, and Chief Ranges, and the Delamar Mountains to the south. To the north, the Fairview Range consists of Devonian to Pennsylvanian carbonate rocks, which are

Section 4.0

interrupted by the Indian Peak Caldera Complex. To the south, the Bristol, Highland and Chief Ranges consist of Cambrian carbonate rocks underlain by Precambrian to Cambrian quartzites. Further south, the Delamar Mountains consist of east-dipping Late Proterozoic to Cambrian rocks and Tertiary volcanic rocks associated with the Caliente and Kane Springs Wash caldera complexes.

• The geometry of Dry Lake and Delamar valleys was defined, in part, by gravity surveys and depth to basement estimates which indicate the presence of low saddles to the north adjoining northern Dry Lake Valley (a.k.a. Muleshoe Valley) and to the south adjoining Delamar Valley. Two grabens are present in Dry Lake Valley, one along the margins of the valley and another along the central axis of the valley. The depth to basement in these areas are estimated to range from about 3,300 to almost 10,000 ft-bgs. In southwest Delamar Valley, a bowl-shaped basin is inferred, and the maximum depth to basement is estimated to be almost 7,000 ft-bgs, with much of the area only about 5,000 ft-bgs.

Hydrogeology

- The basin fill of Dry Lake and Delamar Valleys consists primarily of sands and gravels, with
 increasing clay content and thicker sequences of volcanic tuffs to the south.
- Depth to water in Dry Lake and Delamar Valleys ranges from 3 to 1,316 ft-bgs, and deepens to the south. Water-level elevations range from 6,630 to 3,486 ft-amsl and indicate a north to south gradient and direction of groundwater flow in both the basin fill and carbonate aquifer. This conceptualization of flow is supported by the data and is consistent with previous investigations by Eakin (1966), Scott et al. (1971) and Harrill et al. (1988).

Interbasin Flow

- Interbasin flow is limited by the geologic framework and occurs as a minor amount of inflow from northern Pahroc Valley to Dry Lake Valley, and as outflow from Delamar Valley to Coyote Spring Valley.
- The location of inflow is coincident with a series of northwest-trending right-lateral faults that form the boundary between southern Cave Valley, northern Pahroc Valley, and northern Dry Lake Valley. The inflow is estimated to be about 2,000 afy (Appendix D of Part A).
- Groundwater flows from Dry Lake Valley to Delamar Valley where flow is controlled by the Caliente and Kane Springs Wash caldera complexes to the east, and the South Pahroc Range and associated range-front faults to the west. The geologic framework in this area precludes flow to Pahranagat Valley to the west and Lake and Patterson Valleys to the east.
- Interbasin outflow from Delamar occurs along and across the Pahranagat Shear Zone and into northern Coyote Spring Valley and possibly the very southern part of Pahranagat Valley. This outflow is estimated to be about 24,000 afy and is comprised of the inflow from northern Pahroc Valley and locally-derived recharge generated from within hydrographic area boundaries.

Section 4.0

SE ROA 11936



4-4

Section 4.0

SE ROA 11937

JA_4699

5.0 REFERENCES

- Dixon, G.L., Rowley, P.D., Burns, A.G., Watrus, J.M., and Ekren, E.B., 2007, Geology of White Pine and Lincoln counties and adjacent areas, Nevada and Utah—The geologic framework of regional groundwater flow systems: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. HAM-ED-0001.
- Eakin, T.E., 1966, A regional interbasin groundwater system in the White River area, southeastern Nevada: Water Resources Research, Vol. 2, No. 2, p. 251–271.
- Eastman, H.S., 2007a, Geologic Data Analysis Report for Monitor Well 182W906M in Delamar Valley: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. RDS-ED-0001.
- Eastman, H.S., 2007b, Geologic Data Analysis Report for Monitor Well 182M-1 in Delamar Valley: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. RDS-ED-0002.
- Ekren, E.B., Bucknam, R.C., Carr, W.J., Dixon, G.L., and Quinlivan, W.D., 1976, East-trending structural lineaments in central Nevada: U.S. Geological Survey Professional Paper 986, 16 p.
- Ekren, E.B., Orkild, P.P., Sargent, K.A., and Dixon, G.L., 1977, Geologic map of Tertiary rocks, Lincoln County, Nevada: U.S. Geological Survey Miscellaneous Investigations Series Map I-1041, scale 1:250,000.
- Harrill, J.R., Gates, J.S., and Thomas, J.M., 1988, Major ground-water flow systems in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Investigations Atlas HA-694-C, 4 p., scale 1:1,000,000, two sheets.
- Hess, R., 2004, Nevada oil and gas well database: Nevada Bureau of Mines and Geology Open-File Report 04-1, PDF file, http://www.nbmg.unr.edu/lists/oil/oil.htm.
- Luzier, J.E., 2003, Dry Lake and Delamar Valleys, Nevada Groundwater Hydrology and Development Potential: Luzier Hydrosciences, 32 p.
- Mankinen, E.A., Roberts, C.W., McKee, E.H., Chuchel, B.A., and Morin, R.L., 2007, Geophysical data from Spring Valley to Delamar Valley, east-central Nevada: U.S. Geological Survey Open File Report 2007-1190 [http://pubs.usgs.gov/of/2007/1190/].

NDWR, see Nevada Division of Water Resources.

Section 5.0

SE ROA 11938



- Nevada Division of Water Resources, 2004, Well log database available on March, 2004 at website http://water.nv.gov/Engineering/wlog/wlog.cfm.
- Page, W.R., and Ekren, E.B., 1995, Preliminary geologic map of the Bristol Well quadrangle, Lincoln County, Nevada: U.S. Geological Survey Open-File Report 95-580, 27 p.
- Rowley, P.D., 1998, Cenozoic transverse zones and igneous belts in the Great Basin, western United States--their tectonic and economic implications, *in* Faulds, J.E., and Stewart, J.H., eds., Accommodation zones and transfer zones--The regional segmentation of the Basin and Range province: Geological Society of America Special Paper 323, p. 195-228.
- Rowley, P.D., and Dixon, G.L., 2001, The Cenozoic evolution of the Great Basin area, U.S.A.—New interpretations based on regional geologic mapping, *in* Erskine, M.C., Faulds, J.E., Bartley, J.M., and Rowley, P.D., eds., The geologic transition, High Plateaus to Great Basin—A symposium and field guide (The Mackin Volume): Utah Geological Association and Pacific Section of the American Association of Petroleum Geologists: Utah Geological Association Publication 30, p. 169-188.
- Rowley, P.D., Nealey, L.D., Unruh, D.M., Snee, L.W., Mehnert, H.H., Anderson, R.E., and Gromme, C.S., 1995, Stratigraphy of Miocene ash-flow tuffs in and near the Caliente caldera complex, southeastern Nevada and southwestern Utah, *in* Scott, R.B., and Swadley, W.C., eds., Geologic studies in the Basin and Range—Colorado Plateau transition in southeastern Nevada, southwestern Utah, and northwestern Arizona, 1992: U.S. Geological Survey Bulletin 2056, p. 43-88.
- Rowley, P.D., Shroba, R.R., Simonds, F.W., Burke, K.J., Axen, G.J., and Olmore, S.D., 1994, Geologic map of the Chief Mountain quadrangle, Lincoln County, Nevada: U.S. Geological Survey Geologic Quadrangle Map GQ-1731, scale 1:24,000.
- Scheirer, D.S., 2005, Gravity studies of Cave, Dry Lake, and Delamar Valleys, east-central Nevada: U.S. Geological Survey Open-File Report 2005-1339, 27 p.
- Scott, R.B., Rowley, P.D., Snee, L.W., Anderson, R.E., Harding, A.E., Unruh, D.M., Nealey, L.D., Hudson, M.R., Swadley, W.C., and Ferris, D.E., 1996, Synchronous Oligocene and Miocene extension and magmatism in the vicinity of caldera complexes in southeastern Nevada, *in* Thompson, R.A., Hudson, M.R., and Pillmore, C.L., eds., Geologic excursions to the Rocky Mountains and beyond, Field Trip Guidebook for the 1996 annual meeting, Geological Society of America, Denver, Colorado, October 28-31: Colorado Geological Survey Special Publication 44, 36 p. (CD-ROM).
- Scott, R.B., Gromme, C.S., Best, M.G., Rosenbaum, J.G., and Hudson, M.R., 1995, Stratigraphic relationships of Tertiary volcanic rocks in central Lincoln County, southeastern Nevada, *in* Scott, R.B., and Swadley, W.C., eds., Geologic studies in the Basin and Range—Colorado Plateau transition in southeastern Nevada, southwestern Utah, and northwestern Arizona, 1992: U.S. Geological Survey Bulletin 2056, p. 5-42.

5-2

Section 5.0
- Scott, B.R., Smales, T.J., Rush, F.E., and Van Denburgh, A.S., 1971, Water for Nevada— Report 3– Nevada's Water Resources: Nevada Department of Conservation and Natural Resources, 87 p.
- Swadley, W.C., and Rowley, P.D., 1994, Geologic map of the Pahroc Spring SE quadrangle, Lincoln County, Nevada: U.S. Geological Survey Geologic Quadrangle Map GQ-1752, scale 1:24,000.

USGS, see U.S. Geological Survey.

- U.S. Geological Survey, 2001 Digital elevation model (DEM), Earth Resources Observation and Science (EROS), U.S. Geological Survey, accessed at http://edu.usgs.gov/guides/dem.html.
- U.S. Geological Survey, 2004, National Water Information System (NWIS Web) data accessed at website http://waterdata.usgs.gov/nwis/.
- U.S. Geological Survey, 2007, National Water Information System (NWIS Web) data accessed at website http://waterdata.usgs.gov/nwis/.
- Van Loenen, R.E., 1987, Geologic map of the Mount Grafton wilderness study area, Lincoln and White Pine Counties, Nevada: U.S. Geological Survey Miscellaneous Field Studies Map MF-1938, scale 1:50,000.

Section 5.0

5-3



This Page Intentionally Left Blank

Section 5.0

SE ROA 11941

5-4

Part B - Hydrogeology of Cave, Dry Lake, and Delamar Valleys

Appendix A

Hydrographs for Wells in Cave, Dry Lake, and Delamar Valleys



Historical Water-Level Elevations at 180 N07 E63 14BADB1 USAF (Map ID 180-05)







Historical Water-Level Elevations at 180 N07 E63 15DBAD1 USBLM (Map ID 180-08)



Historical Water-Level Elevations at 180 N08 E64 04ABDD1 USBLM (Map ID 180-14)

A-2



Figure A.1-5







SE ROA 11945

A-3









Historical Water-Level Elevations at 180 N10 E63 25A 1 (Map ID 180-20)

A-4

Appendix A



A-5

SE ROA 11947

JA_4709





Historical Water-Level Elevations at 181 N01 E62 24ABB 1 USBLM (Map ID 181-01)

A-6

Appendix A







Figure A.1-15





Appendix A SE ROA 11950





SE ROA 11951

A-9









SE ROA 11953

JA_4715





Historical Water-Level Elevations at 182W906M (Map ID 182-05)

A-12

Part B - Hydrogeology of Cave, Dry Lake, and Delamar Valleys

Plates



PLATE 1. GEOLOGY AND GEOLOGIC CROSS SECTIONS OF CAVE, DRY LAKE, AND DELAMAR VALLEYS



Part C

Water-Related Effects Analysis Related to Southern Nevada Water Authority Groundwater Applications in Cave, Dry Lake, and Delamar Valleys

Prepared by: Andrew G. Burns, James M. Watrus, and Gary L. Dixon

November 2007

SOUTHERN NEVADA WATER AUTHORITY Groundwater Resources Department Water Resources Division snwa.com

CONTENTS

List of	Figures	iii										
List of	Tables.	v										
List of	Acrony	ms and Abbreviations vii										
1.0	Introduction 1-1											
	1.1	Purpose and Scope 1-1										
2.0	Summa	ary of Water Rights 2-1										
	2.1 2.2 2.3 2.4	Cave Valley Permitted Rights.2-1Dry Lake Valley Permitted Rights2-1Delamar Valley Permitted Rights.2-3Domestic Water Rights.2-3										
3.0	Effects	Analysis										
	3.1 3.2 3.3	Locations of Interest3-1Selected Analytical Model3-43.2.1Theis Analysis3-43.2.2Implementation3-93.2.3Data Requirements3-9Groundwater Development Scenario3-93.3.1Conceptual Model3-103.3.2General Design Information3-103.3.3Aquifer Properties3-103.3.4Site Specific Information3-103.3.3.1.1Site Specific Information for the Carbonate-Rock Aquifer3-113.3.3.1.2Site Specific Information for the Alluvial Aquifer3-113.3.3.1.4Aquifer Property Values Applied to Theis3-133.3.4.1Cave Valley3-133.3.4.2Dry Lake Valley3-153.3.5L imitations3-16										
4.0	Monit	pring and Management										
5.0	4.1 4.2 4.3 4.4	Monitoring Objectives 4-1 Existing Monitoring Sites 4-1 Monitoring Frequency and Data Reporting 4-1 Resource Management 4-5 arv 5-1										
5.0	Julin											



CONTENTS (CONTINUED)

6.0	References
Appen	dix A - NDWR Hydrographic Abstracts for Cave, Dry Lake, and Delamar Valleys
A.1.0	Water Rights
Appen	dix B - Application Points of Diversion Figures
B.1.0	IntroductionB-1
Appen	dix C - Spring Discharge and Temperature Measurements
C.1.0	Introduction
C.2.0	References

ii

Part C - Water-Related Effects Analysis

FIGU Numbe	RES ER TITLE PAGE
2-1	Points of Diversion for Permitted Water Rights in Cave Valley 2-2
2-2	Points of Diversion for Permitted Water Rights in Dry Lake Valley 2-4
2-3	Points of Diversion for Permitted Water Rights in Delamar Valley 2-5
3-1	Locations of Selected Water Rights and Environmental Areas of Concern for Cave Valley
3-2	Locations of Selected Water Rights and Environmental Areas of Concern for Dry Lake Valley
3-3	Locations of Selected Water Rights and Environmental Areas of Concern for Delamar Valley 3-7
3-4	Site-Specific Aquifer Test Locations 3-12
4-1	Existing Monitoring Network 4-4
B.1-1	Area Surrounding Application Point of Diversion 53987 in Cave Valley
B.1-2	Area Surrounding Application Point of Diversion 53988 in Cave Valley
B.1-3	Area Surrounding Application Point of Diversion 53989 in Dry Lake Valley B-4
B.1-4	Area Surrounding Application Point of Diversion 53990 in Dry Lake Valley B-5
B.1-5	Area Surrounding Application Point of Diversion 53991 in Delamar Valley B-6
B.1-6	Area Surrounding Application Point of Diversion 53992 in Delamar Valley





This Page Intentionally Left Blank

iv

SE ROA 11961

JA_4723

Part C - Water-Related Effects Analysis

TABI	LES
NUMB	ER TITLE PAGE
3-1	SNWA Application Points of Diversion and Selected Permitted Underground and Spring Water Rights, Listed by Valley
3-2	Environmental Areas of Concern, Listed by Valley
3-3	Aquifer Properties used in Theis Analysis 3-13
3-4	Simulated Drawdowns at Selected Permitted Underground and Spring Water Rights and Environmental Areas of Concern
4-1	Monitoring Sites in Cave, Dry Lake, and Delamar Valleys and Vicinity 4-2
A.1-1	NDWR Hydrographic Abstract for Cave Valley
A.1-2	NDWR Hydrographic Abstract for Dry Lake Valley
A.1-3	NDWR Hydrographic Abstract for Delamar Valley
C.1-1	Discharge and Temperature Measurements of Selected Springs in Cave, Dry Lake, and Delamar Valleys





This Page Intentionally Left Blank

Part C - Water-Related Effects Analysis

ACRONYMS

BLM	Bureau of Land Management
NAD83	North American Datum of 1983
NDWR	Nevada Division of Water Resources
SNWA	Southern Nevada Water Authority
UTM	Universal Transverse Mercator

ABBREVIATIONS

afy	acre-feet per year
amsl	above mean sea level
bgs	below ground surface
cfs	cubic feet per second
ft	foot
gpm	gallons per minute
m	meter
mi	mile





This Page Intentionally Left Blank

viii

SE ROA 11965

JA_4727

1.0 INTRODUCTION

Part C of this report provides a description of the existing water rights in Cave, Dry Lake, and Delamar valleys, an analysis of the potential effects related to the Southern Nevada Water Authority's (SNWA) applications 53987 through 53992, inclusive, and a monitoring and management plan developed to monitor and manage any effects. The purpose and scope of this part of the report is described in the following section.

1.1 Purpose and Scope

The purpose of the work described in this document is to present an analysis of the potential water-related effects associated with the SNWA applications in Cave, Dry Lake, and Delamar valleys, including a description of existing water rights, an analysis of the potential effects, and a description of a program that would be implemented to monitor and manage effects as they occur. The focus of these analyses is primarily limited to Cave, Dry Lake, and Delamar valleys, although some of the data that were used in the analyses are from locations in adjacent valleys.

The scope of the work involves (1) the compilation and presentation of existing water rights within Cave, Dry Lake, and Delamar valleys, with an emphasis on the underground water rights that will be analyzed as part of the effects analysis; (2) a description of the simplified effects analysis that addresses the proposed pumping associated with the SNWA applications; (3) a discussion of the monitoring and management program associated with development of the groundwater resources in Cave, Dry Lake, and Delamar valleys.

Section 1.0





This Page Intentionally Left Blank

Section 1.0 SE ROA 11967

2.0 SUMMARY OF WATER RIGHTS

This section provides information on the water rights within Cave, Dry Lake, and Delamar valleys, with emphasis on currently committed underground water rights. The following discussion is based on hydrographic abstracts (i.e., formalized descriptions of water-right applications) obtained from the Nevada Division of Water Resources (NDWR) on August 13, 2007 (NDWR, 2007a) and included in Appendix A.

2.1 Cave Valley Permitted Rights

Table A.1-1 contains the hydrographic abstracts for Cave Valley. The NDWR Hydrographic Basin Summary (NDWR, 2007b) indicates 46.58 afy of committed duties for underground rights in Cave Valley.

The majority of the permitted water rights in Cave Valley are associated with springs located primarily in the mountain blocks on the eastern and western sides of the valley; however, there are rights associated with Haggerty and Silver creeks, emanating from the Egan Range, and with North and Sheep creeks, flowing from the western slope of the Schell Creek Range. These creeks flow toward the valley floor on the north end of the valley, enter Cave Valley Wash, and proceed to a terminal depression located at the southern end of the valley.

A majority of the permitted rights in Cave Valley are for stock water, with ten rights for irrigation and one right for domestic use. There are eight underground rights, with each having a manner of use of stock water. There are three reserved rights on springs in the southwest portion of the valley. The reserved rights are held by the Bureau of Land Management (BLM) for stock water.

Figure 2-1 depicts (1) the locations of the points of diversion for permitted water rights in Cave Valley, with each point symbolized by source and labeled with application number and manner of use; and (2) the location of the two SNWA application points of diversion.

2.2 Dry Lake Valley Permitted Rights

Table A.1-2 contains the hydrographic abstracts for Dry Lake Valley. The NDWR Hydrographic Basin Summary (NDWR, 2007b) indicates 56.56 afy of committed duties for underground rights in Dry Lake Valley.

The majority of the permitted rights in Dry Lake Valley are associated with springs located primarily in the mountain block on the eastern and western sides of the valley; however, there are rights associated with Black Canyon, Fairview, and Porphry washes that flow intermittently from the ranges on the eastern edge of the basin. These creeks flow toward the valley floor on the north end of the

Section 2.0

SE ROA 11968

2-1





Figure 2-1 Points of Diversion for Permitted Water Rights in Cave Valley

2-2	Section 2.0
1000	SE ROA 11969

valley and enter Coyote Wash, and proceed to a terminal depression located at the southern end of the valley.

A majority of all the permitted rights in Dry Lake Valley are for stock water, with four rights for irrigation and one right for mining and milling. There are six underground rights; five of which have a manner of use of stock water, and the sixth is for mining and milling. There are four reserved rights on springs. The reserved rights are held by BLM for stock water and other uses.

Figure 2-2 depicts (1) the locations of the points of diversion for permitted water rights in Dry Lake Valley, with each point symbolized by source and labeled with application number and manner of use; and (2) the location of the two SNWA application points of diversion.

2.3 Delamar Valley Permitted Rights

Table A.1-3 contains the hydrographic abstracts for Delamar Valley. The NDWR Hydrographic Basin Summary (NDWR, 2007b) indicates only one committed duty for underground water rights of 7.24 afy in Delamar Valley.

The majority of the permitted rights in Delamar Valley are from springs located primarily in the mountain block on the eastern and western sides of the valley; however, there are rights associated with several reservoirs within the valley. A majority of the rights are for stock water, with one right each for domestic, irrigation, and mining and milling. There is one underground water right with a manner of use of stock water. There is one reserved right on a spring. The reserved right is held by BLM for stock water.

Figure 2-3 depicts (1) the locations of the points of diversion for permitted water rights in Delamar Valley, with each point symbolized by source and labeled with application number and manner of use; and (2) the location of the two SNWA application points of diversion.

2.4 Domestic Water Rights

A domestic well is defined as one well that serves one home (NDWR, 2007c). These wells are exempt from the water-right permitting process when the pumpage does not exceed a daily maximum of 1,800 gallons and the water cannot be furnished by an entity such as a water district or municipality (NDWR, 2007c). This means that domestic wells may not be identified in the hydrographic abstracts of a particular valley. To assess whether domestic wells were located within 2,500 ft of a SNWA application point of diversion, figures were created showing a 2,500-ft buffer around those points of diversion overlain on National Agriculture Imagery Program images (Appendix B). No private lands or homes were visible within the 2,500-ft buffered region.



Section 2.0





Figure 2-2

Points of Diversion for Permitted Water Rights in Dry Lake Valley

Section 2.0 SE ROA 11971



Figure 2-3 Points of Diversion for Permitted Water Rights in Delamar Valley

Section 2.0

SE ROA 11972

2-5



This Page Intentionally Left Blank

Section 2.0 SE ROA 11973

3.0 EFFECTS ANALYSIS

The objective of this effects analysis is to quantitatively estimate the water-level drawdowns at points of diversion associated with currently permitted underground and spring water rights (Section 2.0) and at environmental areas of concern that may result from continuously pumping the wells associated with SNWA's water-right applications in Cave, Dry Lake, and Delamar valleys. The quantitative analysis was performed using a simplified analytical model and a pumping scenario that was developed to evaluate the potential effects associated with pumping SNWA's requested application volume from the application points of diversion. The locations of interest, analytical model, model inputs, and model results are described in the following sections.

3.1 Locations of Interest

Locations of interest to SNWA's groundwater development project in Cave, Dry Lake, and Delamar valleys are the points of diversion associated with currently permitted underground and spring water rights in those valleys as well as environmental areas of concern as identified in SNWA (2007).

Many of the permitted water right locations were excluded from the analysis as a result of their location in relation to the SNWA application points of diversion or priority date. Nearly all of the spring rights in these valleys occur high in the mountain block and their waters are derived from local precipitation. It is therefore assumed, that because these springs are likely not in hydraulic continuity with either the alluvial or carbonate-rock aquifer systems, it is highly unlikely that they would be impacted from pumping on the valley floor and are therefore excluded from the effects analysis. Table 3-1 lists the location information for the selected underground and spring water rights that exist on the valley floors as well as the location information for the SNWA application points of diversion.

Table 3-2 lists the location information for the environmental areas of concern as identified in SNWA (2007). Except for Grassy Spring in Delamar Valley, all of the environmental areas of concern are either located in the mountain block, or are separated from the proposed pumping locations by potential physical barriers (as in the case of unnamed spring at Parker Station). The local nature of these springs is supported by the variable discharge and cold temperatures of the springs (Appendix C). Therefore, these locations are highly unlikely to be effected by SNWA's proposed pumping on the valley floor. Grassy Spring while located at the edge of the mountain block and approximately 7.26 miles from the nearest application, is located along the same fault structure as SNWA application 53992, and therefore this location will be analyzed as part of this work.

The locations of these points of diversion and environmental areas of concern are depicted in Figures 3-1, 3-2, and 3-3.

Section 3.0

3-1

SNWA Application Points of Diversion and Selected Permitted Underground and Spring Water Rights, Listed by Valley (Page 1 of 2) Table 3-1

int of Drillers on to Drillers est Log sed Number			4	1		26	53		1	1		10	44	48 5511	64	12	.85 39357	83	77
Dista from Po	Divers Near Propo		i	i		2.5	3.5		1	1		35.	.9	10.	19,	17.	22	21	18
Distance from Point of Diversion to Nearest Proposed Pumping Well (ft)			1	1		31,516	18,650		1	ł		185,304	33,984	55,341	103,688	93,517	120,623	115,256	97 367
Application Number for Nearest Proposed Pumping Well		Vells	1	Í.	Vater Rights	53987	53987	n Wells	ł	ł	er Rights	53990	53990	53990	53990	53990	53990	53990	53000
Manner of Use		Production /	Municipal	Municipal	and Spring V	Stock	Stock	ed Productio	Municipal	Municipal	d Spring Wat	Stock	Stock	Stock	Stock	Stock	Stock	Stock	Stock
Elevation (ft-amsI)		A Proposed	5,978	6,047	nderground	6,202	6,253	WA Propos	4,791	4,811	erground an	5,667	4,767	4,706	4,975	4,906	5,072	5,442	E ODR
ion	UTM Northing ^a (m)	e Valley SNW	4,248,279	4,257,220	Cave Valley Permitted U	4,238,721	4,244,266	ake Valley SN	4,179,072	4,184,836	ce Valley Und	4,241,005	4,195,177	4,201,362	4,215,376	4,212,756	4,220,143	4,219,938	TOA ANT
Loca	UTM Easting ^a (m)	Cave	684,117	683,439		683,156	680,091	Dry L	672,793	701,088	Dry Lal	695,162	700,487	601,709	692,956	695,348	690,834	702,490	R08 577
Source			Underground	Underground		Underground	Underground		Underground	Underground		Underground	Spring	Underground	Underground	Spring	Underground	Underground	Carina
Priority Date			10/17/1989	10/17/1989		2/27/1922	6/14/1925		10/17/1989	10/17/1989		1/5/1920	9/17/1928	4/26/1960	8/18/1978	8/18/1978	8/18/1978	8/18/1978	41414000
Application Number			53987	53988		6638	7397		53989	53990		5936	8698	18756	35770	35771	35773	35774	UCOCUIN

Section 3.0

SE ROA 11975

Southern Nevada Water Authority - Water Resources Division
SNWA Application Points of Diversion and Selected Permitted Underground and Spring Water Rights, Listed by Valley (Page 2 of 2) Table 3-1

			Loca	ation			Application	Distance from Point of	Distance from Point of	
Application Number	Priority Date	Source	UTM Easting ^a (m)	UTM Northing ^a (m)	Elevation (ft-amsI)	Manner of Use	Number for Nearest Proposed Pumping Well	Diversion to Nearest Proposed Pumping Well (ft)	Diversion to Nearest Proposed Pumping Well (mi)	Drillers Log Numbe
V03840	1/1/1890	Spring	698,549	4,217,584	5,173	Stock	53990	107,763	20.41	1
			Delai	mar Valley SN	WA Propose	d Production	n Wells			
53991	10/17/1989	Underground	683,444	4,157,168	4,913	Municipal	ł	1	+	ł
53992	10/17/1989	Underground	694,792	4,145,180	5,329	Municipal	ł	1	-	1
			Delam	ar Valley Unde	erground and	d Spring Wat	er Rights			
4620	10/8/1917	Spring	693,455	4,139,886	5,024	Stock	53992	17,914	3.39	ł
4622	10/8/1917	Spring	683,947	4,153,060	4,914	Stock	53991	13,578	2.57	1
4894	2/7/1918	Spring	693,098	4,153,581	5,268	Domestic	53992	28,117	5.33	1
4973	3/21/1918	Spring	683,947	4,153,060	4,914	Stock	53991	13,578	2.57	F
5782	9/29/1919	Spring	689,434	4,156,310	4,961	Stock	53991	19,853	3.76	I
5783	9/29/1919	Spring	689,434	4,156,310	4,961	Stock	53991	19,853	3.76	1
10189	12/3/1937	Spring	694,609	4,157,153	5,666	Stock	53991	36,631	6.94	1
11167	9/14/1944	Spring	693,546	4,135,873	5,249	Stock	53992	30,807	5.83	ł
52112	1/1/1880	Spring	600'969	4,157,163	5,766	Stock	53991	37,943	7.19	1
V01400	1/1/1900	Spring	694,609	4,157,153	5,666	Stock	53991	36,631	6.94	ł
V01520	4/1/1900	Spring	693,546	4,135,873	5,249	Stock	53992	30,807	5.83	ţ
V01654	1/1/1900	Spring	688,343	4,158,084	4,949	Stock	53991	16,351	3.10	ł

Part C - Water-Related Effects Analysis

Section 3.0



	Loca	ation		Application	Distance from	Distance from
Name	UTM Easting ^a (m)	UTM Northing ^a (m)	Elevation (ft-amsl)	Number of Nearest Proposed Pumping Well	Nearest Proposed Pumping Well (ft)	Nearest Proposed Pumping Well (mi)
		1	Cave Valley	/		
Unnamed Spring at Parker Station	687,827	4,282,583	6,514	53988	84,448	15.99
Cave Spring	691,761	4,279,249	6,488	53988	77,259	14.63
			Dry Lake Val	ley		
Meloy Spring	700,888	4,236,201	6,178	53990	168,522	31.92
Fence Spring	700,139	4,228,221	6,277	53990	142,373	26.96
Bailey Spring	699,080	4,227,795	6,090	53990	141,096	26.72
Covote Spring	687,693	4,211,513	5,224	53990	97,937	18.55
	1000 00 00 00 00 00 00 00 00 00 00 00 00		Delamar Val	ey		Sector and
Grassy Spring	695,124	4,157,193	5,783	53991	38,317	7.26

Table 3-2 Environmental Areas of Concern, Listed by Valley

Source: Environmental Areas of Concern Identified in SNWA, 2007 ^aNorth American Datum of 1983, Zone 11N

3.2 Selected Analytical Model

The local effects of SNWA's proposed production wells were evaluated using the Theis equation as implemented in RockWorks (RockWare, 2004). Descriptions of the Theis equation, data requirements, the implementation software, and the presentation of results are provided in the following subsections.

3.2.1 Theis Analysis

The local effects of pumping were evaluated with the Theis equation (Theis, 1935) along with the principle of superposition. Drawdowns caused by each pumping well at a given point were calculated separately then summed (superposed) to estimate the total drawdown at that location.

The following solution to the Theis equation was used in the drawdown calculations:

$$h - h_0 = Q/4\pi \bullet T \bullet [W(u)]$$

W(u) = -0.5772 - ln(u) + u - u²/(2×2!) + u³/(3×3!)+...+u²⁰/(20×20!) (3-1)
u = r²S/4Tt

where,

h = Hydraulic head at time t (ft)

Section 3.0 SE ROA 11977







SE ROA 11978





Figure 3-2 Locations of Selected Water Rights and Environmental Areas of Concern for Dry Lake Valley

Section 3.0



Figure 3-3 Locations of Selected Water Rights and Environmental Areas of Concern for Delamar Valley

Section 3.0

SE ROA 11980



Southern Nevada Water Authority - Water Resources Division

- h_0 = Hydraulic head at time zero (ft)
- $Q = Pumping rate (ft^3/day)$
- $T = Transmissivity (ft^2/day)$
- W(u) = Well function
- r = Well radius or interwell distance (ft)
- S = Coefficient of Storage
- t = Time since pumping started (days)

Assumptions associated with the Theis equation are as follows:

- 1. The aquifer has infinite areal extent.
- 2. The aquifer is homogeneous, isotropic, and of uniform thickness.
- 3. The pumping well is fully or partially penetrating.
- 4. Flow to the pumping well is horizontal when the pumping well is fully penetrating.
- 5. The aquifer is confined.
- 6. The flow is unsteady.
- 7. Water is released instantaneously from storage with decline of hydraulic head.
- 8. The diameter of the pumping well is very small so that storage in the well can be neglected.

Other assumptions that are inherent in the Theis equation include: (1) no regional drawdown, (2) no regional gradient, (3) no explicit barriers to flow other than those subsequently specified, (4) no recharge from precipitation, (5) no subsurface inflow or outflow, and (6) no induced recharge. Induced recharge includes stream depletions, reduction in spring flow, and reduction in evapotranspiration. Drawdown at a given location that is caused by several pumping wells is calculated by superposing drawdowns caused by each pumping well. The effect of a flow barrier may be simulated using image wells.

Uncertainties in these calculations may result from the violation of one or more of the Theis equation assumptions. For instance, homogeneity and isotropic characteristics are seldom found in nature, recharge does occur, the full thickness of the aquifer is not known and the well does not fully penetrate the aquifer, a water-level gradient exists, and delayed-yield responses occur. Another common use of the Theis equation is to simulate the response of an unconfined aquifer, which violates one of the main assumptions of the equation. The Theis equation, however, provides good estimates of drawdown in an unconfined aquifer provided that the simulated drawdown is less than 20 percent of the thickness of the aquifer.

Due to the limitations discussed above and further in Section 3.3.5, the use of Theis equation is an approach that can only provide estimates of the potential drawdowns. Geologic features not accounted for in the Theis equation could result in conservative results for certain areas and nonconservative for others. For example, in the case of a single production well, the presence of a barrier would yield larger drawdowns on the production well's side of the barrier, and lesser or no drawdown on the other side. As described in Section 4.0, drawdown will be monitored at several monitor well sites to manage the resource and minimize the effects of the proposed production wells.

3-8

Section 3.0

3.2.2 Implementation

RockWorks is a software application for subsurface data analysis that includes several tools that can be used to develop maps, logs, cross sections, fence diagrams, solid models, and volumetrics. RockWorks can utilize various geologic and hydrologic data. The hydrology module allows for the construction of water table or drawdown maps. RockWorks uses the Theis equation and the principle of superposition to calculate drawdowns due to pumping from multiple wells at specified grid points.

3.2.3 Data Requirements

The relevant information needed in the Theis calculations as implemented by RockWorks for each well location is as follows:

General Information

- Production well radius in feet
- · Production well open interval length in feet
- · Pumping period in days

Aquifer Properties

- Transmissivity in gpd/ft
- Storage coefficient or specific yield

Production Well Information

- Well ID
- Well location coordinates in feet
- Pre-pumping water-level elevation in feet (set to 0 for drawdown calculations)

Pumping Schedule Information

- Pumping rate in gpm
- Duration in days

The production well information and the pumping schedule are known by design. However, the aquifer properties used in this analysis are estimates based on the best available test data. This information is input to RockWorks in table form. In addition, the computational grid must be specified with minimum and maximum coordinates and spacing in feet. This information allows RockWorks to calculate a drawdown value at each grid node for each pumping well. All grids are then superposed, and the drawdowns at each grid point are summed to derive a total drawdown resulting from all pumping wells.

3.3 Groundwater Development Scenario

This section describes the data and information necessary to evaluate the effects of the proposed SNWA groundwater development in Cave, Dry Lake, and Delamar valleys using the Theis equation.



3.3.1 Conceptual Model

The conceptual model applied to the Theis analysis involved two separate analyses one for Cave Valley and a combined analysis for Dry Lake and Delamar valleys. This is consistent with the interpretation for Dry Lake and Delamar valleys that these two hydrographic areas are connected geologically and hydrologically, making them one contiguous basin.

Simulated flow barriers were placed along the eastern and western margins of each of southern Cave, Dry Lake, and Delamar valleys to simulate the lack of interbasin flows in those directions. This interpretation is consistent with the flow interpretations presented in Section 2.0 and 3.0 of Part B of this report. The flow barriers have been simulated in the analysis with the use of image wells. Image wells are imaginary discharging wells, used in the analysis, that are placed across the flow boundaries at the same distance from, and perpendicular to, the boundaries as the application points of diversion. The resulting drawdown at any point within the boundaries is the algebraic sum of the drawdowns produced at that point by the application point of diversion and its image. In this case, the use of image wells results in greater drawdowns within the valley than would be observed without the simulated flow barriers.

While the alluvial and carbonate-rock aquifers were evaluated separately in this analysis, the drawdowns simulated in each were summed for each location of interest. This is a conservative assumption because it is unlikely that pumping an alluvial point of diversion would have much effect on a location of interest sourced in the carbonate-rock aquifer, and visa versa.

3.3.2 General Design Information

The pumping scenario developed for the analysis describes the water-right applications for Cave, Dry Lake, and Delamar valleys. The objective of the scenario is to pump the application volume from the points of diversion and analyze the effects on water levels over a 75-year period. The total pumping period was designed to match the expected life of the equipment and infrastructure. This scenario represents the application points of diversion and therefore, the locations of the points of diversion are fixed at the locations requested in the applications (Table 3-1). The volumes of water to be withdrawn are also fixed under this scenario as the individual application volumes. In this scenario, all of the wells are screened with an open interval of 1,000 ft. It is also assumed for the Theis analysis that the well radii will be 0.83 ft.

3.3.3 Aquifer Properties

3-10

Aquifer properties were derived from the available site-specific data as well as published data for the regional system.

3.3.3.1 Site Specific Information

Site-specific data for the valley-fill and carbonate-rock aquifers were derived from aquifer tests conducted in support of the MX missile-siting investigation. In particular, two aquifer tests conducted in Dry Lake Valley provide useful information regarding the aquifer properties of both the

Section 3.0

alluvial and carbonate-rock aquifers. The results for these two tests have been presented in Ertec (1981a, Table 4-5, p. 45-46), Bunch and Harrill (1984, p. 115-119), Dettinger et al. (1995, Table 1, p. 13-16), and Belcher et al. (2001, hydraulic properties database).

3.3.3.1.1 Site Specific Information for the Carbonate-Rock Aquifer

A carbonate MX well, DL-DT-3, was drilled and tested in northwestern Dry Lake Valley during October through December, 1980 (Figure 3-4). Drill cuttings indicated the well was completed in the Guilmette Formation and/or Simonson Dolomite with an effective open interval from the depth-to-water at 853 ft-bgs to the total depth of the well at 2,395 ft-bgs (Ertec, 1981a, Table 4-9, p. 56). Results of the aquifer test conducted at that location indicated a transmissivity of 13,400 ft²/day (Ertec, 1981a, Table 4-9, p. 56). This was a single well test and no specific yield or storativity numbers were reported.

3.3.3.1.2 Site Specific Information for the Alluvial Aquifer

An alluvial MX well cluster, DL-TW-2 and DL-OW-2, was drilled and tested in southern Dry Lake Valley during January through April, 1980 (Figure 3-4). Drill cuttings and geophysical logging indicated the lithology of the boreholes was poorly sorted to well sorted gravels with less than 30 percent sand and just traces of silt and clay. No confining layers were identified within either borehole (Ertec, 1981b, p. 4-5). The data collected from the observation well were preferred by the authors of the Ertec report and those from the shallow piezometer were preferred over the deep because it was assumed that the shallow piezometer was better developed during air lifting (Ertec, 1981b, p. 30). The shallow peizometer has a perforated interval from 765 to 785 ft-bgs. The cited transmissivity was 3,400 ft²/day and the storativity value cited was the lower bound of the specific yield which was 0.013 (Ertec, 1981b, p. 30). These values may be effected by the limited tested interval.

3.3.3.1.3 Regional Information

Site specific aquifer-property information is very limited within the basins of east central Nevada. To augment that information, additional aquifer-property data were compiled from a much larger region that encompasses areas such as the Death Valley Regional Flow System where many more studies have been conducted. As most of the aquifer tests are not of long enough duration to derive storage parameters that are suitable for the analysis of pumping effects that may last over long durations, the available data were supplemented with literature data, particularly values derived from other studies within the region.

The range of specific yield used for the carbonate-rock aquifer is 0.01 to 0.05, with a 0.03 mid-range value. The minimum value of 0.01 is based on estimates made by Dettinger et al. (1995, p. 32) and used in several studies including the one conducted for Three-Lakes Valley South by SNWA (2005). The maximum value of 0.05 is the same as that used by Schaefer and Harrill (1995, p. 8) in their simulations of the effects of pumping potential production wells associated with the LVVWD water-right applications.

Section 3.0

SE ROA 11984



Figure 3-4 Site-Specific Aquifer Test Locations

Section 3.0 SE ROA 11985

3.3.3.1.4 Aquifer Property Values Applied to Theis

The site specific aquifer property information described in the previous sections was used in the simplified Theis analysis. The transmissivities reported for the carbonate and alluvial aquifers were first converted into hydraulic conductivities by dividing them by the effective open intervals for each well. The hydraulic conductivity values were then multiplied by the 1,000 ft open interval that was assumed for each of the proposed wells. The storativity/specific yield number used for the alluvial aquifer was the same as what was reported in Ertec (1981b, p. 30). There is no site specific information available for the storativity/specific yield of the carbonate-rock aquifer. Therefore, as described previously, an average of values for the regional carbonate-rock aquifer was used for this analysis (Table 3-3).

Aquifer	Calculated Transmissivity ft²/day	Source for Original Transmissivity Used	Storativity/ Specific Yield	Source for Storativity/ Specific Yield Used
Alluvial Aquifer	170,000	Site Specific Information from DL-OW-2	0.013ª	Site Specific Information from DL-OW-2
Carbonate-Rock Aquifer	8,690	Site Specific Information from DL-DT-3	0.03	Average of regional values

Table 3-3 Aquifer Properties used in Theis Analysis

^aThis value was reported as a lower bound (Ertec, 1981b)

3.3.4 Effects Analysis Results

The effects of the production wells associated with SNWA's water right applications in Cave, Dry Lake, and Delamar valleys calculated by the Theis equation are presented in this section. The results are expressed in terms of drawdowns. The drawdowns simulated after 75 years of continuous pumping under the production scenario and described aquifer properties are presented in Table 3-4. As can be seen from the simulation results, the effects of pumping are most significant near the carbonate production wells, after 75 years of continuous pumping. The following sections describe the simulated drawdowns in Cave, Dry Lake, and Delamar valleys.

3.3.4.1 Cave Valley

As indicated in Table 3-4, simulated drawdowns at the end of 75 years of continuous pumping range from 41 ft at water-right number 6638 to 50 ft at water-right number 7397. Water-right numbers 6638 and 7397 are both underground rights located in the southern portion of Cave Valley (Figure 3-1). To further evaluate the effects of the proposed SNWA pumping on underground water rights, the well construction and water-levels must be considered. However, well construction information is not available for these two wells. The simulated drawdowns at these locations are believed to be at the upper bounds of what is expected due to the conservative assumptions in this analysis (i.e., drawdowns in the carbonate rock aquifer and the alluvial aquifer were combined and the storativity in the alluvial aquifer is a lower bound).



Table 3-4 Simulated Drawdowns at Selected Permitted Underground and Spring Water Rights and Environmental Areas of Concern

Application		Loc	ation	Distance from	Simulated Drawdowns
Number or Environmental Area of Concern	Source	UTM Easting ^a (m)	UTM Northing ^a (m)	Nearest Proposed Pumping Well (mi)	at the end of 75 years of Continuous Pumping (ft-bgs)
			Cave Valley		
6638	Underground	683,156	4,238,721	5.97	41
7397	Underground	680,091	4,244,266	3.53	50
			Dry Lake Valle	y .	
5936	Underground	695,162	4,241,005	35.10	8
8698	Spring	700,487	4,195,177	6.44	48
18756	Underground	697,709	4,201,362	10.48	37
35770	Underground	692,956	4,215,376	19.64	21
35771	Spring	695,348	4,212,756	17.71	23
35773	Underground	690,834	4,220,143	22.85	17
35774	Underground	702,490	4,219,938	21.83	18
V03839	Spring	698,577	4,214,407	18.44	21
V03840	Spring	698,549	4,217,584	20.41	19
			Delamar Valle	y	
4620	Spring	693,455	4,139,886	3,39	65
4622	Spring	683,947	4,153,060	2.57	55
4894	Spring	693,098	4,153,581	5.33	64
4973	Spring	683,947	4,153,060	2.57	55
5782	Spring	689,434	4,156,310	3.76	58
5783	Spring	689,434	4,156,310	3.76	58
10189 ^b	Spring	694,609	4,157,153	6.94	61
11167	Spring	693,546	4,135,873	5.83	53
52112 ^b	Spring	695,009	4,157,163	7.19	61
V01400 ^b	Spring	694,609	4,157,153	6.94	61
V01520	Spring	693,546	4,135,873	5.83	53
V01654	Spring	688,343	4,158,084	3.10	56
Grassy Spring	Spring	695,124	4,157,193	7.26	61

^aNorth American Datum of 1983, Zone 11N ^bWater-rights 10189, 52112, and V01400 are all on Grassy Spring

3.3.4.2 Dry Lake Valley

As indicated in Table 3-4, simulated drawdowns at the end of 75 years of continuous pumping range from 8 ft at water-right number 5936 located in the northern portion of Dry Lake Valley to 48 ft at water-right number 8698 which is located approximately 6.44 mi from the nearest SNWA application point of diversion.

To further evaluate the effects of the proposed SNWA pumping on underground water rights, the well construction and water-levels must be considered. Drillers logs could only be found for water-right numbers 18756 and 35773. Water-right number 18756 corresponds to drillers log number 5511 (NDWR, 2007d). According to this drillers log, the well was completed to a depth of 515 ft-bgs with a static depth to water of approximately 398 ft-bgs. Water-right number 35773 corresponds to drillers log number 39357 (NDWR, 2007d). According to this drillers log, the well was completed to a depth of 480 ft-bgs with a static depth to water of 310 ft-bgs. Water-right number 35774 is the Bristol Well (Table D.2-2 of Part A). The Bristol Well was completed to a depth of 80 ft-bgs and has a depth to water of 20 ft-bgs. Therefore, the simulated drawdowns for these locations, 37 ft at 18756, 17 ft at 35773, and 18 ft at 35774, resulting from pumping the proposed production wells, would be unlikely to cause the wells to go dry or even impair their usefulness as stock water wells.

Water-right numbers 8698 and 35771 both have sources identified as springs. Inspection of these points on topographic sheets and aerial imagery suggests that they may be reservoirs fed by springs located within the mountain block to the east (USGS, 1970; 1971). If they are reservoirs, then pumping the proposed production wells would not effect these locations.

3.3.4.3 Delamar Valley

As indicated in Table 3-4, the simulated drawdowns at the end of 75 years of continuous pumping range from 53 ft at water-right numbers 11167 and V01520 to 65 ft at water-right number 4620.

The one environmental area of concern in Delamar Valley is Grassy Spring, which also has water rights associated with it. Water-right numbers 10189, 52112, and V01400 are all on Grassy Spring. Grassy Spring is located in northern Delamar Valley approximately 6.94 mi from the nearest proposed SNWA production well. The simulated drawdown at Grassy Spring is 61 ft. This spring is located over 450 ft higher in elevation than the nearest proposed pumping well and is located within the walls of a caldera complex (Figure 3-3). The discharge measurements and temperatures for Grassy Spring indicate it is derived from local precipitation (Appendix B). These factors suggest a different water source than the source underlying the application point of diversion. Furthermore, the depth to water at the application point of diversion is likely to be on the order of 1,000 ft based on the depth to water of other wells in the valley. Therefore, it is highly unlikely that any effects will occur as a result of pumping the proposed SNWA applications.

The remaining water rights in Delamar Valley are associated with springs. The limited water-level data in Delamar Valley, as described in Section 2.0 of Part B, indicates a deep water table. This would suggest that the springs shown in Figure 3-3 are likely derived from a perched system that would be unaffected by pumping the proposed SNWA applications.

Section 3.0

SE ROA 11988



3.3.5 Limitations

Drawdowns simulated using the Theis equation are only approximations of the potential drawdowns. In reality, the magnitude of the drawdowns caused by pumping at a given observation point depends on several factors:

- · The distance between the pumping well and the observation point
- · The presence of major heterogeneities including barriers to groundwater flow
- · The actual aquifer properties in the region of pumping
- · The magnitude, length, and continuity of pumping

The assumption of continuous pumping is a conservative one. Wells are not designed to produce without any interruptions. Downtime can be accidental or intentional. Accidental interruptions may be due to power outages or equipment malfunction. Intentional interruptions may be imposed for maintenance or management purposes. For example, the aquifer could be pumped on an intermittent basis to lessen the effects of pumping on the aquifer system. Periods of halted pumping would allow the aquifer system to partially or completely recover, depending on their duration, and decrease the cumulative drawdowns in the area.

The Theis solution cannot accurately simulate drawdowns in a complex flow system such as the one comprising the project area and vicinity. It can, however provide conservative approximations of the range of drawdowns that may actually occur.

3-16

Section 3.0

4.0 MONITORING AND MANAGEMENT

SNWA will coordinate the development of a program to monitor long-term fluctuations in water levels and spring discharge at selected monitoring sites in Cave, Dry Lake, and Delamar valleys and vicinity as per the permit conditions set forth by the Nevada State Engineer. The monitoring program will be developed in cooperation with NDWR and other stakeholders with water-related interests in Cave, Dry Lake, and Delamar valleys and vicinity.

4.1 Monitoring Objectives

The objectives of the monitoring program will be to provide long-term monitoring of Cave, Dry Lake, and Delamar valleys and vicinity through evaluation of the monitoring sites of the Joint Funding Agreement between SNWA, USGS, and NDWR and augmentation as necessary.

4.2 Existing Monitoring Sites

SNWA installed two monitor wells in each of Cave, Dry Lake, and Delamar valleys in 2005 as well as a monitor well in northern Pahranagat Valley (Eastman, 2007a through g). Additionally, SNWA cooperatively funds USGS monitoring of wells and springs in Cave, Dry Lake, and Delamar valleys as well as in nearby areas. These monitoring sites are described in Table 4-1 and depicted in Figure 4-1. The monitoring sites are regularly evaluated by SNWA, USGS, and NDWR to ensure data adequacy and appropriate spatial coverage. The monitoring of these sites may be included in the permit conditions prescribed by the Nevada State Engineer.

4.3 Monitoring Frequency and Data Reporting

The proposed monitoring program is envisioned to include depth-to-water measurements collected on a continuous and quarterly basis. Continuous recorders will be installed in the SNWA application production wells and in selected monitor wells to monitor water-level fluctuations, temperature, and barometric pressure. Discrete depth-to-water measurements will be obtained on a regular basis. Spring discharge will be monitored at the springs listed in Table 4-1.

The requirements for submitting monitoring reports will be specified in the Nevada State Engineer approved monitoring plan. SNWA will cooperate in data collection efforts and will compile the data into organized annual reports to be submitted to NDWR and made available to other stakeholders with water-related interests in Cave, Dry Lake, and Delamar valleys and vicinity.

Section 4.0



4-2

Section 4.0

SE ROA	11991
--------	-------

 Table 4-1

 Monitoring Sites in Cave, Dry Lake, and Delamar Valleys and Vicinity

 (Page 1 of 2)

				Loci	ation	-	
Map-ID	USGS Number	Additional Site Identifier	Type	UTM Easting ^a (m)	UTM Northing ^a (m)	Current Monitoring Frequency	Period of Record
		Cave Valley					
180-24	1	180W/501M	Well	688,048	4,273,716	Continuous	2005 to 2007
180-25	j	180VV902M	Well	689,805	4,248,363	Continuous	2005 to 2007
180-05	382810114521501	180 N07 E63 14BADB1 USAF	Well	685,674	4,260,043	Quarterly	1980 to 2007
180-14	383458114473601	180 N08 E64 04ABDD1 USBLM	Well	692,206	4,272,813	Quarterly	1965 to 2007
180-15	383307114471001	180 N08 E64 15BCBC1 USBLM Cave Valley 4	Well	692,845	4,269,374	Quarterly	1990 to 2007
180-19	383632114465801	180 N09 E64 27BCDD1 USBLM	Well	692,983	4,275,700	Quarterly	1964 to 2007
180-20	384207114505601	180 N10 E63 25A 1 Cave Valley 1	Well	686,984	4,285,891	Quarterly	1958 to 2007
		Dry Lake Valley					
181-22	1	181M-1	Well	688,537	4,198,181	Continuous	2005 to 2007
181-23	1	181W909M	Well	698,688	4,174,479	Continuous	2005 to 2007
181-01	375624114444501	181 N01 E62 24ABB 1 USBLM	Well	698,010	4,201,548	Quarterly	1990 to 2007
181-04	380336114473501	181 N02 E64 03B 1 USBLM	Well	693,124	4,215,351	Quarterly	2005 to 2007
181-05	380531114534201	181 N03 E63 27CAA 1 USGS-MX (N. Dry Lake)	Well	684,519	4,218,103	Continuous	1980 to 2007
181-06	380616114494101	181 N03 E64 20BD 1 USBLM - Coyote Well	Well	690,486	4,220,368	Quarterly	1983 to 2007
181-07	380550114412301	181 N03 E65 21D 1 Bristol Well	Well	702,610	4,219,921	Quarterly	1962 to 2007
181-09	381256114500701	181 N04 E64 07DC 1 USGS-MX (Muleshoe Valley)	Well	689,481	4,232,096	Quarterly	1981 to 2007
181-19	374215114453101	181 S03 E64 12AC 1 USGS-MX (S. Dry Lake Well)	Well	697,515	4,175,351	Continuous	1980 to 2007
	-	Delamar Valley					
182-04		182M-1	Well	680,874	4,135,306	Continuous	2005 to 2007
182-05	1	182W906M	Well	690,078	4,133,299	Continuous	2005 to 2007
182-02	372639114520901	182 S06 E63 12AD 1 USGS-MX (Delamar Well)	Well	688,422	4,146,273	Continuous	1980 to 2007
		White River Valley					
207-11	382259115090801	207 N06 E61 18AADA1 NDW-Hot Creek Spring	Spring	661,290	4,249,926	Continuous	1935 to 2007
207-21	382111115055901	207 N06 E61 27AADC1 USGS-MX	Well	666,042	4,246,692	Quarterly	1980 to 2007
207-27	09415510	Preston Big Spring near Preston NV	Spring/ Stream	666,191	4,310,441	Continuous	1947 to 2007
207-28	383540115081801	207 N09 E61 32DABC1 Moorman Spring	Spring	662,053	4,273,440	Biannual	1935 to 2006
207-29	385158115000401	207 N11 E62 04AABA1 Lund Spring	Spring	673,266	4,302,019	Biannual	1910 to 2006

Southern Nevada Water Authority - Water Resources Division

 Table 4-1

 Monitoring Sites in Cave, Dry Lake, and Delamar Valleys and Vicinity

 (Page 2 of 2)

				Loci	ation	Cumont	
Map-ID	USGS Number	Additional Site Identifier	Type	UTM Easting ^a (m)	UTM Northing ^a (m)	Monitoring Frequency	Period of Record
02-30	385507114574801	207 N12 E61 12BDAD1 Cold Springs	Spring	609'299	4,309,454	Biannual	1910 to 2006
15-70	385530115044601	207 N12 E61 12DBDD1 Nicholas Springs	Spring	668,104	4,308,847	Biannual	1910 to 2006
207-32	382526115011401	207 N07 E62 33BCAB1 Flag Spring 1	Spring	672,719	4,254,696	Biannual	1982 to 2006
207-33	382522115012001	207 N07 E62 33BCCB1 Flag Spring 2	Spring	672,576	4,254,570	Biannual	1982 to 2006
207-34	382517115012001	207 N07 E62 33BCCC1 Flag Spring 3	Spring	672,579	4,254,416	Biannual	1949 to 2006
207-35	382624115004001	207 N07 E62 28ABDC1 Butterfield Spring	Spring	673,530	4,256,472	Biannual	1949 to 2006
207-36	381613115110101	207 N05 E60 24CDDA1 Murphy Meadows	Well	658,843	4,237,388	Quarterly	1985 to 2007
207-37	382211115133801	207 N06 E60 22BBBC1 USBLM	Well	654,962	4,248,321	Quarterly	1985 to 2007
207-38	383154115101501	207 N08 E60 24DD 1 USBLM	Well	659,362	4,266,385	Quarterly	1965 to 2007
207-39	383113115061201	207 N08 E61 27DDBA1 USGS-MX	Well	665,638	4,265,125	Quarterly	1979 to 2007
207-40	383113115061202	207 N08 E61 27DDBA2 USGS-MX	Well	665,638	4,265,125	Quarterly	1979 to 2007
207-41	382432115095801	207 N09 E61 07BCCC1 Sorenson Well	Well	659,186	4,280,043	Quarterly	1948 to 2007
207-42	384713115034801	207 N11 E61 25BC 1 USBLM Wilson Meadows West	Well	668,134	4,294,907	Quarterly	1990 to 2007
207-43	383122115083701	207 N11 E61 32B 1 Point	Well	660,910	4,293,336	Quarterly	1947 to 2007
207-44	384640115045001	207 N11 E61 35ACCD1 Public Domain Well 25	Well	666,483	4,293,084	Quarterly	1953 to 2007
207-45	385226115124201	207 N12 E60 27ACBD1 USBLM	Well	655,061	4,304,294	Quarterly	1957 to 2007
207-46	385400115024001	207 N12 E62 18DDAA1 USGS Well 24	Well	670,117	4,307,132	Quarterly	1947 to 2000
		Pahroc Valley					
208-06	380505114593501	208 N03 E62 35BBB 1 USBLM White River MX	Well	675,936	4,217,111	Quarterly	1963 to 2006
208-10	374525115061801	208 S02 E61 23D 2	Well	666,856	4,180,533	Quarterly	1985 to 200
		Pahranagat Valley					
209-04	ł	209M-1	Well	677,377	4,168,166	Continuous	2005 to 200
209-07	09415590	209 S05 E60 10 1 Crystal Spring near Hiko, NV	Spring/ Stream	656,168	4,155,349	Continuous	1912 to 200
209-09	09415640	Ash Springs Creek below Hwy 93 at Ash Springs, NV	Spring/ Stream	659,683	4,147,461	Continuous	1912 to 200

Section 4.0

Part C - Water-Related Effects Analysis

⁴⁻³ SE ROA 11992



4-4

Southern Nevada Water Authority - Water Resources Division



Figure 4-1 Existing Monitoring Network

4.4 Resource Management

SNWA will abide by all permit conditions attached to the water-right permits and work within the water-right administrative process set forth by the State Engineer to (1) manage its use of the water resources and (2) avoid unreasonable lowering of the water table such that adverse impacts to senior water-right holders do not occur. Additionally, SNWA will seek to manage its use of the water resources permitted to it by the State Engineer in a manner that does not adversely affect environmental resources.

Water-resource management scenarios could include the following:

- Spatial redistribution of pumping from production wells that are understood to be causing site-specific effects, to other production wells located farther away from the affected location
- Temporal re-distribution of pumping from continuous to seasonal operations, or other operational scenarios that would reduce effects
- Mitigation of unreasonable effects, as determined by the State Engineer
- · Reduction of production rates to allow for recovery in the aquifer being effected

Section 4.0

4-5



This Page Intentionally Left Blank

Section 4.0 SE ROA 11995

5.0 SUMMARY

Based on abstracts obtained from NDWR, the majority of water rights in Cave, Dry Lake, and Delamar valleys are on springs and are used for the purposes of stock watering. Committed duties for underground rights in these valleys include 46.58 afy in Cave Valley, 56.56 afy in Dry Lake Valley, and 7.24 afy in Delamar Valley.

A simplified analysis using Theis (1935) was conducted to evaluate the potential effects of SNWA's proposed groundwater development project in Cave, Dry Lake, and Delamar valleys. The purpose of the simplified analysis was to evaluate the effects of continuously pumping the application volume from two wells in each of the three basins for a period of 75 years. Although the drawdowns were quantitatively simulated using the Theis equation, the estimates are still considered qualitative because of the many assumptions that are inherent to this equation.

The findings of the quantitative effects analyses are summarized as follows:

- Simulated drawdowns at the end of 75 years of continuous pumping in Cave Valley range from 41 to 50 ft at the two underground water rights located in the southern portion of the valley. No drillers log information was available for these wells.
- Simulated drawdowns at the end of 75 years of continuous pumping in Dry Lake Valley range from 8 to 48 ft. Drillers logs and additional well information was available for three of the five underground water right locations. Based on this information it is unlikely that the three wells would be adversely impacted by the proposed pumping. Two of the valley floor spring water-right locations appear to be reservoirs and therefore no effects are anticipated at those locations. Additional springs in Dry Lake Valley were unlikely to experience effects due to their nature, geographic location, and the geologic complexity between the proposed pumping locations and the springs.
- Simulated drawdowns at the end of 75 years of continuous pumping in Delamar Valley range from 53 to 65 ft. However, the water rights in this valley are associated with springs that are likely from a perched system and are therefore unlikely to be effected by pumping from the proposed SNWA application points of diversion. Grassy Spring was evaluated as part of the Theis analysis and based on its location, nature and the fact that the proposed pumping well location for SNWA likely has a depth to water of 1,000 ft or more, it is highly unlikely that Grassy Spring will be effected by the proposed pumping.
- The estimates of drawdowns presented in this report are conservative. Actual drawdowns are likely to be much lower. This conservative nature is the result of the assumptions that are made in the analysis as well as in using the Theis equation. Examples of these assumptions include: (1) combining the effects of pumping in both the alluvial and carbonate-rock

Section 5.0

SE ROA 11996



aquifers; (2) the aquifer is homogeneous, isotropic, and of uniform thickness; and (3) there is no regional gradient, just to name a few. Based on the list of assumptions involved with the Theis analysis, these results are considered conservative.

SNWA will abide by all permit conditions attached to the water-right permits and work within the water-right administrative process set forth by the State Engineer to (1) manage its use of the water resources, and (2) avoid unreasonable lowering of the water table such that adverse impacts to senior water-right holders do not occur. In support of its resource management program, the SNWA will develop a monitoring program in cooperation with NDWR and other stakeholders with water-related interests in Cave, Dry Lake, and Delamar valleys. The purpose of this program will be to monitor long-term fluctuations in water levels and spring discharge at selected monitoring sites in the vicinity of pumping.

Section 5.0 SE ROA 11997

6.0 REFERENCES

- Belcher, W.R., Elliot, P.E., and Geldon, A.L., 2001, Hydraulic- Property Estimates for Use With a Transient Ground-water Flow Model of the Death Valley Regional Ground-water Flow System, Nevada and California: U.S. Geological Survey Water Resources Investigation Report 01-4120: Carson City, Nevada, 33 p.
- Bunch, R.L., and Harrill, J.R., 1984, Compilation of Selected Hydrologic Data from the MX Missile-siting Investigation, East-central Nevada and Western Utah: U.S. Geological Survey Open-File Report 84-702, 123 p.
- Dettinger, M.D., Harrill, J.R., Schmidt, D.L., and Hess, J.W., 1995, Distribution of Carbonate-Rock Aquifers and the Potential for their Development, Southern Nevada and Adjacent Parts of California, Arizona, and Utah: U.S. Geological Survey Water-Resources Investigations Report 91-4146: Carson City, Nevada, 107 p.
- Eastman, H.S., 2007a, Geologic Data Analysis Report for Monitor Well 182W906M in Delamar Valley: Doc. No. RDS-ED-0001 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007b, Geologic Data Analysis Report for Monitor Well 182M-1 in Delamar Valley: Doc. No. RDS-ED-0002 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007c, Geologic Data Analysis Report for Monitor Well 180W902M in Cave Valley: Doc. No. RDS-ED-0003 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007d, Geologic Data Analysis Report for Monitor Well 180W501M in Cave Valley: Doc. No. RDS-ED-0004 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007e, Geologic Data Analysis Report for Monitor Well 181M-1 in Dry Lake Valley: Doc. No. RDS-ED-0005 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007f, Geologic Data Analysis Report for Monitor Well 209M-1 in Pahranagat Valley: Doc. No. RDS-ED-0006 Southern Nevada Water Authority, Las Vegas, Nevada.
- Eastman, H.S., 2007g, Geologic Data Analysis Report for Monitor Well 181W909M in Dry Lake Valley (Doc. No. RDS-ED-0007): Southern Nevada Water Authority, Las Vegas, Nevada.
- Ertec Western, Inc., 1981a, MX Siting Investigation, Water Resources Program, Technical Summary Report Volume I, Report E-TR-52-I, prepared for the U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 77 p.

Section 6.0

SE ROA 11998



- Ertec Western, Inc., 1981b, Aquifer Testing Dry Lake Valley, Nevada, Report E-81-Nov-I, prepared for the U.S. Department of the Air Force, Ballistic Missile Office, Norton Air Force Base, California, 30 p.
- NDWR, see Nevada Division of Water Resources.
- Nevada Division of Water Resources, 2007a, Hydrographic Abstracts from the Water Rights Database accessed at website http://water.nv.gov/waterRights/permitdb/permitdb_disclamier.cfm on August 13, 2007.
- Nevada Division of Water Resources, 2007b, Hydrographic Basin Summaries by Manner of Use from the Water Rights Database accessed at website http://water.nv.gov/waterRights/permitdb/ permitdb disclamier.cfm on November 6, 2007.
- Nevada Division of Water Resources, 2007c, Overview of Nevada Water Law, Environmental Protection accessed at website http://water.nv.gov/WaterRights/WaterLaw/environmental.cfm on November 6, 2007.
- Nevada Division of Water Resources, 2007d, Well Log Database available on the World Wide Web, accessed November 2007 at http://water.nv.gov/engineering/wlog/wlog.cfm.

RockWare, Inc., 2004, RockWorks2004. Golden, CO.

- Schaefer, D.H., and J.R. Harrill, 1995, Simulated Effects of Proposed Ground-Water Pumping in 17 Basins of East-Central and Southern Nevada. U.S. Geological Survey, Water-Resources Investigations Report 95-4173. Denver, Colorado: U.S. Geological Survey.
- SNWA, see Southern Nevada Water Authority.
- Southern Nevada Water Authority, 2007, Biological Resources for Cave, Dry Lake, and Delamar Valleys. Presentation to the Office of the Nevada State Engineer. Las Vegas, Nevada.
- Southern Nevada Water Authority, 2005, Hydrogeologic Report in Support of Southern Nevada Water Authority's Proposed Groundwater Development Project in Three Lakes Valley South, Clark County, Nevada: Change in Point-of-Diversion Applications 72787-72797, Las Vegas, Nevada.
- Theis, C.V., 1935, The Relation between the Lowering of the Piezometric Surface and the Rate and Duration of Discharge of a Well Using Groundwater Storage, Am. Geophys. Union Trans., vol. 16, pp. 519-524.

USGS, see United States Geological Survey

- U.S. Geological Survey, 1970, Ely Springs Quadrangle, Nevada. 1:24,000 Scale. Denver, CO.
- U.S. Geological Survey, 1971, Coyote Spring Quadrangle, Nevada. 1:24,000 Scale. Denver, CO.

Part C - Water-Related Effects Analysis

Appendix A

NDWR Hydrographic Abstracts for Cave, Dry Lake, and Delamar Valleys

SE ROA 12000

JA_4762

A.1.0 WATER RIGHTS

This appendix contains hydrographic abstracts for Cave, Dry Lake, and Delamar valleys downloaded from the NDWR website on August 13, 2007. The fields and codes in these tables are described as follows:

- · App. Application Number.
- · Change of App. Change of Application Number.
- · Cert. Certificate Number.
- File Date Date of filing.
- Status Status of the Application.
 - ABR Abrogated
 - APP Application
 - CAN Cancelled
 - CER Certificate
 - DEN Denied
 - PER Permit
 - RES Reserved
 - RFA Ready for Action
 - RFP Ready for Action (Protested)
 - VST Vested Right
 - WDR Withdrawn
- · Source Source of the water.
 - LAK Lake
 - OSW Other Surface Water
 - RES Reservoir
 - SPR Spring
 - STR Stream
 - UG Underground
- POD QQ Point of Diversion Quarter Quarter.
- · POD Qtr Point of Diversion Quarter.

Appendix A

SE ROA 12001

A-1



Southern Nevada Water Authority - Water Resources Division

- POD Sec Point of Diversion Section.
- POD Twn Point of Diversion Township.
- POD Rng Point of Diversion Range.
- Div. Rate (cfs) Diversion Rate in cubic feet per second.
- Type of Use The use identified for the water.
 - DOM Domestic
 - IND Industrial
 - IRD Irrigation-DLE
 - IRR Irrigation
 - MM Mining and Milling
 - MUN Municipal
 - OTH Other
 - PWR Power
 - QM Quasi-Municipal
 - STK Stock
- Sup Supplemental, Yes or No.
- Annual Duty Annual duty of water right.
- Duty Units Units associated with the annual duty either AFY for acre-feet per year or AFS for acre-feet per season.
- Owner of Record Owner of the right.

A-2

Appendix A SE ROA 12002

Table A.1-1	DWR Hydrographic Abstract for Cave Valley	(Page 1 of 5)
	NDN	

	Change of App.	Cert.	File Date	Status	Source	POD	POD Qtr	Sec	Twn	Rng	Div Rate (cfs)	Type of Use	Sup	Annual Duty	Duty Units	Owner of Re
	1	1	12/3/1907	CAN	SPR	NE	MN	16	N60	64E	14	PWR	ł	0	ł	Robertson, Edwa
	ł	1	10/2/1908	CAN	SPR	NE	SE	30	11N	63E	0	MM	l	0	1	Hendrix, E.A.
-		1	6/9/1909	CAN	OSW	1	SW	31	10N	64E	0	IRR	ł	0	ł	Barnes, Benn
-	ł	1661	10/21/1914	CER	SPR	NNN	SW	29	11N	63E	0.003	STK	ł.	1.288938	AFS	Carter, Alice
	I.	2334	10/23/1914	CER	SPR	SW	MN	21	11N	63E	0.004	STK	ł	1.657206	AFS	Reid, Robert
1	1	ł	6/15/1917	DEN	SPR	SE	MN	16	N60	64E	1.6	IRR	ł	0	1	Olsen, Casten
	l	643	9/24/1917	CER	STR	SE	SE	Ħ	N60	63E	0.12	IRR	Ĵ	36	AFS	Adams, Myron
1	1	1060	1/31/1918	CER	SPR	SW	NE	16	N60	64E	0.751	IRR	1	225.57	AFS	Mull Revocable Ti 1/13/1999
20.1	ł	540	5/13/1918	CER	SPR	SW	SE	25	08N	64E	0.015	STK	1	7.518805	AFS	Mull Revocable Ti 1/13/1999
	1	542	5/13/1918	CER	SPR	В	B	25	08N	64E	0.015	STK		7.303982	AFS	Mull Revocable T 1/13/1999
	1	707	9/19/1919	CER	SPR	SW	SE	30	11N	63E	0.004	STK	1	1.503761	AFS	Reed, G.M.
	ł	1	9/22/1919	CAN	SPR	SE	SE	12	N60	62E	0.012	STK	1	2.946144	AFS	Gregorio Urrutia (
2	ł	1	9/22/1919	CAN	SPR	SE	NE	32	10N	63E	0.012	STK	ľ	0	į.	Gregorio Urrutia (
	1	ł	9/22/1919	DEN	SPR	SW	NNN	23	10N	64E	0.025	STK	ł	0	1	Gregorio Urrutia (
	ł	1	10/2/1919	DEN	SPR	MN	SE	22	10N	64E	0.5	STK	1	3.007522	AFS	Gregorio Urrutia (
-	t	1	10/2/1919	CAN	SPR	SW	MM	11	N60	64E	0.012	STK	1	2.946144	AFS	Gregorio Urrutia (
~	}	1	11/26/1919	DEN	SPR	SW	SE	25	08N	64E	0.006	STK	1	0	ł	Gregorio Urrulia (
-	ł	ł	11/26/1919	CAN	SPR	NE	MN	13	N60	64E	0.006	STK	ł	0.736536	AFS	Gregorio Urrutia (
-	1	1	12/5/1921	CAN	DUG	ł	ł	ł	06N	63E	0	STK	ł	0	}.	Whipple, J.I.
1	1	1	1/20/1922	WDR	SPR	SE	MN	16	N60	64E	0.8	IRR	1	1	ł	Stephens, Carl W
0	}	2105	2/27/1922	CER	ng	R	SE	21	05N	63E	0.003	STK	1	2.14823	AFA	Jensen, Bruce A.
1	ł	1175	6/14/1925	CER	ng	SW	SE	31	06N	63E	0.015	STK	ł	1.872029	AFS	Jensen, Bruce A.
1.0	ļ	1876	8/20/1925	CER	ng	SW	SE	36	N60	64E	0.012	STK	1	8.961188	AFA	Kirkeby, Gordon
1.5	1	4209	7/26/1929	CER	SPR	SW	ШN	16	N60	64E	0.044	DOM	1	0	1	Great Western M

Appendix A

SE ROA 12003

Part C - Water-Related Effects Analysis



Southern Nevada Water Authority - Water Resources Division

	Owner of Record	Great Western Mining & Development	Great Western Mining & Development	Jensen, Bruce A.	Cave Valley Ranches	Cave Valley Ranches	Cave Valley Ranches	BLM	Cave Valley Ranches Inc.	Gulf Oll Corporation	Lewis, Lou Jeanne	Lewis, Lou Jeanne	Murry Whipple Ranch	Lewis, Jeanne Lou	WX	MX	MX	MX	MX	MX	Southern Nevada Water Authority	Southern Nevada Water Authority			
	Duty Units	1	1	AFA	AFA	AFA	AFS	ł	þ	1	1	ł	1	AFA	AFA	AFA	AFA	1	ł	ť	ł	ł	ļ	AFA	AFA
	Annual Duty	0	0	7,242604	17.922376	17.922376	5.585398	0	0	0	0	0	0	240	79.2	320	126	0	0	0	0	0	0	4343.82	7239.7
	Sup	ł	1	ł	1	1	1	1	1	1	ł	1	1	1	ł	}	ł	1	ł	ł	1	1	ł	1	1
/alley	Type of Use	PWR	MM	STK	STK	STK	STK	IRR	IRR	IRR	IRR	IRR	UNI	IRR	IRR	IRR	IRR	QM	QM	QM	QM	QM	QM	MUN	MUN
or Cave \	Div Rate (cfs)	0	0	0.01	0.025	0.025	0.019	0	0	0	0	0	0	0.89	0.56399999		0.67	0	0	0	٠	0	0	Q	10
-1 ract f	POD Rng	63E	63E	63E	64E	64E	64E	64E	64E	64E	64E	64E	64E	63E	63E	63E	63E	63E	64E	63E	64E	64E	63E	63E	63E
e A.1 Abstr e 2 of	POD Twn	N60	N60	06N	N60	N60	11N	08N	N60	N60	N60	N60	07N	N60	N60	N60	N60	N70	06N	06N	08N	NZ0	NZ0	06N	NLO
Table ohic	POD	4	4	19	14	2	33	6	9	4	16	6	19	e	Ħ	4	11	33	80	21	6	e	14	22	21
ogra	POD	SE	SE	¥	NE	SW	SE	NE	MN	SW	ШЦ	MM	SE	SW	MN	BN	MN	SE	MN	SW	MN	NN	MN	NN	SE
łydro	Pop QQ	NE	Ŋ	SE	MN	SW	NE	SW	SE	MN	SW	MN	SE	SE	SW	SW	SW	MN	SW	SW	BN	SW	R	SW	SE
DWR H	Source	SPR	SPR	SPR	SPR	SPR	SPR	MSO	SPR	SPR	SPR	SPR	ng	STR	SPR	SPR	SPR	ng	DC	DUG	DUG	DG	DUG	DUG	DUG
(DN	Status	DEN	DEN	CER	CER	CER	CER	CAN	CAN	CAN	CAN	CAN	CAN	CER	CER	CAN	CER	WDR	WDR	WDR	WDR	WDR	WDR	RFP	RFP
	File Date	7/26/1929	7/26/1929	10/9/1933	2/8/1934	2/8/1934	10/13/1949	10/26/1960	7/15/1965	7/15/1965	7/15/1965	7/15/1965	4/11/1966	10/15/1969	1/5/1970	1/5/1970	10/5/1973	7/14/1980	7/14/1980	7/14/1980	7/14/1980	7/14/1980	7/14/1980	10/17/1989	10/17/1989
	Cert.	1	1	2135	2269	2270	4059	1	1	1	ł	1	;	8358	8359	ł	9654	1	11	1	1	ł	1	ł	ł
	Change of App.	1	ł	-	ł	ł	;	ł	-	ł	1	t	1	1	1	1	ł	1	ł	1	ł	1	100	ł	ł
	App.	9002	9003	9702	9720	9721	13102	19299	22692	22693	22694	22695	23093	25322	25411	25412	27814	41696	41697	41698	41699	41700	41701	53987	53988

A-4

Appendix A

 Table A.1-1

 NDWR Hydrographic Abstract for Cave Valley

 (Page 3 of 5)

Owner of Record	Lincoln County Water District	Lincoln County Water District	Jensen, Bruce A. and Pamela G.	Jensen, Bruce A. and Pamela G.	Jensen, Bruce A. and Pamela G.	Jensen, Pamela G.	Jensen, Bruce A.	Mull Revocable Trust 1/15/1999	Mull Revocable Trust 1/15/1999	Mull Revocable Trust 1/15/1999	Lewis, Paul	Lewis, Paul	Lewis, Paul	Whipple, Kevin	Cave Valley Ranch, LLC						
Duty Units	AFA L	AFA I	AFA	AFA P	AFA	AFA .	AFA ,	AFA	AFA	AFA	AFA	AFA.	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA
Annual Duty	5210	5210	11.201485	11.201485	80	1280	1280	11.2	11.2	11.2	0	0	0	0	8.071	0	0	0	0	0	0
Sup	1	ł	}	}	1	ł	1	*	*	*	1	ł	1	1	1	1	1	1	1	1	ł
Type of Use	IRR	IRR	STK	STK	IRR	IRR	IRR	STK	STK	STK	IRR	IRR	IRR	STK	QM	OTH	OTH	STK	OTH	OTH	STK
Div Rate (cfs)	10	10	0.0156	0.0156	0.15	3.5	3.5	0.05	0.05	0.05	4	4	4	0.0156	0	0.16	0.07	0.07	0.07	0.16	0.06
POD Rng	63E	64E	63E	64E	64E	63E	63E	64E	64E	63E	63E	63E	63E	64E	64E	64E	64E	64E	64E	64E	64E
POD	O5N	08N	07N	08N	07N	N20	N20	N60	08N	10N	N60	N60	N60	08N	N60						
Sec	œ	6	15	30	33	14	14	27	15	25	4	e	10	4	5	ŝ	5	5	S	5	5
otr Ot	NE	SE	SE	SW	SW	N	NE	NNN	MN	BR	SE	SW	NE	NE	SE	SE	SE	SE	SE	SE	Ľ.
00 00	SE	NE	MN	ß	ß	B	NE	SW	SW	MN	BR	SE	SE	MN	SE	SE	SE	SE	SE	SE	HS.
Source	ng	DUG	ng	DG	SPR	DG	DUG	ng	ng	DG	ng	ng	ng	ng	DG	STR	STR	DC	ng	ng	STR
Status	RFP	RFP	CER	CER	PER	RFP	RFP	PER	PER	PER	RFA	RFA	RFA	RFA	RFA	APP	APP	APP	APP	APP	APP
File Date	12/11/1998	12/11/1998	3/8/2000	3/8/2000	3/8/2000	2/11/2002	2/11/2002	8/19/2005	8/19/2005	8/19/2005	2/9/2006	2/9/2006	2/9/2006	1/3/2007	5/24/2007	8/3/2007	8/3/2007	8/3/2007	8/3/2007	8/3/2007	8/3/2007
Cert	ł	1	16617	16619	1	1	1	ł	1	1	ł	1	ł	1	1	ł	1	+	1	ł	1
Change of App.	1	ł	ł	ţ	1	1	1	1	1	1	1	1	į	ł	1	-	ł	;	1	ł	V01680
App.	64670	64671	56123	66125	66129	68487	68488	73168	73169	73170	73815	73816	73817	75231	75779	76124	76125	76126	76127	76128	76129

Appendix A

SE ROA 12005

Part C - Water-Related Effects Analysis



Southern Nevada Water Authority - Water Resources Division

	Owner of Record	BLM	BLM	BLM	Adams Mcgill Company	Adams Mcgill Company	Cave Valley Ranches	Cave Valley Ranches	Mull Revocable Trust, 1/13/1999	Cave Valley Ranches	Mull Revocable Trust, 1/13/1999	Mull Revocable Trust, 1/13/1999	Cave Valley Ranches Inc.	Mull Revocable Trust, 1/13/1999	1	Mull Revocable Trust, 1/13/1999	Geyser Land & Cattle Co.	Mull Revocable Trust, 1/13/1999	Geyser Land & Cattle Co.	Cave Valley Ranches Inc.	Mull Revocable Trust. 1/13/1999	Adams Mcgill Company	Adams Mcgill Company
	Duty Units	AFA	AFA	AFA	1	ł	}	AFS	AFS	AFS	AFS	ł	ł	AFA	1	ł	}	l	ł	ł	AFA	1	1
	Annual Duty	0	0	0	0	0	0	7.457427	7.457427	7.457427	7.457427	0	0	0	1	0	0	0	0	0	0	0	0
	Sup	1	1	1	1	1	1	1	1	1	ł	ł	1	1	ł	1	1	1	1	ł	1	Ī	ł
Valley	Type of Use	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	ł	STK	STK	STK	STK	STK	IRR	STK	STK
or Cave	Div Rate (cfs)	0.0015	0.0015	0.0015	0.1	0.25	0.5	0.3	0.3	0.3	0.025	1	÷	ł	ł	٢	0.025	0.025	0.025	0.025	0.004	0.25	0.25
-1 act fo 5)	POD	63E	63E	63E	63E	63E	64E	64E	64E	64E	64E	64E	64E	64E	1	64E	64E	64E	64E	64E	64E	63E	63E
e A.1 Absti e 4 of	POD Twn	06N	06N	06N	11N	11N	10N	10N	10N	10N	10N	10N	N60	N60	ł	10N	N60	N60	N60	N60	10N	11N	11N
Table ohic , (Pag	POD	19	31	30	30	20	10	23	22	22	27	34	2	4	ł	26	2	£	14	14	31	15	8
ograp	POD	NE	SE	SW	SE	SE	SW	MM	SE	SE	SW	SE	MN	BR	I	MN	SW	SE	NE	SW	SW	SE	SE
Hydro	POD QQ	SE	SW	RE	NE	BN	NNN	SW	SE	MN	SE	NE	NNN	MN	l	BR	SW	MN	MN	ШN	MN	SW	SE
DWR I	Source	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	STR	STR	STR	STR	STR	SPR	SPR	SPR	SPR	STR	STR	SPR
Z	Status	RES	RES	RES	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	APP	VST	VST	VST	VST	VST	VST	VST	VST
	File Date	4/6/2004	4/6/2004	4/6/2004	8/30/1915	12/8/1916	4/8/1918	12/6/1919	12/6/1919	12/6/1919	12/26/1919	12/29/1919	12/29/1919	1/8/1920	1	1/8/1920	5/3/1920	5/3/1920	5/3/1920	5/3/1920	8/18/1921	4/18/1925	4/18/1925
	Cert.	1	1	1	1	ł	ł	Ţ	1	1	1	1	1	1	76129	ł	1	1	1	ł	Ţ	1	1
	Change of App.	1	1	1	1	1	1	1	ł	1	1	ł	1	1	Changed By:	1	1	1	ł	ł	ł	1	ł
	App.	R09414	R09416	R09417	V01416	V01486	V01559	V01658	V01659	V01660	V01675	V01678	V01679	V01680	V01680	V01681	V01696	V01697	V01698	V01699	V01807	V01878	V01881

Appendix A

SE ROA 12006

up Annual Duty Owner of Record Duty Units	Adams Mcgill Company	0 Adams Mcgill Company	0 Jensen, Bruce A.	0 Jensen, Bruce A.	6.720891 AFS Jensen, Bruce A.	6.720891 AFS Jensen, Bruce A.	0 AFA Lewis, Lou Jeanne	0 Lewis, Lou Jeanne	0 Lewis, Lou Jeanne	Jensen, Bruce A.	Jensen, Bruce A.	3.37579 AFS Jensen, Bruce A.	Jensen, Bruce A.	Jensen, Bruce A.	Jensen, Bruce A.	0 AFA Mull Revocable Trust Dated 1/15/1999	n AFA Mull Revocable Trust				
f Use S	STK -	STK -	STK -	STK -	STK -	STK -	IRR	IRR	IRR -	STK -	STK -	STK -	STK -	STK -	STK	STK	STK	STK .	STK .	STK	STK
Div Rate 1 (cfs) of	0.25	0.05	0.004	0.002	0.25	0.25	0.414	0.414	0.12	0.013	0.013	0.013	0.1	0.05	0.025	0.025	1	0.333	0.333	0.025	0.025
POD Rng	63E	63E	63E	63E	63E	63E	63E	63E	63E	63E	62E	63E	63E	64E	62E	64E	64E	64E	64E	64E	64F
POD	11N	11N	N90	06N	10N	10N	N60	N60	N90	N60	N60	10N	N60	N20	N60	N60	N60	10N	10N	Neo	Neu
POD Sec	10	10	19	30	6	17	4	e	11	6	12	32	32	33	13	2	2	23	22	11	13
POD Qtr	MN	SW	B	SW	SE	SE	NE	SW	MN	SE	SE	SE	ų	SW	MM	SE	NN	MN	SE	SE	NNN
POD	ЧN	SE	SE	NE	MN	SW	SW	SE	SW	BN	MN	SW	N	SE	SE	SW	MN	NN	MM	SW	ų,
Source	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	STR	SPR	SPR	SPR	ads
Status	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST
File Date	4/18/1925	4/18/1925	1/11/1926	1/11/1926	3/21/1927	4/8/1927	11/25/1970	11/25/1970	11/25/1970	3/8/2000	3/8/2000	3/8/2000	3/8/2000	3/8/2000	3/8/2000	8/19/2005	8/19/2005	8/19/2005	8/19/2005	8/19/2005	8/10/2005
Cert.	1	1	;	1	ł	1	1	1	ł	1	1	1	1	-	1	t	ł	Ţ	1	}	
Change of App.	1	1	1	1	1	ł	1	1	ţ	-	ł	l	1	1	1	ţ	1	1	ł	t	1
App.	V01882	V01883	V01964	V01965	V02075	V02079	V02692	V02693	V02694	V09231	V09232	V09233	V09234	V09235	V09236	V09522	V09523	V09524	V09525	V09526	100527

Appendix A

A-7

Part C - Water-Related Effects Analysis

SE ROA 12007



Southern Nevada Water Authority - Water Resources Division

Owner of Record	Ellis, A.C.	E.& F. Mining	Newman, Geo. W.	Carter, Dean	ł		Ť		Cyphers, Robert M.	Butler, N.N.	Butler, N.N.	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishop Church JC LDS	Mackie, A.J.	Corp Presiding Bishop Church JC LDS	Imlay, James W.	Imlay, James W.	Robinson, J.R. Jr.	Culverwell, Wm.	Culverwell, Wm.	Culvervell, Wm.	Culverwell, Wm.	Conoway, John H.	Corp Presiding Bishop Church JC LDS
Units	1	1	ł	AFA	ł	ļ	ł	1	AFS	1	1	AFA	AFA	1	AFA	1	1	ł	1	1	ł	ł	ł	AFA
Annual Duty	0	0	0	663.49618	ł.	+	1		11	1	Ŧ	7.5	7.5	ł	7.5	ţ		ł	0	ł	1	1	1	2.178919
Sup	Ī	Ì	ł	ł	1	ł	1	1	1	1	1	1	1	ł	1	1	1	Į	1	1	1	1	ł	1
Type of Use	IRR	IRR	MM	STK	1	1	1	ł	IRR	MOG	MOD	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK
Div Rate (cfs)	0	0	0	0.91599999	1	ł	1	ł	0.03	0.1	0.1	0.01	0.01	0.025	0	0	0	0	0	0	0.4	0.4	0	0.003
Rng	65E	1	65E	65E (1	1	1	1	66E	65E	65E	64E	64E	64E	64E	64E	64E	65E	65E	65E	65E	65E	65E	65E
Twn	04N	1	05N	01S	1	1)	1	1	04N	04S	04S	03S	02S	03S	015	01S	015	04S	04S	04S	04S	04S	045	04S
Sec	Ĭ.	1	10	1	1	ł	1	1	25	20	20	e	33	21	32	ł	1	1	20	1	1	20	1	31
otr Otr	ł	Ĩ	1	1	1	1	1	Ì	SE	MN	SE	NE	SW	SW	SE	1	1	1	MN	1	ł	SE	1	SE
00 QQ	Ĩ	ł	1	ł	1	1	1	1	SW	NE	щ	RE	SW	NE	MM	1	ł	1	NE	ł	ł	BR	ł	SE
Source	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	RES	RES	SPR	RES	SPR	SPR	SPR	SPR	SPR	SPR	SPR	ng	SPR
Status	CAN	CAN	CAN	CER	CAN	CER	CER	CER	CER	CAN	CAN	CER	CER	CAN	CER	CAN	CAN	DEN	CAN	CAN	DEN	CAN	CAN	CER
File Date	3/9/1907	3/9/1907	6/13/1907	1/10/1908	1	ł	ł	ł	4/26/1915	9/27/1915	9/27/1915	4/10/1916	4/10/1916	4/10/1916	4/10/1916	8/23/1917	8/23/1917	10/1/1917	10/5/1917	10/5/1917	10/5/1917	10/5/1917	11/12/1917	11/12/1917
Cert.	1	3	1	566	36334	52105	52108	52109	1980	1	1	724	725	1	726	1	1	1	1	1	1	1	1	732
Change of App.	1	ł	1	1	Changed By:	Changed By:	Changed By:	Changed By:	-	1	1	1	ł	1	1	1	Ŧ	1		1	1		1	ł
App.	387	388	524	780	780	780	780	780	3368	3592	3593	3875	3876	3877	3878	4550	4551	4612	4616	4617	4618	4619	4694	4696

Table A.1-2 NDWR Hydrographic Abstract for Dry Lake Valley (Page 1 of 11)

Appendix A

Appendix A	
------------	--

Owner of Record	-	Culverwell, William	-	Gardner Ranch Co.	Jeffcott, Vernon	Thorley, Robert A.	Thorley, Robert A.	Highbee, Florence S.	Mathews, Charles Jr.	H.H. Land and Cattle Company	+	West Side Cattle Company	West Side Cattle Company	Goodman, R.F.	Whipple, Deeann	Thorley, Frank A.	Adams McGill Company	Adams McGill Company	Ellsworth, Roy and Mathews, Dan	Corp Presiding Bishop Church JC LDS	H.H. Land and Cattle Company	1
Units	1	AFA	ł	ł	ł	ł	l	AFA	1	AFA	1	1	AFA	AFS	AFS	1	l	AFA	-{	AFA	AFA	ł
Annual Duty	-	1.074115	1	1	0	ł	1	2.178919	1	0	ł	0	44,80594	0.153445	10	1	I	18.075821	ł	6.506068	0	1
Sup	I.	T	1	ł	1	1	1	1	4	1	1	1	1	1	1	1	1	1	ł	1	I.	1
Type of Use	1	STK	1	STK	IRR	STK	STK	STK	STK	STK	ł	IRR	STK	STK	STK	STK	IRR	STK	STK	STK	STK	1
Div Rate (cfs)	ł	0.001	1	0.05	+	÷	0.1	0.003	0.4	0	1	0	0.063	0.002	0	0	0.5	0.025	0.031	0.00899999	0	ł
Rng	1	65E	1	64E	64E	64E	63E	65E	65E	65E	Ţ	65E	65E	65E	64E	64E	64E	64E	64E	63E	63E	1
Twn	ł	04S	1	04S	04N	03S	01N	01N	01N	03S	1	04N	04N	01N	02S	015	05N	05N	02S	02S	02S	ł
PoD	1	31	1	36	36	20	23	23	14	26	1	26	26	Ħ	24	35	14	14	24	33	22	ł
otr atr	1	SE	Ţ	MN	MM	SW	MN	BR	BR	NE	1	BR	SE	MN	NM	SE	MM	MN	MN	SE	MN	ł
POD QQ	1	SW	1	MN	SW	MN	SW	MN	NE	NE	1	SW	SW	SE	SW	SW	SE	SE	SW	RE	SW	ł
Source	SPR	SPR	SPR	SPR	SPR	STR	SPR	SPR	SPR	SPR	SPR	STR	STR	SPR	RES	MSO	DG	DG	MSO	SPR	SPR	SPR
Status	WDR	CER	WDR	CAN	CAN	CAN	DEN	CER	CAN	ABR	CER	CAN	CER	CER	CER	CAN	WDR	CER	DEN	CER	ABR	CER
File Date	1	11/12/1917	-	11/12/1917	1/21/1918	2/13/1917	2/23/1918	3/14/1918	3/14/1918	3/21/1918	1	5/7/1918	8/12/1918	1/16/1919	1/25/1919	1/25/1919	1/5/1920	1/5/1920	4/30/1920	5/6/1920	5/6/1920	1
Cert.	52111	733	52110	ł	ł	ł	ł	525	1	734	52106	1	1924	526	1119	ŀ	t	854	1	1053	1054	52103
Change of App.	Changed By:	1	Changed By:	1	1	1	ł	1	-	t	Changed By:	1	ł	1	ł	1	ł	ł	ł	1	-	Changed By:
App.	4696	4697	4697	4699	4855	4906	4934	4961	4962	4972	4972	5057	5200	5356	5371	5372	5935	5936	6076	6094	6095	6095

NDWR Hydrographic Abstract for Dry Lake Valley Table A.1-2

A-9 SE ROA 12009

Part C - Water-Related Effects Analysis



Southern Nevada Water Authority - Water Resources Division

	cord			A.	forn			0.	ank of		lines	ſS				Land &	A.		Land	u	Α.	Bishop	E
	Owner of Re	Jeffcott, Vernon	Jeffcoat, Vernor	Thorley, Robert	Nevada Silver F Mining Co.	Brodie, W.E.	Williams, Alex E	Comet Mines C	Federal Land B. Berkeley	Nevada Lead Company	Pioche Union N. Company	Robison Brothe	Clark, Douglas	Clark, Douglas	Hamilton, R.H.	Imperial Farms Cattle Co., Inc.	Skinner, Lucille	Swallow, R.T.	Imperial Farms and Cattle Co.	Whipple, Deeat	Thorley, Frank	Corp Presiding Church JC LDS	Whipple, Deear
	Units	1	1	1	1	1	AFA	AFA	AFA	I	1	AFA	AFA	AFA	ł	1	1	AFA	AFA	AFA	1	AFA	AFS
	Annual Duty	1	0	ł	t	l	10.74115	18.075821	1.595828	1	ł	1.565139	2.608565	4.848862	}	ł	1	40.325346	7.242604	12.060777	ļ	6.782269	0.675158
_	dns	1	Ī	1	ł	1	ł	ł	1	1	1	1	1	ł	1	1	T	1	1	ł	1	1	1
Valley	Type of Use	STK	STK	STK	MM	STK	STK	MM	STK	MM	MM	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK
Jry Lake	Div Rate (cfs)	0.5	0	0.167	0.5	0.1	0.015	0.025	0.002	0.25	0.33	0.002	0.006	0.012	0.003	0.5	0.5	0.5	0.01	0.017	0.03	0.00899999	0.001
t for [POD Rng	64E	64E	66E	65E	65E	66E	66E	65E	65E	66E	63E	63E	63E	63E	64E	65E	64E	64E	65E	65E	64E	65E
A.1-2 stract	POD Twn	03S	03S	02S	04N	04N	02S	01N	01N	04N	01N	02N	01N	01N	02N	N90	06N	06N	N90	01S	01N	03S	03S
able Abs	POD	21	21	33	34	35	33	32	22	36	26	13	22	22	34	15	ł	15	ŝ	5	2	11	4
T aphic (P	POD	MN	MN	SE	SW	ŝ	SW	SW	SW	MN	SE	SW	NE	MM	¥	MN	1	NE	SE	SE	NE	MN	RE
Irogr	POD	NE	NE	NNN	MN	SE	NE	SW	NE	SW	1	MM	SE	SE	NE	SE	ł	NNN	NE	NE	WN	NE	SE
WR Hyd	Source	SPR	SPR	SPR	SPR	SPR	SPR	NG	SPR	MSO	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	DUG	MSO	SPR
NDN	Status	DEN	DEN	DEN	WDR	CAN	CER	CER	CER	CAN	CAN	CER	CER	CER	DEN	DEN	MDR	WDR	CER	CER	DEN	CER	CER
	File Date	11/27/1920	11/29/1920	1/15/1921	4/6/1921	5/5/1921	1/27/1922	7/17/1922	10/23/1922	3/19/1924	5/9/1924	5/16/1924	11/4/1925	11/4/1925	11/4/1925	12/8/1927	12/8/1927	8/28/1928	8/19/1928	9/17/1928	9/15/1930	8/11/1932	5/6/1933
	Cert.	ł	ł	1	3	ł	835	1629	971	l	1	1466	2209	2210	ł	ł	1	1	8146	5705	ł	2107	2293
	Change of App.	1	1	1	1	1		1	ł	ł	l	1	1	ł	Ì	1	1	1	1	1	1	1	-
	App.	6337	6338	6376	6432	6454	6619	6718	6803	7064	7111	7117	7563	7564	7565	8405	8406	8669	8670	8698	9335	9618	9660

Appendix A

SE ROA 12010

JA_4772

 Table A.1-2

 NDWR Hydrographic Abstract for Dry Lake Valley

 (Page 4 of 11)

Owner of Record	BLM	Jones, Erastus L.	Whipple, Deeann	Whipple, Deeann	Corp Presiding Bishop Church JC LDS	Bleak, Juanita W. and Wheller, Casey L.	Whipple, Deeann	Conaway, John H.	Comet Coalition Mines Co.	Thorley, Frank	Thorley, Frank	Byron, A.	Higbee, E. Edwin, Kristine H.	Higbee, E. Edwin and Kristine H.	Ercanbrack, Byron A.	Higbee, E. Edwin and Kristine H.	Higbee, E. Edwin and Kristine H.	Thorley, Robert A.	Ercanbrack, Bryon A.	Jones, H. Wendell	Delmue, Albert	Barnett, Cora A.
Units	AFA	AFA	AFS	AFS	AFA	AFA	AFS	AFA	1	AFS	AFA	1	AFA	AFA	1	AFA	AFA	AFA	AFA	AFS	AFA	ł
Annual Duty	361.5	3	5.21713	5.21713	7.181226	2.240297	6.720891	4.480594	0	5.21713	4.480594	ł	1.872029	1.503761	1	2.792699	1.626517	13.441782	0	3.98957	10.833217	1
Sup	1	ł	1	1	1	1	1	1	1	1	T	1	1	1	1	1	- Į-	ł	ł	ł	T	ł
Type of Use	STK	STK	STK	STK	STK	STK	STK	STK	MM	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	IRD
Div Rate (cfs)	0.5	0	0.015	0.015	0.01	0.003	0.013	0.1	2	0.013	0.013	0.003	0.003	0.003	0.025	0.00899999	0.002	0.1	0.015	0	0.015	0
Rng	65E	64E	65E	65E	65E	66E	65E	64E	66E	63E	63E	63E	63E	63E	64E	63E	63E	66E	63E	66E	64E	64E
Twn	02N	04S	01N	01N	04S	01N	02S	04S	01S	02N	02N	035	02S	02S	02S	02S	02S	015	02S	02S	01N	03S
Sec	9	2	22	23	თ	26	33	2	o	34	35	10	32	34	29	27	27	80	27	11	24	21
Otr	MN	IJ	SW	ų	SW	NE	SE	MN	SW	SE	MM	SW	SE	SW	SE	MN	SW	SE	MN	NN	MN	NN
00 QQ	NNN	SW	NE	NNN	NE	SE	SE	MN	SW	SW	MN	NE	MN	SW	SE	B	RE	SE	Щ	SW	Ш	NN
Source	ng	MSO	SPR	SPR	SPR	SPR	RES	ng	ng	SPR	SPR	SPR	SPR	SPR	0SW	SPR	SPR	MSO	SPR	STR	ng	DC
Status	WDR	DEN	CER	CER	CER	CER	CER	WDR	WDR	CER	CER	CAN	CER	CER	DEN	CER	CER	CAN	CAN	CER	CER	WDR
File Date	3/6/1936	1/7/1937	5/17/1937	5/17/1937	10/17/1941	12/3/1943	5/10/1944	1/19/1946	2/15/1947	2/6/1948	2/6/1948	6/18/1948	6/18/1948	6/18/1948	6/18/1948	1/8/1949	3/7/1949	4/11/1949	5/19/1949	12/24/1952	4/26/1960	3/11/1964
Cert.	ł	1	2355	2356	2805	3063	2826	1	1	3583	3584	1	4390	4391	ł	4501	4502	ł	ł	4712	5059	1
Change of App.	1	1	1	1	ł	1	1	1	1	1	1	1	1	1	1	6497	11308	I	6497	ļ	1	ł
App.	9965	10070	10119	10120	10747	11033	11118	11490	11780	12246	12247	12509	12511	12512	12514	12793	12840	12879	12920	14732	18756	21868

Appendix A

SE ROA 12011

Part C - Water-Related Effects Analysis


Southern Nevada Water Authority - Water Resources Division

A-12

Appendix A

Table A.1-2	VDWR Hydrographic Abstract for Dry Lake Valley (Page 6 of 11)
-------------	--

Owner of Record	Delmue, Frank	Delmue, Frank	Delmue, Frank	Steward, Robert	Imperial Farms Land and Cattle Co.	Imperial Farms Land and Cattle Co.	Geyser Ranch Limited Partnership													
Units	ł	1	1	-	AFA	AFA	AFA	AFA	AFA	1	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA
Annual Duty	0	0	0	0	1.350316	1.350316	1.350316	1.350316	1.350316	ł	11.416308	15.835524	4.91024	3.253034	3.253034	3.253034	3.253034	10.096681	4,848862	3.253034
Sup	ł	1	ł	1	T	ł	ł	ł	1	ł	1	1	ł	1.	1	1	1	1	1	1
Type of Use	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK
Div Rate (cfs)	0	0	0	0	0.25	0.25	0.25	0.25	0.25	0.25	0	0	0.007	0.005	0.005	0.005	0.005	0.014	0.007	0.005
Rng	65E	65E	65E	65E	65E	65E	65E	65E	65E	65E	64E	64E	65E							
Twn	03N	04N	04N	05N	05N	05N	05N	05N	05N	06N	N90	06N	05N	04N						
Sec	21	ъ	35	10	14	15	16	21	22	21	35	12	26	26	22	22	22	26	26	35
otr Otr	1	1	SE	SE	SW	SW	З	N2	SE	t	ţ	1	SW	MM	SE	ŝ	BR	SE	NW	SE
00 00	ł	1	NE	SE	ł	ł	ł	1	ł	1	ł	ł	SE	MN	SE	NE	SE	MM	SE	NE
Source	ng	SPR	SPR	SPR	SPR	SPR	SPR	MSO	SPR	SPR	RES	RES	RES	SPR						
Status	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CAN	CER							
File Date	3/9/1978	3/9/1978	3/9/1978	3/15/1978	3/15/1978	3/15/1978	3/15/1978	3/15/1978	3/15/1978	4/24/1978	4/26/1978	4/26/1978	8/7/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978
Cert.	1	1	1	1	ł	1	1	1	1	j	ł	1	10175	10204	10205	10206	10207	10208	10209	10210
Change of App.	ł	1	1	1	1	ł	1	1	1	1	1	ł	1	1	ł	ť	ł	1	ł	1
App.	35100	35101	35102	35120	35127	35128	35129	35130	35131	35328	35334	35336	35696	35761	35762	35763	35764	35766	35767	35768

Appendix A

SE ROA 12013

Part C - Water-Related Effects Analysis



Southern Nevada Water Authority - Water Resources Division

	Owner of Record	Geyser Ranch Limited Partnership	Steward, Robert	Steward, Robert	Geyser Ranch Limited Partnership													
	Units	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA						
	Annual Duty	3.253034	3.191656	3.253034	3.253034	3.191656	3.191656	10.096681	1.166182	1.350316	11.416308	4.050948	3.222345	3.222345	3.222345	4.848862	5.800221	4.848862
	Sup	1	1	1	ł	1	I	1	1	1	1	1	1	}	1	1	Ц.	ł
Valley	Type of Use	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK						
Iry Lake	Div Rate (cfs)	0.005	0.004	0.005	0.005	0.004	0.004	0.014	0.002	0.002	0.25	0.006	0.004	0.004	0.004	0.25	0.008	200.0
for D	POD Rng	64E	64E	64E	65E	64E	65E	65E	65E	65E	64E	64E	65E	65E	65E	65E	64E	65E
A.1-2 tract	POD	OIN	02N	02N	02N	03N	03N	04N	05N	05N	N90	06N	04N	04N	04N	04N	05N	04N
able Abs	POD Sec	-	4	11	19	20	21	ŝ	21	10	35	12	თ	16	15	12	18	26
T aphic (P	POD	NE	NE	SW	SW	MN	SE	SE	ų	SW	SW	SW	SE	SE	SW	MM	NE	SE
rogr	POD	SW	SW	MN	SE	SE	MN	SE	MM	NE	NE	SE	SE	NNN	MN	1	RE	SE
NR Hyd	Source	RES	DG	SPR	RES	DG	ng	SPR	SPR	SPR	RES	RES	SPR	SPR	SPR	SPR	SPR	SPR
ND	Status	CER	CER	CER	WDR	CER	CER	CER	CER	CAN	CER	CER						
	File Date	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	8/18/1978	9/6/1978	9/6/1978	9/6/1978	9/6/1978	10/3/1978	10/3/1978	10/3/1978	11/20/1978	11/20/1978	11/20/1978
	Cent	10186	10869	10211	10187	10870	10871	10212	10288	10289	Ī	10215	10217	10218	10220	ł	10222	10223
	Change of App.	1	1	ł	1	ł	1	ł	1	1	-	1	1	1	1	-	1	ł
	App.	35769	35770	35771	35772	35773	35774	35775	35843	35844	35849	35851	35951	35952	35954	36178	36179	36180

A-14

Appendix A

Table A.1-2	WR Hydrographic Abstract for	(Page 8 of 11)
	NDN	

Dry Lake Valley

Owner of Record	Steward, Robert	Carter, Dean	MX	U.S. Government	MX	U.S. Government	Meadow Valley Land 8 Cattle Co.	Hatch, Roger	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishor Church JC LDS	Corp Presiding Bishor Church JC LDS	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishop Church JC LDS	H.H. Land and Cattle Company	H.H. Land and Cattle Company	Southern Nevada Water Authority	Southern Nevada Water Authority
Units	AFA	AFA	AFA	1	1	AFA	AFA	AFS	AFA	AFA	AFA	AFA	AFA						
Annual Duty	1.350316	3.222345	3808.934546		ł	217	1280	0.92067	6.506068	0.460335	2.178919	1.442383	1.442383	9.421523	6.506068	1.074115	2,178919	4343.905194	7239.84199
Sup	1	1	1	1	ł	ł	1	Ŧ	1	1	1	1	1	T	1	1	1	1	1
Type of Use	STK	STK	ΜØ	MØ	QM	MØ	IRD	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	MUN	MUN
Div Rate (cfs) 0	0.002	0.004	5.5	+	1	1	Q	0.006	0.00899999	0.001	0.003	0.002	0.002	0.013	0.00899999	0.001	0.003	ø	10
Rng	65E	63E	64E	64E	65E	64E	64E	65E	63E	65E	65E	65E	65E	65E	64E	65E	65E	64E	65E
Twn	05N	01S	03S	06N	05N	04N	03S	01N	02S	03S	03S	03S	04S	04S	04S	04S	04S	02S	02S
Sec	23	33	12	9	9	7	12	22	22	21	21	28	16	29	25	31	31	30	80
otr Otr	SW	MN	BR	MN	SW	SE	BR	SE	MN	RE	SE	NE	MN	MN	SE	SE	SE	SW	SE
00 00	SE	MN	SW	SW	BR	SW	SW	NN	SW	SW	MN	MN	MN	SW	SE	SW	SE	SE	ШN
Source	SPR	SPR	ng	ng	ng	ng	ng	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	ng	ng
Status	CER	CAN	WDR	WDR	WDR	WDR	DEN	CER	CER	CER	CER	CER	CER	CER	CER	WDR	WDR	RFP	RFP
File Date	11/20/1978	12/22/1978	1/30/1980	7/14/1980	7/14/1980	7/14/1980	4/26/1982	1/20/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	5/18/1988	10/17/1989	10/17/1989
Cert.	10295	ł	1	1	1	1	ł	13590	13775	13776	13777	13778	13779	13780	13781	ł	1	1	1
Change of App.	;	780	1	1	1	1	ł	ł	6095	V01549	780	4972	V01027	780	780	4697	4696	1	ł
App.	36183	36334	40433	41732	41733	41734	45588	51776	52103	52104	52105	52106	52107	52108	52109	52110	52111	53989	53990

Appendix A

SE ROA 12015

Part C - Water-Related Effects Analysis

A xibnaqqA

91-4

App.	Change of App.	Cert.	File Date	Status	Source	QQ	POD	POD Sec	POD	POD Rng	Div Rate (cfs)	0.	Type f Use	Type f Use Sup	Type Sup Annual Duty	Type Sup Annual Duty Units
ω	1	1	7/1/1994	DEN	UG	NW	SW	б	04S	1000	64E	64E 27627	64E 27627 MUN	64E 27627 MUN	64E 27627 MUN	64E 27627 MUN
50180	-	1	7/1/1994	DEN	UG	SE	SE	6	04S	σ	đi đi	4E 27627	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN
60181	1	ł	7/1/1994	DEN	NG	ŝ	WW	7	04S	ç,	ħ	4E 27627	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN
60182	1	1	7/1/1994	DEN	NG	NW	WW	80	04S	0	¥n	4E 27627	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN	4E 27627 MUN
60189	1	ł	7/1/1994	DEN	NG	NW	SW	IJ	04S	127.	64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR
60190	1	4	7/1/1994	DEN	NG	SE	SE	თ	04S		64E	64E 0	64E 0 PWR	64E 0 PWR	64E 0 PWR	64E 0 PWR
60191	ţ	1	7/1/1994	DEN	UG	NN	WN	25	04S		63E	63E 27627	63E 27627 PWR	63E 27627 PWR	63E 27627 PWR	63E 27627 PWR
60192	Ľ	1	7/1/1994	DEN	NG	ŝ	WW	7	04S		64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR
60193	1	1	7/1/1994	DEN	NG	WN	NW	00	04S	10.1	64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR
64668	1	1	12/11/1998	RFP	UG	SW	Zm	20	01S	100	65E	65E 10	65E 10 IRR	65E 10 IRR	65E 10 IRR 5120	65E 10 IRR 5120 AFA
64669	1	ľ	12/11/1998	RFP	NG	SW	NE	33	05N		64E	64E 10	64E 10 IRR	64E 10 IRR	64E 10 IRR 5120	64E 10 IRR 5120 AFA
69878	Ĭ	1	4/18/2003	RFP	UG	SE	SE	6	04S	-	64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR 0	64E 27627 PWR 0 AFA
69879	Ĭ	1	4/18/2003	RFP	DG	SW	SE	13	04S	-	63E	63E 27627	63E 27627 PWR	63E 27627 PWR	63E 27627 PWR 0	63E 27627 PWR 0 AFA
69880	ļ	ł	4/18/2003	RFP	UG	WN	WN	18	04S		64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR 0	64E 27627 PWR 0 AFA
69883	ł	ł	4/18/2003	RFP	BN	WN	SW	5	04S		64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR 0	64E 27627 PWR 0 AFA
69884	î.	1	4/18/2003	RFP	UG	WN	WN	8	04S	1.1	64E	64E 27627	64E 27627 PWR	64E 27627 PWR	64E 27627 PWR 0	64E 27627 PWR 0 AFA
69885	I	ł	4/18/2003	RFP	NG	SE	WN	7	04S	100	64E	64E 27627	64E 27627 MUN	64E 27627 MUN	64E 27627 MUN 0	64E 27627 MUN 0 AFA
69886	Ì	1	4/18/2003	RFP	UG	SE	SE	5	04S	1.1	64E	64E 27627	64E 27627 MUN	64E 27627 MUN	64E 27627 MUN 0	64E 27627 MUN 0 AFA
69887	ł	Ī	4/18/2003	RFP	UG	WN	WS	5	04S	-	64E	64E 27627	64E 27627 MUN	64E 27627 MUN	64E 27627 MUN 0	64E 27627 MUN 0 AFA
88869	1	ł	4/18/2003	RFP	UG	SE	WN	7	04S	-	4E	34E 27627	34E 27627 PWR	34E 27627 PWR	34E 27627 PWR 0	34E 27627 PWR 0 AFA

Table A.1-2 NDWR Hydrographic Abstract for Dry Lake Valley (Page 9 of 11)

uoisivid səənosəR vəter Authority - Water Resources Division SE ROA 12916 LI-A

V01288 V01289 V01290	V01288 V01289	V01288		V01287	V01268	V01267	V01265	V01250	V01135	V01134	V01027 Changed By:	V01027 Changed By:	V01027	R09411	R09410	R05989	R04778	69891	06869	68869	App. Change of App.	
ļ		ł	1	1	ł	1	ł	ł	ł	ł	52112	52107	ł	ł	1	1	1	1	1	ł	Cert.	
112111217	VIDAHOAN	2/24/1914	2/24/1914	11/6/1913	11/20/1913	11/17/1913	11/17/1913	6/8/1913	9/30/1912	9/30/1912	ţ	Ì	3/10/1911	3/23/2004	3/23/2004	6/21/1993	2/8/1989	4/18/2003	4/18/2003	4/18/2003	File Date	
	VST	VST	VST	VST	VST	VST	VST	VST	VST	VST	CER	CER	VST	RES	RES	RES	RES	RFP	RFP	RFP	Status	ND
	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	DG	UG	UG	Source	WR Hyd
	WS	SW	WW	WN	SE	NE		SE	SE	WS	ł	1	WN	WN	SE	SE	SE	WN	SW	WW	QQ	drogr
İ	WW	SE	SE	SW	SW	SW	SW	SE	NE	WN	ł	ł	WN	SE	SE	WW	NE	WN	SE	WW	POD	aphi (P
t	15	ω	16	15	13	21	28	24	36	31	1	}	16	30	5	10	8	18	13	8	POD Sec	c Abs
Î	05N	04N	05N	04N	02N	01N	OIN	02N	02N	02N	1	Î	04N	04N	04N	04S	S90	04S	04S	04S	POD	A.1- strac
Î	65E	65E	65E	65E	63E	63E	63E	65E	65E	66E	ł	1	65E	65E	65E	65E	65E	64E	63E	64E	POD	2 11)
	0.1	0.013	0.013	0.013	0.013	0.1	0.05	0.013	0.033	0.025	ł	ł	0.011	0.0015	0.0015	0	0.002	27627	27627	27627	Div Rate (cfs)	Dry Lake
	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	1	1	STK	STK	STK	OTH	OTH	MUN	MUN	MUN	Type of Use	Valle
	}	1	1	1	1	-	1	1	ł	ĺ	1	ł		ł	ł	ł	1	1	1	1	Sup	4
	4.480594	5.616087	1	5.616087	9.2067	6.720891	2.240297	1.381005	2.792699	3.928192	1	1	7.948451	0	0	0.214823	1.288938	0	0	0	Annual Duty	
I	AFA	AFA	1	AFA	AFA	AFA	AFA	AFA	AFS	AFS	1	ł	AFA	AFA	AFA	AFA	AFA	AFA	AFA	AFA	Units	
	Adams McGill Company	Geyser Ranch Limited Partnership	Adams McGill Company	Geyser Ranch Limited Partnership	Adams McGill Company	Adams McGill Company	Adams McGill Company	Delmue, Joseph	Lyttle, Edwin	Lyttle, Edwin	1	1	Corp Presiding Bishop Church JC LDS	BLM	BLM	BLM	BLM	Jacob, Beverly Joan	Jacob, Beverly Joan	Jacob, Beverly Joan	Owner of Record	

Part C - Water-Related Effects Analysis

SE ROA 12017

A xibneqqA

81-4

-1

A xibneqqA

				ND	WR Hy	drogr	aphi (P	age	11 of	11)	Dry Lake	Valle	×			
App.	Change of App.	Cert.	File Date	Status	Source	QQ	Qtr POD	POD	POD	POD	Div Rate (cfs)	Type of Use	Sup	Annual Du	4	ty Units
/01295	1	1	2/21/1914	VST	SPR	SE	NE	32	05N	65E	0.25	IRR	ł	60	_	AFS
V01296	ł	ł	2/24/1914	VST	SPR	Nim	NE	00	04N	65E	0	IRR	1	12		AFS
V01297	ľ	1	2/21/1914	VST	SPR	SW	SE	30	04N	65E	0	IRR	1	1.6		AFS
V01299	1	ł.	3/10/1914	VST	SPR	WN	SW	29	04N	65E	0.013	STK	1	4.480594		AFA
V01300		{	3/10/1914	VST	SPR	NW	SW	33	04N	65E	0.013	STK	ţ.	1		1
V01301	1	I	2/21/1914	VST	SPR	NE	NE	18	05N	64E	0.038	STK	Т.	3.37579		AFA
V01302	Ĩ	T	2/21/1914	VST	SPR	NE	SE	33	04N	65E	0.125	STK	1	4,480594	the second se	AFA
V01459	ł	ł	4/12/1916	VST	SPR	SE	SW	33	02S	66E	0.125	STK	Ì	3.37579	the second se	AFS
V01549	t.	ł	3/21/1918	ABR	SPR	NE	WW	24	03S	65E	0	STK	1	0		AFS
V01549	Changed By:	52104	1	CER	SPR	1	ł	I	Ī	ł		1	1	ł	-	ł
V01787	11	1	1/27/1922	VST	SPR	NE	WS	33	02S	66E	0.025	STK	ł	6.720891	-	AFA
V02350	3	4	9/8/1947	VST	SPR	NE	NE	ω	04S	65E	0.017	STK	1	1.135493		AFA
V02351	ſ	1	9/8/1947	VST	SPR	WS	SE	14	04S	65E	0.017	STK	1	1.350316	-	AFA
V03839	}	- {	2/1/1982	VST	SPR	WW	SW	თ	02N	65E	0.004	STK	1	2.025474		AFS
V03840	1	1	2/1/1982	VST	SPR	SW	SW	30	03N	65E	0.004	STK	3	2.025474		AFS
V04697	Ì	ą.	9/16/1988	VST	SPR	WN	WW	35	01N	65E	0.05	STK	1	12.306289		AFA
1000540		ł	5/19/1994	VST	SPR	WW	WW	35	01N	65E	0.05	STK	1	;		1

uoisivid səənosən vətev - Vatev Authority - Vater Resources Division SE BOA 12018

6		Н	
		14	

A xibneqqA

4695	4693	4683	4680	4644	4632	4628	4622	4621	4620	4462	3879	3476	3475	3475	3475	3475	3271	3270	3270	1532	App.	
ł	I	-	-	ţ	ł	;	-	1	ł	1	-	1	Changed By:	Changed By:	Changed By:	-	1	Changed By:	1	1	Change of App.	
731	730	L E	1	ł	704	$\left(\mathbf{I} \right)$	729	728	727	3186	1090	ł	12388	11705	9713	427	1923	52113	269	1	Cert.	
11/12/1917	11/12/1917	11/8/1917	11/8/1917	10/17/1917	10/13/1917	10/10/1917	10/8/1917	10/8/1917	10/8/1917	6/8/1917	4/10/1916	6/30/1915	ł	1	ł	6/30/1915	2/15/1915	1	2/15/1915	11/26/1909	File Date	1
CER	CER	DEN	DEN	CAN	CER	CAN	CER	CER	CER	CER	CER	CAN	ABR	CAN	CER	CER	CER	CER	ABR	CAN	Status	NDA
SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	RES	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	Source	WK HY
WS	SW	SE	SE	SW	SW	SE	1	ł	SE	NE	SE	NE	ł	1	ł	SW	NE	1	SE	1	QQ	arog
SW	SW	SE	SE	WS	SE	SE	1	1	NE	WN	SE	NE	1	1	ł	SE	NE	1	SE	1	Qtr POD	rapn
32	32	15	15	31	24	29	T	2	33	24	30	16	ł	1	ł	17	-	ł	Ø	1	POD	(Pag
05S	055	07S	07S	05S	05S	035	05S	05S	06S	05S	03S	05S	1	1	1	S90	S90	1	05S	04S	Pop	e 1 o
65E	65E	64E	64E	65E	62E	63E	63E	64E	64E	63E	63E	63E	ł	1	I	65E	62E	ł	65E	64E	POD	f7)
0.011	0.011	0.1	0.1	ŗ	0.003	0.025	0.002	0.003	0.001	0.025	0.00899999	0.025	1	ł	1	0.037	0.012	ł	0	0	Div Rate (cfs)	or Delama
STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	ł	ł	ł	IRR	STK	I	STK	IRR	Type of Use	r valley
}	ł	1	1	ł	1	1	1	ł	ł	ł	ł	1	1	1	1	ł	1	1	1	ł	Sup	
0	7.97914	1	1	0	2.117540969	0	0.552402	2.178919	0.705847	18.014443	D	0	1		1	25.15	8.961188	1	0	0	Annual Duty	
1	AFA	ł	1	1	AFA	1	AFA	AFA	AFS	AFA	1	1	1	đ	1	AFS	AFA	1	AFA	1	Units	
Gardner Ranch Company	Duffin, Maine R.	Jones, J.	Jones, J.	Jeffcott, Vernon	Nevada Rock and Sand Corporation	Conway, J.H.	Gardner Ranch Company	Duffin Jr., Press W.	Gardner Ranch Co.	Corp Presiding Bishop Church JC LDS	Corp Presiding Bishop Church JC LDS	Mathews, William Jr.	-	1	ł	LDS	Pace, Sid	1	H.H. Land and Cattle Company	Edwards, JNO. L.	Owner of Record	

Part C - Water-Related Effects Analysis

A	xibneqqA
---	----------

٨-20

6886	6885	6576	6202	6201	6113	5783	5782	5344	5318	5317	5316	5301	5092	5013	4973	4950	4904	4903	4902	4898	4894	4753	App.	
	1	Ĩ			ł	1	;	1	ł	ł	1	1	1	1	:	ł		1	1	1	ł	1	Change of App.	
1000	1225	1500	1	ł	ł	1006	1005	ł	582	ł	581	736	ł	ł	735	ł	1	1	ł	1	ł	ł	Cert.	
ADDIADOD	4/28/1923	10/13/1921	7/12/1920	7/12/1920	5/12/1920	9/29/1919	9/29/1919	12/26/1918	11/20/1918	11/20/1918	11/20/1918	10/25/1918	6/5/1918	4/22/1918	3/21/1918	3/6/1918	2/13/1917	2/13/1917	2/13/1918	2/11/1918	2/7/1918	12/3/1917	File Date	
200	CER	CER	DEN	WDR	DEN	CER	CER	CAN	CER	WDR	CER	CER	CAN	CAN	CER	CAN	CAN	CAN	CAN	CAN	PER	DEN	Status	ND
סדר	RES	RES	UG	UG	SPR	SPR	SPR	RES	RES	RES	RES	RES	WSO	RES	SPR	SPR	SPR	SPR	SPR	SPR	SPR	SPR	Source	NRHY
n n	WN	WN	1	SE	SE	WN	WW	ł	SW	SE	WN	SE	1	SW	1	SE	1	Ì.	3	SW	SE	1	QOP	drog
NE	SE	WS	Ţ	NE	NE	NE	NE	1	SE	SE	NE	NE	1	WS	1	SE	1	ł	1	WW	SE	1	Qtr POD	rapr
ŝ	7	21	1	33	7	7	7	16	26	36	31	16	1	2	1	15	1	1	1	2	16	34	POD	(Pag
075	07S	04S	05S	05S	055	055	05S	05S	04S	04S	04S	055	07S	08S	05S	07S	055	1	05S	05S	055	04S	POD	e 2 o
63E	63E	63E	63E	64E	64E	64E	64E	64E	63E	63E	64E	64E	63E	62E	63E	64E	64E	1	64E	63E	64E	63E	POD	f 7)
0	0	0.014	0	0	4	0.015	0.012	0	0.01	0.025	0	0.014	0	0	0.002	0.2	0.025	0.025	0.025	1.6	0.025	0.5	Div Rate (cfs)	or Delama
STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	STK	IRR	STK	STK	STK	IRR	DOM	STK	Type of Use	ir valley
ŀ	1	T	1	1	1	1	ł	ł	1	1	1	1	1	1	1	ł	1	ł	1	}	ł	ł	Sup	
9.973925	9.973925	4.971618	0	0	0	10.863906	9.053255	9.973925	3.98957	0	15.682079	10.004614	10	9.973925	1.135493	80	0	0	0	0	0	0	Annual Duty	
AFA	AFA	AFS	1	1	1	AFA	AFA	AFA	AFS	1	AFS	AFA	AFA	AFA	AFA	AFS	1	ł	ł	1	ł	1	Units	
LDS	LDS	Nevada Rock and Sand Corporation	Cutler, Warren	Cutler, Warren	Norris, A.H.	Duffins, Mamie R.	Duffins, Mamie R.	Adams, John A.	Corp Presiding Bishop Church JC LDS	Henrie, James	Carter, Dona	Duffin, Mame R.	Adams, John A.	Richard, J.W.	Gardner Ranch Co.	Horn, Cyrus	Burt, L.L.	Burt, L.L.	Burt, L.L.	Duffin, Mamie Ryan	Sawyer, Tillie B.	Henrie and Thiriot	Owner of Record	

uoisivid səənəsər Mater Authority - Water Resources Division SE ROA 12920

A xibnaqqA

				NDV	VR Hy	drog	raph	Tabl (Pag	e 3 o	1-3 act fo f 7)	or Delamar	Valley				
App.	Change of App.	Cert.	File Date	Status	Source	QQ	POD	POD	POD	POD Rng	Div Rate (cfs)	Type of Use	Sup	Annual Duty	Units	Owner of Record
6887	ł	ł	4/28/1923	WDR	RES	WS	SE	22	07S	64E	0	STK	ł	9.973925	AFA	Conaway, John H.
6888	1	ł	4/28/1923	WDR	RES	NE	SE	28	07S	63E	0	STK	1	0	ł	Conaway, John H.
7287	1	ł	1/14/1925	CAN	SPR	1	1	1	05S	65E	0	DOM	ł	Ł	ł	Farling, John W.
8800	1	ł	12/28/1928	CAN	RES	SE	SE	07	06S	63E	0.001	STK	1	0.982048	AFA	Conway, John
8801	1	ł	12/28/1928	CAN	RES	SE	SW	33	06S	63E	0.001	STK	}	0	ł	Conway, John
8921	I	1700	5/24/1929	ABR	SPR	SW	SW	20	05S	64E	0	STK	1	0	AFA	H.H. Land and Cattle Company
8921	Changed By:	52115	ł	CER	SPR	1	1	1	1	ł	ł	ł	1	ł	ł	-
9285	1	1	6/28/1930	DEN	SPR	SW	SW	32	05S	65E	n –	MM	1	0	ł	Jones, C.R.
9286	4	ł	6/28/1930	DEN	SPR	NE	SE	13	06S	65E	0.1	MM	}	0	1	Jones, C.R.
9287	1	}	6/28/1930	DEN	SPR	WN	SE	13	06S	64E	0.2	MM	ł	0	I	Jones, C.R.
9288	1	ł	6/28/1930	DEN	SPR	SW	SE	24	590	64E	0.1	MM	ł	1	ł	Jones, C.R.
9289	1	ł	6/28/1930	DEN	0SM	SW	SW	31	05S	65E	0.55	MM	ł	0	ł	Jones, C.R.
9659	1	2109	5/12/1933	CER	OSW	WW	WW	12	06S	63E	0.00899999	STK	1	5.984355	AFA	LDS
9713	3475	2423	11/25/1933	CER	SPR	NE	SW	17	06S	65E	0.05	MM	Ð	25	AFS	LDS
9793	I	ł	8/27/1934	DEN	0SM	WW	SW	10	S90	63E	0	STK	1	9.973925	AFS	Adams, W.B.
9794	1	}	8/27/1934	DEN	RES	NE	NE	7	07S	63E	0	STK	ł	9.973925	AFS	Adams, W.B.
9834	1	1	1/28/1935	WDR	UG	SW	SE	cn	290	65E	o	MM	1	0	1	Caliente Cyaniding Company
10069	1	ł	1/7/1937	DEN	WSO	WW	SE	13	04S	63E	0.027	STK	1	ł	ł	Jones, E.I.
10088	1	2622	2/10/1937	CER	SPR	Z	NE	4	290	62E	0.003	STK	1	2.240297	AFA	Nevada Rock and Sand Corporation
10189	1	2403	12/3/1937	CER	SPR	NE	SE	ω	05S	64E	0.025	STK	1	18,10651	AFA	Corp Presiding Bishop Church JC LDS
10189	Changed By:	52366	ł	DEN	SPR	1	1	1	1	1	1	ł	1	1	1	
10440	ł	2720	11/15/1939	CER	RES	SW	SE	20	07S	64E	0.003	STK	1	2.792699	AFA	LDS

Part C - Water-Related Effects Analysis

A xibriadrif	A	XI	ouə	ddy
--------------	---	----	-----	-----

A-22

App.	Change of App.	Cert.	File Date	ND	NR Hy Source	drog QD QD	POD Ottr	Tab Inic A (Pag Pop Sec	Je 4 (POD Twn	of 7) Rng	or Delam	ar Va	ley e of	lley ^{e of} Sup	lley e of Sup Annual	lley e of Sup Annual Units
-	1	2595	8/21/1940	CER	OSM	NW	NE	24	04S	ø	ĥ	4E 0.003	4E 0.003 STK	4E 0.003 STK	4E 0.003 STK 2.240297	4E 0.003 STK 2.240297 AFA
10627	1	2615	2/21/1941	CER	SPR	NE	WN	17	08S	64	m	E 0.01	E 0.01 STK	E 0.01 STK	E 0.01 STK 7.41	E 0.01 STK 7.41 AFA
10629	1	2596	2/24/1941	CER	SPR	SW	WN	20	06S		SE	35E 0.006	55E 0.006 STK	55E 0.006 STK	55E 0.006 STK 4.327149	55E 0.006 STK 4.327149 AFA
10637	1	ţ	3/24/1941	DEN	LAK	SE	NE	24	07S		62E	62E 0	62E 0 STK	62E 0 STK	62E 0 STK 39.987767	62E 0 STK 39.987767 AFS
10638	1	ł	3/24/1941	DEN	MSO	SE	SE	9	290		63E	63E 0.055	63E 0.055 STK	63E 0.055 STK	63E 0.055 STK 39.987767	63E 0.055 STK 39.987767 AFS
10654	5878	2633	4/28/1941	CER	OSW	SW	NE	16	075		63E	63E 0.007	63E 0.007 STK	63E 0.007 STK	63E 0.007 STK 5.002307	63E 0.007 STK 5.002307 AFA
10659	1	2637	5/10/1941	CER	RES	NE	SE	17	075		63E	63E 0	63E 0 STK	63E 0 STK	63E 0 STK 9.973925	63E 0 STK 9.973925 AFA
10736	1	2668	10/2/1941	CER	RES	NN	NE	35	048		63E	63E 0.005	63E 0.005 STK	63E 0.005 STK	63E 0.005 STK 3.98957	63E 0.005 STK 3.98957 AFA
10789	t	2722	3/4/1942	CER	OSM	NE	SE	26	S90		63E	63E 0.012	63E 0.012 STK	63E 0.012 STK	63E 0.012 STK 8.961188	63E 0.012 STK 8.961188 AFA
10887	1	ł	1/26/1942	CAN	SPR	NE	SE	13	06S		64E	64E 0.025	64E 0.025 STK	64E 0.025 STK	64E 0.025 STK 0	64E 0.025 STK 0
10888	į	1	10/26/1942	CAN	SPR	SW	SE	13	06S		64E	64E 0.025	64E 0.025 STK	64E 0.025 STK	64E 0.025 STK 0	64E 0.025 STK 0
10889	1	ł	10/26/1942	CAN	SPR	WN	SE	13	890	~	64E	64E 0.025	64E 0.025 STK	64E 0.025 STK	64E 0.025 STK 0	64E 0.025 STK 0
10896	Î	1	12/3/1942	CAN	SPR	SE	SE	9	078		64E	64E 0.025	64E 0.025 STK	64E 0.025 STK	64E 0.025 STK	64E 0.025 STK
11167	1	3073	9/14/1944	CER	SPR	SE	SE	9	078		64E	64E 0.003	64E 0,003 STK	64E 0.003 STK	64E 0.003 STK 2.178919	64E 0.003 STK 2.178919 AFA
11377	ţ	1	10/1/1945	DEN	SPR	WN	SE	17	230	0,	65E	65E 0.011	65E 0.011 STK	65E 0.011 STK	65E 0.011 STK 2.178919	65E 0.011 STK 2.178919 AFA
11378	T	4047	10/1/1945	CER	SPR	SE	WN	19	065		65E	65E 0.002	65E 0.002 STK	65E 0.002 STK	65E 0.002 STK 1.442383	65E 0.002 STK 1.442383 AFA
11378	Changed By:	52116	1	WDR	SPR	1	1	1	1	1	1	1	-	1		
11387	1	1	10/13/1945	WDR	SPR	SW	WN	20	065	0,	65E	65E 0.034	65E 0.034 STK	65E 0.034 STK	65E 0.034 STK 24.612578	65E 0.034 STK 24.612578 AFA
11525	1	3356	3/26/1946	CER	SPR	WS	WN	17	085	02	64E	64E 0.003	64E 0.003 STK	64E 0.003 STK	64E 0.003 STK 2.240297	64E 0.003 STK 2.240297 AFA
11705	3475	ł	10/5/1946	CAN	UG	WS	SE	13	065	07	64E	64E 0.04	64E 0.04 STK	64E 0.04 STK	64E 0.04 STK	64E 0.04 STK
12388	3475	4085	3/29/1948	ABR	SPR	SW	ŝ	17	065	03	65E	65E 0	65E 0 STK	65E 0 STK	65E 0 STK 0	65E 0 STK 0 AFS
12388	Changed By:	51260	I	CER	SPR	;	1	1	1	5	1	1	1			

uoisivid səənusək Aster Authority - Water Resources Division SE ROA 12922 A-23

A xibnaqqA

4 UG SE SE 23 04S 63E 5.4 4 UG SW NW 14 05S 69E 3.4 4 UG SE NE 12 06S 63E 0 R UG SE NE 12 06S 63E 0 V UG SE NE 17 06S 65E 0.017 R SPR SW SE 12 05S 64E 0.001 R SPR NW SW 2 05S 64E 0.008 R SPR SE SE 6 05S 65E 0.008	N UG SE SE 23 04S 63E 5.4 N UG SW NW 14 05S 69E 3.4 N UG SE NE 12 06S 63E 0 AN UG SE NE 17 06S 65E 0.017 ER SPR SW SE 17 06S 65E 0.001 ER UG SE SE 12 07S 64E 0.003 ER SPR NW SW 2 05S 65E 0.008 ER SPR NE SE 6 0.55 65E 0.003	AN UG SE SE 23 04S 63E 5.4 AN UG SW NW 14 05S 69E 3.4 DR UG SE NE 12 06S 63E 0 EN UG SE NE 12 06S 63E 0 EN UG SE NE 12 06S 63E 0 AN UG SE NE 12 06S 63E 0 EN UG SE NE 17 06S 65E 0.017 ER SPR SW SE 12 07S 64E 0.001 ER SPR NW SW 2 05S 64E 0.008 ER SPR NE SE 6 05S 65E 0.003 ER SPR NE NE 20 05S 65E 0.002	CAN UG SE SE 23 04S 63E 5.4 CAN UG SW NW 14 05S 69E 3.4 CAN UG SE NE 12 06S 69E 3.4 WDR UG SE NE 12 06S 63E 0 DEN UG SE NE 12 06S 63E 0 CER SPR SW SE 17 06S 65E 0.017 CER SPR NW SE 12 07S 64E 0.0017 CER SPR NW SW SE 12 07S 64E 0.0017 CER SPR NW SW SE 6 $05S$ 64E 0.003 CER SPR NE SE 6 $05S$ 65E 0.003 CER SPR NE NE 20 $05S$
NE 12 06S 63E 5 NE 12 06S 63E 0 V SE 17 06S 65E 0.017 V SE 17 06S 65E 0.017 V SE 12 07S 64E 0.017 SE SE 12 05S 64E 0.013 SE SE 6 05S 65E 0.008	NE 12 06S 63E 5 NE 12 06S 63E 0 V SE 17 06S 65E 0.017 V SE 17 06S 65E 0.017 SE 12 07S 64E 0.017 SE SE 6 05S 64E 0.008 SE SE 6 05S 65E 0.008 NE 18 05S 65E 0.003	NE 12 06S 63E 5 NE 12 06S 63E 0 V SE 17 06S 65E 0.017 V SE 17 06S 65E 0.017 V SE 12 07S 64E 0.017 V SW 2 05S 64E 0.008 V SW 2 05S 64E 0.008 SE 18 05S 65E 0.003 NE 18 05S 65E 0.003 NE 20 05S 65E 0.003	NE 12 06S 63E 5 NE 12 06S 63E 0 V SE 17 06S 65E 0.017 V SE 17 06S 65E 0.017 V SE 12 07S 64E 0.017 V SE 12 05S 64E 0.017 V SE 12 05S 64E 0.008 V SE 6 05S 64E 0.008 E NE 18 05S 65E 0.003 E NE 20 05S 65E 0.002 V NE 20 05S 65E 0.002 V NE 20 05S 65E 0.002
SE 17 06S 65E 0.017 SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 SW 2 05S 64E 0.008 SE 6 05S 64E 0.008	SE 17 06S 65E 0.017 SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 SW 2 05S 64E 0.008 SW 2 05S 64E 0.008 SE 6 05S 65E 0.008 NE 18 05S 65E 0.003	SE1706S65E0.017SE1706S65E0.017SE1207S64E0.01SW205S64E0.008SE605S65E0.008NE1805S65E0.003NE2005S65E0.002	SE1706S65E0.017SE1706S65E0.017SE1207S64E0.01SW205S64E0.008SE605S65E0.008NE1805S65E0.002NE2005S65E0.002SE1706S65E0.002
r SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 V SW 2 05S 64E 0.008 SE 5 05S 65E 0.008	I SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 / SW 2 05S 64E 0.008 SE 6 05S 65E 0.008 NE 18 05S 65E 0.003	I SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 I SW 2 05S 64E 0.008 I SE 6 05S 65E 0.008 I NE 18 05S 65E 0.003 I NE 18 05S 65E 0.003 I NE 20 05S 65E 0.003	I SE 17 06S 65E 0.017 SE 12 07S 64E 0.01 I SW 2 05S 64E 0.008 I SW 2 05S 64E 0.008 I SE 6 05S 65E 0.008 I NE 18 05S 65E 0.003 V NE 20 05S 65E 0.002 V SE 17 06S 65E 0.002
SE SE 12 07S 64E 0.01 NW SW 2 05S 64E 0.008 SE SE 6 05S 65E 0.008	SE SE 12 07S 64E 0.01 NW SW 2 05S 64E 0.008 SE SE 6 05S 65E 0.008 NE NE 18 05S 65E 0.003	SE SE 12 07S 64E 0.01 NW SW 2 05S 64E 0.008 SE SE 6 05S 65E 0.008 NE NE 18 05S 65E 0.003 SW NE 20 05S 65E 0.003	SE SE 12 07S 64E 0.01 NW SW 2 05S 64E 0.008 SE SE 6 05S 65E 0.008 NE NE 18 05S 65E 0.003 SW NE 20 05S 65E 0.002 NW SE 17 06S 65E 0.002
\$PR NW SW 2 05S 64E 0.008 \$PR SE SE 6 05S 65E 0.008	spr NW SW 2 05S 64E 0.008 spr SE SE 6 05S 65E 0.008 spr NE NE 18 05S 65E 0.003	\$PR NW SW 2 05S 64E 0.008 \$PR SE SE 6 05S 65E 0.008 \$PR NE NE 18 05S 65E 0.003 \$PR SW NE 20 05S 65E 0.003	iPRNWSW205S64E0.008ipRSESE605S65E0.008ipRNENE1805S65E0.003ipRSWNE2005S65E0.002spRNWSE1706S65E0.002
SPR SE SE 6 05S 65E 0.008	SPR SE SE 6 05S 65E 0.008 SPR NE NE 18 05S 65E 0.003	SPR SE SE 6 05S 65E 0.008 SPR NE NE 18 05S 65E 0.003 SPR SW NE 20 05S 65E 0.002	SPR SE SE 6 05S 65E 0.008 SPR NE NE 18 05S 65E 0.003 SPR SW NE 20 05S 65E 0.002 SPR NW SE 17 06S 65E 0.002
	SPR NE NE 18 05S 65E 0.003	SPR NE NE 18 05S 65E 0.003 SPR SW NE 20 05S 65E 0.002	SPR NE NE 18 05S 65E 0.003 SPR SW NE 20 05S 65E 0.002 SPR NW SE 17 06S 65E 0.002

Part C - Water-Related Effects Analysis

A	XIDU	addy
v	a state of	

₩-24

69882	69881	64679	64678	60194	60188	60187	60186	60185	60184	60183	54144	53992	53991	52366	52118	App.	
ł	1	ł	ł	1	Ĵ.	ł	Ĩ	T	ľ	l	ł	Ì	ł	10189	ł	Change of App.	
1	1	1	{	ł	1	1	1	I	ł	ł	1	İ	1	Í.	14740	Cert.	
4/18/2003	4/18/2003	12/11/1998	12/11/1998	7/1/1994	7/1/1994	7/1/1994	7/1/1994	7/1/1994	7/1/1994	7/1/1994	11/6/1989	10/17/1989	10/17/1989	8/3/1988	5/18/1988	File Date	n.
REP	RFP	RFP	RFP	DEN	RFP	RFP	DEN	CER	Status	ND							
511	UG	UG	UG	UG	UG	ug	NG	UG	NG	UG	UG	UG	UG	SPR	SPR	Source	NR Hy
WW	SE	SW	SE	SW	SE	WW	NW	SE	WW	WS	SE	NE	ŝ	NW	NE	2 POP	drog
WW	SW	WN	ŝ	ŝ	SW	WN	WW	SW	WN	SE	NE	NE	NE	WS	WN	Qtr Qtr	Iraph
20	24	24	30	13	24	18	25	24	18	13	12	15	4	2	17	POD	(Pag
DAS.	04S	07S	05S	04S	290	290	05S	05S	06S	POD Twn	e 6 o						
ASS	63E	63E	64E	63E	63E	64E	63E	63E	63E	63E	63E	64E	63E	64E	65E	POD Rng	f 7)
77677	27627	10	10	27627	27627	27627	27627	27627	27627	27627	0.2	10	თ	0.025	0.001	Div Rate (cfs)	r Delama
PWR	PWR	IRR	IRR	PWR	PWR	PWR	MUN	MUN	MUN	MUN	MM	MUN	MUN	STK	STK	Type of Use	r Valley
1	1	1	1	1	- [1	ł	1		1	1	ľ	1	1	1	Sup	
0	0	5120	5120	l	ł	1	ł	ł	ł	ł	1	7239.7	4343.82	18.10651	0.337579	Annual Duty	
AFA	AFA	AFA	AFA	ł	ł	1	ł	1	I	1	ł	AFA	AFA	AFA	AFA	Units	
Jacob, Beverly Joan	Jacob, Beverly Joan	Lincoln County Water District	Lincoln County Water District	Spencer, Robert Wallace	Spencer, Robert Wallace	Spencer, Robert Wallace	Spencer, Robert Wallace	Spencer, Robert Wallace	Spencer, Robert Wallace	Spencer, Robert Wallace	Delamar Minerals Company	Southern Nevada Water Authority	Southern Nevada Water Authority	H.H. Land and Cattle Company	Corp Presiding Bishop Church JC LDS	Owner of Record	

uoisinid seonosea veter Authority - Water Resources Division SE ROA 12924

and the second se	

A xibneqqA

				NDV	VR Hy	drog	raph	(Pag	e 7 o	f7)	r Delama	r Valley				
App.	Change of App.	Cert.	File Date	Status	Source	POD	Pop	POD	POD	POD Rng	Div Rate (cfs)	Type of Use	Sup	Annual Duty	Units	Q
69892	1	1	4/18/2003	RFP	UG	SE	SW	24	04S	63E	27627	MUN	ł	0	AFA	Jaco
69893	ł	1	4/18/2003	RFP	UG	WW	WW	25	04S	63E	27627	MUN	ł	0	AFA	Jaco
R04339	+	ł	5/3/1985	RES	SPR	WW	WS	ω	05S	64E	0	OTH	1	0	1	BLM
V01022A01	ł	ł	2/6/1911	VST	SPR	1	SE	36	05S	64E	0	STK	ł	0	AFA	LDS,
v01022a02	1	1	2/6/1911	VST	SPR	£	SE	36	S90	64E	0	STK	ł	0	AFA	Linco
V01022A03	ł	ſ	2/6/1911	VST	SPR	ł	WS	31	S90	65E	0.0625	STK	1	0	AFA	Duffi
V01022A04	:	1	2/6/1911	VST	SPR	1	WS	33	S90	65E	0.0125	STK	1	0	AFA	Ballo
V01022A05	1	ł	2/6/1911	VST	SPR	ł	SE	33	05S	65E	0.025	STK	ł	0	AFA	LDS,
V01022A06	ł	ł	2/6/1911	VST	SPR	1	ł	1	1	ł	0	STK	1	0	AFA	LDS,
V01398	1	ł	7/3/1915	ABR	SPR	SE	WW	11	05S	64E	0	STK	1	0	AFA	H.H.
V01398	Changed By:	52114	1	CER	SPR	ł	ł	ł	1	ł	l	ł	ł	1	1	1
V01399	ł	1	7/3/1915	VST	SPR	SE	ŝ	თ	05S	65E	0.025	STK	1	0	1	Henri
V01400	ł	1	7/12/1915	VST	SPR	NE	SE	ω	05S	64E	0.025	STK	ł	0	1	Duffir
V01418	1	1	9/14/1915	VST	SPR	SW	WS	34	05S	62E	0.025	STK	ł	D	1	Henri
V01419	ł	1	9/2/1915	VST	SPR	SW	WW	20	05S	65E	0.025	STK	ł	ł	1	Mack
V01420	ł	1	9/14/1915	VST	SPR	WN	WS	27	05S	62E	0.025	STK	1	0	1	Henrie
V01449	1	1	3/13/1916	VST	SPR	SW	SW	25	03S	62E	0.1	STK	ł	0	1	Neva
V01520	1	ł	10/8/1917	VST	SPR	SE	SE	9	07S	64E	0	STK	1	1	1	Gardr
V01550	1	I	3/21/1918	VST	SPR	SE	SE	15	07S	64E	0.025	STK	1	0.122756	AFA	Duffin
V01598	1	T	10/25/1918	VST	SPR	SE	SW	4	06S	64E	0,025	STK	1	ł	1	Linco
V01654	1	1	11/29/1919	VST	SPR	SW	SE	36	04S	63E	0.025	STK	ł	0.675158	AFA	Duffin
	-	ł	7/10/1923	VST	SPR	NE	WS	17	290	65E	0.05	STK	Ĺ	1.687895	AFA	Horn,

Part C - Water-Related Effects Analysis



This Page Intentionally Left Blank

A-26

Appendix A

Part C - Water-Related Effects Analysis

Appendix B

Application Points of Diversion Figures

SE ROA 12027

JA_4789

B.1.0 INTRODUCTION

This appendix contains figures showing a 2,500 ft buffer around SNWA's application points of diversion with imagery provided from the National Agriculture Imagery Program.





B-1





Figure B.1-1 Area Surrounding Application Point of Diversion 53987 in Cave Valley



Figure B.1-2 Area Surrounding Application Point of Diversion 53988 in Cave Valley

Appendix B

SE ROA 12030

B-3





Figure B.1-3 Area Surrounding Application Point of Diversion 53989 in Dry Lake Valley

B-4



Figure B.1-4 Area Surrounding Application Point of Diversion 53990 in Dry Lake Valley

Appendix B

SE ROA 12032

B-5





Figure B.1-5

Area Surrounding Application Point of Diversion 53991 in Delamar Valley



B-6

Appendix B



Figure B.1-6 Area Surrounding Application Point of Diversion 53992 in Delamar Valley

Appendix B

SE ROA 12034

B-7



This Page Intentionally Left Blank

B-8

Appendix B SE ROA 12035

Part C - Water-Related Effects Analysis

Appendix C

Spring Discharge and Temperature Measurements

SE ROA 12036

JA_4798

C.1.0 INTRODUCTION

This appendix describes spring discharge and temperature measurements at selected springs within Cave, Dry Lake, and Delamar valleys.





C-2

Appendix C

SE ROA 12038

and Delamar Valleve Drv Lake of Selected Springs in Cave Table C.1-1



Southern Nevada Water Authority - Water Resources Division

C.2.0 REFERENCES

- Bunch, R.L., and Harrill, J.R., 1984, Compilation of Selected Hydrologic Data from the MX Missile-Siting Investigation, East-Central Nevada and Western Utah, U.S. Geological Survey Open-File Report 84-702, 123 p.
- Carpenter, E., 1915, Ground Water in Southeastern Nevada: U.S. Geological Survey, Water-Supply Paper 365, 86 p.
- Hess, J.W., and Mifflin, M.D., 1978, A Feasibility Study of Water Production from Deep Carbonate Aquifers in Nevada, Desert Research Institute Publication No. 41054, 136 p.
- Pavelko, M.T., 2007, Spring Database for the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah: U.S. Geological Survey Data Series 272, 10 p.
- U.S. Geological Survey, 2004, National Water Information System (NWIS Web) data available on the World Wide Web, accessed (June 10, 2004, May 8, 2006, November 21, 2007, November 27, 2007) at URL http://waterdata.usgs.gov/nwis/.



This Page Intentionally Left Blank

Appendix C SE ROA 12040

C-4



Southern Nevada Water Authority

Hydrologic Data Analysis Report for Test Well 184W105 in Spring Valley Hydrographic Area 184



August 2009

This document's use of trade, product, or firm names is for descriptive purposes only and does not imply endorsement by the Southern Nevada Water Authority. Although trademarked names are used, a trademark symbol does not appear after every occurrence of a trademarked name. Every attempt has been made to use proprietary trademarks in the capitalization style used by the manufacturer.

Suggested citation:

Prieur, J.P., Farnham, I.M., and Fryer, W., 2009, Hydrologic data analysis report for Test Well 184W105 in Spring Valley: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. DAR-ED-0002, 81 p.



Hydrologic Data Analysis Report for Test Well 184W105 in Spring Valley Hydrographic Area 184

By: James P. Prieur,¹ Irene M. Farnham,¹ and William Fryer²

August 2009

- 1. Southern Nevada Water Authority, Las Vegas, NV
- 2. S.M. Stoller Corporation, Las Vegas, NV

SOUTHERN NEVADA WATER AUTHORITY Groundwater Resources Department Water Resources Division ♦ snwa.com

CONTENTS

List of	Figures	
List of	Acrony	ms and Abbreviations
ES.1.0	Execut	ive Summary
1.0	Introdu	iction
	1.1 1.2 1.3	Program Objectives.1-1Testing and Monitoring Program1-1Report Organization1-2
2.0	Well S	ite Description
	2.12.2	Hydrogeologic Setting2-12.1.1Regional Hydrogeologic Setting2-12.1.2Local Hydrogeologic Setting2-3Well Data2-52.2.1Test Well 184W1052-52.2.2Monitor Wells 184W506M and 184W504M2-52.2.3Water-Level Data2-9
3.0	Test D	escription and Background Data
	3.1 3.2 3.3 3.4	Site Activities3-1Test Equipment and Site Layout3-1Discharge Information3-1Instrumentation and Background Data3-2
4.0	Well H	lydraulics and Performance Testing
	4.1 4.2	Development4-14.1.1Development Results4.1Step-Drawdown Test4.2.1Well Performance and Specific Capacity4.2.2Well Loss Analysis4-3
5.0	Consta	nt-Rate Test Evaluation
	5.1 5.2 5.3 5.4	Data Review and Adjustments5-1Constant-Rate Test Data5-2Analytical Model Selection5-2Constant-Rate and Recovery-Test Analysis5-65.4.1Test Analysis Methodology5-65.4.2Test Analysis Result Summary5-85.4.3Barker GRFM Analysis5-85.4.4Cooper-Jacob Analysis5-10Discussion5-15
	5.5	Discussion





6.0	Water Chemistry			
	6.1	Ground	Indwater Sample Collection and Analysis	
	6.2	EPA Drinking Water Standards		
	6.3	Ground	Groundwater-Chemistry Results.	
		6.3.1	Field Results	6-2
		6.3.2	Major Constituents	6-3
		6.3.3	Trace and Minor Constituents	6-5
		6.3.4	Stable Isotopes and Environmental Tracers	6-5
			6.3.4.1 Hydrogen and Oxygen Isotopes	6-5
			6.3.4.2 Carbon Isotopes	6-6
			6.3.4.3 Chlorine-36/Chloride Ratios	6-7
			6.3.4.4 Strontium and Uranium Isotopes	6-7
		6.3.5	Radiological Parameters.	6-8
		6.3.6	Organic Compounds	6-8
	6.4	Summary		6-8
7.0	Refer	ences		7-1

Appendix A - CD-ROM Contents

A.1.0	Introd	Introduction	
	A.1.1	Photos	.A-1
	A.1.2	Read-Me File	.A-4
	A.1.3	Background Water-Level Data	.A-4
	A.1.4	Barometric-Pressure Data	.A-4
	A.1.5	Step-Drawdown Test Data	.A-4
	A.1.6	Constant-Rate Test	.A-4
	A.1.7	AQTESOLV	.A-4
	A.1.8	Water Chemistry	.A-5

Appendix B - Water-Chemistry Data

B .1.0	References		B-6
---------------	------------	--	-----

FIGUF Numbe	RES R TITLE PAGE
2-1	SNWA Exploratory and Test Wells in Spring Valley (as of August 2008) 2-2
2-2	Surficial Geology and Structural Features at Monitor Well 184W506M and Test Well 184W105
2-3	Test Well 184W105 Construction Schematic
2-4	Borehole Stratigraphic Column of Test Well 184W105 2-8
2-5	Monitor Well 184W506M Construction Schematic
2-6	Borehole Stratigraphic Column of Monitor Well 184W506M 2-12
2-7	Test Well 184W105 Historic Hydrograph
2-8	Monitor Well 184W506M Historic Hydrograph 2-14
2-9	Monitor Well 184W504M Historic Hydrograph 2-14
3-1	Hydrograph for Background Well 184W504M During Test Period
3-2	Local Barometric-Pressure Variation and Groundwater-Level Measurements at Monitor Well 184W506M
4-1	Linear Plot of Drawdown for Each Pumping Interval During Step-Drawdown Testing of Test Well 184W105
4-2	Linear Plot of Step-Test Drawdown and Depth-to-Pumping Level Various Discharge Rates for Test Well 184W105
4-3	Step-Test Specific Capacity versus Discharge Rate for Test Well 184W105
4-4	Evaluation of Head Loss Coefficients Using Hantush-Bierschenk Method from Step-Drawdown Test Results
5-1	Log-Log Data Plot of Drawdown versus Time from Monitor Well 184W506M 5-3
5-2	Semi-Log Data Plot of Drawdown versus Time from Monitor Well 184W506M 5-3
5-3	Log-Log Data Plot of Drawdown versus Time from Test Well 184W105 5-4
5-4	Semi-Log Data Plot of Drawdown versus Time from Test Well 184W105



FIGUI Numbe	RES (CONTINUED) R TITLE PAGE	
5-5	Monitor Well 184W506M Recovery Data Presenting Residual Drawdown versus the Log of the Ratio of t/t'	
5-6	Optimal Barker GRFM Solution Pumping Period Log-Log Plot	
5-7	Optimal Barker GRFM Solution Pumping Period Semi-Log Plot	
5-8	Optimal Barker GRFM Solution Drawdown Derivative for Monitor Well 184W506M	
5-9	Optimal Barker GRFM Solution Drawdown Derivative for Test Well 184W105 5-12	
5-10	Optimal Barker GRFM Solution, Test Well 184W105 Well Losses Removed 5-12	
5-11	Optimal Barker GRFM Optimal Solution Recovery Period	
5-12	Cooper-Jacob Analysis, Monitor Well, Early-Time	
5-13	Cooper-Jacob Analysis, Monitor Well, Late-Time	
6-1	Piper Diagram Illustrating Relative Major-Ion Compositions	
6-2	Stiff Diagrams Illustrating Major-Ion Concentrations	
6-3	Plot of δD versus $\delta^{18}O$	
A-1	184W105 Test Well Site, Facing Southwest	
A-2	184W105 Test Wellhead Equipment and Piping LayoutA-2	
A-3	184W105 Test Wellhead Equipment with Generator	
A-4	Discharge Piping, Facing East from Well Site 184W105A-3	
A-5	Discharge Location East of U.S. Highway 93 for Hydrologic Testing Performed at Test Well 184W105	
TAB Nume	ELES BER TITLE PA	GE
-------------	--	-----
2-1	Test Well 184W105 Borehole and Well Statistics	2-6
2-2	Monitor Well 184W506M Borehole and Well Statistics	-10
2-3	Measuring-Point Information	-13
4-1	Step-Drawdown Test Analysis	4-5
5-1	Measurement and Parameter Values Used for Analysis	5-6
5-2	Summary of Optimal Analysis Results	5-8
6-1	Trace Elements Present in Higher Concentrations in the Monitor Wells than in the Test Wells	6-5
B-1	Field and Analytical Results, Analytical Methods, Reporting Limits, and Maximum Contaminant Levels for Inorganic, Stable Isotopic, and Radiological Constituents in Groundwater Samples from Test Well 184W105 and Monitor Well 184W506M	3-1
B-2	Organic Compounds Analyzed in Groundwater Samples from Test Well 184W105, Including the EPA Method, Reporting Limit, and Maximum Contaminant Level	3-4





This Page Left Intentionally Blank

ACRONYMS

Barker GRFM	Barker generalized radial flow model
EPA	U.S. Environmental Protection Agency
GSLDFS	Great Salt Lake Desert Flow System
HA	hydrographic area
HSLA	high strength low alloy
MCL	maximum contaminant level
MS	mild steel
NAD83	North American Datum of 1983
NDOT	Nevada Department of Transportation
NTU	nephelometric turbidity unit
SNWA	Southern Nevada Water Authority
TDH	total dynamic head
UTM	Universal Transverse Mercator

ABBREVIATIONS

°C	degrees Celsius
afa	acre-feet per annum
amsl	above mean sea level
bgs	below ground surface
cm	centimeter
ft	foot
gal	gallon
gpm	gallons per minute
I.D.	inside diameter (of casing)
in.	inch
in. Hø	inches of mercury
L	liter
L lb	liter pound
L lb m	liter pound meter
L lb mEq	liter pound meter milliequivalent
L lb mEq mg	liter pound meter milliequivalent milligram
L lb m mEq mg mi	liter pound meter milliequivalent milligram mile
L lb mEq mg mi min	liter pound meter milliequivalent milligram mile minute
L lb m mEq mg mi min ml	liter pound meter milliequivalent milligram mile minute milliliter





ABBREVIATIONS (CONTINUED)

mrem	millirem
μg	microgram
μS	microsiemen
O.D.	outside diameter (of casing)
‰	per mil
pmc	percent modern carbon
pCi	picocurie
psi	pounds per square inch
rpm	revolutions per minute
yr	year

ES.1.0 EXECUTIVE SUMMARY

In support of its Clark, Lincoln, and White Pine Counties Groundwater Development Project, Southern Nevada Water Authority (SNWA) installed test and monitor wells in Spring Valley (Hydrographic Area 184) to evaluate hydrogeologic conditions. This report documents the collection, analysis, and evaluation of data obtained during the well development and hydraulic testing of Test Well 184W105 and Monitor Well 184W506M located in western Spring Valley, White Pine County, Nevada. This report also presents groundwater-level data collected at the site post-test through August 2008.

The development and hydraulic testing program of Test Well 184W105 was performed from February 26 through March 10, 2007. The test well and associated Monitor Well 184W506M are completed stratigraphically in the Ely Limestone to a depth of 1,160 ft bgs. The wells are completed in an unconfined, fractured, carbonate-rock aquifer system. Static depth to water in the test well is approximately 208 ft bgs.

The development phase pumping extracted 8,241,000 gallons of water and improved specific capacity, a ratio of discharge (*Q*) to drawdown (*s*) in the test well, from 56 to 70 gpm/ft at 2,400 gpm for a 24 percent improvement. A five-interval step-drawdown test was conducted at discharge rates ranging from 2,300 to 3,700 gpm to estimate the optimal pumping rate, evaluate well loss coefficients, and determine the discharge rate for the constant-rate test.

A 72-hour constant-rate test was performed at a target discharge rate of 3,000 gpm. Hydrogeologic data and diagnostic log-log and derivative drawdown data plots indicated that a dual-porosity conceptual model is the most appropriate primary solution method. The Barker generalized radial flow model was applied to the site data as the primary analytical solution. A secondary analytical solution using the Cooper-Jacob semi-log straight-line approximation was also performed for comparison. Analyses were performed using AQTESOLV evaluation software.

Results of the Barker analysis using the optimal best-fit of all site pumping and recovery data indicate an estimated hydraulic conductivity (*K*) of approximately 60.5 ft/day and a specific storage of 1.0×10^{-5} ft⁻¹. This equates to a storativity (*S*) of 9.46×10^{-3} assuming a saturated thickness of 946 ft. Matrix hydraulic conductivity (*K'*) for a fracture spacing of 10 ft is 6.30×10^{-3} ft/day. A secondary analysis for transmissivity (*T*) using the Cooper-Jacob solution indicates that the flow regime may have been dominated by radial flow near the end of the test period. Results of the analysis indicate a *T* of approximately 5.4×10^4 ft²/day and *S* of 2.05×10^{-4} . Assuming a saturated thickness of 946 ft, the resulting *K* value is 57 ft/day. A sensitivity analysis was performed for varying fracture spacing, anisotropy ratios, and matrix skin factors. The estimated effective saturated thickness used has a direct proportional relationship to the *K* value derived from *T*. Partial penetration of the test well was also evaluated.

Executive Summary

ES-1



Specific capacity during the last 12 hours of the 3,000 gpm, 72-hour constant-rate test ranged from 54.32 to 55.62 gpm/ft. The optimal initial operational pumping rate is projected to range up to 3,500 gpm based upon test results. A total of 22,652,000 gallons of water was extracted throughout the well development and hydraulic testing program.

Groundwater samples were collected from Test Well 184W105 and Monitor Well 184W506M and analyzed for a suite of chemical parameters. Stabilization of the water-quality parameters, measured in the field, was observed prior to sample collection. The chemistry of these samples was compared to that of other SNWA wells in the vicinity. All samples exhibited a calcium-magnesium-bicarbonate facies characteristic of groundwater of a carbonate-rock aquifer. Light stable isotope (δD and $\delta^{18}O$) compositions, typical of recharge at high elevations and cold temperatures, were observed for all groundwater samples. The isotopic composition of chloride ($^{36}Cl/Cl$) was also consistent with precipitation in the southwestern United States. The isotopic compositions of carbon (^{14}C and $\delta^{13}C$) and strontium ($^{87}Sr/^{86}Sr$) were indicative of groundwater interaction with carbonate minerals along the flow path.

1.0 INTRODUCTION

In support of its Clark, Lincoln, and White Pine Counties Groundwater Development Project, Southern Nevada Water Authority (SNWA) installed test and monitor wells in Spring Valley to evaluate hydrogeologic conditions. This report documents the collection, analysis, and evaluation of data obtained during the well development and hydraulic testing of Test Well 184W105 and Monitor Well 184W506M located in Hydrographic Area (HA) 184, Spring Valley, Nevada. The two wells are completed within the unconfined, fractured carbonate aquifer of the Ely Limestone stratigraphic unit. This report also presents groundwater-level data collected at the site post-test through August 2008. A separate document entitled *Geologic Data Analysis Report for Monitor Well 184W506M and Test Well 184W105 in Spring Valley* (Eastman and Muller, 2009) includes the documentation and detailed results for the drilling program, including evaluation of lithology, structural features, drilling parameters, and geophysical logs.

1.1 Program Objectives

The objectives of developing Test Well 184W105 were to remove any remaining drilling fluids and improve the hydraulic connection with the formation. This phase of development consisted of pump and surge activities and was in addition to the airlifting and swabbing development that were performed immediately after well installation.

Hydraulic testing was performed to evaluate well performance and to provide data on the hydraulic properties of the carbonate-rock aquifer in the vicinity of the test well. Groundwater samples were also collected for laboratory analysis to evaluate the groundwater chemistry of the aquifer in the vicinity of the well.

1.2 Testing and Monitoring Program

The well development and hydraulic testing program was performed from February 26 through March 10, 2007, and consisted of the following activities:

- Final well development, using surging methods
- Well hydraulic testing and performance evaluation, using a five-interval step-drawdown test
- Aquifer-property evaluation testing, using a 72-hour constant-rate test and subsequent water-level recovery measurements
- Collection of groundwater samples for laboratory chemical analysis

1-1



A complete schedule of test program activities is presented in Section 3.1.

Monitor Well 184W506M is part of the Spring Valley regional baseline water-level monitoring network. Water-level data have been collected continuously from this location since its testing.

1.3 Report Organization

This report is divided into seven sections and two appendixes.

Section 1.0 presents introductory information about the testing program and this report.

Section 2.0 describes the well site hydrogeology and summarizes the well construction, borehole lithology, and water-level data for the test and monitor wells.

Section 3.0 describes the test program and presents information on test instrumentation and background data.

Section 4.0 presents the analysis and evaluation of the results from the test well development and performance step-drawdown testing.

Section 5.0 presents the analysis and evaluation of the constant-rate aquifer test.

Section 6.0 presents the groundwater-chemistry results and evaluation.

Section 7.0 provides a list of references cited in this report.

Appendix A presents site photos and documentation of site physical and transducer test data. The data package on the CD-ROM includes regional background monitor well water levels, barometric pressure, and hydrologic data collected from the test and monitor wells.

Appendix B presents the water-chemistry laboratory data reports.

2.0 Well Site Description

SNWA Test Well 184W105 is located on the west side of Spring Valley, on Bureau of Land Management property, approximately 14 mi north of the Lincoln County and White Pine County boundary in Section 26, T12N, R66E. Access to the site is from U.S. Highway 93 along a dirt road to the west approximately one half mile. A topographic map with the site location and other SNWA test and monitor wells installed as of August 2008 is presented on Figure 2-1.

Two monitor wells were used during testing for observation and background control purposes. Monitor Well 184W506M, located 212 ft to the west-northwest of the test well, was used as an observation well during testing. Monitor Well 184W504M, used to observe background conditions during testing, is located approximately 8 mi south of the test well.

2.1 Hydrogeologic Setting

This section presents the regional and local hydrogeologic setting of the Test Well 184W105 well site. Previous studies and reports that detail the regional hydrogeology are referenced. A description of the local hydrogeologic setting is provided and is based on field mapping and review of existing hydrogeologic and geophysical information.

2.1.1 Regional Hydrogeologic Setting

Spring Valley, located in east-central Nevada, is approximately 120 mi in length and averages approximately 16 mi in width. The valley is located within the Basin and Range province and is an up-gradient basin within the Great Salt Lake Desert Flow System (GSLDFS). It is bounded by the Schell Creek Range to the west, the Antelope Range to the north, the Snake Range and Limestone Hills to the east, the Wilson Creek Range to the south, and the Fortification Range to the southwest. Adjacent valleys are shown in Figure 2-1.

The primary aquifer systems within Spring Valley are carbonate and basin fill, with a volcanic aquifer occurring in the southwest portion of the valley. Extensive north-south-trending range-front faults and related structures are the primary control of groundwater flow in the carbonates and are present on both the east and west sides of the valley. The local discharge of groundwater in south-central Spring Valley is through the basin fill generally toward the center axis of the valley with discharge occurring through evapotranspiration. Regional groundwater flow in the southern portion of Spring Valley is postulated to occur south of the Snake Range through fractures in the carbonates of the Limestone Hills into Hamlin Valley and through to Snake Valley.

Numerous studies related to Spring Valley and adjacent basins have been performed since the late 1940s. These studies have included water-resource investigations, geologic and hydrogeologic

2-1





Figure 2-1 SNWA Exploratory and Test Wells in Spring Valley (as of August 2008)

investigations, recharge and discharge estimations, and other hydrologic studies. The regional hydrogeologic framework and a summary of results of previous studies have been presented in several reports. These reports include:

- Water Resources Appraisal of Spring Valley, White Pine and Lincoln Counties, Nevada (Rush and Kazmi, 1965)
- *Major Ground-Water Flow Systems in the Great Basin Region of Nevada, Utah, and Adjacent States* (Harrill et al., 1988)
- Geologic and Hydrogeologic Framework for the Spring Valley Area (SNWA, 2006a)
- Summary of Groundwater Water-Rights and Current Water Uses in Spring Valley (SNWA, 2006b)
- Water Resources Assessment for Spring Valley (SNWA, 2006c)
- Geology of White Pine and Lincoln Counties and Adjacent Areas, Nevada and Utah—The Geologic Framework of Regional Groundwater Flow Systems (Dixon et al., 2007)
- Water Resources of the Basin and Range Carbonate-Rock Aquifer System, White Pine County Nevada, and Adjacent Areas in Nevada and Utah (Welch et al., 2008)
- 2008 Spring Valley Hydrologic Monitoring and Mitigation Plan Status and Data Report (SNWA, 2009)

2.1.2 Local Hydrogeologic Setting

The site location was selected after conducting a geologic reconnaissance of the area including field mapping, review of regional geophysical and well data, and evaluation of surface structural features using aerial photography. Regional data and geologic mapping in the vicinity indicate the presence of faulting and related structures at the site.

Quaternary surface alluvium overlays the Pennsylvanian-Permian Ely Limestone at the well site. The Ely Limestone is light-olive gray to medium-gray limestone composed almost entirely of organic detritus and is approximately 1,900 to 2,500 ft thick in this area (Hose and Blake, 1976). This formation also includes yellowish-gray to tan, silty limestone layers with some portions that are dolomitic. The Pennsylvanian-Permian Ely Limestone is inferred to be underlain by Mississippian Scottie Wash Quartzite, which is underlain in turn by the Mississippian Chainman Shale. The two units may act as local and regional aquitards.

The test and monitor wells are situated along strike and within the damage zone of faults exposed in the nearby Ely Limestone outcrops to the northwest. The orientation of the faults relative to the test and monitor well, including measured dips, is presented in Figure 2-2. A detailed hydrogeologic cross section through the test site was not presented because of the limited available data. A more







Figure 2-2 Surficial Geology and Structural Features at Monitor Well 184W506M and Test Well 184W105

JA_4821

detailed discussion of local geologic structure is presented in the Geologic Data Analysis Report (Eastman and Muller, 2009).

2.2 Well Data

Test Well 184W105 and Monitor Well 184W506M are completed in the unconfined, fractured carbonate-rock aquifer, stratigraphically in the Pennsylvanian-Permian Ely Limestone. Unsaturated Quaternary surface alluvium overlays the carbonate rock at this location to a depth of 15 to 35 ft.

Detailed geologic data for lithologic and hydrogeologic evaluation were collected during drilling and field mapping. This included collection and identification of drill cuttings, documentation of drilling parameters including penetration rate, fluid loss and mud viscosity, and downhole geophysical logging. A detailed presentation and analysis of the geologic data at this site, including local structural features, are presented in the Geologic Data Analysis Report for this site (Eastman and Muller, 2009). Summary data for these wells are provided in Sections 2.2.1 and 2.2.2 of this report.

Based on the drill cuttings for the two wells, the Ely Limestone is commonly silty in the upper third of the boreholes and becomes less silty and more cherty with depth. This variation is in accord with the description of the Ely Limestone by Drewes (1967). Fracturing is commonly evident in the cuttings, often associated with clay-rich intervals and/or abundant calcite veinlets. Fracturing with calcite and/or clay zones was noted in both wells from about 200 to 300, 460 to 550, 750 to 940, and 980 to 1,040 ft bgs.

Geophysical data indicate open fractures associated with clay-rich intervals between 490 and 575 ft bgs in both boreholes. The data also suggest open fracturing from 660 to 725 and 795 to 870 ft bgs in Monitor Well 184W506M and from 760 to 810 and at 950 ft bgs in Test Well 184W105.

2.2.1 Test Well 184W105

Test Well 184W105 was drilled to a total depth of 1,160 ft bgs between October 20 and November 6, 2006, using mud rotary techniques. A 40-in. O.D. conductor casing was placed to a depth of 59.3 ft bgs and grouted in place. After the borehole was advanced to completion depth, downhole geophysical logging was performed. A 20-in. O.D. completion string, including approximately 696.78 ft of Ful-Flo louvered screen, was then installed. The gravel pack extends from a depth of 50 ft bgs to the bottom of the borehole. A summary chart of Test Well 184W105 drilling and well construction statistics is presented in Table 2-1, and a well construction schematic is presented on Figure 2-3. The borehole lithologic log for Test Well 184W105 is presented in Figure 2-4.

2.2.2 Monitor Wells 184W506M and 184W504M

Monitor Well 184W506M was completed at a depth of 1,160 ft bgs between October 9 and October 19, 2006. A 20-in. O.D. conductor casing was set to a depth of 77.5 ft bgs and grouted in place. A 14.75-in. borehole was then advanced to completion depth. The 8-in. nominal-diameter completion string, including approximately 690 ft of slotted casing, was placed in the open borehole. No gravel pack was used in the well. A summary chart of well drilling and well construction statistics



LOCATION DATA Estimated Coordinates	N 4,306,176 m; E 713,991 m (UTM, Zone 11, N	IAD83)			
Ground Elevation	6,007.303 ft amsl				
DRILLING DATA Spud Date	10/20/2006				
Total Depth (TD)	1,160 ft bgs				
Date TD Reached	10/30/2006				
Date Well Completed	11/7/2006				
Hole Diameter	46-in. from 0 to 60 ft bgs 26-in. from 60 to 1,160 ft bgs				
Drilling Techniques	Conventional Circulation from 0 to 160 ft bgs Reverse Circulation from 160 to 1,160 ft bgs				
Drilling Fluid Materials Used	Aqua Clear (5) 25 galGranulated Chlorine (1) 5 galMax-Gel (50) 50-lb bagsGel (653) 50-lb bagsSoda Ash (66) 50-lb bagsBiCarb (1,084) 50-lbDrisPac (28) 50-lb bagsCalcium (10) 50-lb bagsQuick Gel (1084) 50 lb bagsCement (7) Supersacks				
Drilling Fluid Properties	Viscosity Range = 34 to 75 sec/qt Weight Range = 8.5 to 10.0 lbs Filtrate Range = 5.1 to 28 ml Filter Cake Range = 1/32 to 3/32 in.	Average = 45.8 Average = 9 Average = 13 Average = 1.6/32nd			
CASING DATA	40-in. MS Conductor Casing from 0 to 59.3 ft bgs 20-in. HSLA Completion Casing from +2 to 1,135.13 ft bgs				
WELL COMPLETION DATA 60 ft of 3-in. gravel sounding tube from 0 to 60 ft 419.62 ft of blank HSLA 20-in. casing from +2 to 417.62 ft bgs 696.78 of 20-in. Ful-Flo louver screen from 417.62 to 1,114.4 ft bgs 20.40 ft blank 20-in. sump MS casing from 1,114.4 to 1,134.80 ft bgs 0.33 ft bullnose CS casing from 1,134.80 to 1,135.13 ft bgs Cement, Plug and Gravel Pack Depth 0 to 60 ft on outside of conductor casing (cement) 50 to 1,160 ft from bottom of Conductor Casing to TD (1/2 in. gravel pack) 45 to 50 ft bgs sand 0 to 45 ft bgs grout outside of completion casing, and inside of conductor					
MONITOR WELL	Static Water Level: 208.69 ft bgs (9/23/08) Groundwater Elevation: 5,798.61 ft amsl				
DRILLING CONTRACTOR Lang Exploration Drilling					
GEOPHYSICAL LOGS BY	ervices (Prescott, Arizona)				
OVERSIGHT	Southern Nevada Water Authority				

Table 2-1Test Well 184W105 Borehole and Well Statistics



Figure 2-3 Test Well 184W105 Construction Schematic

Depth ft bgs)	Lithology	Unit	Lithologic Description
0		QTa	Recent and older silty gravel (GM) consisting of limestone clasts. 15 ft thick.
		₽Pe	White to light gray limestone, bleached and finely crystallized. 35 ft thick.
100-	1	FB	Light to dark gray, varicolored fault breccia with limestone fragments in sandy to clayey calcareous matrix, partly to moderately cemented. 80 ft thick.
			Tan to orange, pale tan, light brown limestone, silty to argillaceous and fossiliferous. Calcite and occasional silica veinlets. 40 ft thick.
200-			Light tan to tan, light gray to gray limestone, silty to argillaceous, and occasionally fossiliferous; tan to gray, pale tan chert, abundant 170 to 180 ft bgs. Calcite veinlets on fractures. 60 ft thick.
			Light tan to tan, light gray to gray limestone, micritic to recrystallized often silty to argillaceous; minor pale tan chert. Calcite veinlet orange to tan clay and occasional limonite on fractures. 60 ft thick.
300-			Limestone as above, with minor chert; occasional limy siltstone beds. Intervals of abundant pale yellow, tan to orange clay, calcareous, silty, occasional limonite. 70 ft thick.
400-			Tan to gray, brownish gray limestone, microcrystalline to micritic, often crystalline, occasionally sucrosic, occasionally fossiliferous and argillaceous; minor chert. Occasional calcite veinlets, weak limonite. 60 ft thick.
			Tan to gray, brownish gray limestone, micritic to crystalline, occasionally sucrosic, fossiliferous, or argillaceous. Minor pale orange, tan, light gray clay, calcareous, limonitic. Occasional to common calcite veinlets. 40 ft thick.
			Limestone, as above, fossiliferous or argillaceous; gray to tan, brown chert. Minor to moderate pale orange, tan, light gray clay, calcareous, limonitic, 20 ft thick.
500-	1		Tan to light gray limestone, micritic to crystalline; minor chert. Occasionally fossiliferous. Common calcite veinlets. Clay common fractures. 30 ft thick.
			Light gray, gray, light tan, tan, orange limestone, micritic to recrystallized. Intervals of abundant light brown to tan clay 510 to 520 and 550 to 560 ft bgs. 50 ft thick.
600-	1		Tan, tan-gray, orange-tan limestone, micritic, often sandy to silty, occasionally fossiliferous; minor chert. Occasional to abundant calcite veinlets. 60 ft thick.
		PPe	Light gray to tan, gray, brown limestone, micritic to microcrystalline, recrystallized, occasionally sandy to silty, and occasionally fossiliferous. Occasional calcite veinlets, abundant 660 to 670 ft bgs with iron oxides. 50 ft thick.
700-			Limestone as above; abundant translucent gray, tan to gray, pinkish tan chert, 710 to 730 ft bgs. Variable calcite veinlets and occasional orange to pale tan clay. 60 ft thick.
			Limestone as above, no chert. Occasional orange to reddish orange clay. 20 ft thick.
1.43	/		Light tan to tan, light gray to gray, pinkish gray limestone, micritic to microcrystalline, recrystallized, fossiliferous. Minor clay. Calcite veinlets common, often with limonite. 20 ft thick.
800-			Limestone, as above, occasional fossil fragments; moderate chert. Light gray to orange-gray clay. Calcite veinlets rare, chalcedonic silica in chert. 20 ft thick.
			Limestone, as above, fossiliferous zones; translucent gray, tan-gray, brown-gray chert. Abundant calcite veinlets and veins, often with limonite; occasional silica veinlets. 60 ft thick.
			Pale pink to tan, tan-gray limestone, micritic, to microcrystalline. 20 ft thick.
900-	1		Light gray to tan, gray, and brown limestone, micritic to sucrosic, occasionally fossiliferous; minor translucent very dark gray to light gray chert. Calcite veinlets common. Minor clay. 60 ft thick.
4			Light gray to gray to brownish gray limestone, micritic to sucrosic, occasionally fossiliferous; minor translucent gray to tan-gray che White to grange clay with calcite fragments 950 to 960 ft bgs. Occasional calcite veinlets, 30 ft thick
			Pinkish gray, tan, gray to brown-gray, brown limestone, micritic to microcrystalline, often silty; abundant gray to tan-gray chert, 980 to 990 ft bgs. Occasional calcite veinlets, 30 ft thick.
1000-			Tan, gray to brown-gray, brown limestone, micritic to microcrystalline, fossiliferous; minor to moderate gray chert. Occasional calcite veinlets, common 990 to 1,000 ft bgs. 30 ft thick.
			Tan, gray to brown-gray, brown limestone, micritic to microcrystalline, often silty; occasional limy siltstone; moderate chert 1,050 to 1,060 ft bgs. 40 ft thick.
1100-			Gray to brown-gray limestone, as above; moderate to abundant translucent pale tan, tan-gray, gray, brown-gray chert. 50 ft thick.

Tan, gray to brown-gray, brown limestone, as above, often very fossiliferous. Occasional calcite veinlets. 50 ft thick.



Figure 2-4 Borehole Stratigraphic Column of Test Well 184W105



2-8



for Monitor Well 184W506M is presented in Table 2-2, and a well construction schematic is presented on Figure 2-5. The borehole lithologic log for Monitor Well 184W506M is presented in Figure 2-6.

Monitor Well 184W504M was monitored during the hydraulic testing to observe regional groundwater trends and to identify outside influences affecting regional water levels, such as changes in barometric pressure, earthquakes, and lunar effects. The hydrologic conditions affecting the water levels in this well are expected to be the same as those affecting the test well. This well is also completed in the unconfined, fractured carbonate-aquifer system. The 8-in.-diameter well is completed at a depth of 1,020 ft bgs with an open borehole interval of 60 to 1,040 ft bgs.

2.2.3 Water-Level Data

Depth-to-water measurements were obtained at the wells relative to a marked temporary or permanent reference measuring point. Professional survey elevations for the measuring points and ground-surface elevations for the wells are presented in Table 2-3.

Static groundwater-elevation data have been collected on a continuous basis at Monitor Wells 184W506M and 184W504M from just preceding the test to present. These wells are currently equipped with In-Situ Level TROLL 700 integrated transducers. Physical measurements are collected from the test well on a six-week to quarterly frequency. The two monitor wells are included in the SNWA regional groundwater monitoring network.

Static groundwater elevation is approximately 5,798 to 5,799 ft amsl at Test Well 184W105, which corresponds to a depth to water of approximately 208 to 208.5 ft bgs. Static groundwater elevation at Monitor Well 184W506M is approximately 5,799 ft amsl, which corresponds to a depth to water of 215 ft bgs. Background well 184W504M static groundwater elevation is approximately 5,800 ft amsl and approximately 100 ft bgs. Period-of-record hydrographs for the wells are presented on Figures 2-7 through 2-9. The hydrograph for the background well highlights time intervals during this test and a later unrelated test performed at another test well, 184W103, located adjacent to 184W504M. Static water levels have remained within a narrow range since the test period. A detailed background hydrograph at 184W504M during the testing period is presented in Section 3.4.



	A			
LOCATION DATA Estimated Coordinates	N 4,306,214 m; E 713,940 m (UTM, Zone 11	I, NAD83)		
Ground Elevation	6,014.0370 ft amsl			
DRILLING DATA Spud Date	10/09/2006			
Total Depth (TD)	1,160 ft bgs			
Date TD Reached	10/17/2006			
Date Well Completed	10/19/2006			
Hole Diameter	26-in. from 0 to 79.5 ft bgs 14.75-in. from 79.5 to 1,160 ft bgs			
Drilling Techniques	Drilling TechniquesConventional Circulation from 0 to 220 ft bgsReverse Circulation from 220 to 1,160 ft bgs			
Drilling Fluid Materials Used	Max-Gel = (42) 50-lb bagsBiCarb (5) 50-lb bagsSoda Ash = (12) 50-lb bagsQuick Gel (98) 50-lb bagsDrisPac = (14) 50-lb bagsCalcium (4) 50-lb bagsEZ-Mud = (7) 5-gal bucketsCement (25) 98 lb sacksGel (1915) 50-lb bagsCement (3) Supersacks			
Drilling Fluid Properties	Viscosity Range = 32 to 166 sec/qt Weight Range = 8 to 9 lbs Filtrate Range = 10 to 22.8 ml Filter Cake Range = 1/32 to 4/32 in.	Average = 48.5 Average = 8.9 Average = 15.2 Average = 1.96/32nd		
CASING DATA	20-in. MS Conductor Casing from 0 to 77.5 ft bgs 8-in. MS Completion Casing from +2.4 to 1,140.33 ft bgs			
WELL COMPLETION DATA	432.03 ft of blank MS 8-in. casing from +2.4 to 429.63 ft bgs 690.37 ft of slotted MS 8-in. casing from 429.63 to 1,120 ft bgs 20 ft blank sump MS 8-in. casing from 1,120 to 1,140 ft bgs 0.33 ft bullnose CS casing from 1,140 to 1,140.33 ft bgs			
	Cement Depth 0 to 79.5 ft on outside of conductor casing			
WATER	Static Water Level: 215.35 ft bgs Groundwater Elevation: 5,798.68			
DRILLING CONTRACTOR	Lang Exploration Drilling			
GEOPHYSICAL LOGS BY	Raymond Federwisch, Geophysical Logging Services (Prescott, Arizona)			
OVERSIGHT	Southern Nevada Water Authority			

Table 2-2Monitor Well 184W506M Borehole and Well Statistics



Figure 2-5 Monitor Well 184W506M Construction Schematic

t bgs)	Lithology	Unit	Lithologic Description
0		QTa	Recent and older silty gravel (GM) consisting of limestone and ash-flow tuff clasts. 15 ft thick.
		₽Pe	White to light gray limestone, bleached, finely recrystallized. 45 ft thick.
100	1	FB	Light to dark gray, blue-gray fault breccia with limestone fragments in a sandy to clayey calcareous matrix, partly cemented. 50 ft thick.
100-		PPe	Light tan to gray, orange limestone, fine-grained micritic, occasionally coarser where brecciated (~10%). Calcite veinlets and calcite on healed fractures. Weak pervasive limonite. 20 ft thick.
		FB	Fault breccia as above. 10 ft thick. Tan to orange limestone, micritic to fine sucrosic, commonly silty with tan-orange siltstone beds. Occasional to abundant calcite veinlets and limonite on fractures. 30 ft thick.
200-	V		Tan, gray, olive-gray, brownish gray limestone, micritic to fine sucrosic, occ detrital, fossiliferous, and often silty; minor tan to brown, brownish gray siltstone. Occasional calcite veinlets and limonite on fractures. 80 ft thick.
300-			Tan to olive gray, orange, brownish gray limestone, micritic to fine sucrosic, fossiliferous, commonly silty; occasional horizons with tan to tan-gray chert. Occasional to moderate calcite and silica veinlets, limonite on fractures. Minor clay. 50 ft thick.
			Tan to olive gray, orange, brownish gray limestone, micritic to fine sucrosic, fossiliferous, commonly silty; occasional horizons with light gray to gray chert. Occasional to abundant calcite veinlets and limonite on fractures. Minor clay. 50 ft thick.
5			Light to medium gray, tan limestone, micritic to fine sucrosic, commonly silty with up to 30% tan to orange siltstone beds. Occasion to common calcite veinlets and limonite on fractures. Pale yellow to gray, orange clay on fractures and surfaces. 20 ft thick.
400-			Light to medium gray, tan limestone, micritic to crystalline. Occasional to common calcite veinlets and limonite on fractures. 30 ft the Light tan to gray, brownish gray to olive-gray limestone, micritic to crystalline, occasionally sucrosic, often silty, detrital; pale tan chert to 15%. Moderate to abundant orange to yellow clay, calcareous, often limonitic. Occasional to common calcite veinlets. Occasional limonite on fractures. 30 ft thick.
			Limestone as before, minor chert. Occasional brecciated fragments, 450 to 460 ft bgs. 30 ft thick.
	/		White to gray, light tan, tan, orange limestone, micritic to recrystallized, fossiliferous, commonly silty. Abundant light gray to cream clay, 470 to 480 ft bgs. Occasional limonite and manganese oxides. 40 ft thick.
500-			White to tan, orange-tan, gray limestone, micritic, fossiliferous. Minor clay. Occasional limonite and manganese oxides. 20 ft thick.
			White to gray, light tan, tan limestone, micritic. Intervals of abundant light brown to tan clay. Occasional calcite veinlets. 40 ft thick.
600-			Light gray to tan, gray, brown limestone, micritic, commonly sandy to silty and often fossiliferous. Occasional calcite veinlets. Abundant tan to orange clay, 620 to 630 ft bgs. 70 ft thick.
		PPe	Light gray to tan, gray, brown limestone, micritic, commonly sandy to silty with occasional siltstone beds, and often fossiliferous. Light to dark gray, tan chert. Occasional calcite veinlets. 70 ft thick.
700-			Light tan to light brown, light gray to gray limestone, micritic to sucrosic, crystalline, occasionally fossiliferous; occasional siltstone. Zones of abundant tan to light brown, yellow-orange to orange clay, calcareous. 30 ft thick.
			Gray, brownish gray, tan-gray limestone, micritic, often fossiliferous; chert as above. 70 ft thick.
800-	1		Gray, brownish gray, tan-gray limestone, micritic, occasionally fossiliferous, often silty; light gray to gray, tan chert. Calcite veinle common, often vuggy. 80 ft thick.
900-			Gray to brown limestone, micritic, occasionally fossiliferous, often silty; minor chert as above. Calcite veinlets common, often vuggy. 30 ft thick.
			Gray to brown limestone, micritic, occasionally fossiliferous; occasional limy siltstone. Light to dark gray, tan chert to 10%. Calcite veinlets common, often vuggy. 30 ft thick.
6			Light to dark gray, brown-gray, tan limestone, micritic to sucrosic, occasionally fossiliferous. Minor to moderate tan, gray to dark g chert. 50 ft thick.
1000-			Limestone as above with clay and occasional limonite; minor to moderate pale tan, tan-brown chert. Calcite veinlets common to 1,020 ft bgs, then occasional. Abundant clay 990 to 1,000 ft bgs, with abundant chert an silica veinlets. 50 ft thick.
			Gray to brown-gray, tan limestone, micritic, often argillaceous and silty, occasionally fossiliferous; light to dark gray chert. Minor of Occasional calcite veinlets. 40 ft thick.
1100-			Gray to brown-gray, tan limestone, micritic, occasionally fossiliferous; light to dark gray chert. Occasional calcite veinlets. 40 ft thick.



Light to dark gray, brown limestone, micritic, occasionally fossiliferous, occasional calcite veinlets. 40 ft thick.



Figure 2-6 Borehole Stratigraphic Column of Monitor Well 184W506M



2-12



Table 2-3 Measuring-Point Information

Well ID	Well Use During Testing	UTM Northing (m) ^a	UTM Easting (m) ^a	Temporary MP (ft amsl)	Permanent MP (ft amsl)	Ground Surface Elevation (ft amsl)
184W105	Test Well	4,306,176	713,991	6,013.10	6,009.15	6,007.30
184W506M	Observation Well	4,306,214	713,940	6,016.59	6,016.44	6,014.04
184W504M	Background Well	4,293,712	713,647	5,901.44	5,901.44	5,900.11

^aUniversal Transverse Mercator, North American Datum of 1983, Zone 11N, Meters MP = Measuring Point



Figure 2-7 Test Well 184W105 Historic Hydrograph



Figure 2-8 Monitor Well 184W506M Historic Hydrograph



Note: An additional aquifer test was performed on Well 184W103 located near 184W504M as indicated on figure.

Figure 2-9 Monitor Well 184W504M Historic Hydrograph

3.0 Test Description and Background Data

This section describes the activities, pump equipment, and monitoring instrumentation associated with development and testing of 184W105. Background hydrologic data and regional trends are also presented and evaluated in this section.

3.1 Site Activities

The following summarizes the development and testing activities performed in 2007 at the well site:

- February 26 to 27: Developed the test well using surge and pump methods. The well was developed at rates ranging from 1,500 to 2,625 gpm.
- February 28: Performed initial step-drawdown test at discharge rates ranging from 1,750 to 2,500 gpm.
- March 2: Performed 12-hour constant-rate test at 2,300 gpm.
- March 3: Reset pump intake 100 ft deeper.
- March 4 to 5: Performed an additional 21 hours of development using surging methods. Pumped at rates ranging from 2,400 to 3,640 gpm.
- March 6: Performed step-drawdown test at rates ranging from 2,300 to 3,700 gpm.
- March 7 to 10: Performed 72-hour constant-rate test at 3,000 gpm and collected recovery data.

3.2 Test Equipment and Site Layout

A Johnson Pump Company vertical line shaft turbine pump was used in Test Well 184W105. The intake was initially set at 283 ft bgs and then lowered to 383 ft bgs prior to the step-drawdown test. The test well transducer was set at 278 ft below the measuring point and then lowered to 358 ft when the intake was lowered. A pump discharge line check valve was not used during the test to allow more effective development activities.

3.3 Discharge Information

Pumped water was discharged east of the site through approximately 2,000 ft of 12-in.-diameter piping. To comply with Nevada Department of Transportation (NDOT) permit requirements, the line



was reduced to an 8-in.-diameter assembly in order to pass through a culvert under U.S. Highway 93. The discharge pipe/line then was expanded back to 12-in. diameter east of the highway. The discharge point was located near the edge of the NDOT right-of-way on the east side of U.S. Highway 93.

A total of 22,652,000 gallons was pumped over the course of the development and testing periods for Test Well 184W105.

3.4 Instrumentation and Background Data

Regional and background water levels were continuously recorded prior to, during, and after the test period. Groundwater levels in Test Well 184W105 were recorded during the test period using an In-Situ HERMIT 3000 Data Logger. Test Well 184W105 was equipped with an In-Situ 250 psi pressure transducer. Monitor Well 184W506M and background well 184W504M were equipped with an In-Situ Level TROLL 700 integrated transducer. Barometric pressure was recorded at the test well and at evapotranspiration (ET) station SV1 located approximately 8 mi south-southeast of the test well.

Manual measurements were performed at both the test and monitor wells using an Enviro-Tech 1,000-ft electronic water-level indicator probe at prescribed time intervals and in accordance with SNWA standards. Groundwater-chemistry samples were collected and analyzed on site regularly for pH, conductivity, temperature, and turbidity throughout the testing period. Program test data are presented in data files on the CD-ROM that accompanies this report.

Data collected from background well 184W504M were used to identify any regional trend in groundwater level during the test period. A depth-to-water hydrograph for background well 184W504M during the testing period is presented on Figure 3-1.

The hydrograph for background well 184W504M indicates no significant trend that would influence the results of the tests. During the constant-rate test, an average daily cycle of water-level change of 0.10 ft was observed. This background change is insignificant with respect to the magnitudes of drawdowns observed during testing and is not incorporated as an adjustment to the test records used for the analysis of the test.

Figure 3-2 presents a plot of barometric-pressure data and groundwater-level measurements in Monitor Well 184W506M collected during the constant-rate test. The barometric-pressure record, recorded at Test Well 184W105 and ET station SV1, covers the time period during the constant-rate test. During the record period, the barometric pressure varied by approximately 0.18 in. Hg. This equates to 0.20 ft of head, assuming 100 percent barometric efficiency of the well. The amount and duration of change in barometric pressure did not significantly influence the test results, as shown on Figure 3-2. Any barometric effect in this hydrogeologic setting is insignificant with respect to the magnitudes of drawdown observed during testing.

No other outside influences, such as the existence of other pumping wells in the vicinity of Test Well 184W105, were identified.



Hydrologic Data Analysis Report for Test Well 184W105 in Spring Valley

Figure 3-1 Hydrograph for Background Well 184W504M During Test Period



Figure 3-2 Local Barometric-Pressure Variation and Groundwater-Level Measurements at Monitor Well 184W506M



The respective borehole deviations for wells 184W105 and 184W506M are presented in the geophysical logs in the Closure Distance plots provided in the Geologic Data Analysis Report (Eastman and Muller, 2009). Evaluation of borehole deviation and depth to groundwater indicated negligible influence on depth-to-water measurement results.

Transducer data collected in the wells were compared to manually collected data. Only minor inconsistencies were identified, and these were within the accuracy range of the instrumentation. No variation between the transducer and manually collected data was observed that would influence the test results.

4.0 Well Hydraulics and Performance Testing

This section presents development results and analysis of the step-drawdown well performance testing.

4.1 Development

Prior to this phase of development, the well was initially developed after drilling using a dual-swab technique. A dual swab was used prior to and after placement of the gravel pack. AQUA-CLEAR PFD, a polymer dispersant, was added to the well to break up residual drilling mud, and a final swab was performed the length of the screen. Test Well 184W105 was developed using a surging and pumping technique. The well was pumped at a constant rate for a short period of time (usually under an hour) until turbidity data reached a certain low threshold and then surged repeatedly. Water-level and groundwater-chemistry data were collected during the pumping period. Specific capacity (discharge [Q] in gpm/drawdown[s] in ft) was determined during and at the end of each pumping period to evaluate development effectiveness and the need for additional development.

4.1.1 Development Results

A total of 8,241,000 gallons of water was pumped during this phase of pump development. Development at this site was very effective. The specific capacity at 2,400 gpm improved from 55.9 gpm/ft (42.9 ft of drawdown) on February 26, 2007, to 69.5 gpm/ft (34.5 ft of drawdown) on March 5, 2007, for a 24.3 percent improvement.

4.2 Step-Drawdown Test

A step-drawdown test was performed using five different pumping rates ranging from 2,300 to 3,700 gpm. The pumping periods ranged from 60 to 90 minutes in duration and were continuous. Figure 4-1 presents a graph showing plots of the drawdown versus time for each pumping interval.

4.2.1 Well Performance and Specific Capacity

Well specific capacity is a measure of the well's productivity and efficiency. Specific capacity usually decreases to some degree with time and increased discharge rate. Graphs of drawdown versus discharge rate and specific capacity versus discharge rate are presented on Figures 4-2 and 4-3, respectively.

Results of the step-drawdown test indicate a productive well with specific capacity values of 50 to 70 gpm/ft for associated short-term pumping rates of 3,500 to 2,300 gpm, respectively. Specific



Figure 4-1 Linear Plot of Drawdown for Each Pumping Interval During Step-Drawdown Testing of Test Well 184W105



Linear Plot of Step-Test Drawdown and Depth-to-Pumping Level Various Discharge Rates for Test Well 184W105



Step-Test Specific Capacity versus Discharge Rate for Test Well 184W105

capacity during the last 12 hours of the 72-hour, 3,000-gpm constant-rate test ranged from 54.33 to 55.62 gpm/ft. Based on these results, an operational pumping rate could range up to 3,000 to 3,500 gpm.

4.2.2 Well Loss Analysis

The drawdown observed in a pumping well is the effect of aquifer and well losses. The aquifer loss is the theoretical drawdown expected at the pumping well in a perfectly efficient well where flow is laminar. The well loss is the additional drawdown in the pumping well caused by the turbulent flow and frictional head loss effects in or adjacent to the well. Loss components are also classified as linear and nonlinear losses. Linear well losses are usually caused by damage to the formation during drilling, residual drilling fluids not removed during well development, or head losses as groundwater flows through the gravel pack and screen. Nonlinear head losses are caused by turbulent flow occurring inside the well screen pump column and the fracture zone adjacent to the well. Higher turbulent well losses caused by the formation are expected to occur more often in a fractured bedrock aquifer than in granular porous media.

Determination of well loss allows the calculation of a drawdown and specific capacity expected in the pumping well at various discharge rates. Evaluation of well loss also includes the evaluation of turbulent flow with increased pumping rate. Generally, specific capacity decreases at higher pumping rates due to increase of turbulent flow at the well. The evaluation of well loss allows for better projection of the optimal pumping rate and estimation of actual drawdown in the aquifer near the

well, removed from the effects of losses caused by pumping and well inefficiencies, friction loss, and turbulent flow.

Head loss coefficients are calculated by the equation:

$$s = BQ + CQ^2 \tag{Eq. 4-1}$$

where,

- s = Drawdown in the pumping well
- B = Linear loss coefficient
- C = Nonlinear well loss coefficient caused by turbulent flow

Q = Discharge rate

Results of the evaluation and a graph of specific drawdown (drawdown/discharge) versus discharge rate used to evaluate head loss coefficients using the Hantush-Bierschenk method (Bierschenk, 1963; Hantush, 1964) are presented in Figure 4-4. Evaluation using the Rorabaugh method (Rorabaugh, 1953) was also performed and compared to the results of other analysis methods.



Figure 4-4 Evaluation of Head Loss Coefficients Using Hantush-Bierschenk Method from Step-Drawdown Test Results

The loss coefficient for *B* is 0.0039218 and *C* equals 4.5×10^{-6} using the Hantush-Bierschenk Method. R^2 is the coefficient of determination, which is the proportion of variability in a data set. Using these values, specific capacity and drawdown estimates can be projected for any pumping rate using the equation

$$Q/s = 1/(4.5 \times 10^{-6}Q + 0.00392)$$
 (Eq. 4-2)

The reliability of the projection is highest within the discharge testing range of the step-drawdown test. Results from applying the Rorabaugh method calculated *C* as equal to 4.69×10^{-6} .

The percent of head loss attributed to laminar flow can also be estimated using the equation

$$((BQ)/(BQ+CQ^2)) \times 100$$
 (Eq. 4-3)

Table 4-1 shows that the nonlinear losses compose about 73 to 81 percent of the drawdown, the percentage increasing with increasing production rate. This analysis indicates that the nonlinear losses are substantial, which should be reflected in a substantial well loss contribution to pumping-well drawdown. Evaluation assumed a saturated thickness of 946 ft.

Well efficiency can be evaluated by estimating the drawdown in the test well if there were no well losses. Well efficiency can also be calculated using an estimated T or, if multiple observation wells were present and a distance drawdown graph prepared, projecting estimated drawdown at the test well. The calculations are more reliable if no cascading water is entering the borehole, which commonly occurs in a fractured bedrock aquifer system. Based on the preferred analysis presented in Section 5.4, the drawdown at the end of the test period was estimated to be approximately 22 ft. The actual drawdown observed was about 53 ft, yielding an estimated efficiency of 42 percent using well loss calculated from the step-drawdown test results. This is within an expected range for wells completed in fractured aquifers.

Q (gpm)	s (ft)	s/Q (ft/gpm)	Nonlinear Losses (ft)	Linear Losses (ft)	Total Losses (ft)	Nonlinear Total (%)
2,300	32.80	0.0142603	23.81	9.02	32.83	73
2,500	38.11	0.0152427	28.13	9.80	37.93	74
3,000	50.83	0.0169439	40.50	11.77	52.27	77
3,500	69.70	0.0199152	55.13	13.73	68.85	80
3,700	75.43	0.0203863	61.61	14.51	76.12	81

Table 4-1Step-Drawdown Test Analysis



This Page Left Intentionally Blank

5.0 CONSTANT-RATE TEST EVALUATION

This section summarizes the collection of hydraulic testing data, selection of the analytical solutions for analysis of drawdown and pumping data, and the results of the 72-hour constant-rate and recovery test at Test Well 184W105.

5.1 Data Review and Adjustments

Water-level data were collected with transducer and physical methods using the instrumentation described in Section 3.4. The physical measurements were used to confirm the transducer data. No significant variation between the two data sets was observed. Data collection time intervals were logarithmic and in accordance with SNWA and industry standards.

Outside effects, such as changes in barometric pressure, regional water-level trends, and precipitation events, were monitored during the test period. No influences that would significantly affect the test results were identified. No other pumping wells were present in the area to influence the test results. A detailed discussion of background data and outside influences is presented in Section 3.4.

The target discharge rate for the constant-rate test was 3,000 gpm; however, three minor discharge flow adjustments occurred during the test. The first was a readjustment down and back up at approximately 1,100 minutes elapsed time into the test at 07:00 on March 8, 2007, due to a stuck rpm meter on the generator. A second flow readjustment downward occurred at 2,400 minutes at 04:00 on March 9, 2007, when the flow meter indicated 3,050 gpm. A reading of 2,900 gpm was made at approximately 2,600 minutes into the test at 07:30 on March 9, 2007, and flow was adjusted up to 3,000 gpm. The observed variations were approximately 1.7 to 3.3 percent of the target discharge rate. Totalizer readings indicated a total volume of 13,177,000 gallons pumped during the 72-hour test, which averages 3,050 gpm for the duration of the test or a 1.7 percent variance from the target discharge rate. The flow variations had no significant effect on the test analysis. However, in order to capture the variation during the test and incorporate it into the curve-fitting, the incremental average flow rate was calculated from the totalizer record; the running average rate was then calculated, and a set of incremental rates at 0.5-day time steps was calculated based on the running average. This captures the variation of the applied stress while smoothing the instantaneous adjustments.

During the initial minute of the test, small variations in drawdown were observed. These were the result of water filling the pump column and pressure variations at the flow control valve.

Vertical flow losses within the well were considered during analysis. Upward flow within the well screen and casing to the pump intake is subject to friction losses that are a function of the screen and casing diameters, friction coefficient, and flow rate. Since the flow rate varies along the depth of the well screen because of distributed water intake along the screen, the losses vary with depth.



Assuming consistent hydraulic conductivity along the well depth and no vertical gradient, the friction losses due to upward flow in the well were calculated to reach a maximum of just under 1 ft at total depth of the screen. A nominal friction factor of 2X smooth pipe was used for the screen. Relative to the drawdown observed during the test, this influence is negligible, and the total length of the screened interval was considered to have been stressed equally.

Minor smoothing of the transducer data record was performed to average noise in the test well data record. The record indicates some noise in the drawdown data caused by turbulence at the pump intake and shifts in drawdown associated with minor variation of the production rate described earlier in this section. The synthetic production record was developed to smooth the shifts while preserving the total volumes recorded periodically.

The data logger time for recovery at Monitor Well 184W506M lagged from the test well record by about 7.44 seconds (0.000086 days) based on the start of recovery. The monitor-well recovery record was shifted by this amount.

Early-time recovery data after cessation of pumping are obscured because the pump was not fitted with a check valve. After the pump was stopped, the water column in the pump column flowed back into the well. This created a short-term injection pulse into the well that is superimposed on the recovery. Examination of the recovery response indicates that this pulse almost instantaneously raises the water level above the original static water level. The recovery water level then decays back to the aquifer recovery response. This effect is observed in both the test well and in the monitor well and does not influence the analysis of the recovery data after the pulse reaches equilibrium.

5.2 Constant-Rate Test Data

The constant-rate test was performed for a duration of 72 hours at a target pumping rate of 3,000 gpm. A summary of drawdown data for Monitor Well 184W506M and Test Well 184W105 is presented graphically in log-log and semi-log form on Figures 5-1 through 5-4. Transducer and physical test data are presented in Appendix A. Recovery data were collected immediately upon cessation of pumping activities. Recovery data are presented in a plot of residual drawdown versus log of t/t' (elapsed time from beginning pumping/time of recovery) in Figure 5-5.

5.3 Analytical Model Selection

The analytical model used for the evaluation of the site data was selected based upon the conceptual model of site hydrogeologic conditions and diagnostic log-log and drawdown derivative plots. A dual-porosity model was selected as the primary evaluation method because of the presence of saturated fractured bedrock encountered at the site and the drawdown response curves observed. The drawdown curve and derivative plot are representative of the signature of a dual-porosity system, which would be expected in fractured carbonate bedrock. Initial response in the main fracture network would start to occur as borehole storage effects diminish in early time. A mid-time transition, semi-stabilization period then occurs during which water in the formation matrix material is released to the fracture network and the drawdown curve flattens. Rate of release would be dependent upon the matrix skin effect. As pumping continues, release of matrix water decreases,





Figure 5-1 Log-Log Data Plot of Drawdown versus Time from Monitor Well 184W506M



Semi-Log Data Plot of Drawdown versus Time from Monitor Well 184W506M





Figure 5-3 Log-Log Data Plot of Drawdown versus Time from Test Well 184W105



Semi-Log Data Plot of Drawdown versus Time from Test Well 184W105

JA_4845


Note: t = Elapsed time since pumping began; t' = elapsed time since pumping stopped.



drawdown increases, and the flow regime approaches radial flow conditions. This is illustrated in Figure 5-2.

The Barker generalized radial flow model (Barker GRFM) (Barker, 1988), which is a generalized radial flow model for an unsteady, confined, fractured media, dual-porosity conceptual analytical model, was selected as the primary solution. This analytical model is equivalent to the Moench (1984) fractured media, dual-porosity, radial flow model. However, the Barker GRFM incorporates a flow dimension term. Flow dimension (n) provides adjustment of the response for variation in the flow geometry, ranging from n = 1 for linear flow, to n = 2 for radial flow, to n = 3 for spherical flow. This parameter has application to situations in which a linear feature, such as a fault, may affect the drawdown response or conversely as an adjustment for partial penetration (shift to slightly spherical flow dimension) effects that cannot be estimated in advance. An analytical model with these features that also incorporates specific aspects of unconfined aquifer response, such as delayed gravity drainage or dewatering, is not available. A dual-porosity solution is more appropriate over an unconfined solution, such as the Neuman solution (Neuman, 1975), which considers only delayed response or gravity drainage of the formation. Given that the water table was located within fractured carbonate with low storage, the delayed gravity drainage effect would not be expected to be as substantial as dual-porosity effects. In an unconfined condition, a correction equation for dewatering (Kruseman and De Ridder, 1994, p. 101) was applied to the drawdown response before analysis to account for the variation in effective saturated thickness influencing the test. This approach provides for bounding of the effect of dewatering and was applied in this solution. The aquifer test analysis software AQTESOLV V4.50 (Duffield, 1996-2007) was used for curve fitting.

5-5



General assumptions associated with the Barker GRFM solution are that:

- An aquifer has infinite extent and uniform extent of flow.
- Pumping and observation wells are fully penetrating.
- An aquifer is confined with single or dual porosity.
- Matrix blocks are slab shaped or spherical.
- Flow is unsteady.

The flow dimension may be adjusted to compensate for spherical flow caused by partial penetration. The dewatering correction may be applied to compensate for unconfined response.

The complexities of the aquifer system do not fully conform to the assumptions of the analytical model. However, the Barker GRFM solution is the most appropriate of the analytical solutions available for the observed hydrogeologic conditions at this test location. While the assumptions related to aquifer and flow conditions are not perfectly satisfied, they are sufficiently satisfied to provide a reasonable estimate of aquifer parameters.

Cooper-Jacob semi-log straight-line approximation (Cooper and Jacob, 1946) was used as a secondary evaluation solution method. This approach was used to fit early- and late-time data. For a homogeneous, radial-flow, dual-porosity system, the early-time (after casing storage, but before matrix effect) and late-time (after the matrix effect period approaching radial-flow conditions) slopes on a semi-log plot would be similar.

5.4 Constant-Rate and Recovery-Test Analysis

5.4.1 Test Analysis Methodology

The data logger records of pressure transducer output were used to create AQTESOLV input files of the drawdown and recovery data. The time representing the measurement at the start of identifiable drawdown at the test well was used as the start time to determine the elapsed time and drawdown magnitude. The basic input measurement and parameter values used for analysis are shown in Table 5-1.

r(w) Radius of the well	1.08 ft					
r(c) Radius of the well casing 0.83 ft						
r(e) Radius of the production tubing	0.42 ft					
r Radial distance from 184W105 to 184W506M ^a	212 ft					
b Aquifer saturated thickness ^b 946 ft						
b' Fracture spacing	3.3, 10 ft					

 Table 5-1

 Measurement and Parameter Values Used for Analysis

^aSurface measurement

^bStatic water level to bottom of the borehole

Parameter symbols used in this section are presented below:

- K = Aquifer/ fracture hydraulic conductivity (ft/day)
- K' =Matrix hydraulic conductivity (ft/day)
- n = Flow dimension; 1 = linear, 2 = radial, and 3 = spherical (dimensionless)
- Q = Pumping discharge rate (gpm)
- Sf = Fracture skin factor (dimensionless)
- Ss = Fracture-specific storage (ft⁻¹)
- Ss' = Matrix-specific storage (ft⁻¹)
- Borehole skin factor or well loss coefficient value (dimensionless) Sw =
- = Drawdown at pumping well S
- Time t =
- = Transmissivity (ft^2/day) Т
- = Storativity (dimensionless) S

A sensitivity analysis was performed on three primary parameters to evaluate effects on fracture hydraulic conductivity (K). These parameters were (1) dewatering correction for drawdown, (2) fracture spacing, and (3) matrix-specific storage (Ss'). The correction for dewatering was considered because the aquifer is unconfined, and drawdown in the vicinity of the well was a significant fraction of the aquifer saturated thickness. Average fracture spacing is estimated from borehole geophysics and for practical purposes has nonuniform spacing and characteristics, and for comparison purposes, fracture spacing of 3.3 ft and 10 ft was used for analysis to evaluate sensitivity to the parameter. There is also no independent data for anisotropy of vertical/horizontal hydraulic conductivity. The sensitivity to this anisotropy was checked for general effect on the solution and was determined to be negligible. Because the well is located in a fault zone and faulting in the test area is high-angle, vertical hydraulic conductivity may be expected to be relatively high and default anisotropy of 1 is judged reasonable. The sensitivity to Ss' was evaluated progressively in conjunction with the correlated parameter K.

The Barker GRFM solution was fitted to the drawdown and recovery responses of both the test well and the monitor well sequentially and iteratively to determine the model parameter set that would best fit all of the data. Initially, the paired drawdown responses could be fitted similarly well with a wide range of K, with differences in values for the other parameters adjusting the fit. The selection of the most representative set of parameter values depends upon the conceptual model for the aquifer system, the constraints placed upon storage parameter values, and interpretation of well borehole skin as related to nonlinear flow losses at the test well distorting actual drawdown near the test well.

The monitor well response provides information on the formation hydraulic properties independent of linear and nonlinear head losses associated with the pumping well and theoretically provides the information necessary to determine storage. However, the information from the single monitor well is not as definitive as multiple observation wells to evaluate and define asymmetry and horizontal anisotropy.

5-7



5.4.2 Test Analysis Result Summary

The Barker GRFM solution was derived through an extensive iterative analysis process that converged to provide an optimal match for all test data. The primary solution was verified through application of a more simplified Cooper-Jacob secondary solution. Results of the Barker GRFM and Cooper-Jacob solutions are summarized in Table 5-2. The optimal solution analysis plots for each method are presented below.

Primary Solution Barker GRFM Analysis										
Fracture Spacing (ft)	<i>K</i> (ft/day)	Ss (ft ⁻¹)	K∕ (ft/day)	Ss´ (ft⁻¹)	n	Sf	Sw	<i>T</i> ª <i>(</i> ft²/day)		
3.3	64.00	1.25 × 10 ⁻⁷	4.83 × 10 ⁻⁴	1.00 × 10 ⁻⁵	2.20	2.84	6.25	60,544		
10	60.50	1.53 × 10 ⁻⁷	6.30 × 10 ⁻³	1.00 × 10 ⁻⁵	2.20	5.00	5.73	57,233		
		Secondary Sol	lution Cooper	Jacob Analys	sis					
Analysis TimeK (ft/day)LocationST (ft²/day)								y)		
Early-Time	56.36	Monitor Well	2	53,320						
Late-Time	57.83	Monitor Well		NA	54,710					

Table 5-2Summary of Optimal Analysis Results

^aAssume saturated thickness of 946 ft to derive *T*.

NA = Not applicable

5.4.3 Barker GRFM Analysis

The Barker GRFM solution was fitted to the data iteratively, applying constraints successively to refine the fit and produce an overall model that was consistent with all site and literature data and to determine the parameter range in which the solution is optimized. The model fit to all of the data and constraints is optimal within a relatively restricted range for the major parameters. The initial fitting was first to the observation well drawdown, then to the test well drawdown, then to the observation well recovery, and then to the test well recovery. Fitting started with a radial flow system (n = 2), Sf = 0, and well loss coefficient Sw = 0. The flow dimension was maintained at n = 2 until the final model fitting. Fitting started from lower fracture hydraulic conductivity (K) values, which required matrix-specific storage (Ss') that was greater than could be supported by reported value ranges (independent of the site) on carbonates. Also, these parameter sets did not simulate the observed recovery of the test well. Through an iterative process, analysis matched the test well recovery with an increase in the K value. Constraints were then imposed on the Ss and Ss'values, allowing an Ss range from $1 \times 10^{-8} - 1 \times 10^{-9}$, and an Ss' maximum value of 1×10^{-5} . Fitting is not sensitive to Ss values in the range 1×10^{-7} to 1×10^{-10} ; however, fitting is very sensitive to the Ss'value. The K' and Ss' parameters are highly correlated, so there is no unique solution, and the external constraint for Ss' is important.

Constraints on fracture-specific storage (*Ss*) for carbonates for Nevada are based on information from Kilroy (1992) ($1.06 \times 10^{-7} - 4.57 \times 10^{-8}$ ft⁻¹), Galloway and Rojstaczer (1989) (1.1×10^{-11} ft⁻¹), and Bredehoeft (1997) (9×10^{-10} ft⁻¹). Matrix-specific storage in carbonates is several orders of magnitude larger that fracture-specific storage, so the overall storage (sum of fracture and matrix storage) is similar to the matrix storage. Specific storage can be equated to storativity (*S*) as the product of specific storage and aquifer thickness. SNWA has documented ranges of storativity for the lower and upper carbonate aquifers from 8.14×10^{-3} to 1.70×10^{-9} . For the test well, with a nominal aquifer thickness of 946 ft, the *Ss* would be 8.6×10^{-6} to 1.8×10^{-12} ft⁻¹. This indicates that there is considerable latitude in *Ss* values, with the upper bound for the range for values about 1×10^{-5} . General information from Freeze and Cherry (1979) for carbonate compressibility for jointed rock can be used to calculate the theoretical *Ss*, which extends the upper range to about 3×10^{-5} ft⁻¹.

With Ss and Ss' values constrained, the K value required to fit the monitor well drawdown was determined. Then, the difference in drawdown magnitude between the monitor well and the test well had to be accounted for. The step-drawdown test analysis indicated that a large proportion of the test well drawdown was nonlinear losses, which typically are due to well losses. However, the well construction provides substantial screen-open area, and the gravel pack likewise should not be restrictive because of extensive well development. Consequently, the well losses are mainly attributed to the turbulent flow in the near-well radius that results from converging flow in the fractures, which are restrictive. The large proportion of drawdown attributed to nonlinear losses equates to a large well loss coefficient value (Sw). In turn, these large well losses account for the great difference in drawdown between the monitor well and the test well. A small increase of the flow dimension (n) above 2 (radial flow) helped tune the drawdown difference between the wells. This may indicate a partial penetration effect because the well does not fully penetrate the formation into a confining unit. The fact that the well is completed in a high-angle fault zone suggests that the vertical K within the tested area, primarily the fault zone, could be similar to the horizontal K, and production could have induced vertical flow within the formation from below the bottom of the well. The parameters K, n, and Sw are highly correlated, and a unique solution is not identifiable. However, the confidence intervals for these parameters for the Barker GRFM solution are not wide, so the given values are approximately optimal.

There is no quantitative information, such as spacing and aperture, on the hydraulically active fractures. During analysis, fracture spacing of 1 (nominal), 3.3, and 10 ft were used. The results are presented in Table 5-2 for both the 3.3- and 10-ft spacing, providing an indication of the effect of fracture spacing on the parameter values. In general, fracture *K* must be increased to compensate for fewer fractures and *Ss*'. Fracture spacing of 10 ft is probably the best general estimate for the well, based upon available data.

The Barker GRFM solution optimal aquifer hydraulic conductivity (*K*), which is dominated by fracture hydraulic conductivity, ranged from 60.50 to 64.00 ft/day using a fracture spacing sensitivity analysis range of 10 and 3.3 ft, respectively. Matrix hydraulic conductivity (*K'*) ranged from 4.83×10^{-4} to 6.30×10^{-3} ft/day. Fracture-specific storage ranged from 1.25×10^{-7} to 1.53×10^{-7} ft⁻¹. Matrix-specific storage of 1.00×10^{-5} ft⁻¹ relates to aquifer storativity of 9.46×10^{-3} , assuming a saturated thickness of 946 ft. Increased estimated saturated thickness would equate to a theoretical proportional increase in aquifer storativity.

5-9



Log-log and semi-log time drawdown plots for the pumping period using the optimal Barker GRFM solution with a fracture spacing of 10 ft are presented in Figures 5-6 and 5-7, respectively. Derivative drawdown versus time for Monitor Well 184W506M and Test Well 184W105 are presented in Figures 5-8 and 5-9, respectively. The derivative drawdown response in the monitor well is consistent with a dual-porosity fractured bedrock system.



Optimal Barker GRFM Solution Pumping Period Log-Log Plot

Well loss analysis of Test Well 184W105 is presented in Section 4.2.2. An evaluation and removal of well loss components are presented in Figure 5-10, which provides an indication of drawdown in the formation in the vicinity of the test well outside of the drawdown distortion caused by well losses from turbulent flow and well construction. This calculation of drawdown without well losses provides a more realistic value of aquifer drawdown in the vicinity of the test well during testing.

Analysis results of recovery data collected from the test and monitor well is presented in Figure 5-11. This figure presents a plot of residual drawdown versus $\log t/t'$ (ratio of total pumping elapsed time to time since pumping stopped). In this plot, initial recovery is to the right and later recovery is to the left.

5.4.4 Cooper-Jacob Analysis

The Cooper-Jacob secondary solution at the monitor well, where well loss does not distort the drawdown, compared favorably to the Barker GRFM results. Transmissivity values of 53,320 and 54,710 ft²/day were derived from the early-time and late-time data, respectively. Using a saturated



Figure 5-7 Optimal Barker GRFM Solution Pumping Period Semi-Log Plot



Optimal Barker GRFM Solution Drawdown Derivative for Monitor Well 184W506M





Figure 5-9 Optimal Barker GRFM Solution Drawdown Derivative for Test Well 184W105



Optimal Barker GRFM Solution, Test Well 184W105 Well Losses Removed

Section 5.0 SE ROA 12091





thickness of 946 ft resulted in a hydraulic conductivity of 56.36 to 57.83 ft/day. The hydraulic conductivity value derived from transmissivity using the Cooper-Jacob method is directly related to the effective aquifer saturated thickness used. Storativity was calculated using early-time data and is estimated to be 2.05×10^{-4} . This corresponds to a fracture-specific storage of 2.16×10^{-7} /ft assuming an effective saturated thickness of 946 ft, which would dominate early pumping time.

The Cooper-Jacob straight-line analysis of the semi-log plot for time versus drawdown of Monitor Well 184W506M early-time and late-time data is presented in Figures 5-12 and 5-13, respectively. The early-time aquifer response can only be evaluated after the casing storage effects are past and before dual-porosity matrix flow begins. The initial phase of early-time data at the start of a test is usually affected by flow system instability associated with variation in discharge rates prior to stabilization and borehole storage effects where water is removed from the storage within the well. The test well response data do not have a sufficient early-time record for analysis because of the influence of wellbore storage. The Cooper-Jacob straight-line solution was fitted to the stabilized slope just before matrix flow began. The late-time data, after the second day of pumping, were also fitted with a Cooper-Jacob straight-line solution. It was not definitive that the late-time drawdown response had stabilized sufficiently to accurately determine the stabilized slope. However, the early-time and late-time fitted slopes are similar, which is expected for a dual-porosity response after matrix effect is complete and radial flow is reached. This suggests that both periods achieved near stabilization. Additional longer-term pumping would be needed to confirm that the straight line indicating radial flow continues or whether any boundary conditions are encountered. The T values determined for both analyses are similar to the optimal solution from the dual-porosity analysis. The





Figure 5-12 Cooper-Jacob Analysis, Monitor Well, Early-Time



Cooper-Jacob Analysis, Monitor Well, Late-Time

early-time solution yields an *S* value that is also similar to the storage determined for the preferred dual-porosity analysis. The late-time *S* value is not applicable because of the offset resulting from the matrix-dominated flow period.

5.5 Discussion

Analysis of the test results indicates an optimal K and S value based upon the data collected during the 72-hour constant-rate test and subsequent recovery period. The carbonate aquifer system at the site is complex, with two primary fracture zones identified as hydrologically connected to the wells. The results of the testing provide a composite hydraulic conductivity over the length of the saturated interval of the wells.

The controlling factor for determination of K from T from the Cooper-Jacob secondary solution is the estimated saturated thickness (b). The highest K results from assuming a full penetration of the saturated zone by the pumping well, resulting in a b of 946 ft. Proportional lower K values would be derived from this secondary method with larger b estimates. Specific storage derived from the Barker GRFM solution multiplied by the saturated thickness results in storativity. The thicker the saturated thickness the larger the storativity value for the aquifer, assuming a consistent specific storage value.

It is significant that the recovery curves for both wells are almost identical once past the effect of the pump column injection phenomenon, which is caused by the return of water in the pump column to the formation after the pump is stopped. This occurs because no check valve was used in the pump column. The crux of determining a solution that coordinates all aspects of the test data is evaluation of the site hydrogeologic conditions, simulating both the large difference in drawdown between the test well and the monitor well through identification of well loss and the almost identical recoveries.

The test provided representative data about the aquifer system without outside pumping or natural hydrologic variation influence. Diagnostic data plots and site hydrogeologic conditions were indicative of a dual-porosity aquifer system. The plots indicate the early-time wellbore storage effects, fracture network response phase, transition zone of matrix or delayed response phase, and system equilibrium reflected in the suggested late-time equivalent radial flow. No significant recharge or barrier condition boundaries were identified in the data results.

The short-term pumping period, availability of one observation well, and expected aquifer heterogeneities limit the ability to scale results to determine horizontal anistropy or evaluate potential boundary conditions. The presence of boundaries and/or higher or lower hydraulic-conductivity zones that may appear after extended pumping cannot be evaluated until extended pumping is performed. Additional analysis and review should be performed as longer-term operational pumping data become available for the well site or as additional regional hydrogeologic data are obtained.



This Page Left Intentionally Blank

6.0 WATER CHEMISTRY

Groundwater-chemistry data for Test Well 184W105 and Monitor Well 184W506M are presented within this section. Additional data for other SNWA wells located within the vicinity of these wells (see Figure 2-1) are also presented for comparison.

6.1 Groundwater Sample Collection and Analysis

Water samples were collected from Test Well 184W105 on March 8, 2007, at 08:00 after pumping over 22 million gallons (following well development, step-drawdown testing, and a portion of the constant-rate test). For these samples, turbidity, pH, specific conductance, dissolved oxygen, and temperature were measured in the field. With the exception of dissolved oxygen, these parameters were also measured periodically during well development and testing. Sampling and field measurement of the water-quality parameters were performed using the National Field Manual for the Collection of Water-Quality Data (USGS, 2007) as the basis. All measurement equipment was calibrated according to the manufacturers' calibration procedures. Samples were sent to Weck Laboratories, Inc., (Weck) for analysis of a large suite of parameters including major solutes, minor and trace constituents, radiological parameters, and organic compounds. Weck is certified by the State of Nevada and performs all analyses according to U.S. Environmental Protection Agency (EPA) methods or methods published in Standard Methods for the Examination of Water and Wastewater (Eaton et al., 2005). The parameters analyzed and the corresponding analysis method are presented in Tables B-1 and B-2. Weck provided all sample containers and preservatives. Radiation Safety Engineering, Inc., and Frontier Analytical Laboratory were contracted by Weck for the analysis of radiological parameters and dioxin, respectively. In addition, samples were collected for analysis of oxygen and hydrogen isotopes by University of Waterloo's Environmental Isotope Laboratory, carbon isotopes by University of Arizona's NSF-Arizona Accelerator Mass Spectrometry Laboratory, chlorine-36 by Purdue University's Purdue Rare Isotope Measurement (PRIME) Laboratory, and strontium and uranium isotopes (and uranium concentration) by the USGS Earth Surface Processes Radiogenic Isotope Laboratory.

Water samples were collected from Monitor Well 184W506M on October 31, 2006, at 11:53 after pumping approximately 469,000 gallons. Samples were sent to Weck for analysis of major solutes and trace and minor constituents. A sample was also collected for the analysis of oxygen and hydrogen isotopes by University of Waterloo's Environmental Isotope Laboratory (Table B-1). The pH, specific conductance, and temperature associated with these samples were measured in the field. Monitor Well 184W506M was used as the water source for drilling Test Well 184W105 and as the main water source for drilling background well 184W504M. The water source for drilling Monitor Well 184W506M was the small well at Harbecke Ranch.



For comparison, the groundwater chemistry of additional wells in the area are presented in this section. The wells, all drilled by the SNWA (see Figure 2-1), were completed in a carbonate-rock aquifer to the following depths:

184W101	1,760 ft bgs
184W502M	1,828 ft bgs
184W103	1,046 ft bgs
184W504M	1,040 ft bgs
184W105	1,160 ft bgs
184W506M	1,160 ft bgs

6.2 EPA Drinking Water Standards

The national maximum contaminant levels (MCLs) for drinking water, established by the EPA and authorized by the Safe Drinking Water Act, are presented in Tables B-1 and B-2. These national health-based standards are to protect against both naturally occurring and man-made contaminants that may be found in drinking water. Also presented in Table B-1 are the secondary drinking water standards established by the EPA. These are nonenforceable guidelines that regulate contaminants that may cause cosmetic or aesthetic effects in drinking water. A single constituent, di(2-ethylhexyl) phthalate (DEHP), exceeded the primary drinking water standards for the groundwater of Test Well 184W105; no constituent exceeded the secondary MCL. Groundwater samples taken from Monitor Well 184W506M exceeded the secondary MCL for aluminum. These exceedances will be discussed further in Sections 6.3.3 and 6.3.6.

6.3 Groundwater-Chemistry Results

In this section, the field measurements and analytical results for the groundwater of Monitor Well 184W506M and Test Well 184W105 are presented and compared to those of groundwater samples from four wells within the vicinity.

6.3.1 Field Results

Field measurements of turbidity, pH, specific conductance, and temperature were performed periodically throughout well development and testing of Test Well 184W105 and for the samples collected for laboratory analysis (see Table B-1). For Test Well 184W105, these parameters stabilized within the first hour of the constant-rate test. Measurements ranged from 0.38 to 1.25 nephelometric turbidity units (NTUs) (turbidity), 7.44 to 8.21 (pH), 322 to 383 μ S/cm (specific conductance), and 12.6°C to 15.0°C (temperature) over the remaining period of pumping (71 hours) with no observable trends. Field measurements made at the time of sample collection are reported as 0.41 NTU, 282 μ S/cm, 7.8, 13.0°C, and 5.08 mg/L for turbidity, specific conductance, pH, water temperature, and dissolved oxygen concentration, respectively.

During the 8-hour constant-rate test for Monitor Well 184W506M, field measurements of pH, specific conductance, and temperature ranged from 8.04 to 8.18, 394 to 374 μ S/cm (slight decreasing trend), and 12.0°C to 15.6°C, respectively. No turbidity or dissolved oxygen concentration

measurements were performed for the groundwater of Monitor Well 184W506M. Field measurements made at the time of sample collection are reported as 385 μ S/cm, 8.1, and 12.7°C for specific conductance, pH, and water temperature, respectively.

When compared to Test Well 184W105 and Monitor Well 184W506M, the water temperatures in the deeper wells were significantly higher, 24.1°C (184W101) and 20.5°C (184W502M), but were quite similar in 184W103 (12.0°C) and 184W504M (12.1°C). In general, the specific conductivities were greater in the monitor wells, 394 μ S/cm (184W502M), 333 μ S/cm (184W504M), and 385 μ S/cm (184W506M), than in the test wells, 359 μ S/cm (184W101), 263 μ S/cm (184W103), and 282 μ S/cm (184W105). The higher specific conductivities observed for the groundwater from 184W101 and 184W502M are attributed to increased mineral dissolution in the warmer groundwater. The pH values ranged from 7.5 (184W504M) to 8.5 (184W502M) with no clear trend between the monitor and test wells.

6.3.2 Major Constituents

The concentration of the major constituents in groundwater samples from Test Well 184W105 and Monitor Well 184W506M are presented in Table B-1. Major constituents are defined as those commonly present in groundwater at concentrations greater than 1 mg/L and typically include bicarbonate (HCO₃), calcium (Ca), chloride (Cl), magnesium (Mg), potassium (K), silica (SiO₂), sodium (Na), sulfate (SO₄). The sum of the charge of major cations should equal the sum of the charge of the major anions in solution (in milliequivalents per liter [mEq/L]); thus, calculation of the anion-cation (charge) balance is used to assess the accuracy of the analyses and to ensure that the full suite of anions and cations present as major constituents in the groundwater have been included in the analyses. The charge balance for Test Well 184W105 and Monitor Well 184W506M groundwater analyses, 2.0 and 3.2 percent, respectively, indicate that the analyses were performed adequately (Table B-1).

To illustrate the relative major-ion compositions in these groundwater samples, a Piper diagram is presented in Figure 6-1. A Piper diagram consists of two triangular plots presenting the major cations (left triangle) and major anions (right triangle) in percent milliequivalents. The two triangular plots are then projected to a central diamond where the relative abundance of all major ions is presented. A Piper diagram is used to evaluate similarities in groundwater major-ion compositions, to identify the hydrochemical water type representing the aquifer(s) from which the groundwater was collected, and to assess possible evolutionary trends that have occurred along a flowpath. As shown in Figure 6-1, the relative concentrations of major ions are similar for all six groundwater samples. The groundwater samples all represent a calcium-magnesium-bicarbonate facies that is typical of dissolution of calcite and dolomite in waters of a carbonate-rock aquifer. The relative concentrations of sodium plus potassium (Na + K) tend to be slightly greater in the groundwater samples from the monitor wells than in that of the associated test wells.

Stiff diagrams for these groundwater samples are presented in Figure 6-2. Major solutes are presented in a Stiff diagram so that their relative proportions are identified by their shape and the magnitude of the concentrations by its size. As apparent in the Stiff diagrams in Figure 6-2, groundwater from the four wells, 184W105, 184W103, 184W506M, and 184W504M, are nearly identical with a somewhat greater concentration of sodium in the monitor wells. The concentrations



Figure 6-1 Piper Diagram Illustrating Relative Major-Ion Compositions



Figure 6-2 Stiff Diagrams Illustrating Major-Ion Concentrations

of calcium and bicarbonate are greater in the groundwater samples from Test Well 184W101 and Monitor Well 184W502M. This may be attributed to increased dissolution of carbonate minerals in the deeper and warmer groundwater of these wells.

6.3.3 Trace and Minor Constituents

The concentrations of trace elements in the groundwater from Test Well 184W105 and Monitor Well 184W506M are presented in Table B-1. The dominant trace element present in the groundwater from Test Well 184W105 is strontium, which is consistent with the relatively high concentration of strontium in carbonate rocks (i.e., limestone) (Drever, 1988). Relatively higher concentrations of aluminum, iron, manganese, and zinc were observed in the groundwater from Monitor Well 184W506M (Table B-1) when compared to the concentrations in the groundwater of Test Well 184W105. In fact, the concentrations of these elements are consistently higher in the monitor wells than in the test wells (Table 6-1). The elevated concentration of these elements in the groundwater of the monitor wells is therefore thought to result from interaction with the casing used for the monitor wells and is not expected to reflect naturally occurring concentrations in the groundwater.

Well Name	Concentration (µg/L)								
	Aluminum	Iron	Manganese	Zinc					
184W506M	320	300	62	29					
184W105	26	<20	0.78	<5					
184W502M	180	5,700	39	56					
184W101	8.4	<20	2.8	<5					
184W504M	130	500	24	55					
184W103	<5	<20	1.8	5.5					

Table 6-1Trace Elements Present in Higher Concentrationsin the Monitor Wells than in the Test Wells

6.3.4 Stable Isotopes and Environmental Tracers

The stable hydrogen, oxygen, and carbon isotopic compositions of the groundwater samples from Test Well 184W105 and the stable hydrogen and oxygen isotopic compositions of the groundwater samples of Monitor Well 184W506M are presented in Table B-1. Table B-1 also presents chlorine-36, strontium-87/86, and uranium-234/238 data for the groundwater samples collected from Test Well 184W105.

6.3.4.1 Hydrogen and Oxygen Isotopes

Stable isotopes of hydrogen and oxygen behave conservatively in most groundwater systems and therefore can be used to indicate groundwater source, trace groundwater flowpaths, evaluate possible mixing of groundwater along a flowpath, and evaluate water budgets. Isotopic concentrations are



reported using delta notation (δD and $\delta^{18}O$) as the relative difference between the isotopic ratio (D/¹H or ¹⁸O/¹⁶O) measured for the sample and that of the Vienna Standard Mean Ocean Water (VSMOW) reference standard. The analytical precisions for δD and $\delta^{18}O$ are typically $\pm 1\%$ and $\pm 0.2\%$, respectively.

The analytical results for δD and $\delta^{18}O$ for Test Well 184W105 and Monitor Well 184W506M are presented in Table B-1 and Figure 6-3 (mean value). Figure 6-3 also presents data for the four SNWA wells in the vicinity along with the Global Meteoric Water Line ($\delta D = 8\delta^{18}O + 10$) (Craig, 1961). These groundwater samples exhibit similar relatively light stable isotope ratios that are typical of recharge at high elevations and cold temperatures. The samples all plot slightly below the Global Meteoric Water Line, suggesting that the water underwent only slight evaporation prior to recharging.



Plot of δD versus $\delta^{18}O$

6.3.4.2 Carbon Isotopes

The isotopic composition of stable carbon (δ^{13} C) in groundwater is used to assess the extent of isotope mass transfer that occurred along a groundwater flowpath. Corrections based on this assessment can then be applied to Carbon-14 (¹⁴C) data to determine the age of the groundwater. The δ^{13} C composition is reported as the relative difference between the isotopic ratio, ¹³C/¹²C, for the sample and that of the Pee Dee Belemnite (PDB) reference standard. The analytical precision for δ^{13} C is typically ± 0.3‰. Carbon-14 is reported as percent modern carbon (pmc), where modern carbon is defined as the approximate ¹⁴C activity of wood grown in 1890 (13.56 disintegrations per

minute per gram of carbon), before the dilution of ${}^{14}C$ in the atmosphere by burning fossil fuels. The analytical precision for ${}^{14}C$ in these groundwater samples is ± 0.1 pmc.

Relatively similar values of δ^{13} C and 14 C were measured in the groundwater of the test wells: 184W101 (-5.8‰, 4.93 pmc), 184W103 (-6.7‰, 10.37 pmc), and 184W105 (-5.8‰, 6.09 pmc); carbon isotopes were not measured for the monitor wells. The low 14 C and relatively heavy values of δ^{13} C suggest that the groundwater has interacted with isotopically heavy and 14 C-free carbonate minerals. From these data, it appears that water-rock interaction has occurred to a lesser extent along the groundwater flowpath to Test Well 184W103 as compared to the other test wells. This suggests a shorter residence time for this groundwater. Further evaluation of groundwater flowpaths is required to assess the extent of these reactions and to accurately estimate the groundwater age.

6.3.4.3 Chlorine-36/Chloride Ratios

The ratio of atoms of chlorine-36 to chloride (³⁶Cl/Cl) can be used to trace groundwater flow. Dominant factors controlling the observed ³⁶Cl/Cl ratios and Cl concentrations are the initial values inherited during recharge, the progressive dissolution of Cl-rich (low ³⁶Cl) carbonate rocks along the groundwater flowpath, and the mixing of water with different ³⁶Cl/Cl ratios (Moran and Rose, 2003). The interpretation of ³⁶Cl/Cl data requires knowledge of the compositions of the recharge water and the potential mixing components along the groundwater flow path. The ³⁶Cl/Cl ratio in precipitation varies with distance from the ocean and has not been previously evaluated in this region. Ratios measured in recently recharged groundwater and soils throughout the southwestern United States of 500×10^{-15} to 880×10^{-15} have been reported (Davis et al., 1998; Phillips, 2000).

The ³⁶Cl/Cl ratios are consistent with precipitation in the southwestern United States. Of the three test wells, the ³⁶Cl/Cl ratios are the lowest (429.2×10^{-15}) and the chloride concentrations the greatest (7.5 mg/L) for 184W105, as compared to 486.1×10^{-15} and 4.6 mg/L for 184W101 and 545.1 $\times 10^{-15}$ and 5.2 mg/L for 184W103. This suggests greater water-rock interaction and a longer residence time for the groundwater from Test Well 184W105.

6.3.4.4 Strontium and Uranium Isotopes

The ratio of radiogenic to nonradiogenic strontium (87 Sr/ 86 Sr) has been used to identify groundwater sources, to evaluate potential mixing components, and to identify rock types through which groundwater has flowed. Groundwater 87 Sr/ 86 Sr ratios (0.70928) for Test Well 184W105 are quite similar to those of test wells 184W101 (0.71054) and 184W103 (0.70902) and to those expected from water-rock interaction with marine carbonates (0.707 to 0.709) (Peterman et al., 1970; Burke et al., 1982).

The ratio of uranium-234 activity to that of uranium-238 (234 U/ 238 U Activity Ratio) has also been used to evaluate groundwater flow systems. As with other chemical constituents, the 234 U/ 238 U activity ratios are relatively similar for the groundwater samples from test wells 184W105 (2.08), 184W101 (2.97), and 184W103 (3.75).



6.3.5 Radiological Parameters

Radiological parameters were analyzed in groundwater from Test Well 184W105, and the corresponding results are presented in Table B-1. The reported activity for each of these parameters is consistent with background concentrations in natural groundwater.

6.3.6 Organic Compounds

A large suite of organic compounds was analyzed for groundwater samples collected from Test Well 184W105. The corresponding minimum detection levels and MCLs (if applicable) are presented in Table B-1. With the exception of DEHP, no organic compounds were detected. DEHP was detected at a concentration of 6.1 μ g/L, which is slightly above the MCL of 6.0 μ g/L (Table B-1). This compound is present in many plastic products and may have been introduced from tubing or other plastic materials during groundwater sampling. No analyses for organic compounds were performed for the groundwater of Monitor Well 184W506M.

6.4 Summary

Groundwater samples were collected from Test Well 184W105 and Monitor Well 184W506M and analyzed for a suite of chemical parameters. Field measurement of water-quality parameters was also performed during aquifer testing and used to demonstrate stabilization of the water chemistry prior to collection of the samples. The resulting data were compared to data from samples collected from other SNWA wells in the vicinity; all wells were completed in a carbonate-rock aquifer. As is characteristic of dissolution of calcite and dolomite in waters of a carbonate-rock aquifer, the groundwater represents a calcium-magnesium-bicarbonate facies. The relative concentrations of sodium plus potassium (Na + K) tend to be slightly greater in the groundwater samples from the monitor wells than in those of the associated test wells. Similar relatively light stable isotope ratios, typical of recharge at high elevations and cold temperatures, were observed for all of the groundwater samples evaluated. The ³⁶Cl/Cl ratio measured for the sample collected from Test Well 184W105 was consistent with precipitation in the southwestern United States, and the low ¹⁴C and relatively heavy values of δ^{13} C suggest that the groundwater has interacted with isotopically heavy and ¹⁴C-free carbonate minerals. The ⁸⁷Sr/⁸⁶Sr ratios were similar between the samples collected from the test wells and were typical of water-rock interaction with marine carbonates. The $^{234}U/^{238}U$ activity ratios were also relatively similar for the groundwater samples of the test wells. The samples from the monitor wells were not analyzed for 36 Cl/Cl, δ^{13} C, 14 C, 87 Sr/ 86 Sr, or 34 U/ 238 U activity ratios.

The data were also evaluated with respect to the EPA Safe Drinking Water Act standards. For Test Well 184W105, a single constituent, DEHP, exceeded the primary drinking water MCL, and no constituent exceeded the secondary MCL. Groundwater from Monitor Well 184W506M exceeded the secondary MCL for aluminum. Both of these exceedances are attributed to sampling or to the well construction and are not considered to reflect the natural water.

7.0 REFERENCES

- Barker, J.A., 1988, A generalized radial-flow model for hydraulic tests in fractured rock: Water Resources Research, Vol. 24, No. 10, p. 1796-1804.
- Bierschenk, W.H, 1963, Determining well efficiency by multiple step-drawdown tests: International Association Science Hydrology Publication, Vol. 64, p. 493-507.
- Bredehoeft, J.D., 1997, Fault permeability near Yucca Mountain: Water Resources Research, Vol. 33, Issue 11, p. 2459–2463.
- Burke, W.H., Denison, R.E., Hetherington, E.A., Koepnick, R.B., Nelson, N.F., and Otto, J.B., 1982, Variation of seawater ⁸⁷Sr/⁸⁶Sr throughout Phanerozoic time: Geology, Vol. 10, p. 516-519.
- Cooper, H.H., and Jacob, C.E., 1946, A generalized graphical method for evaluating formation constants and summarizing well-field history: American Geophysical Union Transactions, Vol. 27, No. 4, p. 526-534.
- Craig, H., 1961, Isotopic variations in meteoric waters: Science, Vol. 133, p. 1702-1703.
- Davis S.N., Cecil, L.D., Zreda, M., and Sharma, P, 1998, Chlorine-36 and the initial valve problem: Hydrogeology Journal, Vol. 6, No 1, p. 104-114.
- Dixon, G.L., Rowley, P.D., Burns, A.G., Watrus, J.M., Donovan, D.J., and Ekren, E.B., 2007, Geology of White Pine and Lincoln counties and adjacent areas, Nevada and Utah—The geologic framework of regional groundwater flow systems: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. HAM-ED-0001, 157 p.
- Drever, J.I., 1988, The geochemistry of natural waters. Second edition: Englewood Cliffs, New Jersey, Prentice Hall.
- Drewes, H., 1967, Geology of the Conners Pass Quadrangle, Schell Creek Range, east-central Nevada: U.S. Geological Survey Professional Paper 557, 93 p.
- Duffield, G.M., 1996-2007, HydroSOLVE, Inc, AQTESOLV Version 4.50 Professional software.
- Eastman, H.S., and Muller D.C., 2009, Geologic Data Analysis Report for Monitor Well 184W506M and Test Well 184W105 in Spring Valley: Southern Nevada Water Authority, Las Vegas, Nevada, Doc. No. RDS-ED-0011, 41 p.



Eaton, A.D., Clesceri, L.S., Rice, E.W., Greenberg, A.E., and Franson, M.H., eds., 2005, Standard methods for the examination of water and wastewater. Twenty-first edition: Washington, D.C., American Public Health Association.

Freeze, R.A., and Cherry, J.A., 1979, Groundwater: Englewood Cliffs, New Jersey, Prentice-Hall.

- Galloway, D., and Rojstaczer S., 1989, Analysis of the frequency response of water levels in wells to Earth tides and atmospheric loading, *in* Proceedings 1988 Canadian/American Conference on Hydrogeology: Baniff, Alberta, Canada, National Water Well Association, Dublin, Ohio, p. 100-113.
- Hantush, M.S., 1964, Hydraulics of wells, *in* Advances in Hydroscience: Chow, V.T., eds., Vol. I, Academic Press, New York and London, p. 281-432.
- Harrill, J.R., Gates, J.S., and Thomas, J.M., 1988, Major ground-water flow systems in the Great Basin region of Nevada, Utah, and adjacent states: U.S. Geological Survey Hydrologic Investigations Atlas Report No. HA-694-C, scale 1:1,000,000, two sheets.
- Hose, R.K., and Blake, M.C., 1976, Geology and mineral resources of White Pine County, Nevada—Part I, Geology: Nevada Bureau of Mines and Geology Bulletin 85, p. 1–35.
- Kilroy, K.C., 1992, Aquifer storage characteristics of Paleozoic carbonate rocks in southeastern Nevada estimated from harmonic analysis of water-level fluctuations [Ph.D. dissertation]: University of Nevada, Reno, 77 p.
- Kruseman, G.P., and De Ridder, N.A., 1994, Analysis and evaluation of pumping test data, second edition, Publication 47: International Institute for Land Reclamation and Improvement, Netherlands, 377 p.
- Moench, A.F., 1984, Double-porosity models for a fissured groundwater reservoir with fracture skin: Water Resources Research, Vol. 20, No. 7, p. 831–846.
- Moran J.E., and Rose, T.P., 2003, A chlorine-36 study of regional groundwater flow and vertical transport in southern Nevada: Environmental Geology, Vol. 43, p. 592-605.
- Neuman, S.P., 1975, Analysis of pumping test data from anisotropic unconfined aquifers considering delayed gravity response: Water Resources Research, Vol. 11, Issue 2, p. 329–342.
- Peterman, Z.E., Hedge, C.E., and Tourtelot, H.A., 1970, Isotopic composition of strontium in sea water throughout Phanerozoic time: Geochimica et Cosmochimica Acta, Vol. 34, p. 105-120.
- Phillips, F.M., 2000, Chlorine-36—environmental tracers in subsurface hydrology: P.G. Cook and A.L. Herczeg, ed., Kluwer, Boston.
- Rorabaugh, M.J., 1953, Graphical and theoretical analysis of step-drawdown test of artesian well, *in* Proceedings of the American Society of Civil Engineers, Vol. 79, separate no. 362, 23 p.

- Rush, F.E., and Kazmi, S.A.T., 1965, Water resources appraisal of Spring Valley, White Pine and Lincoln Counties, Nevada: Nevada Department of Conservation and Natural Resources, Water Resources Reconnaissance Series Report 33, 36 p.
- SNWA, see Southern Nevada Water Authority.
- Southern Nevada Water Authority, 2006a, Geologic and hydrogeologic framework for the Spring Valley area—Presentation to the Office of the Nevada State Engineer: Southern Nevada Water Authority, Las Vegas, Nevada, 122 p.
- Southern Nevada Water Authority, 2006b, Summary of groundwater water-rights and current water uses in Spring Valley—Presentation to the Office of the Nevada State Engineer: Southern Nevada Water Authority, Las Vegas, Nevada, 97 p.
- Southern Nevada Water Authority, 2006c, Water resources assessment for Spring Valley— Presentation to the Office of the Nevada State Engineer: Southern Nevada Water Authority, Las Vegas, Nevada, 167 p.
- Southern Nevada Wather Authority, 2009, 2008 Spring Valley hydrologic monitoring and mitigation plan status and data report: Doc. No. WRD-ED-0004, Las Vegas, Nevada, 109 p.
- U.S. Geological Survey, 2007, National field manual for the collection of water-quality data [Internet]: U.S. Geological Survey Techniques of Water-Resources Investigations, Book 9, chaps. A1-A9, available from http://pubs.water.usgs.gov/twri9A.

USGS, see U.S. Geological Survey.

Welch, A.H., Bright, D.J., and Knochenmus, L.A., eds., 2008, Water Resources of the Basin and Range Carbonate-Rock Aquifer System, White Pine County, Nevada, and adjacent areas in Nevada and Utah: U.S. Geological Survey Open-File Report 2007-5261, 97 p.



This Page Left Intentionally Blank

Hydrologic Data Analysis Report for Test Well 184W105 in Spring Valley

Appendix A

CD-ROM Contents

SE ROA 12108

A.1.0 INTRODUCTION

This appendix describes the digital contents of the CD-ROM that accompanies this report. The CD-ROM contains background water-level, barometric-pressure, step-drawdown test, and constant-rate test data. This CD-ROM also includes an electronic copy of the groundwater-chemistry data, as well as the AQTESOLV input files for the step-drawdown and constant-rate tests.

A.1.1 Photos

The following photos show an overview of the site (Figure A-1), the pump and motor setup (Figure A-2), the site setup (Figure A-3), discharge line (Figure A-4), and energy dissipation at the termination of the discharge line for erosion prevention (Figure A-5).



Figure A-1 184W105 Test Well Site, Facing Southwest







Figure A-2 184W105 Test Wellhead Equipment and Piping Layout



Figure A-3 184W105 Test Wellhead Equipment with Generator



Figure A-4 Discharge Piping, Facing East from Well Site 184W105



Figure A-5 Discharge Location East of U.S. Highway 93 for Hydrologic Testing Performed at Test Well 184W105



A.1.2 Read-Me File

Included on the CD-ROM is a text file version of this appendix that describes the contents of the CD-ROM. There is also an index of the files and folders in the form of a PDF document.

A.1.3 Background Water-Level Data

A spreadsheet containing the continuous water-level data and corresponding chart from SNWA Monitor Well 184W504M. This well was used to monitor background conditions during development and testing at Test Well 184W105.

A.1.4 Barometric-Pressure Data

Barometric-pressure data are located in the continuous record data files associated with Test Well 184W105. An In-situ HERMIT 3000 data logger recorded the barometric pressure during the development and testing at well 184W105. Barometric data from SNWA ET site SV1 are also included. These data can be found in files labeled "184W105 XDR PRIMARY Data CR.xls" for the constant-rate test and "184W105 Man PRIMARY XDR Data Develop and STEP.xls" for the development and the step-drawdown test.

All barometric-pressure data are reported in inches Hg.

A.1.5 Step-Drawdown Test Data

A summary spreadsheet for the initial step test, which compiles all of the manual data, including charts, is labeled "184W105 Man Data Step Summary.xls." The manual and continuous record of the water levels for the final step test in Test Well 184W105 is provided in the spreadsheet labeled "184W105 Man PRIMARY XDR Data Develop and STEP.xls."

A.1.6 Constant-Rate Test

The constant-rate test data from Test Well 184W105 are provided in the spreadsheets labeled "184W105 Man Data 12hr CR 2300 gpm.xls" for the manual data for the 2300 gpm test; "184W105 Man PRIMARY Data CR.xls" for the manual data for the 3,000 gpm test; and "184W105 XDR PRIMARY Data CR.xls" for the continuously recorded transducer data for the 3000 gpm test. The constant-rate test data from the observation well 184W506M are provided in the spreadsheets labeled "184W506M Man PRIMARY Data CR.xls" for the manual data and "184W506M XDR Chart CR.xls" for the continuously recorded transducer data.

A.1.7 AQTESOLV

The input files for using AQTESOLV software for aquifer analysis are provided. The input files are in the form of Excel spreadsheets with water-level and discharge data for both the step-drawdown and constant-rate tests. AQTESOLV files have also been included with basic information, such as casing, borehole, and downhole equipment radius, as well as approximate saturated thickness.

A-4

A.1.8 Water Chemistry

The laboratory results from Weck Labs, Inc., are included in PDF format and labeled "184W105_WL_Chemistry.pdf" for well 184W105 and "184W506M_WL_Chemistry.pdf" for well 184W506M.



This Page Left Intentionally Blank

Hydrologic Data Analysis Report for Test Well 184W105 in Spring Valley

Appendix B

Water-Chemistry Data

SE ROA 12115

Table B-1 Field and Analytical Results, Analytical Methods, Reporting Limits, and Maximum Contaminant Levels for Inorganic, Stable Isotopic, and Radiological Constituents in Groundwater Samples from Test Well 184W105 and Monitor Well 184W506M (Page 1 of 3)

Constituent Name	Unit	Analysis Method	RL	184W105 3/8/2007 08:00	184W506M 10/31/2006 11:53	Primary MCL	Secondary MCL		
Field Measured									
рН	units	Field		7.8	8.1		6.5 to 8.5		
Conductivity	μS/cm	Field		282	385				
Dissolved Oxygen	mg/L	Field		5.08					
Temperature	°C	Field		13.0	12.7				
Turbidity	NTU	Field		0.41					
	Stabl	e Isotopes and E	Environi	mental Tracers					
Carbon-14 (¹⁴ C)	pmc	NA		6.09					
Carbon-13/12 (δ ¹³ C)	per mil (‰)	NA		-5.8					
Chlorine-36/Chloride (³⁶ Cl/Cl)	ratio	NA		4.292 × 10 ⁻¹³					
Hydrogen-2/1 (δD)	per mil (‰)	NA		-112.8/-112.2	-111.8/-111.7				
Oxygen-18/16 (δ^{18} O)	per mil (‰)	NA		-14.84	-15.09/-14.97				
Strontium-87/86	ratio	NA		0.70928					
Uranium-234/238	Activity Ratio	NA		2.0803					
		Major	Solutes						
Alkalinity Bicarbonate	mg/L as HCO ₃	SM 2320B	2	200	190				
Alkalinity Carbonate	mg/L as CaCO ₃	SM 2320B	2	ND	4.5				
Alkalinity Hydroxide	mg/L as CaCO ₃	SM 2320B	2	ND	ND				
Alkalinity Total	mg/L as CaCO ₃	SM 2320B	2	170	160				
Calcium	mg/L	EPA 200.7	0.1	35	36				
Chloride	mg/L	EPA 300.0	0.5	7.5	7.9		250		
Fluoride	mg/L	EPA 300.0	0.1	0.16	0.25	4	2.0		
Magnesium	mg/L	EPA 200.7	0.1	20	20				
Nitrate	mg/L as N	EPA 353.2	0.1	0.7	0.7	10			
Potassium	mg/L	EPA 200.7	1	1.8	1.8				
Silica	mg/L	EPA 200.7	0.1	17	17				
Sodium	mg/L	EPA 200.7	1	9.3	16				
Sulfate	mg/L	EPA 300.0	0.5	16	17		250		
Cation/Anion Balance	%	Calculation		2	3.2				

B-1



Table B-1

Field and Analytical Results, Analytical Methods, Reporting Limits, and Maximum Contaminant Levels for Inorganic, Stable Isotopic, and Radiological Constituents in Groundwater Samples from Test Well 184W105 and Monitor Well 184W506M (Page 2 of 3)

Constituent Name	Unit	Analysis	ы	184W105 3/8/2007	184W506M 10/31/2006	Primary	Secondary			
	Onit		RL 0	00.00	11.55	MCL	MCL			
Aluminum, total	μg/L	EPA 200.8	5	26	320		50 to 200			
Antimony, total	μg/L	EPA 200.8	0.5	ND	ND	6				
Arsenic, total	μg/L	EPA 200.8	0.4	2.4	1.8	10				
Arsenic (III)	μg/L	EPA 200.8	1	2.4						
Arsenic (V)	μg/L	EPA 200.8	1	ND						
Barium, total	μg/L	EPA 200.8	0.5	92	47	2,000				
Beryllium, total	μg/L	EPA 200.8	0.1	ND	ND	4				
Boron, total	μg/L	EPA 200.7	10	40	54					
Bromide	μg/L	EPA 300.1	10	66	72					
Cadmium, total	μg/L	EPA 200.8	0.1	ND	ND	5				
Chlorate	μg/L	EPA 300.1	10	ND	ND					
Chromium, total	μg/L	EPA 200.8	0.2	3.6	2	100				
Chromium (VI)	μg/L	EPA 218.6	0.3	1.8 ^a						
Chromium (III)	μg/L	Calculation	0.2	1.8						
Copper, total	μg/L	EPA 200.8	0.5	ND	2.7	1,300 ^b	1,000			
Iron, total	μg/L	EPA 200.7	20	ND	300		300			
Lead, total	μg/L	EPA 200.8	0.2	0.46	0.78	15 ^b				
Lithium, total	μg/L	EPA 200.7	10	ND	ND					
Manganese, total	μg/L	EPA 200.8	0.2	0.78	62		50			
Mercury, total	μg/L	EPA 245.1	0.1	ND	ND	2.0				
Molybdenum, total	μg/L	EPA 200.8	0.1	2.2	2.1					
Nickel, total	μg/L	EPA 200.8	0.8	ND	1.4					
Nitrite	mg/L as N	EPA 353.2	0.1	ND		1				
Orthophosphate	μg/L as P	EPA 365.1	2	ND						
Phosphorus, total	μg/L as P	EPA 365.1	10	ND						
Selenium, total	μg/L	EPA 200.8	0.4	2.1	1.5	50				
Silver, total	μg/L	EPA 200.8	0.2	ND	ND		100			

Appendix B SE ROA 12117 Table B-1 Field and Analytical Results, Analytical Methods, Reporting Limits, and Maximum Contaminant Levels for Inorganic, Stable Isotopic, and Radiological Constituents in Groundwater Samples from Test Well 184W105 and Monitor Well 184W506M (Page 3 of 3)

		Analysis		184W105 3/8/2007	184W506M 10/31/2006	Primary	Secondary		
Constituent Name	Unit	Method	RL	08:00	11:53	MCL	MCL		
Trace and Minor Constituents (Continued)									
Strontium, total	μg/L	EPA 200.7	5	170	190				
Thallium, total	μg/L	EPA 200.8	0.2	ND	ND	2			
Uranium, total	μg/L	NA		2.82		30			
Vanadium, total	μg/L	EPA 200.8	0.5	3.7	3.0				
Zinc, total	μg/L	EPA 200.8	5	ND	29		5,000		
		Miscellaneou	is Paran	neters					
Total Dissolved Solids	mg/L	SM 2540C	10	200	250		500		
Total Organic Carbon	mg/L	SM 5310C	0.3	ND	1.4				
Total Suspended Solids	mg/L	EPA 160.2	5	ND	25				
Hardness	mg/L as CaCO ₃	EPA 200.7	1	170					
Langelier Index	@ 60°C	SM 2330B	-10	0.7					
Langelier Index	@ Source Temp.	SM 2330B	-10	0.073					
MBAS	mg/L	SM 5540 C	0.05	ND					
Cyanide	mg/L	SM 4500CN E	0.01	ND		0.2			
		Radiochemic	al Parar	neters					
Gross Alpha	pCi/L	EPA 900.0	1	3.1 ± 0.82		15			
Gross Beta	pCi/L	EPA 900.0	0.87	1.8 ± 0.56		4 mrem/yr			
Radium, total gross	pCi/L	EPA 903.1		0.5 ± 0.1		5			
Radium-226	pCi/L	EPA 903.1		0.5 ± 0.1					
Radium-228	pCi/L	EPA 904	0.4	ND					
Radon	pCi/L	SM 7500		353 ± 37					
Strontium-90	pCi/L	EPA 905.0	0.6	ND					
Tritium	TU	NA	0.8	ND					
Tritium	pCi/L	EPA 906.0	340	ND					
Uranium	pCi/L	EPA 200.8	2	0.13		30 µg/L			

^aHolding time was exceeded.

^bReported value is the action limit. MBAS = Methylene blue active substances mrem/yr = Millirem per year NA = Not available ND = Not detected

RL = Reporting limit

SM = Standard method (Eaton et al., 2005)

TU = Tritium Unit



Table B-2 Organic Compounds Analyzed in Groundwater Samples from Test Well 184W105, Including the EPA Method, Reporting Limit, and Maximum Contaminant Level (Page 1 of 2)

Chlorinated Pesticides by EPA 508 (µg/L)										
Analyte	RL	MCL	Analyte	RL	MCL	Analyte	RL	MCL		
Aldrin	0.075		Endosulfan II	0.01		PCB 1016 Aroclor	0.1			
BHC (Alpha)	0.01		Endosulfan sulfate	0.05		PCB 1221 Aroclor	0.1			
BHC (Beta)	0.05		Endrin	0.1	2	PCB 1232 Aroclor	0.1			
BHC (Delta)	0.05		Endrin aldehyde	0.05		PCB 1242 Aroclor	0.1			
Chlordane (tech)	0.1	2	Heptachlor	0.01	0.4	PCB 1248 Aroclor	0.1			
Chlorothalonil	5		Heptachlor Epoxide	0.01	0.2	PCB 1254 Aroclor	0.1			
4,4'-DDD	0.02		Hexachlorobenzene	0.5	1.0	PCB 1260 Aroclor	0.1			
4,4'-DDE	0.01		Hexachlorocyclopentadiene	1	50	Propachlor	0.5			
4,4'-DDT	0.02		Lindane	0.2	0.2	Toxaphene	1	3		
Dieldrin	0.02		Methoxychlor	10	40	Trifluralin	0.01			
Endosulfan I	0.02		Polychlorinated biphenyls (PCBs)	0.5	0.5					
Organic Compounds by EPA 525.2 (μg/L)										
Alachlor	0.1	2	Di(2-ethylhexyl) phthalate	3	6	Prometon	0.2			
Atrazine	0.1	3	Diazinon	0.1		Prometryn	0.1			
Benzo(a)pyrene	0.1	0.2	Dimethoate	0.2		Simazine	0.1	4		
Bromacil	1		Metolachlor	0.1		Thiobencarb	0.2			
Butachlor	0.2		Metribuzin	0.1						
Di(2-ethylhexyl) adipate	5	400	Molinate	0.1						
		Pu	rgeable Organic Compounds	by EPA	A 524.2 (µ	u g/L)				
tert-amyl Methyl Ether	3		Di-isopropyl ether	3		1,2,3-Trichlorobenzene	0.5			
Benzene	0.5	5	1,1-Dichloroethane	0.5		1,2,4-Trichlorobenzene	0.5	70		
Bromobenzene	0.5		1,2-Dichloroethane	0.5		Methyl tertiary butyl ether (MTBE)	3			
Bromochloromethane	0.5		1,1-Dichloroethylene	0.5	5	Naphthalene	0.5			
Bromodichloromethane	0.5		cis-1,2-Dichloroethylene	0.5	7	n-Propylbenzene	0.5			
Bromoform	0.5		trans-1,2-Dichloroethylene	0.5	70	Styrene	0.5	100		
2-Butanone	5		Dichlorodifluoromethane	0.5	100	Tetrachloroethylene	0.5	5		
n-Butylbenzene	0.5		1,2-Dichloropropane	0.5		1,1,1,2-Tetrachloroethane	0.5			
sec-Butylbenzene	0.5		1,3-Dichloropropane	0.5	5	1,1,2,2-Tetrachloroethane	0.5			
tert-Butylbenzene	0.5		2,2-Dichloropropane	0.5		Toluene	0.5	1,000		
tert-Butyl Ethyl ether	3		1,1-Dichloropropene	0.5		1,1,1-Trichloroethane	0.5	200		
Carbon tetrachloride	0.5	5	cis-1,3-Dichloropropene	0.5		1,1,2-Trichloroethane	0.5	5		
Chlorobenzene	0.5	100	trans-1,3-Dichloropropene	0.5		Trichloroethylene	0.5	5		
Chloroethane	0.5		total-1,3-Dichloropropene	0.5		Trichlorofluoromethane	5			
Table B-2 Organic Compounds Analyzed in Groundwater Samples from Test Well 184W105, Including the EPA Method, Reporting Limit, and Maximum Contaminant Level (Page 2 of 2)

Purgeable Organic Compounds by EPA 524.2 (µg/L) (Continued)								
2-Chloroethylvinyl ether	1		Ethylbenzene	0.5	700	1,2,3-Trichloropropane	0.5	
Chloroform	0.5		Hexachlorobutadiene	0.5		1,1,2-Trichloro-1,2,2-trifluoroethane	10	
2-Chlorotoluene	0.5		2-Hexanone	5		1,2,4-Trimethylbenzene	0.5	
4-Chlorotoluene	0.5		Isopropylbenzene	0.5		1,3,5-Trimethylbenzene	0.5	
Dibromochloromethane	0.5		p-Isopropyltoluene	0.5		Vinyl chloride	0.5	2
Dibromomethane	0.5		Methyl bromide	0.5		Xylene (m,p) isometric pair	0.5	
m-Dichlorobenzene	0.5		Methyl chloride	0.5		Xylenes, total	0.5	10,000
o-Dichlorobenzene	0.5	600	Methylene chloride	0.5	5	o-Xylene	0.5	
p-Dichlorobenzene	0.5	75	4-Methyl-2-pentanone	5				
Chlorinated Acids by EPA 515.3 (µg/L)								
2,4,5-T	0.2		Acifluorfen	0.5		Dichlorprop	0.3	
2,4,5-TP (Silvex)	0.2	50	Bentazon	2		Dinoseb	0.5	7
2,4-D	0.5	70	Dalapon	0.5	200	Pentachlorophenol	0.2	1
2,4-DB	2		DCPA	0.1		Picloram	1	500
3,5-Dichlorobenzoic acid	1		Dicamba	0.6				
N-Methylcarbamoyloximes and N-Methylcarbamates by EPA 531.1 (µg/L)								
3-Hydroxycarbofuran	2		Baygon	5		Methomyl	2	
Aldicarb	2		Carbaryl	2		Oxamyl (Vydate)	2	200
Aldicarb sulfone	2		Carbofuran	5	40			
Aldicarb sulfoxide	2		Methiocarb	3				
Organics by Other EPA Methods (µg/L)								
Glyphosate (EPA 547)	5	700	Diquat (EPA 549.2)	4	20	1,2-Dibromo-3-chloropropane (EPA 504.1)	0.01	0.2
Endothall (EPA 548.1)	45	100	Dioxin (EPA 1613)	5 pg/L	30 pg/L	Ethylene dibromide (EPA 504.1)	0.02	0.05

RL = Reporting Limit





B.1.0 References

Eaton, A.D., Clesceri, L.S., Rice, E.W., Greenberg, A.E., and Franson, M.H., eds., 2005, Standard methods for the examination of water and wastewater. Twenty-first edition: Washington, D.C., American Public Health Association.